ORIGINAL

07 JAN 19 PM 1:5

RECEIVED-FPSC

BEFORE THE FLORIDA PUBLIC SERVICE COMMISSION

)

)

)

)

In re: Progress Energy Florida, Inc.'s Petition for Approval of Southeast Supply Header long-term fuel transportation contracts.

Docket No. 060793-EI

Filed: January 19, 2007

PROGRESS ENERGY FLORIDA, INC.'S NOTICE OF SERVICE OF RESPONSES TO STAFF'S FIRST DATA REQUEST

Progress Energy Florida, Inc. ("PEF"), by and through its undersigned counsel, hereby provides notice of service of its Responses to Staff's First Data Request, by hand delivery on Lisa C. Bennett, Esq., Florida Public Service Commission, Office of General Counsel, 2540 Shumard Oak Boulevard, Tallahassee, Florida 32399-0850. A copy of PEF's Responses to Staff's First Data Request with confidential portions redacted is filed herewith with the Commission Clerk.

Respectfully submitted this 19th day of January, 2007.

Kenneth A. Hoffman, Esq. Marsha E. Rule, Esq. Rutledge, Ecenia, Purnell & Hoffman, P.A. P. O. Box 551 Tallahassee, FL 32302 (850) 681-6788 (Telephone) (850) 681-6515 (Telecopier)

- - and - -

John T. Burnett, Esq. P. O. Box 14042 St. Petersburg, FL 33733-4042 (727) 820-5184 (Telephone) (727) 820-5249 (Telecopier)

Attorneys for Progress Energy Florida, Inc. DOCUMENT NUMBER-DATE

00541 JAN 195

CMP _____ COM _____ CTR _____ ECR _____ GCL _____ GCL _____ OPC _____ RCA _____ SCR _____ SGA ____

- SEC _)___
- OTH _____

FPSC-COMMISSION CLERK

CERTIFICATE OF SERVICE

I HEREBY CERTIFY that a copy of the foregoing was furnished by Hand Delivery to the following this 19th day of January, 2007:

Lisa C. Bennett, Esq. Office of General Counsel Florida Public Service Commission 2540 Shumard Oak Boulevard Tallahassee, FL 32399-0850

.

Kenneth A. Hoffman, Esq.

progressenergy\sesh\noticeofservicejan1907.wpd

PROFESSIONAL ASSOCIATION ATTORNEYS AND COUNSELORS AT LAW

STEPHEN A. ECENIA RICHARD M. ELLIS KENNETH A. HOFFMAN LORENA A. HOLLEY MICHAEL G. MAIDA MARTIN P. MCDONNELL J. STEPHEN MENTON POST OFFICE BOX 551, 32302-0551 215 SOUTH MONROE STREET, SUITE 420 TALLAHASSEE, FLORIDA 32301-1841

> TELEPHONE (850) 681-6788 TELECOPIER (850) 681-6515

> > January 19, 2007

R. DAVID PRESCOTT HAROLD F. X. PURNELL MARSHA E. RULE GARY R. RUTLEDGE MAGGIE M. SCHULTZ

GOVERNMENTAL CONSULTANTS PARSONS B. HEATH MARGARET A. MENDUNI

REDACTED

Lisa C. Bennett, Esq. Florida Public Service Commission Office of General Counsel 2540 Shumard Oak Boulevard Tallahassee, Florida 32399-0850

Re: Docket No. 060793-EI - Petition for approval of long-term fuel transportation contracts with Duke Energy Southeast Supply Header, LLC and CenterPoint Energy Southeastern Pipelines Holding, L.L.C. ("SESH Pipeline Contracts"), by Progress Energy Florida, Inc.

Dear Ms. Bennett:

By this letter, Progress Energy Florida, Inc. ("PEF") hereby provides its Responses to the Commission Staff's First Data Request:

1. Why does PEF seek approval of its long-term fuel supply and transportation contracts with the Southeast Supply Header, LLC (SESH), as opposed to only seeking Commission approval of the SESH project's costs as prudent and reasonable for fuel clause cost recovery purposes? In responding to this question, please include responses to the following questions:

A.) If the Commission only approves PEF's SESH project for cost recovery purposes and does not approve the long-term fuel supply and transportation contracts, will PEF continue with the SESH project? Please explain.

PEF's Response: PEF will have to seriously weigh the risks associated with going forward with this project in the absence of Florida Public Service Commission ("PSC" or "Commission") approval. Simply finding that the costs are of a nature recoverable through the fuel clause and not addressing the prudence of entering into a long-term contract for which all terms are known now could expose the Company to significant financial risk. PEF has filed the final versions of the SESH contracts with final terms and conditions set forth therein and all material facts regarding these long-term contracts that are known at this time have been presented to the Commission. As the Commission knows, these types of costs were already found to be of a nature recoverable through the fuel clause for FPL in Docket No.

Lisa C. Bennett, Esq. Page 2 January 19, 2007

060001-EI. There is past PSC precedent and policy with regard to seeking pre-approval of the terms and conditions of long-term fuel related contracts. Further, everything the Commission needs to make a decision on this matter is currently available. It is important to note that PEF is not requesting the Commission to approve anything with regard to the future management of these contracts. Rather, PEF is only requesting the Commission find that entering into these contracts was a prudent decision based on the facts known at the time the contracts were executed.

B.) Why would Commission approval of the long-term fuel supply and transportation contracts be in the public interest?

PEF's Response: Approval of the long-term contracts is in the best interest of the public for several reasons. Absent these contracts, by 2009, approximately 78% of PEF's transportation capacity will be sourced from the off-shore Mobile Bay area. The SESH contracts will cut our projected reliance on off-shore Gulf of Mexico production in 2009 by half. This is important due to the fact that demand for gas is projected to increase significantly in the future, while production from the Mobile Bay area is decreasing. With this in mind, it is prudent for PEF to seek out and have available alternate production supply. In addition, the production in the off-shore Mobile Bay area is highly susceptible to prolonged shut downs during extreme weather events. This was illustrated in the 2005 storm season and is discussed in some depth in the report titled "Impact of the 2005 Hurricanes on the Natural Gas Industry in the Gulf of Mexico Region" produced by the DOE, a copy of which is provided herewith as Exhibit A. The Southeast Supply Header will provide access to on-shore natural gas supply basins. It is important to understand that the transportation from Mobile Bay into the State of Florida has been very reliable during extreme weather events, it is primarily the production that has been significantly impacted. This new source would offset the risk associated with natural gas supply interruptions from extreme weather events. Another benefit to the public interest is the potential for downward pressure on gas prices in the Mobile Bay area. PEF believes that adding access to extra supply will place downward pressure on the price for natural gas coming out of the Mobile Bay area. One of the fundamental rules of economics is that with all else being equal, an increase in supply will decrease the market clearing price for a product.

C.) Please cite all relevant examples that PEF believes serves as precedent for the Commission approving fuel transportation contracts (as opposed to Commission approval of cost recovery only).

PEF's Response: In Docket No. 041414-EI, PEF filed a petition with the Commission requesting approval of the terms and conditions of the contracts with BG LNG Services, LLC, Southern Natural Gas Company and Florida Gas Transmission (the "Cypress Project") for long-term natural gas supply and transportation. The Commission approved the long term fuel supply and transportation contracts comprising the Cypress Project in Order No. PSC-05-0721-FOF-EI issued July 5, 2005. This is a particularly relevant example in that it had to do with contracts that were the best and most prudent alternative not based solely on price, but only after non-price factors were considered.

Lisa C. Bennett, Esq. Page 3 January 19, 2007

D.) If the Commission is being asked to approve a contract, shouldn't the Commission be involved in the contract negotiations between PEF and the contracting party?

PEF's Response: It is PEF's position that contract negotiations are a utility management function. The role of the Commission is to review and determine the prudence of such actions and the resulting contracts based on the facts known at the time the contracts are signed. If the Commission were involved in the negotiation process, it would hamper the ability of the Commission to serve in an objective, unbiased capacity in reviewing and determining the prudence of the contracts after the fact.

E.) If the answer to the above is yes, would this open the door to PEF inviting the Commission into other contractual negotiations and management decisions?

PEF's Response: N/A

F.) If Commission found portions of the language within the contract objectionable, would PEF amend the contract to include the Commission's concerns?

PEF's Response: The contracts represent a lengthy process of evaluation and negotiation between PEF and the counterparty. If the Commission found certain language objectionable, PEF would first try to explain why the language is in place and try to alleviate the Commission's concerns. If the Commission still had problems with the language, PEF would look into any feasible options to address the Commission's concerns including possible language modification. That being said, it is important to understand that these contracts represent extended negotiations that are difficult to modify where both sides have had to compromise on various, interrelated provisions. A unilateral modification imposed by the Commission could impact the negotiated balance of considerations each party provides and receives under these negotiated agreements.

2. Please refer to Exhibit KF-2. Also refer to witness Fonvielle's testimony, page 9, lines 15 through 23 and continuing on to page 10, lines 1 through 7, and to page 7 of witness Portuondo's testimony:

A.) Other than the cost estimates represented on this schedule, are there any other costs associated with PEF's SESH project that are proposed to be charged to the fuel cost recovery clause? Please explain.

PEF's Response: Please refer to SESH's Pro Forma Tariff (hereinafter referred to as the "Tariff"), attached hereto as **Exhibit B**. PEF anticipates recovery of the variable usage rates referenced in Rate Schedule FTS Section 3.2(c) and further defined in Sections 8 and 22 of the Tariff (Original Sheet Nos. 9, 71, 92, 93, 94 and 95). In addition to normally recurring pipeline fixed and variable charges, PEF would be subject to non-recurring charges such as the penalty charges defined in Sections 13.8 and 23.4

Lisa C. Bennett, Esq. Page 4 January 19, 2007

of the Tariff (Original Sheet Nos. 83 and 97), and other charges specifically approved by FERC in the future.

B.) On page 9 of witness Fonvielle's testimony, lines 17 through 19, he states the following: "PEF's participation in the SESH Pipeline Project will result in, but not be limited to, two types of invoiced costs to be passed through the fuel clause: (1) fixed demand costs and, (2) variable commodity costs." Why would PEF not limit the costs proposed to be passed through the fuel cost recovery clause to these two components?

PEF's Response: As stated in Section 3.2 of the proposed Rate Schedule FTS (see Sheet No. 9 of the Tariff attached hereto as **Exhibit B**), there are other applicable charges and surcharges that are contemplated but not currently known at this time referenced in Sections 8 and 22 of the proposed Tariff General Terms and Conditions.

C.) What other known costs or potential costs does witness Fonvielle propose for recovery through the fuel cost recovery clause?

64

PEF's Response: None known or currently unknown that have not been addressed in PEF's responses to the above referenced Requests Nos. 2(A) and 2(B).

D.) Please provide an analysis of the variable costs on KF-2 showing the rates and calculation of the variable costs and stating all assumptions.

PEF's Response: The variable costs shown on Exhibit KF-2 are comprised of the following three components:

Lisa C. Bennett, Esq. Page 5 January 19, 2007

Key assumptions used in PEF's model are shown on Appendix A, page 16 of the Business Analysis Package attached as **Confidential Exhibit C**.

E.) Has the FERC set recourse rates for the SESH pipeline? If yes, what are these rates? If no, when does PEF expect the FERC to set recourse rates for the SESH pipeline?

PEF's Response: No. The Tariff was filed as Exhibit "P" to the SESH Certificate Application filed with FERC on December 18, 2006. It is typical for FERC to approve the proposed tariff at the time the 7c Certificate is approved and issued. As to when it will be approved, this question can only be answered by the FERC.

Recourse rates as filed are shown on the Tariff Original Sheet Nos. 4, 5, and 6.

- FTS Max Reservation Rate (Fixed Cost) = \$.3827/dt (100% LF Rate)
- FTS Usage-1 Rate (Variable Cost) = .0064/dt
- FTS ACA Surcharge (Variable Cost) = \$.0000/dt (see Original Sheet No. 93, Section 22.1)

Gas Transporters Use (%) (Variable Fuel Retention = 0.70% (will be a fuel tracker with annual trueup, see Tariff Original Sheet No. 94, Section 22.2)

3. Four new LNG import terminals are proposed for the Gulf coast (Conoco, Freeport, TX; Exxon, Ingleside, TX; Sempra, Hackberry, LA; Cheniere, Sabine, LA). Also, AES has proposed a LNG import terminal for the Bahamas. These proposed terminals are expected to begin service by 2010:

A.) If these new terminals begin service as proposed, what effect will that have on the Southeast Supply Header pipeline?

PEF's Response: If new LNG terminals are eventually permitted and constructed in the Gulf region, they should provide incremental supply to many interstate pipelines including pipelines that are planned to be interconnected o the SESH pipeline. PEF would be able to access this incremental supply directly or indirectly through the SESH pipeline capacity.

B.) What effect will new LNG terminals on the Gulf coast have on PEF's use of the SESH for supply?

PEF's Response: The effect of additional LNG supply in the Gulf region cannot be known at this time. As with domestic supply sources, LNG supply is and will likely continue to be sold at a price that will fluctuate daily based on an observable market index, such as the NYMEX Henry Hub contract, with a basis differential to account for locational differences. As stated in response

Lisa C. Bennett, Esq. Page 6 January 19, 2007

to question 3(A) above, PEF would have direct or indirect access to Gulf LNG supply and could contract for this supply potentially utilizing the SESH pipeline if it is the best cost option after considering all price and non-price terms.

C.) For each LNG terminal project listed above, what is PEF's current understanding of the status of the project?

PEF's Response: The following information was obtained from each company's website and from FERC filing documents.

ConocoPhillips, Freeport, TX

- In late 2004, ConocoPhillips signed an agreement with Freeport LNG Development, LP to participate in the proposed Quintanna (Freeport), TX terminal.
- Federal Energy Regulatory Commission (FERC) FERC Application in Docket No.: CP03-75-000

• Construction Status (as filed by ConocoPhillips on January 11, 2007)

Activity	% Completion
Storage and Vaporation	
Tanks	52.9%
Balance of Plant	41.5%
Marine Terminal & Berthing Area	78.4%
Pipeline	98.7%
Wetland Mitigation	95.0%

Expected to be in service in Early 2008.

The Project will connect directly with Kinder Morgan TX, Houston Pipeline and Dow Pipeline. It will indirectly interconnect to FGT via Kinder Morgan TX and Houston Pipeline at existing interconnects in Texas. There will be no direct connection with Gulfstream Natural Gas System.

Lisa C. Bennett, Esq. Page 7 January 19, 2007

- Freeport LNG Expansion L.P.'s FERC Application, Docket No.: CP05-361-000.
- Filed with FERC on May 26, 2005.

Exxon, Ingleside (Vista Del Sol), TX

- Vista Del Sol LNG Terminal located in San Patricio County, TX
- Filed in FERC Docket No. CP04-395-000
- FERC Order issuing Certificate issued on June 20, 2005
- Facility will be connected with Vista del Sol Pipeline (25 miles in length)
- There is no direct connection with either FGT or Gulfstream Natural Gas System planned.
- The facility is expected to be operational in the 2008/09 timeframe, with a processing capacity of 1 billion cubic feet per day (bcfd) of LNG.
- No construction activity to date.
- Possible sale of project to a Dutch subsidiary of the Carlyle Group and Riverstone Holdings as reported in the Friday, January 12, 2007, Gas Daily.

Sempra, (Cameron LNG Terminal) Hackberry, LA

- Cameron LNG Terminal located in Cameron Ph, LA
- Filed in FERC Docket No. CP02-378-000
- FERC Order issuing Certificate on September 11, 2003
- The Cameron LNG receipt terminal and associated facilities will be built in Cameron, Louisiana, which is located approximately 148 miles east of Houston, Texas, and 230 miles west of New Orleans, Louisiana. The \$750 million project will have the capacity to regasify up to 1.5 billion cubic feet (Bcf) of natural gas per day, and the site can accommodate expansion up to 2.65 Bcf per day. The terminal is projected to begin commercial operations in late 2008.
- On Aug. 1, 2005, Sempra LNG signed a 20-year Capacity Agreement with ENI S.p.A. for 40% of the capacity of the Cameron LNG receipt terminal (approximately 600 Mmcfd).
 - The Engineering, Construction and Procurement (EPC) contract was signed in December 2004 with the joint venture Aker Kvaerner / IHI. A Notice To Proceed with construction was granted in August 2005.
 - Construction at the Cameron LNG terminal site began in September 2005.

Lisa C. Bennett, Esq. Page 8 January 19, 2007

- The FERC application was submitted in December 2005, requesting approval to expand to 2.65 Bcf/d send out capacity.
- In addition to the terminal, Sempra Pipelines & Storage will also be constructing a 35mile pipeline to transport natural gas from the facility to connect with existing interstate pipelines to the north which will include an interconnect with FGT in Calcasieu Ph, LA. It will not connect directly with Gulfstream Natural Gas System.

Cheniere Energy, (Sabine Pass LNG) Sabine, TX

- Sabine Pass LNG Terminal located in Jefferson County, TX
- Filed on December 22, 2003, in FERC Docket No. CP04-47 et.al.
- FERC Order issuing Certificate issued on June 20, 2005
- Facility will be connected with Sabine Pass Pipeline (16 miles in length)
- The facility is expected to be operational in the 2008 timeframe, with a processing capacity of 2.6 billion cubic feet per day (bcfd) of LNG.
- The facility is to be expanded to accommodate an incremental 1.4 Bcf/d sendout and is expected to be operational in 2009.
 - Cheniere will connect to a 16-mile pipeline from the Sabine Pass LNG terminal and expects the pipeline to be operational in the fourth quarter of 2007. This pipeline will commence at the proposed Sabine Pass LNG, L.P. LNG import terminal near Sabine Lake in Cameron Parish, Louisiana and will run eastward along a corridor that will allow for interconnection points with interstate and intrastate natural gas transmission pipelines in South Louisiana, including Natural Gas Pipeline Company of America, Transcontinental Gas Pipeline Corporation, Tennessee Gas Pipeline Company, Florida Gas Transmission Company (Cameron Ph, LA), and Bridgeline Holdings,
 - L.P. The project will not interconnect with Gulfstream Natural Gas System.
 - As of December 1, 2006, construction of the project is 50.9% complete.

AES LNG (Ocean Express Pipeline) Ocean Cay Terminal, Bahamas

- Filed with the FERC in February 2002 in Docket No. CP02-90
- LNG Terminal awaiting Bahamian Governmental Approval
- Construction has not started.
- The facility was expected to be operational in the 2008/09 timeframe but those dates appear very optimistic.
- On December 21, 2006, AES requested that the FERC grant a four (4) year extension of the in-service date to January 29, 2011, due to unexpected delays in securing final project approval from the Bahamian Government.
- Project's future remains uncertain.

Lisa C. Bennett, Esq. Page 9 January 19, 2007

4. Did PEF consider new LNG sources from new terminals on the Gulf coast as an alternative to the SESH project? Please explain.

<u>PEF's Response</u>: Yes. As mentioned on pages 11 and 12 of the Direct Testimony of Kent Fonvielle and page 9 of the Business Analysis Package attached hereto as Confidential Exhibit C, PEF considered potential LNG sources in the Gulf region as well as additional LNG from the Elba Island LNG terminal.

While additional LNG will further reduce PEF's dependence on offshore Mobile Bay area production, by 2009 existing LNG contracts with BG will make up **structure** percent of PEF's overall supply portfolio. Therefore, purchasing additional LNG supplies will not in and of itself increase supply diversity. Additionally, LNG cargos can be re-routed during situations when prices are more favorable in world markets other than the United States (i.e., gas prices more attractive in Europe). Furthermore, Gulf LNG terminals would likely experience the same disruptions, simultaneous with curtailments of offshore production, due to extreme weather events in the Gulf. Finally, when compared to onshore supply, LNG introduces incremental geopolitical risks, where the curtailment of gas exports could occur due to foreign country events such as strikes, wars, terrorism, etc.

5. Please refer to page 6, lines 12 through 16, of Javier Portuondo's testimony and to Kent Fonvielle's testimony, page 7, lines 14 through 21 and continuing on to page 8, lines 1 through 6. For gas to meet increasing gas requirements, what are PEF's plans for transporting this additional gas on pipelines in Florida? As part of this answer, please provide a schedule showing additional firm pipeline capacity for the years 2008 through 2012.

PEF's Response: With the SESH pipeline capacity PEF will begin purchasing a portion of its natural gas in the Perryville Hub area. This onshore gas will be transported to the Mobile Bay area using the SESH pipeline, where it will feed into the two existing interstate pipelines that supply gas to the State of Florida: Florida Gas Transmission and Gulfstream Natural Gas System. PEF will use its firm transportation rights on FGT and Gulfstream to transport this gas to its plants in Florida. Additional firm pipeline capacity currently under contract for PEF is shown in the table below. The capacity on each pipeline during a specific period is as stated and is not additive to earlier periods shown.

Lisa C. Bennett, Esq. Page 10 January 19, 2007



6. Please refer to page 6 of witness Portuondo's testimony, lines 9 through 16 and to page 9, lines 4 through 12 of witness Fonvielle's testimony regarding potential savings for PEF's customers as a result of the SESH project. Also, regarding costs, please refer to page 7 of witness Portuondo's testimony and to page 9 of witness Fonvielle's testimony, lines 15 through 23 and continuing on to page 10, lines 1 through 7.

A.) Please explain how the potential savings are calculated. Include total savings, how the savings were calculated, and the comparison to PEF's estimated annual project costs.

PEF's Response: As can be seen from the graph on page 6 of the attached Business Analysis Package, the Mobile Bay basis (premium above NYMEX) has been increasing over the past few years. Basis increases during periods when demand for gas outstrips supply in a given region. Therefore, new infrastructure projects such as the SESH pipeline that increase available supply, should help to suppress the increasing basis and may act to lower the basis over time. The actual future basis, with and without the incremental supply brought to the region by the proposed SESH pipeline, is not known and therefore PEF cannot calculate these potential savings.

A second opportunity to create savings is to procure gas supply via the SESH pipeline in the Perryville Hub area at a basis lower than the Mobile Bay basis.

Lisa C. Bennett, Esq. Page 11 January 19, 2007

B.) How can the SESH project potentially cause a lower overall cost of gas for PEF's customers?

PEF's Response: See response to 6(A) above.

C.) Will the SESH project increase PEF's delivered price of gas? Please explain.

PEF's Response: The future cost of natural gas delivered to PEF's generating plants is affected by many variables, but is primarily driven by changes in the underlying commodity price as measured by the NYMEX contract. PEF will continue to manage volatility in the price of gas through our hedging program. In addition, PEF will seek to offset the increase in fixed pipeline costs through procurement strategies such as those described in 6(A) above.

7. Will the SESH project allow PEF to negotiate better non-price terms for natural gas supply contracts? Please explain.

PEF's Response: The SESH project will not necessarily lead to better non-price terms in future gas supply deals, although it will increase the pool of suppliers available to PEF and thus increase competition. In addition, the onshore gas available via the SESH capacity will be less subject to disruption from severe weather events affecting offshore production platforms.

8. Will the SESH project cause PEF to buy less off-shore gas? Please explain.

PEF's Response: Yes. The SESH capacity will allow PEF to buy less offshore gas than otherwise would be required to support the gas generation fleet in the future.

9. What part of new PEF demand for natural gas will be supplied by Mobile Bay?

Lisa C. Bennett, Esq. Page 12 January 19, 2007

PEF's Response: The following table depicts the contribution of each supply region as a percentage of PEF's projected gas usage over time. PEF's projected gas usage includes existing gas-fired generation and planned gas-fired generating units.



10. PEF's proposed share of the SESH project capacity is 200,000 MMBtu by 2009. The projected in-service date is around mid-2008.

PEF's Response: PEF's proposed share of SESH project capacity is as follows:

6/1/2009 – 5/31/2010: 150,000 Dths/da	у
6/1/2010 – 5/31/2022: 200,000 Dths/da	y
6/1/2022 - 5/31/2023: 50,000 Dths/day	

A.) As of the SESH project's mid-2008 in-service date, does PEF intend to fully use the 200,000 MMBtu firm capacity?

PEF's Response:

B.) Will PEF have excess capacity initially or at various times? If no, please explain. If yes, what are PEF's plans for the excess capacity?



Lisa C. Bennett, Esq. Page 13 January 19, 2007

C.) Referring to Exhibit KF-2, please provide the analysis assuming 80% and 90% utilization of pipeline capacity.

PEF's Response: See attached **Confidential Exhibits D and E** which show projected annual costs assuming 80% and 90% utilization, respectively.

D.) How will PEF's participation in the SESH project be affected if the proposed inservice date is moved back to the end of 2008?

PEF's Response: PEF's contract with SESH has several conditions precedent to monitor progress of the project and allow termination of the contract if certain critical milestones are not met. Assuming these earlier milestones are met, PEF would begin taking service at the later in-service date of the project.

E.) What is the SESH pipeline project's status regarding FERC regulatory approval? As part of the response to this question, please provide a timeline of the significant FERC regulatory actions, both historical and expected.

<u>PEF's Response</u>:

- National Environmental Policy Act (NEPA) pre-filing filed in Docket No. PF06-28
- NEPA pre-filing request submitted on May 5, 2006
- Certificate application was filed by SESH with the FERC on December 18, 2006
- Docket Nos. CP07-44, CP07-45, CP07-46 and CP07-47 were assigned to the subject NGA Section 7c Certificate Application
- Noticed by the FERC on December 28, 2006 with a comment deadline date of January 18, 2007. Likely will take eight (8) months for approval from FERC but only FERC knows how long it will take. (See FERC Notice attached hereto as **Exhibit F**)

F.) Other than FERC regulatory approval, are there remaining regulatory approvals for the SESH pipeline project? If yes, please provide a summary of the remaining regulatory approvals.

PEF's Response: This information is not known to PEF but is likely available from SESH.

11. If the Commission chooses to move consideration of this petition past the March 13, 2007 Agenda Conference, what risks would this create for PEF regarding terms in the contract and whether PEF would proceed with the contract/project?

PEF's Response:

Lisa C. Bennett, Esq. Page 14 January 19, 2007

12. Please refer to Kent Fonvielle's testimony, page 10, lines 8 through 22, and pages 11 and 12. Of the various options that PEF considered, was the SESH the lowest cost alternative? Please explain. Also, provide an analysis showing the cost of each alternative considered.

<u>PEF's Response</u>: No. Although SESH was not the lowest cost alternative, it was determined to be the most cost-effective option considering all price and non-price factors as further described below.



13. Please provide complete copies of any studies or analyses done by PEF or on behalf of PEF since January 1, 2005 that consider LNG as a future source of supply of natural gas for PEF.

PEF's Response: See page 9 of the attached Business Analysis Package.

14. Please provide complete copies of any cost/benefit analyses that justify the SESH project as the lowest cost alternative for new natural gas supply for PEF.

PEF's Response: See pages 10 & 11 of the attached Business Analysis Package.

Lisa C. Bennett, Esq. Page 15 January 19, 2007

15. Please provide complete copies of any cost/benefit analyses or savings calculations regarding the SESH project's potential to result in savings for customers.

PEF's Response: Please refer to pages 3 - 7 of the attached Business Analysis Package for a discussion of the overall benefits of the SESH pipeline project. Also, please refer to response to question 6(A) and to pages 23 - 24 of the attached Business Analysis Package for a discussion and analysis of potential savings that may be derived.

16. Please refer to page 6 of witness Portuondo's testimony, lines 9 through 16 and to page 9, lines 4 through 12 of witness Fonvielle's testimony. Please provide any analyses of the impact the SESH project will have on gas prices in the Mobile Bay area. Also, please provide copies of any documentation that these witnesses relied upon in making these specific statements about savings and "lower overall cost of gas for PEF's customers."

PEF's Response: PEF has done no formal analysis of the impact the SESH project will have on gas prices in the Mobile Bay area. This project is not being proposed to lower the price of gas available in Mobile Bay. Absent any price impact on natural gas out of Mobile Bay, PEF believes this is a worthwhile project and will benefit PEF's customers through improved reliability and supply diversity. With regard to the statements referenced above in witness Portuondo and Fonvielle's testimony, our belief that there may be a positive impact on the price of gas out of Mobile Bay is founded in economic theory. The law of supply and demand dictates that with all else equal, an increase in supply will result in a lower market clearing price for a commodity. In fact, the existence of OPEC illustrates the influence supply of a commodity can have on its price. This pipeline will increase the supply of gas available to the Mobile Bay area as well as anyone connecting to the Mobile Bay area. While it is not possible to make a reliable prediction on price impact, it makes sense that at any given point in the future, with demand for natural gas at a given level, one would expect the price coming out of Mobile Bay to be lower with an increased supply available.

17. Please provide complete copies of all workpapers, analyses, and source documents associated with the testimony and exhibits of Javier Portuondo.

PEF's Response: Documents relied on concerning the purpose and scope of the SESH project are being provided as part of the response to question number 18. Please see copy of the DEP report titled "Florida's Energy Plan" issued on January 17, 2006 referenced in Mr. Portuondo's testimony, attached hereto as **Exhibit G**. Please also see the copy of the DOE report titled "Impact of the 2005 Hurricanes on the Natural Gas Industry in the Gulf of Mexico Region", attached as **Exhibit A**. This report contains useful background information with regard to natural gas production and susceptibility to extreme weather events which was considered as Mr. Portuondo's testimony was prepared.

Lisa C. Bennett, Esq. Page 16 January 19, 2007

18. Please provide complete copies of all workpapers, analyses, and source documents associated with the testimony and exhibits of Kent Fonvielle.

PEF's Response: See attached files. (Exhibit H).

19. Please provide complete copies of studies done by or for PEF addressing gas procurement reliability and SESH.

PEF's Response: See attached files. (Confidential Exhibit C).

20. Please provide a copy of Section 3 of Rate Schedule FTS included in the Transporter's tariff.

PEF's Response: Please see PEF's response to Request No. 2F above and the Tariff attached hereto as **Exhibit B**. Section 3 of Rate Schedule FTS is contained in Original Sheet Nos. 9, 10, and 11.

Sincerely,

Kenneth A. Noffman Co-Counsel for Progress Energy Florida, Inc.

KAH/rl

cc: John Burnett, Esq., Co-Counsel for Progress Energy Florida, Inc.

progressenergy\sesh\responsetofirstdata.wpd

IMPACT OF THE 2005 HURRICANES

on the Natural Gas Industry IN THE GULF OF MEXICO REGION



PATHS OF HURRICANE KATRINA AND RITA



About This Report

This report was prepared by the U.S. Department of Energy's Office of Fossil Energy with support from the Energy Information Administration's Office of Oil and Gas. The Office of Fossil Energy supports technology research and policy options to ensure clean, reliable and affordable supplies of oil and natural gas for American consumers working closely with the National Energy Technology Laboratory, which is the Department's lead center for the research and development of advanced fossil energy technologies. The Energy Information Administration (EIA) is the independent statistical and analytical agency within the Department of Energy.

Design and Editorial Support

Science Applications International Corporation (SAIC)

Select photos and graphics courtesy of: Enbridge Offshore Pipelines, LLC; Enterprise Products Partners, LP; Excelerate Energy, LLC; Panhandle Energy Company; and Technology & Management Service, Inc. (TMS)

EXECUTIVE SUMMARY

This report summarizes the findings of the Department of Energy's (DOE) monitoring of the impact of hurricanes Katrina and Rita on the natural gas industry in the Gulf of Mexico region from late August 2005 through early March 2006. During that time, DOE coordinated with other Federal agencies and various natural gas industry personnel to track storm recovery efforts on a daily basis and identify disrupted natural gas flows and possible bypasses.

These monitoring efforts provided further insights into the complex supply delivery operation associated with offshore natural gas production. They also highlighted the importance of accurate, timely data and identified specific data elements that would be pertinent during other supply-related emergencies. Some of these lessons learned are particularly relevant as another hurricane season is underway.

Hurricanes and tropical storms are plentiful in the Gulf of Mexico, which is a major source of U.S. natural gas. The Federal Outer Continental Shelf (OCS) in the Gulf of Mexico provides about 10 billion cubic feet (Bcf) of gas per day or 20 percent of all the natural gas produced domestically. In a 4-week period in August and September 2005, hurricanes Katrina and Rita dealt a one-two punch to natural gas industry operations in the Gulf region. All aspects of the industry were affected, with the storms causing destruction and substantial damage to offshore production platforms and pipelines as well as onshore production wells, pipelines, processing plants, and other infrastructure supporting the Gulf production and delivery system.

The Minerals Management Service (MMS) estimated that 3,050 of the Gulf's 4,000 platforms and 22,000 of the 33,000 miles of Gulf pipelines were in the direct path of either Hurricane Katrina or Hurricane Rita. In addition, 47 major natural gas processing plants and 17 natural gas liquids fractionation sites located within the 70 counties and parishes along the Gulf Coast of Texas, Louisiana, Mississippi, and Alabama were threatened by the storms' approach. These facilities have the capacity to process 22.8 Bcf per day.

$\ensure{\ensuremath{\mathsf{Figure}}\xspace}$ $\ensuremath{\mathsf{Gulf}}\xspace$ Platforms and Paths of Hurricanes Katrina and Rita

The damage from the two storms varied, but together caused the biggest disruption to operations that the industry has ever seen. Hurricane Katrina, which was a Category 5 storm (winds greater than 155 miles per hour) when it entered the Gulf in late August 2005, destroyed 44 platforms and damaged 20 others. It also damaged at least 100 pipelines in Federal waters (based on data as of March 8, 2006), 36 of which were large-diameter lines (10 inches or larger), and resulted in the shut-in of at least 8 natural gas processing plants.

Hurricane Rita, which was a Category 4 hurricane (winds between 131 and 155 miles per hour) when it entered the Gulf just a few weeks after Katrina, destroyed 69 platforms and damaged 32 others. It also damaged at least 83 offshore pipelines, 28 of which were largediameter lines. As a consequence of both hurricanes, virtually all large natural gas processing plants in the area from Galveston Bay, Texas, through Mississippi were shut down. A total of 27 plants were affected, or nearly 75 percent of total processing capacity in the region, and operations were disrupted at several fractionators (natural gas liquids plants).

The offshore and onshore service industry supporting OCS natural gas production and deliveries was also devastated by the hurricanes. The network of work boats, crews, divers, supplies, and equipment needed to assess the damage and perform repairs to platforms and pipelines was shredded. In addition, docks and fleets were destroyed, electric power was lost on a wide-scale basis, and transportation fuels were not available for the boats, helicopters, and ground transportation vital to the recovery. Even the basic necessities of food, water, and shelter were not available in large areas of the hurricanes' impact zones. This damage to the support service industry hindered the natural gas industry's ability to recover from the storms.

Two liquefied natural gas (LNG) import terminals (the Panhandle Energy/Trunkline LNG terminal in Lake Charles, Louisiana, and Excelerate Energy's offshore LNG operation, Gulf Gateway Energy Bridge) were also

FINAL REPORT 2006

in the paths of hurricanes Katrina and Rita but suffered little or no damage. However, the navigation channel to the Lake Charles terminal was closed for several days after Rita because of debris in the shipping channel, and gas gathering pipelines serving Gulf Gateway Energy Bridge LNG operations were affected by the storms.

Storm recovery efforts have highlighted the complexity and inter-relatedness of the natural gas supply industry. Natural gas in the Gulf comes from wells as deep as 5 miles below the water's surface. Offshore production platforms are connected to downstream facilities by a series of gathering pipelines that transport the natural gas produced at these points to progressively larger diameter lines. The largest-diameter lines transport the consolidated natural gas stream to onshore compression stations, dehydration and separation facilities, processing plants, and eventually to transmission lines for delivery to end users.

Damage to any of these components affects the others. For example, reduced production flow meant that some operational processing plants were inactive because they had no gas to process. In other cases, production flow had to be redirected around damaged processing plants to other facilities or to other pipelines with access to processing capability. At times pipeline flow directions were reversed and temporary bypasses were utilized. Since all aspects of the Gulf natural gas delivery system – production, pipelines, and processing – are tied closely together, the recovery process for each of these three links is tied to the recovery of the other two links.

As of March 8, 2006, when DOE ended its active monitoring of storm recovery efforts, production had returned to about 9 Bcf per day, and flow rates of primary pipelines within the hurricane impact area were stabilizing within normal ranges. All but 2¹ of the 47 gas processing plants in the area were operating, although at reduced flow levels in aggregate. The remaining damage to production platforms, pipelines, and related infrastructure was expected to take several months to repair. In some cases, the decision to repair or how to repair had not been made, or the priority to proceed was low, based on economic considerations. Damage assessments and information about recovery efforts throughout the whole process, from production to pipelines to processing plants, were often difficult to obtain by DOE staff. This was because of concerns about competitive advantage, limiting the responses to requests for company-specific information, as well as the lack of information about the extent and effects of the damages given the enormous impacts of the hurricanes.

The MMS released almost daily reports of evacuation and production shut-in statistics for the platforms in the Federal OCS from late August until mid December and then weekly reports through the end of February, followed by biweekly reports until May 3, 2006, and two additional reports on June 1 and June 21, 2006. DOE combined the MMS production data with data on pipeline and processing plant damages and repairs to provide daily reports on the status of the whole production and delivery process. These daily reports also included input from industry contacts and web-based pipeline informational postings.

It is noteworthy that as late as early May, when MMS issued its final damage assessment report, operators were still revising assessments and evaluating the economics of repair, abandonment, or expansion. On May 1, 2006, MMS reported that, based on additional industry investigations and reports, more than twice the number of pipelines were damaged than had been identified in January (457 vs. 183), and the number of damaged primary lines was revised from 64 to 101. MMS also reported that shut-in gas production was almost 1.3 Bcf per day as of May 3, 2006, or about 13 percent of daily gas production in the Federal waters of the Gulf.

In its final shut-in statistics report on June 21, 2006, MMS estimated that about 9.4 percent of daily gas production remained shut in as of June 19, 2006. Cumulative shut-in gas production from August 26, 2005, through June 19, 2006, was 803.6 Bcf, which is equivalent to about 22 percent of yearly gas production in the Federal OCS in the Gulf.

THE ABCS OF NATURAL GAS SUPPLY IN THE GULF REGION

Storm recovery efforts after hurricanes Katrina and Rita have highlighted the complexity and interrelatedness of the natural gas supply industry in the Gulf of Mexico region, as well as its flexibility. A highly complex infrastructure is required for producing natural gas offshore, gathering and moving the gas onshore, processing the gas and separating natural gas liquids, and delivering pipeline-quality gas to the long-distance transmission pipelines that transport natural gas to markets throughout the United States. Compressor stations along the lines maintain the pressure to keep the gas flowing, and storage facilities feed and receive gas for system balancing and supply. The transmission lines serve a number of distribution company main lines that branch into yet smaller lines that serve residential, industrial and commercial customers. The Gulf of Mexico region also has two liquefied natural gas (LNG) import terminals, which play an important role in diversifying and expanding natural gas supplies. The LNG is transported by ship, stored at low temperatures and atmospheric pressure in super-insulated tanks, and then regasified and fed into the natural gas pipeline system.

Each element of this complex, integrated system is dependent upon adjacent elements, so damage to one



Floure 2: Gulf of Mexico Natural Gas Supply Schematic

FINAL REPORT 2006

portion of the system can affect the entire chain. The system is generally capable of dealing with short-term interruptions or disturbances, but the major disruptions in the aftermath of hurricanes Katrina and Rita required extensive industry coordination, particularly since support services were overloaded. Producers, pipelines, and processors had to work together to optimize repair efforts and move as much natural gas to market as possible. Understanding the basics of the natural gas supply process can help put these recovery efforts in context.

Production: The Gulf of Mexico region produces about 20 percent of U.S. natural gas

Oil and natural gas in the Gulf come from wells as deep as 5 miles below the water's surface. These wells flow to production platforms where the primary separation of gas, oil, and water is completed. The type of production platform depends on the water depth of the producing area.

Platforms in shallow water (less than 1,000 feet) are generally fixed platforms, compliant towers, or tensionleg platforms that are attached to the seafloor. Platforms in deep water generally involve floating structures that can receive production from subsea wells. Deepwater platforms far offshore typically pipe gas and oil to a "hub platform" in shallower water before sending it to shore.

Typically 5 to 20 wells are drilled from production platforms, and there are about 4,000 platforms in the Gulf of Mexico. High-pressure gas (1,100 pounds per square inch) may go directly into a pipeline, while lowpressure gas must be compressed at a central platform.

Gathering. The Gulf Region has about 33,000 miles of gathering pipelines

A network of pipelines is used to collect the natural gas from multiple platforms in multiple fields and deliver it onshore through a process called gathering. The gathering system may include additional platforms, metering stations, compressors, and processing equipment. Gathering systems can serve multiple fields and multiple producers. A series of gathering pipelines transports the natural gas produced at the various points to progressively larger diameter lines.



Figure 3. Components of Shallow Water Gas Production Platforms



Figure 4. Gathering Systems Collect Gas from Multiple Fields

The largest-diameter lines transport the consolidated natural gas stream to onshore compression stations, dehydration and separation facilities, processing plants, and eventually to transmission lines for delivery to end users.



Figure 5. Enbridge: One of Several Systems

The Federal Energy Regulatory Commission (FERC) has jurisdiction over interstate transmission lines but not over gathering lines. In some cases, however, gathering systems may connect to transmission pipelines while still offshore. These offshore transmission lines provide distinct corridors for moving gas from multiple gathering systems.

Processing: The Gulf Coast area has more than 45 major natural gas processing plants and 17 fractionators The consolidated gas stream from the Gulf moves to natural gas processing plants located onshore in the Gulf of Mexico coastal region, mainly in Louisiana and Texas. These natural gas plants process the gas to remove water, contaminants, and hydrocarbon liquids for the primary purpose of preparing the gas and liquids for end use by the consumer.



Plance 6. Enterprise Natural Gas Processing Plant

用品牌

Some plants process gas from wells in state waters, marshlands, and onshore gas fields, as well as the gas from offshore fields. Gas from multiple gas gathering systems is dehydrated, "sweetened" (removal of hydrogen sulfide gas and carbon dioxide), and processed to remove heavier hydrocarbons such as ethane, propane, butane, and pentanes. These "natural gas liquids" (NGLs) are sent to a fractionator (by pipeline or truck), and the clean dry gas, which is now mostly methane, is metered, compressed, and sent to a transmission line.

The fractionation facilities' products are shipped to regional petroleum and petrochemical industry customers by pipeline, truck, and barge. The petroleum refining process also produces NGLs, and refineries often have fractionators, which may also process NGLs from gas plants.



Figure 7. Promix Fractional Plant

Transmission, Storage couldebt behave

8

Major transmission lines link the Gulf Coast with points north and east and local distribution systems Larger diameter pipelines are designed to transport volumes of natural gas from the Gulf of Mexico to markets in the Northeast, Midwest, and points inbetween. A primary component of transmission lines are compressor stations that initially pressurize the gas feeding the lines and then recompress the gas at various points along the lines to maintain the line pressure and flow rate. Gas storage facilities, found in both producing and consuming regions, feed the transmission system. Most storage in the Gulf region is in caverns formed in subsurface salt formations by dissolving with water. Market hubs have developed where multiple interstate pipelines meet, allowing for the efficient metering, storage, and trading of gas volumes among producers, pipelines, marketers, and customers.

The transmission lines also serve a number of distribution company main lines that branch into yet smaller lines which serve residential, industrial, and commercial customers. These distribution pipelines are regulated by state public utility commissions.

Figure 8. Major Interstate Gas Pipelines Connect the Gulf Coast to Northeast and Midwest Markets



FINAL REPORT 2006



Figure 9 Gulf Platforms Damaged or Destroyed by Hurricane Katrina or Hurricane Rita

PRODUCTION

The Gulf of Mexico (Gulf) region is an important source of U.S. energy supply.² Before hurricanes Katrina and Rita, the Federal Outer Continental Shelf (OCS) in the Gulf provided about 27 percent (1.5 million barrels per day) of all the oil and 20 percent (10 billion cubic feet (Bcf) per day) of all the natural gas produced domestically.

Hurricanes Katrina and Rita cut right through the major oil and natural gas production operations in the Gulf. As noted earlier, an estimated 3,050 of the Gulf's 4,000 platforms were in the direct path of either Hurricane Katrina or Hurricane Rita. Virtually all production was shut-in in anticipation of the storms, and employees were evacuated from platforms and working rigs. According to the Minerals Management Service (MMS), more than 90 percent of the manned platforms and 85 percent of the working rigs were evacuated at one time.³

Damage Assessment

Because of the large amount of infrastructure in the storms' paths, damages from each of the hurricanes were extensive. Combined, the damages from the two hurricanes were far worse than anything previously experienced by the petroleum industry in the Gulf of Mexico. Hurricane Katrina, which was a Category 5 hurricane when it entered the Gulf, destroyed 44 platforms and damaged 20 others, while Hurricane Rita, which was a Category 4 storm when it entered the Gulf, destroyed 69 platforms and damaged 32 others (Figure 9). In comparison, only 7 platforms were destroyed when Hurricane Ivan hit the Gulf in 2004 as a Category 4 storm, because Ivan's path bypassed the major production infrastructure.





Figure 16. Federal Offshore Gulf of Mexico Natural Gas Production Recovery after the 2005 Hurricanes

Almost 80 percent of the destroyed platforms were older "end of life" platforms not built to MMS 1988 design standards. MMS does not expect these platforms, which represented less than 1 percent of the Gulf's gas production, to be repaired.⁴ Repairs on the remaining destroyed platforms were expected to be completed no earlier than the summer of 2006. The same schedule applies to the platforms that sustained structural damage or required engineering services before repairs could begin.

Recovery

Natural gas production in the Gulf declined from an average of approximately 10 Bcf per day to less than 2 Bcf per day after Katrina made landfall on August 29, 2005 (Figure 10). This low point in Gulf gas production was a result of shutting down all the production platforms threatened by Katrina. Repairs were undertaken as soon as possible after the storm, and gas production returned to approximately 6.5 Bcf per day by mid-September. Then Hurricane Rita hit on September 24, 2005, further damaging some of the platforms recovering from Katrina and causing new damage in other parts of the Gulf. After Rita struck, Gulf gas production was reduced to about 2 Bcf per day.

The industry's initial recovery after the second of the two hurricanes, Rita, was slower than what it experienced after Katrina. While with Katrina, Gulf gas production rebounded to 6.5 Bcf per day within 3 weeks, after Rita, Gulf gas production did not reach 6.5 Bcf per day for almost 2 months.

Gulf gas production returned to 8 Bcf per day by the end of December 2005, and had reached about 9 Bcf per day as of March 8, 2006, as more difficult repairs and longerterm solutions were required. The remaining damage to production platforms and infrastructure will likely take several months to repair. MMS reported on May 1, 2006, that it has approved four replacement platforms that were proposed by operators to take the place of eight destroyed platforms. MMS also reported that shut-in gas production was less than 1 Bcf per day as of June 19, 2006, which is equivalent to about 9 percent of daily gas production in the Gulf.

III PIPELINE Infrastructure

The pipeline system within the Gulf of Mexico (Gulf) comprises approximately 33,000 miles of pipelines that link the estimated 4,000 operating platforms to onshore elements. The hurricanes caused damage to and shut in most pipelines within more than 50 miles on either side of their respective paths. The pipeline system is composed of surface-level piping, valves, metering points, compressors, and dehydration and separation facilities, as well as sub-sea piping and valves, all working in harmony for the sole purpose of keeping natural gas flowing. Secondary lines (typically less than 20 inches) feed the larger diameter primary lines (20 to 36 inches in diameter) that transport the natural gas directly to points on shore.

Damage Assessment

The MMS estimated that 22,000 miles of pipeline were in the direct path of the two hurricanes. These pipelines transported approximately 67 percent of the natural gas produced in the Gulf. The path maps of Rita and Katrina indicate that the storms impacted the coast line at the west and east borders of Louisiana, respectively. If one were to draw a line 50 miles west of Rita's path and another 50 miles east of Katrina's path, the area bounded by these two lines would represent the area with the majority of the storm damage (Figure 11).

Approximately three dozen primary pipelines transport natural gas from the Gulf to the shorelines of Louisiana, Texas, Mississippi, and Alabama. Two-thirds of these primary lines enter Louisiana and thus were within the hurricane impact area. Most primary lines in the hurricane impact area experienced damage from the storms that either restricted or completely halted operations.

Most of the secondary lines were also located within the hurricane impact area and experienced similar levels of damage to those of the primary lines. At least two secondary lines in the Mobile Bay Area, located approximately 25 miles outside of the hurricane impact zone, experienced damage.



Figure 11. Hurricane Impact Area

The dozen or so primary lines located outside the hurricane impact area transport gas to Mississippi, Alabama, and Texas. Most of these pipelines had minimal damage from the storms and were fully operational when the platforms and processing plants in these areas resumed service.

The damage from the two storms to the pipelines varied in pattern, but there was a degree of similarity. In general, approximately half of the pipeline breaches occurred within an area that also experienced damaged or destroyed platforms.

All but a half dozen of the pipeline breaches occurred in the waters of the continental shelf (i.e., in water depths of 200 feet or less) with half of the continental shelf breaches located within 25 miles of the transition from deep water. As the continental shelf is closer to the shoreline in the eastern part of the hurricane impact area, approximately half of the breaches occurred within 25 miles of the shoreline in South Timberline and Main Pass areas or in the waters surrounding the Plaquemines Peninsula and Lafourche Parish.

Katrina caused almost double the number of natural gas pipeline breaches caused by Rita. Katrina's breaches, being in the eastern portion of the damage area, generally occurred closer to the shore, while the pipeline breaches caused by Rita were more randomly distributed along the path of the hurricane.

Recovery

Under normal conditions, primary and secondary pipelines flow at 50 to 60 percent of design capacity; however, the actual utilization of the line capacity can vary significantly, depending on circumstances. During the recovery period from the hurricanes, many lines operated at or beyond their design capacity, while others had low flow rates for periods lasting from several days to more than a week. Several lines had such low flow rates that they were shut down.

In spite of the operational problems, natural gas continued to flow because of the many interconnections that link the pipelines together. Companies used this interconnectivity to flow natural gas around nonoperational damaged pipeline sections and pathways, to bypass non-operational plants or compressor stations, and to reroute flow to onshore operating facilities. This was accomplished through a great deal of teamwork and creativity such as reversing flow direction and utilizing bypasses. The interconnectivity of primary pipelines, which was the most important to maintaining gas flow, is greater onshore and declines as one moves away from shore. Onshore interconnects are typically less likely to suffer failure because of hurricane events, and repairs are not as difficult to make as those to offshore pipelines.

Even with the pipeline network flexibility and the cooperative efforts by operators, it was not physically possible or economical to reroute all available Gulf gas production. As a result, the damage sustained by the pipeline network materially reduced production from the Gulf.

Flow rates of primary lines located within the hurricane impact area were stabilizing within normal ranges by early March 2006. At that time, operators were still considering how or whether to repair the remaining damaged facilities. Most of the remaining damage was located near the outer fringes of the lines or at end segments that connect to platforms with low production volumes.

In some cases parties have decided, because of economics, not to repair shared assets (e.g., platforms) that were part of the pipeline infrastructure. This sometimes resulted in a work around, such as laying pipe on the sea floor, to avoid a more costly investment. The return on pipeline investment is recovered through gas flow rather than gas price. Therefore the depletion rates of producing areas are critical factors in the decision to repair, abandon, or expand facilities. Repairs to the secondary pipelines, which affect small portions of overall scheduled capacity for a pipeline, were prioritized based on the availability of work crews and economics (see box, "Factors Affecting Recovery Efforts").

In tracking recovery efforts on a line-by-line basis, DOE considered two events to be significant: (1) a pipeline's return to partial flow, and (2) when a pipeline reached and held a consistent flow rate. DOE also closely monitored a number of factors that could restrict the pipeline from operating at normal capacities and attempted to measure their effects.

Just after the hurricanes, the pipelines had the capability to transport approximately 4.5 Bcf per day, or enough flow capacity to handle all available production from the Gulf at that date. The capacity to flow was maintained by several pipelines, most of which were located outside of the hurricane impact area. Had there been more production available, the pipelines located within the hurricane impact area most likely would not have been able to handle the volume or reroute the flow to onshore facilities. The pipelines at this time were at least a partial constraint in the production and supply chain. By the end of October, many of the primary lines in the hurricane impact area were well on the way to recovery or fully operational. Recovery continued to progress through December, and by January, the recovery had reached a plateau.

As of March 8, 2006, when DOE discontinued its active monitoring of repair efforts, there were still several primary pipelines with damage to sections that cut off production. There was also damage remaining to some secondary pipelines and outlying laterals from both primary and secondary lines. Final repairs to these lines most likely will take several months to complete. Competition is extensive for the crews and equipment needed for making repairs and for extending lines to new producing areas.

Factors Affecting Recovery Efforts

Several factors have affected the return to service of various line segments. The depth of water in which the damage resides, the length of the effected zone, size of the pipeline, and the complexity of the zone (e.g., interconnects, sea valves) are all factors in determining the time, complexity, and costs required to repair or replace damaged areas. There are a limited number of crews that can make repairs in deep water. Initially after the hurricanes, many of these crews were under contract and committed to specific companies, thus affecting repair scheduling. Also, producers sometimes outbid pipeline operators for the services of these specialized crews. Other factors affecting repair schedules included those instances when damages turned out to be more extensive than original assessments revealed, which lengthened the time required to complete repairs. Rough seas at times also contributed to extending the required repair time.

Long lead times required for delivery of repair materials and components also hindered repair efforts. Many components such as valves and flanges must be custom-built or are in short supply. Destroyed or damaged platforms also presented major problems, as connections needed to be isolated and lines dewatered. Pipeline breaches create the need to dewater the line and remove contaminants before operation is restarted. The only facilities for treating large volumes of water from the pipeline are located onshore. This means that the sea water must be routed through the pipe to shore without damaging other sections of the pipeline during the transporting process.

As late as early May, pipeline operators were still revising assessments of damage to some outer-lying pipelines. These assessments, for the most part now complete, have revealed that several pipelines experienced more extensive damage than the original assessments indicated, and many more pipelines were damaged than first reported. On May 1, 2006, MMS reported that, based on additional industry investigations and reports, more than twice the number of pipelines were damaged than had been identified in January (457 vs. 183), and the number of damaged primary lines was upped from 64 to 101.⁵ MMS reported that 32 of the damaged primary lines had returned to service.

W. Natural Gas Processing

The natural gas product fed into the mainline gas transportation system in the United States must meet specific quality measures for the pipeline grid to operate properly. Consequently, natural gas produced at the wellhead, which in most cases contains contaminants⁶ and natural gas liquids,⁷ must be processed, i.e., cleaned, before it can be safely delivered to the high-pressure, long-distance pipelines that transport the product to the consuming public. Natural gas that is not within certain specific gravities, pressures, Btu content range, or water content levels will cause operational problems, pipeline deterioration, or can even cause pipeline rupture.

The processing of wellhead natural gas into pipelinequality dry natural gas can be quite complex and usually involves several processes to remove: (1) oil; (2) water; (3) elements such as sulfur, helium, and carbon dioxide; and (4) natural gas liquids. In addition to those four processes, it is often necessary to install scrubbers and heaters at or near the wellhead. The scrubbers serve primarily to remove sand and other large-particle impurities. The heaters ensure that the temperature of the natural gas does not drop too low and form a hydrate with the water vapor content of the gas stream. These natural gas hydrates are crystalline ice-like solids or semisolids that can impede the passage of natural gas through valves and pipes.

Within the 70 counties and parishes located along the Gulf Coast States of Texas, Louisiana, Mississippi, and Alabama, there were 47 major natural gas processing plants (each with a processing capability of 100 million cubic feet (MMcf) per day or greater) when the hurricanes struck. Total processing capacity for the major plants was 22,841 MMcf per day. Of the 47 plants,





9 also included on-site fractionation facilities, that is, they had the capability to extract individual natural gas liquid components including ethane, propane, butane, and pentane, which are sold as separate commodities.

The other 38 natural gas processing plants traditionally confine their operations primarily to extracting pipelinequality natural gas (residue gas), and secondarily the production of a natural gas liquids mixture stream (often referred to as Y-grade) that is transported to a separate facility for fractionation via rail, truck, barge, or pipeline, mostly the latter.

In the area surveyed, there are at least 17 fractionation sites, the largest being the Mont Belvieu complex, located in southeast Texas about 60 miles from the Louisiana border. More than a dozen pipelines in the Texas-Louisiana area transport Y-grade liquids (a mixed heavy hydrocarbons liquid stream under pressure) to the Mont Belvieu area for fractionation. Conversely, multiple types of transportation infrastructure (pipeline, truck, rail, or barge) are available to move finished NGL products and refinery grade propylene to area markets as feedstock for chemical, refining, and industrial markets. Mont Belvieu and most of the other local fractionation facilities maintain, or have access to, salt cavern facilities which are used for storing unprocessed mixed liquids and individual extracted NGL products.

Matural Sas Processing Piencs

Damage Assessment

Although the processing/treatment segment of the natural gas industry generally receives little public attention, its overall importance to the natural gas industry became readily apparent in the aftermath of hurricanes Katrina and Rita in August and September 2005. Damage caused by the hurricanes resulted in a number of shut-ins to gas processing plants.

FINAL REPORT 2006

Causes of the shut-ins varied—based on either internal or external conditions. Internal conditions refer to damage directly affecting the gas processing plants, including flooding, debris, and destruction of equipment. External conditions refer to closures caused by lack of electricity, inaccessibility of the plant site because of road damage or other problems, lack of upstream supplies to the processing plant caused by production shut-ins or pipeline problems, and downstream problems related to the disposal of natural gas liquids or Y-grade liquids.

Downstream conditions are critical to an operating gas processing plant whether the flow from the gas processing plant is raw gas for processing at a fractionator (i.e., the Y-grade liquids) or processed liquids that are ready to be marketed. If the typically large volumes of co-products cannot be properly disposed of, the gas processing plant cannot operate. The liquids problems were caused either by problems affecting the pipelines exiting the gas processing plant or at the downstream facility itself. After Hurricane Katrina made landfall in late August, there were at least eight gas processing plants known to have been closed down.⁸ These eight plants represented 6,615 MMcf per day of capacity that had a pre-hurricane flow of 4,158 MMcf per day. When Hurricane Rita struck the Gulf producing areas less than a month later, the cumulative damage from both storms was much greater.

At the end of September 2005, 27 gas processing plants with 16,796 MMcf per day of capacity were shut in, or almost 75 percent of total capacity for all major plants in the region (Figure 12).⁹ The 27 shut-in gas processing plants included virtually all the large plants in the area from Galveston Bay, Texas, through Mississippi (Figure 13). Only 20 of the major plants, with a capacity of 6,045 MMcf per day, were active at the end of September.

Eleven of the twenty-seven plants were inactive because of internal factors, which forced the shut-in of 7,665 Bcf per day of capacity. Most of the facilities with internal



Figure 10. Gulf Processing Plants Affected by Hurricane Katrina or Hurricane Rita

problems appear to have endured some amount of flooding from either storm surge or rain penetration. The reports on water damage did not necessarily include a clear designation regarding the nature of the problem. A number of facilities were flooded so badly that inspections were not conducted or completed for weeks. The major internal problems were debris and damage to equipment, especially controls and electrical equipment. There were few reported problems involving damage to moving equipment or structures.

In addition to the shut-in capacity as a result of internal problems, an even larger amount of capacity was offline although the plants themselves were reported as operational. At the end of September 2005, there were 16 gas processing plants that were reported as operational but were not active because of problems outside the plant. Causes for these 16 shut-ins included: lack of power (6 plants), upstream supplies being unavailable (10 plants), problems with disposal of co-products downstream (6 plants), and one case of bypass because of market conditions.¹⁰

Recovery

As a group, the plants affected by external problems were reactivated most quickly, with roughly two-thirds of those being back on-line by the end of October 2005. The plants with internal problems required the longest recovery periods. Only one plant with internal problems became active again during October, and even at the end of December, 7 of the original 11 remained inactive.

Recovery during October 2005 was the most active of any month after the hurricanes in terms of the number of plants returned to service, as the number of inactive plants declined from 27 at the end of September to 15 by the end of October. At the end of October, 8,335 MMcf per day of capacity was inactive, with 14,506 MMcf per day active (Figure 12).

Almost half the remaining inactive gas processing plants returned to operation during November, as the count fell from 15 to 8 by the end of the month. Seven gas processing plants were inactive because of internal problems. By the end of November, there were 39 active plants with capacity of 17,366 MMcf per day and inactive capacity of 5,475 MMcf per day. The situation was unchanged at the end of the calendar year as no additional plants were activated during December, although flows through the active plants did increase.

In January, four more plants with a capacity of 2,425 MMcf per day were restored back to active status. By the end of January, four gas processing plants were still inactive. Included in the group of four was BP's Grand Chenier plant with capacity of 950 MMcf per day, which was being decommissioned as announced in late January. As of March 8, 2006, 45 of the 47 plants were restored to active status. The final reactivation of a major processing plant occurred on April 2, 2006, when the Stingray plant resumed processing, bringing capacity of all major active plants to 21,891 MMcf per day.

Natural Gas Fractionarcrs (Natural Cas Liquids Plants)

Damage Assessment and Recovery

Fractionators in Texas

While the path of Hurricane Katrina did not take it anywhere near the Mont Belvieu complex in east Texas, this area was within the edge of Hurricane Rita's path. Most of the fractionation, storage, and pipeline facilities in the area reported no sustainable major damage. However, the absence of water, which could not be pumped to three of the plants because the local electric power grid was down, did force temporary closures. (All plants in the Mont Belvieu area went into a planned shutdown mode prior to the expected hurricane landfall.)

One fractionator, which had its own on-site well water source, was able to return to operations within a day or so. The other three facilities at Mont Belvieu did not reopen for about 8 days. Some flows were directed to area storage facilities that were still operational.

There was an indirect impact from the storm on parts of the area's pipeline network. Although a number of pipelines suffered no major damage themselves, a lack of electric utility service and/or damage to power facilities at pump stations along the pipeline system caused a temporary shutdown of operations.

Other fractionators are located in the Beaumont area of east Texas (Jefferson County), where certain facilities
FINAL REPORT 2006

were significantly impacted by Hurricane Rita. Response to the damage to the fractionators varied. In mid-October, Y-grade deliveries from some gas processing plants in the area were redirected to Mont Belvieu for fractionation or into temporary storage. In at least one case where both gas processing and refinery facilities were damaged, recovery of refinery operations was given priority. Consequently, the repairs to the fractionator facility were not completed until late November. Today all fractionation and refinery operations are back to normal.

Fractionators in Louisiana

The hurricanes caused numerous difficulties for several fractionators in southern Louisiana. A number of fractionators suspended operations at least temporarily because of up- or down-stream problems. Upstream problems included shut-in offshore production, pipeline damage, or loss of supply flow because of inactive gas processing plants.

Fractionators also were affected if the downstream pipelines or storage were unavailable or customers were unable to receive the NGLs. These difficulties then might result in other bottlenecks. For example, in the aftermath of the hurricanes, operations on certain pipelines were disrupted at least briefly, with the result that natural gas processing plants that used these lines for transportation of their raw NGL mixture had to cease gas processing operations because of a lack of liquids storage facilities.

V. LNG

Liquefied natural gas (LNG) plays an important role in diversifying and expanding natural gas supplies. LNG arriving in the continental United States enters through one of five LNG receiving and regasification terminals located along the Atlantic and Gulf coasts. The United States currently has six LNG terminals that receive, store, and regasify LNG—four on the mainland, one in the offshore Gulf of Mexico, and one in Puerto Rico.

Damage Assessment

The Panhandle Energy/Trunkline LNG terminal in Lake Charles, Louisiana, and the Excelerate Energy sub-sea

Gulf Gateway Energy Bridge 116 miles off the Louisiana coast were in the paths of hurricanes Katrina and Rita. Assessments indicate the hurricanes had only minor impacts on these two LNG import terminals. The Lake Charles terminal has the capacity to import 1.5 billion cubic feet (Bcf) of gas per day and received 103.8 Bcf in 2005. The Gulf Gateway Energy Bridge facility has the capability to import 0.5 Bcf per day.

The Lake Charles terminal is connected to the Gulf of Mexico by a 48-mile (80 km) ship channel. The channel is dredged to a depth of 40 feet (12 m) and is 400 feet (120 m) wide with no overhead navigational obstructions. The facility is connected to the mainline transmission system of Trunkline Gas Company, LLC by 45 miles (72 km) of 30-inch diameter pipeline with a capacity of 1.2 Bcf per day (9.1 mmtpa).



Figure 14 Lake Charles LNG Terminal and Surrounding Pipelines

From September 26 to October 3, 2005, the Lake Charles navigation channel and turning basin were closed because of debris in the waterway and dockage area that required removal. In addition there was limited operational capability to transfer LNG from tanker to storage because of the lack of commercial electric power.



Figure 15. Gulf Gateway Energy Bridge and Surrounding Pipelines

On October 3, 2005, the navigational channel and turning basin reopened to deep draft marine traffic for daylight transits. However, the Trunkline LNG terminal was operating on backup generator power thereby limiting cargo transfers and regasification rates below normal operating inventories. The terminal's LNG inventory increased to 70 percent of capacity (4.4 Bcf) upon completion of cargo transfer from LNG tank ships. On October 5, 2005, the Trunkline LNG terminal at Lake Charles was operational at the pre-hurricane normal levels. Full commercial power was restored. Full regasification and send-out capability and LNG cargo transfers were completed in evening hours.

LNG operations at Excelerate Energy's Gulf Gateway Energy Bridge deepwater port were also affected by the paths of both Katrina and Rita. However, the initial diving assessment results indicated little or no damage was done to the facility. Gas gathering pipelines Sea Robin and Blue Water, which service the Gulf Gateway terminal, were impacted by the hurricanes and the pipelines were in-operative through mid-November. The Gulf Gateway facility is operational; however no shipments had been scheduled because of the unavailability of spot market cargos.

V). INFORMATION COLLECTION

Natural Gas Production Information Collection

The primary source of natural gas production data for the Federal Outer Continental Shelf (OCS) in the Gulf of Mexico is the Department of the Interior's Minerals Management Service (MMS),¹¹ which has oversight and regulatory responsibility for petroleum operations on the Federal OCS. As part of its normal operations, MMS collects monthly production information for all natural gas wells operating on platforms in the Federal OCS. This information is stored in MMS's Technical Information Management System (TIMS). It is made available to the public on a monthly basis on the MMS website, although data are only available 2 to 3 months after the month in question.

FINAL REPORT 2006

In response to the damage inflicted by the hurricanes, MMS conducted an emergency data collection effort to provide more detailed and real-time information on the impacts of the hurricanes on Federal OCS Gulf production. Here, MMS used TIMS data plus direct contact with operators of the natural gas wells and platforms to assess: numbers of platforms damaged or destroyed, shut-in production, pipeline segments that have facility measuring points (FMPs), and daily volumes at FMPs.¹² With this emergency data collection effort MMS was able to issue a daily report on how much natural gas production was shut in for the Gulf in aggregate.

Natural Gas Pipeline Information Collection

The Federal Energy Regulatory Commission (FERC) requires that natural gas interstate pipeline companies provide complete and timely information about available and released transportation capacity on user-friendly, Internet accessible informational listings that are accessible by all customers on an equal basis. These webbased electronic information profiles provide a wealth of information, including the Daily Scheduled Capacity and Available Capacity on a receipt and delivery point basis listed as Operationally Available by Segment or by Location. The listings also contain Total Design Capacity for each point.

Natural Gas Processing Plant Information Collection

Information on a limited number of data items is sufficient to represent the condition of the processing plant segment of the industry. There is no public source of information regarding plant status and operations that is available on a timely basis. After the hurricanes, information on the larger processing plants was provided by industry sources.

19

VII. Endnotes

- 1. Most of the information in this report is compiled through March 8, 2006, with few exceptions. A key exception concerns the final reactivation of gas processing plants which occurred with the startup of the Stingray plant on April 2, 2006, bringing the count of active plants to 46. As noted in the section on gas processing plants, the Grand Chenier plant was decommissioned so it will not be reactivated.
- 2. The Gulf of Mexico region in this section refers to the Federal Outer Continental Shelf (OCS) in the Gulf of Mexico, which is under the jurisdiction of the U.S. Department of the Interior, Minerals Management Service. The Federal Government administers the submerged lands, subsoil, and seabed, lying between the seaward extent of the States' jurisdiction and the seaward extent of Federal jurisdiction. State jurisdiction extends 3 nautical miles (3.3 miles) off the coastline, excepting Louisiana where it extends 3 imperial nautical miles (3.45 miles), and Texas and the west coast of Florida where jurisdiction extends 9 nautical miles.

Federal jurisdiction is defined under accepted principles of international law. The seaward limit is defined as the farthest of 200 nautical miles seaward of the baseline from which the breadth of the territorial sea is measured or, if the continental shelf can be shown to exceed 200 nautical miles, a distance not greater than a line 100 nautical miles from the 2,500-meter isobath or a line 350 nautical miles from the baseline. Outer Continental Shelf limits greater than 200 nautical miles but less than either the 2,500 meter isobath plus 100 nautical miles or 350 nautical miles are defined by a line 60 nautical miles seaward of the foot of the continental slope or by a line seaward of the foot of the continental slope connecting points where the sediment thickness divided by the distance to the foot of the slope equals 0.01, whichever is farthest.

 Minerals Management Service, Press Release: "Impact Assessment of Offshore Facilities from Hurricanes Katrina and Rita," January 19, 2006.

- Minerals Management Service, Press Release: "Office of the Secretary, Interior Secretary Gale Norton Reports on Gulf of Mexico Energy Status," October 4, 2005.
- Minerals Management Service, Press Release: "MMS Updates Hurricanes Katrina and Rita Damage," May 1, 2006.
- Includes non-hydrocarbon gases such as water vapor, carbon dioxide, hydrogen sulfide, nitrogen, oxygen, and helium.
- Ethane, propane, and butane are the primary heavy hydrocarbons (liquids) extracted at a natural gas processing plant, but other petroleum gases, such as isobutane, pentanes, and normal gasoline, also may be processed.
- 8. This information is as of September 19, 2005. It is based on reports by the Minerals Management Service, staff reports from the DOE Office of Fossil Energy, data compiled by the Energy Information Administration, and trade press reports.
- 9. Many of the gas processing plants ceased operations in advance of the hurricanes for safety reasons. Those plants that were unaffected by the storm opened promptly thereafter.
- 10. Some plants were confronted with more than one problem, so the sum of the plants exceeds the total.
- Department of the Interior, Minerals Management Service (MMS), Minerals Revenue Management, Royalty-In-Kind Program, and Offshore Mineral Management, Production and Development, Structural & Technical Support, Surface Commingling & Production Measurement, and Pipeline Section.
- 12. When events occur in the Gulf of Mexico region, MMS Continuity of Operations Plan (COOP) requires natural gas companies to report evacuated rigs and platforms and shut-in oil (BOPD) and natural gas (MMcf/d) volumes. The data are required to be reported by 11:30 CST and a daily report, aggregated by districts, is posted on the MMS website (www.mms.gov) at 13:00 CST.

20 соявляется сталя



SOUTHEAST SUPPLY HEADER PROJECT

Docket No. CP07-____000

EXHIBIT P

SESH Rate Derivation and Pro Forma Tariff

	EXH	BIT	
oles.	A		
tabl	15		

Docket No. CP07-____ Exhibit P

Rates and Tariff

Table of Contents

<u>Schedule</u>

Explanatory Notes	1
Total Cost of Service and Rate Design	2
Operation and Maintenance Expenses	3
Depreciation Expense and Other Taxes	4
Rate Base and Return	5
Federal and State Income Taxes	6
Rate of Return	7
Tariff	8

Docket No. CP07-____ Exhibit P Schedule 1 Page 1 of 3

Explanatory Notes

Rate Derivation

SESH has calculated its maximum rates for FTS firm transportation service as shown on Schedule 2 of this Exhibit P. SESH's maximum rates were developed, using the Commission's SFV methodology.

The cost of service which SESH used for developing the FTS firm reservation rate was its total fixed cost of service, less \$1,500,000 which was assigned to interruptible service. The resulting FTS reservation cost of service was divided by a volume determinant equal to SESH's total capacity of 1,032,951 dth per day, times 12 months, as shown on lines 13–17 of Schedule 2.

The usage rate for FTS service was developed by dividing SESH's variable cost of service by a design determinate equal to SESH's total capacity of 1,032,951 dth per day, times 365 days, times a 70% load factor. This calculation is shown on lines 18-21 of Schedule 2.

The FTS usage-2 rate was set equal to the 100% load factor rate for firm FTS service.

The usage rates for interruptible ITS and PALS services were set to equal the 100% load factor FTS rate.

Docket No. CP07-____ Exhibit P Schedule 1 Page 2 of 3

Explanatory Notes

Cost of Service

SESH's cost of service is based on its total capital cost as presented in Exhibit K herein. It proposes to use a 1.67% depreciation rate as described in Exhibit O.

SESH's proposed 10.0% rate of return was calculated on schedule 7 of this Exhibit P, based on the company's proposed 50%/50% capital structure. A debt cost of 6.5% was applied, as described in Exhibit L-1 herein.

A return on equity of 13.5% is being requested, based on SESH's level of business and financial risk. The Commission has recently approved returns on equity at the 14% level for new startup pipeline projects similar to SESH. The Gulfstream Natural Gas System, L.L.C. received authority to use a 14% ROE rate, in its initial rates, in a Commission order issued April 28, 2000 in Docket No. CP00-6-000 (91 FERC 61,119).

Income taxes have been included in SESH's cost of service, consistent with the Commission's policy statement in Docket No. PL05-005¹. Under that policy statement, partnerships are permitted recovery of income taxes in cost of service, to the extent of their ownership by tax paying corporations. Both of the SESH partners are ultimately owned by tax paying corporations. Income taxes have been calculated as shown on schedule 6, using a 35% federal income tax rate and a 6% state income tax rate.

Ad valorem taxes were developed, as shown on schedule 4, by applying separate ad valorem factors to the gross plant balances in each state. The factors were developed based on the individual assessment ratios and tax rates in each county.

An estimate of operation and maintenance was included based on an engineering estimate of direct operation and maintenance expenses for similar facilities of SESH's operating partner Duke Energy Gas Transmission, LLC. Additional administrative and general expenses have been included based on forecasted overhead costs for the partnership.

¹ Inquiry Regarding Income Tax Allowances, Docket No. PL05-5-000 (111 FERC ¶61,139)

Docket No. CP07-____ Exhibit P Schedule 1 Page 3 of 3

Explanatory Notes

SESH Tariff

Included in Exhibit P hereto is the *pro forma* FERC Gas Tariff whereby SESH will provide transportation service. The terms and conditions of the tariff are structured to conform to the requirements of the Commission's Order Nos. 636, *et seq.* and Order No. 637. The tariff includes the proposed Rate Schedule FTS, under which SESH will render firm transportation service to the SESH Customers and the proposed Rate Schedule ITS, under which SESH will render interruptible transportation service. The tariff also provides for the ability to negotiate rates.

Southeast Supply Header, LLC Total Cost of Service and Rate Design

	(1)	(2)	(3)	(4)
Line				
<u>No.</u>	Description	<u>2008</u>	<u>2009</u>	<u>2010</u>
4	Operation and Maintenance Furness	£44.000 7 04		¢40.004.540
1	Operation and Maintenance Expense	\$11,283,791	\$11,735,143	\$12,204,548
2	Teves Other then Income	\$13,025,015 #0,000,070	\$13,025,015	\$13,025,015
3	Faderal Income	\$9,920,856	\$10,123,794	\$10,330,971
4	Pederal Income Taxes	\$28,365,188	\$27,303,413	\$26,053,490
5	State Income Taxes	\$5,172,983	\$4,979,346	\$4,751,396
6	Return	<u>\$77,307,805</u>	<u>\$74,386,519</u>	<u>\$70,947,579</u>
1	Total Cost of Service	\$145,075,638	\$141,553,230	\$137,312,999
8	Cost of Service Breakdown For Rate Design			
ğ	Eixed Cost of Service For FTS	\$141 877 264		
10	Cost of Service Allocation to Interruntible	\$1,500,000		
11	Variable Cost of Service	\$1,000,000 \$1,608,374		
12	Total	\$145.075.638		
12	10(4)	φ1 4 0,070,000		
13	FTS Reservation Rate			
14	Fixed Cost of Service For FTS	\$141,877,264		
15	Capacity (dth/d)	1,032,951		
16	Design Determinant	12,395,412		
17	Reservation Rate (\$/dth)	\$11,446		
18	FTS Usage Rate			
19	Variable Cost of Service	\$1,698,374		
20	Design Determinant (In 15 x 365 x 70%)	263,918,981		
21	Usage Rate (\$/dth)	\$0.0064		
22	FTS Usage-2 Rate	\$0.3827		
23	ITS Usage Rate (\$/dth)	\$0 3827		
20		ψ0.0027		
24	PALS Usage Rate(\$/dth)	\$0.3827		

Southeast Supply Header, LLC Operation and Maintenance Expenses

	(1)	(2)	(3)	(4)
Line				
<u>No.</u>	Description	<u>2008</u>	<u>2009</u>	<u>2010</u>
1	Direct O&M			
2	Compressor Station -Labor	\$1,323,660	\$1,376,607	\$1,431,671
3	Compressor Station -M&O	\$1,698,374	\$1,766,309	\$1,836,961
4	Pipeline -Labor	\$1,463,186	\$1,521,713	\$1,582,582
5	Pipeline -M&O	\$1,877,397	\$1,952,493	\$2,030,593
6	M&R Station -Labor	\$31,851	\$33,125	\$34,450
7	M&R Station -M&O	\$40,868	<u>\$42,503</u>	<u>\$44,203</u>
8	Total Direct O&M	\$6,435,336	\$6,692,749	\$6,960,459
9	Administrative and General	<u>\$4,848,455</u>	<u>\$5,042,393</u>	<u>\$5,244,089</u>
10	Total O&M	\$11,283,791	\$11,735,143	\$12,204,548

Southeast Supply Header, LLC Depreciation Expense and Other Taxes

	(1)		(2)	(3)	(4)
Line <u>No.</u>	Description		2008	2009	<u>2010</u>
1 2 3 4	Depreciation Expense Depreciable Plant Depreciation Rate Depreciation Exp.		\$779,941,000 <u>1.67%</u> \$13,025,015	\$779,941,000 <u>1.67%</u> \$13,025,015	\$779,941,000 <u>1.67%</u> \$13,025,015
5	Taxes Other than Income:				
6 7 9	<u>Gross Plant</u> Louisiana Mississippi Alabama Total Gross Plant		\$131,487,841 \$568,029,707 <u>\$82,488,452</u> \$782,006,000	\$131,487,841 \$568,029,707 <u>\$82,488,452</u> \$782,006,000	\$131,487,841 \$568,029,707 <u>\$82,488,452</u> \$782,006,000
10 11 12 13 14	<u>Ad Valorem Taxes</u> Louisiana Mississippi Alabama Total Ad Valorem Taxes	2.37% 0.95% 1.46%	\$3,121,137 \$5,373,477 <u>\$1,200,207</u> \$9,694,821	\$3,183,560 \$5,480,946 <u>\$1,224,211</u> \$9,888,717	\$3,247,231 \$5,590,565 <u>\$1,248,695</u> \$10,086,492
15 16 17	<u>Payroll Taxes</u> Labor Cost Payroll Taxes	8.02%	<u>\$2,818,697</u> \$226,035	<u>\$2,931,445</u> \$235,076	<u>\$3,048,703</u> \$244,479
18	Total Taxes Other than Income		<u>\$9,920,856</u>	<u>\$10,123,794</u>	<u>\$10,330.971</u>

Southeast Supply Header, LLC Rate Base and Return

	(1)		(2)	(3)	(4)
Line <u>No.</u>	Description		2008	2009	<u>2010</u>
1 2 3 4	<u>Rate Base</u> Gas Plant in Service Accumulated Depreciation Net Plant		\$782,006,000 <u>(\$6,512,508)</u> \$775,493,492	\$782,006,000 (<u>\$19,537,523)</u> \$762,468,477	\$782,006,000 (<u>\$32,562,538)</u> \$749,443,462
5 6	Working Capital Materials & Supplies @	0.313%	\$2,443,970	\$2,541,729	\$2,643,398
7 8	Accum. Deferred Income Taxes Total Rate Base		<u>(\$4,859,413)</u> \$773,078,050	<u>(\$21,145,012)</u> \$743,865,194	<u>(\$42,611,066)</u> \$709,475,795
9	Return @	10.00%	<u>\$77,307,805</u>	<u>\$74,386,519</u>	<u>\$70,947,579</u>

Southeast Supply Header, LLC Federal and State Income Taxes

	(1)		(2)	(3)	(4)
Line <u>No.</u>	Description		2008	2009	<u>2010</u>
1	Return		\$77,307,805	\$74,386,519	\$70,947,579
2 3 4 5	Adjustments Interest and Debt Expense Amortization of Equity AFUDC Total Adjustments		(\$25,125,037) <u>\$495,438</u> (\$24,629,599)	(\$24,175,619) <u>\$495,438</u> (\$23,680,181)	(\$23,057,963) <u>\$495,438</u> (\$22,562,525)
6	Net Taxable Income		\$52,678,206	\$50,706,338	\$48,385,054
7	Federal Income Tax @	35.00%	\$28,365,188	\$27,303,413	\$26,053,490
8	Pre-FIT (Lines 6 and 7)		\$81,043,394	\$78,009,751	\$74,438,544
9	State Income Tax @	6.00%	\$5,172,983	\$4,979,346	\$4,751,396

Southeast Supply Header, LLC Rate of Return

Line <u>No.</u>	(1) <u>Description</u>	(2) Capitalization <u>Ratios</u>	(3) Component <u>Cost</u>	(4) Return <u>Component</u>
1	Long-Term Debt	50.00%	6.50%	3.250%
2	Equity	<u>50.00%</u>	13.50%	<u>6.750%</u>
3	Total	100.00%		10.000%

.

FERC GAS TARIFF

PRO FORMA ORIGINAL VOLUME NO. 1

OF

SOUTHEAST SUPPLY HEADER, LLC

FILED WITH

FEDERAL ENERGY REGULATORY COMMISSION

Communications Concerning the Tariff Should Be Addressed To:

5400 Westheimer Court Houston, TX 77056

Telephone Number: Facsimile Number:

Preliminary Statement	DESCRIPTION		_	Sheet No.
Negotiated Rates	Preliminary Statement . System Map Statement of Rates Capacity Release Rates. Additional Charges and S		· · · · · · · · ·	· · · 2 · · · 3 · · · 4 · · · 5
Rate Schedule:	Negotiated Rates	••••••	• • • • • • •	• • • • 7
FTS Firm Transportation Service 8 ITS Interruptible Transportation Service. 14 PALS Parking and Lending Service. 17 General Terms and Conditions Index. 23 Form of Service Agreement Index: 160	Rate Schedule:			
General Terms and Conditions Index	FTS Firm Transportation ITS Interruptible Transp PALS Parking and Lending	Service ortation Service. Service	· · · · · · · · ·	8 14 17
Form of Service Agreement Index:	General Terms and Condit	ions Index		23
FTS Service Agreement	Form of Service Agreemen	t Index:		160
	FTS Service Agreement . PALS Service Agreement. Capacity Release Umbrell LINK® Service Agreement ITS Service Agreement .		 	161 167 170 180 183

Original Sheet No. 2

PRELIMINARY STATEMENT

Southeast Supply Header, LLC ("SESH") owns and operates a natural gas pipeline company engaged in the business of transporting natural gas in interstate commerce under authorization granted by, and subject to the jurisdiction of, the Federal Energy Regulatory Commission. Its pipeline system extends in an easterly direction from the Perryville Hub, Louisiana to various points in Mississippi and Alabama terminating at an interconnection with Gulfstream Natural Gas System, L.L.C. near Coden, Alabama.

The location of SESH's system is shown on the map included herein.

Services will be provided under specific Agreements and rate schedules and SESH reserves the right to limit its Agreements for transportation of gas to Shippers acceptable to it after consideration of its existing commitments, delivery capacity, Delivery Point, credit-worthiness of Shippers, and other factors deemed pertinent by SESH, consistent with the terms and conditions of this tariff.

Nothing in this tariff is intended to inhibit the development of, or discriminate against the use of, imbalance management services provided by third parties or SESH's Shippers. Any party interested in providing imbalance management services must coordinate with SESH.

Issued by: Issued on:

Original Sheet No. 3

System Map

(A System Map reflecting the pipeline route as constructed will be set forth on this tariff sheet.)

Issued by: Issued on:

Original Sheet No. 4

٦

Ra	ate Schedule	Maximum Rate Per Dth	Minimum Rate Per Dth
RATI	E SCHEDULE FTS 1/		
1.	Reservation Rate Per Month Per Dth of MDQ	\$11.446	\$ 0.0000
2.	Usage-1 Rate Per Dth	\$ 0.0064	\$ 0.0064
3.	100% L.F. Rate	\$ 0.3827	\$ 0.0064
4.	Usage-2 Rate Per Dth	\$ 0.3827	
RAT	E SCHEDULE ITS 1/		
1. 2.	Usage-1 Rate Per Dth Usage-2 Rate Per Dth	\$ 0.3827 \$ 0.3827	\$ 0.0064
RAT	E SCHEDULE PALS 1/		
1.	Usage Rate Per Dth	\$ 0.3827	\$ 0.0064
1/	Backhaul rate is equal to the	e Forward haul m	rate.
		,	

Issued by: Issued on:

Original Sheet No. 5

R.	ate Schedule	Maximum Rate Per Dth	Minimum Rate Per Dth
MAX RAT:	IMUM DAILY CAPACITY RELEASE E 1/		
1.	Reservation Rate Per Month Per Dth of MDQ	\$ 0.3763	\$ 0.0000
2.	Usage-1 Rate per Dth	\$ 0.0064	\$ 0.0064
3.	100% L.F. Rate	\$ 0.3827	\$ 0.0064
	Backhaul fate is equal to t	ne Forward ha	ul rate.
		ne Forward ha	ul rate.
	Backmaul fate is equal to t	ne Forward ha	ul rate.
	Backmaul fate is equal to t	ne Forward ha	ul rate.
	Backmaul fate is equal to t	ne Forward ha	ul rate.

Original Sheet No. 6

ADDI SURC FTS	TIONAL CHARGES OR CHARGES APPLICABLE TO AND ITS SERVICE	Maximum Rate Per Dth	Minimu Rate Per Dt
1.	Annual Charge Adj. (ACA)	\$ 0.0000	\$ 0.000
2.	Gas for Transporter's Use (%) 1/	0.70 Per	cent
1/	Transporter's Use (%) will not be appl Transportation.	ied to Backhaul	

Original Sheet No. 7

STATEMENT OF NEGOTIATED RATES FOR TRANSPORTATION OF NATURAL GAS

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

		RATE SCHEDULE FTS Firm Transportation Service		
1.	AVAII	LABILITY		
	This Rate Schedule is available to any Shipper for the Transportation of Gas by Transporter, subject to the following limitations:			
	(a)	Transporter has determined that it has sufficient available and uncommitted capacity to perform service requested by Shipper; and		
	(d)	Shipper and Transporter have executed an Agreement under this Rate Schedule.		
2.	APPLI	ICABILITY AND CHARACTER OF SERVICE		
	(a)	This Rate Schedule shall apply to all Transportation Service rendered by Transporter for Shipper pursuant to the executed Agreement under this Rate Schedule.		
	(b)	Transportation Service under this Rate Schedule shall consist of: (1) the receipt of Gas on behalf of Shipper, (2) the Transportation of Gas, and (3) the Tender of Gas for delivery by Transporter to Shipper, or for Shipper's account up to Shipper's MDQ.		
	(c)	Transportation Service rendered under this Rate Schedule shall be firm, up to the Transportation Path MDQ specified in the executed Agreement.		

Original Sheet No. 9

٦

		RATE SCHEDULE FTS Firm Transportation Service (CONTINUED)
3.	RATES	
	Each Month,	Shipper shall pay to Transporter the following rates:
	3.1 Reserv	vation Rates.
	(a)	A Reservation Rate, as stated in the Statement of Rates for Transportation of Natural Gas, shall be paid each Month for each Dekatherm of Shipper's MDQ.
	3.2 Usage	Rates.
	(a)	The Usage-1 Rate, as stated in the Statement of Rates for Transportation of Natural Gas, multiplied by that portion of the total quantity of Gas deliveries on any Day pursuant to the Shipper's Agreement which is not in excess of the lower of (i) 110% of the scheduled quantities of Gas under the Agreement for such Day or (ii) the MDQ in effect under the Agreement for such Day.
	(d)	The Usage-2 Rate, as stated in the Statement of Rates for Transportation of Natural Gas, multiplied by that portion of the total quantity of Gas deliveries on any Day pursuant to the Shipper's Agreement which is greater than the lower of 110% of (i) the scheduled quantities of Gas under the Agreement for such Day or (ii) the MDQ in effect under the Agreement for such Day.
	(c)	Other Applicable Charges or Surcharges. All applicable surcharges or charges, including, but not limited to those contained in Sections 8 and 22 of the General Terms and Conditions, and as stated in the Statement of Rates for Transportation of Natural Gas multiplied by each Dekatherm of Gas delivered.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

RATE SCHEDULE FTS Firm Transportation Service (CONTINUED)

- 3.3 Transporter's Use. Each Shipper will furnish Transporter fuel at the nominated Receipt Point(s). The amount of fuel furnished to Transporter will be based on the applicable percentage for Transporter's Use, as calculated pursuant to Section 22.2 of the General Terms and Conditions and as stated in the Statement of Rates for Transportation of Natural Gas.
- 3.4 Negotiated Rates. Shipper and Transporter may mutually agree, pursuant to the provisions of Section 30 of the General Terms and Conditions, to a negotiated rate, which rate shall be less than, equal to, or greater than Transporter's Maximum Recourse Rate, but shall not be less than the minimum rate. Any such rates may be based upon a rate design other than straight fixed variable (SFV). Such negotiated rate shall be set forth in an executed Agreement and/or in Transporter's Tariff and shall be agreed to in a written agreement between Transporter and Shipper.
- 3.5 Discounted Rates. Subject to any limitations agreed to by Shipper and Transporter, Transporter may from time to time and at any time selectively adjust any or all of the rates charged to any individual Shipper for any and all of the transportation paths for which a Maximum Rate and Minimum Rate are stated in the Statement of Rates for Transportation of Natural Gas of this Tariff or a superseding Tariff; provided, however, that such adjusted rate(s) shall not exceed the applicable Maximum Rate(s), nor shall they be less than the Minimum Rate(s), set forth on such sheets. Transporter shall have the right to charge the Maximum Rate at any time as a condition for new service, or for continuation of service under an existing Agreement. Transporter shall make all information filings and/or postings on LINK® as required by the Commission's regulations with respect to any charges at less than the Maximum Rate.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

RATE SCHEDULE FTS Firm Transportation Service (CONTINUED)

4. RECEIPT AND DELIVERY POINTS

- 4.1 Receipt Point Availability.
 - Subject to the availability of capacity at any (a) specific point and to the Receipt Point MDQs that apply to Agreements under this Rate Schedule that were executed by Transporter and Shipper on or before December 29, 2006, Shipper shall have access to all Receipt Points on Transporter's system on a Priority Class One basis. Transporter, subject to all necessary regulatory approvals and agreements with interconnecting parties, intends to establish receipt point interconnects with the following interstate pipelines at locations proximate to Transporter's system: CenterPoint Energy Gas Transmission Company's ("CEGT") Carthage to Perryville system ("CEGT-Line CP"); CEGT's Line FM 63 ("CEGT FM 63"); Columbia Gulf Gas Transmission ("Columbia Gulf"); Florida Gas Transmission Company ("FGT"); Gulfstream Natural Gas System, L.L.C. ("Gulfstream"); Gulf South Pipeline Company, L.P. ("Gulf South"); Southern Natural Gas Company ("Sonat"); Tennessee Gas Pipeline Company ("Tennessee"); Texas Eastern Transmission, LP ("Texas Eastern"); and Transcontinental Gas Pipe Line Corporation ("Transco").
 - (b) The following shall apply only with respect to Agreements under this Rate Schedule that were executed by Transporter and Shipper on or before December 29, 2006:

The Receipt Point MDQ at each of the following Receipt Points shall be equal to the MDQ specified in the applicable Agreement: (i) CEGT-Line CP; (ii) Gulf South's (East Texas Expansion) pipeline; (iii) Columbia Gulf's system near Perryville, Louisiana (subject to completion of the interconnection between Transporter's system and Columbia Gulf's system) and (iv) the proposed Continental Connector pipeline sponsored by a subsidiary of El Paso Corporation, if such

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	RATE SCHEDULE FTS Firm Transportation Service (CONTINUED)		
	pipeline is constructed and connected with Transporter's system within five (5) years after the date on which service is first initiated on Transporter's system.		
	4.2 Shipper's Primary Delivery Point(s) and/or Shipper's Transportation Path will be listed in Exhibit B attached to Shipper's Agreement. Shipper shall have the right to utilize all other Delivery Point(s) as Secondary Delivery Point(s), subject to available capacity and the provisions of the General Terms and Conditions.		
	4.3 Upon five (5) Business Days prior notice, Shipper shall have the right to redesignate any points listed on Exhibit Bas Primary Delivery Point(s), subject to available capacity and the provisions of the General Terms and Conditions; provided, however, if Shipper is paying a Negotiated Rate for service under the Agreement and requests to change its Primary Delivery Point under the Agreement, then unless otherwise agreed to in writing by Shipper and Transporter the rate applicable for service to such new Primary Delivery Point shall be the maximum Recourse Rate. Furthermore, Shipper shall have the right to utilize all other Delivery Point(s) as Secondary Delivery Point(s), subject to available capacity and the provisions of the General Terms and Conditions.		
	4.4 Subject to mutual agreement and the provisions of Section 12 of the General Terms and Conditions, Shipper shall agree with Transporter as to the minimum delivery pressure at the Delivery Point. Such pressure shall be set forth in Exhibit Bto the Agreement.		
5.	COMMISSION AND OTHER REGULATORY FEES		
	nippers will reimburse Transporter for any separately stated ees required by the Commission or any other federal or state egulatory body.		
6.	GENERAL TERMS AND CONDITIONS		
	All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule.		

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	RATE SCHEDULE ITS Interruptible Transportation Service			
1.	AVAILABILITY			
	This Rate Schedule is available to any Shipper for the Transportation of Gas by Transporter when Shipper and Transporter have executed an Agreement under this Rate Schedule			
2.	APPLICABILITY AND CHARACTER OF SERVICE			
	(a) This Rate Schedule shall apply to all Transportation Service rendered by Transporter for Shipper pursuant to the executed Agreement under this Rate Schedule.			
	 (b) Transportation Service under this Rate Schedule shall consist of: (1) the receipt of Gas on behalf of Shipper, (2) the Transportation of Gas, and (3) the Tender of Gas for delivery by Transporter to Shipper, or for Shipper's account. 			
	(c) Transportation Service rendered under this Rate Schedule shall be interruptible. Interruptible service shall be available only to the extent of available capacity as it may be from Day to Day and from time to time within the Gas Day, under current conditions and shall be offered in accordance with the priorities established in the General Terms and Conditions of Transporter's Tariff.			
3.	RATES			
	Each Month, Shipper will pay Transporter the following rates:			
	3.1 Usage-1 Rate. A rate equal to the applicable Usage Rates either as stated in the Statement of Rates for Transportation of Natural Gas multiplied by each Dekather of Gas delivered or as agreed upon in writing between Transporter and Shipper, but in no event will such rate b less than the minimum rate as stated in the Statement of Rates for Transportation of Natural Gas.			
	3.2 Usage-2 Rate. A rate equal to that stated in the Statement of Rates for Transportation of Natural Gas, multiplied by that portion of the total quantity of Gas			

Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

RATE SCHEDULE ITS Interruptible Transportation Service (CONTINUED)

deliveries on any Day pursuant to the Agreement, which is greater than the lower of 110% of (i) the scheduled quantities of Gas under the Agreement for such Day or (ii) the MDQ in effect under the Agreement for such Day.

- 3.3 Other Applicable Charges or Surcharges. All applicable surcharges or charges, including, but not limited to, those contained in Sections 8 and 22 of the General Terms and Conditions and as stated in the Statement of Rates for Transportation of Natural Gas multiplied by each Dekatherm of Gas delivered.
- 3.4 Transporter's Use. Each Shipper will furnish Transporter fuel at the nominated Receipt Point(s). The amount of fuel furnished to Transporter will be based on the applicable percentage for Transporter's Use, as calculated pursuant to Section 22.2 of the General Terms and Conditions and as stated in the Statement of Rates for Transportation of Natural Gas.
- 3.5 Negotiated Rates. Shipper and Transporter may mutually agree, pursuant to the provisions of Section 30 of the General Terms and Conditions, to a negotiated rate, which rate shall be less than, equal to, or greater than Transporter's applicable maximum Recourse Rate, but shall not be less than the minimum rate. Such negotiated rate shall be set forth in an executed Agreement and/or in Transporter's Tariff and shall be agreed to in a written agreement between Transporter and Shipper.
- 3.6 Discounted Rates. Subject to any limitations agreed to by Shipper and Transporter, Transporter may from time to time and at any time selectively adjust any or all of the rates charged to any individual Shipper for any and all of the transportation routes for which a Maximum Rate and Minimum Rate are stated in the Statement of Rates for Transportation of Natural Gas of this Tariff or a superseding Tariff; provided, however, that such

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

RATE SCHEDULE ITS Interruptible Transportation Service (CONTINUED)

adjusted rate(s) shall not exceed the applicable Maximum Rate(s), nor shall they be less than the Minimum Rate(s), set forth on such sheets. Transporter shall have the right to charge the Maximum Rate at any time as a condition for new service, or for continuation of service under an existing Agreement. Transporter shall make all information filings and/or postings on LINK® required by the Commission's regulations with respect to any charges at less than the Maximum Rate.

4. COMMISSION AND OTHER REGULATORY FEES

Shippers will reimburse Transporter for any separately stated fees required by the Commission or any other federal or state regulatory body.

5. GENERAL TERMS AND CONDITIONS

All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

 AVAILABILITY This Rate Schedule is available to any Shipper for the parkin and lending of Gas from Transporter, subject to the following limitations: Transporter has determined that it is operationally abl to render such service; Shipper and Transporter have executed an Agreement unde this Rate Schedule. APPLICABILITY AND CHARACTER OF SERVICE This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule. Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule and shall consist of:		Parking and Lending Service
 This Rate Schedule is available to any Shipper for the parkin and lending of Gas from Transporter, subject to the following limitations: 1.1 Transporter has determined that it is operationally abl to render such service; 1.2 Shipper and Transporter have executed an Agreement unde this Rate Schedule. 2. APPLICABILITY AND CHARACTER OF SERVICE 2.1 This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule. 2.2 Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall be interruptible and shall consist of: (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper for receipt op parked quantities on Transporter holding the parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 	1.	AVAILABILITY
 1.1 Transporter has determined that it is operationally abl to render such service; 1.2 Shipper and Transporter have executed an Agreement under this Rate Schedule. 2. APPLICABILITY AND CHARACTER OF SERVICE 2.1 This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule. 2.2 Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall consist of: (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper for receipt point(s) nominated by Shipper for receipt of parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 		This Rate Schedule is available to any Shipper for the parking and lending of Gas from Transporter, subject to the following limitations:
 1.2 Shipper and Transporter have executed an Agreement under this Rate Schedule. 2. APPLICABILITY AND CHARACTER OF SERVICE 2.1 This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule. 2.2 Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall be interruptible and shall consist of: (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper to receipt of parked quantities; (2) Transporter holding the parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 		1.1 Transporter has determined that it is operationally able to render such service;
 APPLICABILITY AND CHARACTER OF SERVICE This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule. Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall be interruptible and shall consist of: (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper to receipt point(s) nominated by Shipper for receipt of parked quantities; (2) Transporter holding the parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 		1.2 Shipper and Transporter have executed an Agreement under this Rate Schedule.
 2.1 This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule. 2.2 Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall be interruptible and shall consist of: (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper to receipt point(s) nominated by Shipper for receipt of parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 	2.	APPLICABILITY AND CHARACTER OF SERVICE
 2.2 Service under this Rate Schedule shall consist of eithe parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall be interruptible and shall consist of: (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper to receipt point(s) nominated by Shipper for receipt of parked quantities; (2) Transporter holding the parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 		2.1 This Rate Schedule shall apply to service which is rendered by Transporter for Shipper pursuant to an executed Agreement under this Rate Schedule.
 (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivere by Shipper to receipt point(s) nominated by Shipper for receipt of parked quantities; (2) Transporter holding the parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked. 		2.2 Service under this Rate Schedule shall consist of either parking or lending of Gas during any Day, or part thereof. Service rendered by Transporter under this Rate Schedule shall be interruptible and shall consist of:
		 (a) Parking Service. Parking Service is an interruptible service which provides for (1) the receipt by Transporter of Gas quantities delivered by Shipper to receipt point(s) nominated by Shipper for receipt of parked quantities; (2) Transporter holding the parked quantities on Transporter's pipeline system; and (3) return of the parked quantities to Shipper, provided, however, that Transporter is not obligated to return parked quantities on the same Day and at the same point(s) at which the Gas is parked.

Issued by: Issued on:

Original Sheet No. 18

RATE SCHEDULE PALS Parking and Lending Service (CONTINUED)

- (b) Lending Service. Lending Service is an interruptible service which provides for (1) Shipper receiving Gas quantities from Transporter at delivery point(s) nominated by Shipper for delivery of loaned quantities of Gas; and (2) the subsequent return of the loaned quantities of Gas to Transporter, provided, however, that Transporter is not obligated to accept return of loaned Gas on the same Day and at the same point(s) at which the Gas is loaned.
- (c) If the Shipper receives parked quantities or returns loaned quantities at point(s) other than the point(s) at which the park or loan occurred, then Shipper and Transporter shall enter into a separate Transportation Agreement(s) to effectuate receipt or delivery of Gas from or to the new point(s).
- 2.3 Service rendered under this Rate Schedule shall be provided for a minimum of a one (1) Day term. The term shall be set forth on the Agreement executed between Shipper and Transporter.
- 2.4 Transportation of Gas quantities for or on behalf of Shipper to or from the designated point(s) under the Agreement shall not be performed under this Rate Schedule. Shipper shall make any necessary arrangements with Transporter and/or third parties to receive or deliver Gas quantities at the designated point(s) for Parking or Lending Service hereunder.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

		RATE SCHEDULE PALS Parking and Lending Service (CONTINUED)
RATES		
Each rates	Month, :	Shipper shall pay to Transporter the following
3.1	Usage	Rates.
	(a)	The applicable Usage Rate, as stated in the Statement of Rates for Transportation of Natural Gas, which shall be paid for each Dekatherm of Gas parked or loaned each Day at each point by Transporter for or on behalf of the account of Shipper multiplied by the highest balance of Gas guantities parked and/or loaned by Shipper on such Day;
	(b)	Other Applicable Charges or Surcharges. All applicable surcharges or charges including, but not limited to, those charges under Sections 8 and/or 22, and as stated in the Statement of Rates for Transportation of Natural Gas multiplied by each Dekatherm of Gas parked or loaned each Day at each point by Transporter for or on behalf of the account of Shipper.
3.2	Trans furni	porter's Use. Shipper shall not be required to sh fuel for service under this Rate Schedule.
3.3.	Negot agree Gener which Trans less be se Trans agree	iated Rates. Shipper and Transporter may mutually , pursuant to the provisions of Section 30 of the al Terms and Conditions, to a negotiated rate, rate shall be less than, equal to, or greater than porter's Maximum Recourse Rate, but shall not be than the minimum rate. Such negotiated rate shall t forth in an executed Agreement and/or in porter's Tariff and shall be agreed to in a written ment between Transporter and Shipper.
	RATES Each rates 3.1 3.2 3.3.	RATES Each Month, rates: 3.1 Usage (a) (b) 3.2 Trans furni 3.3. Negot agree Gener which Trans less be se Trans agree
٦

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

 3.4 Discounted Rates. Subject to any limitations agreed to by Shipper and Transporter, Transporter may from time to time and at any time selectively adjust any or all of the rates charged to any individual Shipper for service under this Rate Schedule for which a Maximum Rate and Minimum Rate are stated in the Statement of Rates for Transportation of Natural Gas of this Tariff or a superseding Tariff; provided, however, that such adjusted rate(s) shall not exceed the applicable Maximum Rate(s), nor shall they be less than the Minimum Rate(s), nor shall they be less than the Minimum Rate(s), set forth on such sheets. Transporter shall have the right to charge the Maximum Rate at any time as a condition for new service, or for continuation of service under an existing Agreement. Transporter shall make all information fillings and/or postings on LINK® required by the Commission's regulations with respect to any charges at less than the Maximum Rate. 4. OPERATIONAL REQUIREMENTS OF TRANSPORTER 4.1 Shipper may be required, upon notification from Transporter, to cease or reduce deliveries to, or receipts from, Transporter hereunder within a Day consistent with Transporter's operating requirements. Further, Shipper may be required to return loaned quantities or remove parked quantities upon notification by Transporter. Such notification shall, at a minimum, be provided by posting on LINK®, and may also be provided by other means of communication. Transporter's notification shall be returned, consistent with Transporter's notification, but in no event shall the specified time be sconer than the next Day after Transporter's notification, subject to the following conditions: 		RATE SCHEDULE PALS Parking and Lending Service (CONTINUED)
 4. OPERATIONAL REQUIREMENTS OF TRANSPORTER 4.1 Shipper may be required, upon notification from Transporter, to cease or reduce deliveries to, or receipts from, Transporter hereunder within a Day consistent with Transporter's operating requirements. Further, Shipper may be required to return loaned quantities or remove parked quantities upon notification by Transporter. Such notification shall, at a minimum, be provided by posting on LINK®, and may also be provided by other means of communication. Transporter's notification shall specify the time frame within which parked quantities shall be removed and/or loaned quantities shall be returned, consistent with Transporter's operating conditions, but in no event shall the specified time be sooner than the next Day after Transporter's notification, subject to the following conditions: 	3.4	Discounted Rates. Subject to any limitations agreed to by Shipper and Transporter, Transporter may from time to time and at any time selectively adjust any or all of the rates charged to any individual Shipper for service under this Rate Schedule for which a Maximum Rate and Minimum Rate are stated in the Statement of Rates for Transportation of Natural Gas of this Tariff or a superseding Tariff; provided, however, that such adjusted rate(s) shall not exceed the applicable Maximum Rate(s), nor shall they be less than the Minimum Rate(s), set forth on such sheets. Transporter shall have the right to charge the Maximum Rate at any time as a condition for new service, or for continuation of service under an existing Agreement. Transporter shall make all information filings and/or postings on LINK® required by the Commission's regulations with respect to any charges at less than the Maximum Rate.
4.1 Shipper may be required, upon notification from Transporter, to cease or reduce deliveries to, or receipts from, Transporter hereunder within a Day consistent with Transporter's operating requirements. Further, Shipper may be required to return loaned quantities or remove parked quantities upon notification by Transporter. Such notification shall, at a minimum, be provided by posting on LINK®, and may also be provided by other means of communication. Transporter's notification shall specify the time frame within which parked quantities shall be removed and/or loaned quantities shall be returned, consistent with Transporter's operating conditions, but in no event shall the specified time be sconer than the next Day after Transporter's notification, subject to the following conditions:	4. OPER	ATIONAL REQUIREMENTS OF TRANSPORTER
	4.1	Shipper may be required, upon notification from Transporter, to cease or reduce deliveries to, or receipts from, Transporter hereunder within a Day consistent with Transporter's operating requirements. Further, Shipper may be required to return loaned quantities or remove parked quantities upon notification by Transporter. Such notification shall, at a minimum, be provided by posting on LINK®, and may also be provided by other means of communication. Transporter's notification shall specify the time frame within which parked quantities shall be removed and/or loaned quantities shall be returned, consistent with Transporter's operating conditions, but in no event shall the specified time be sconer than the next Day after Transporter's notification, subject to the following conditions:

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

RATE SCHEDULE PALS Parking and Lending Service (CONTINUED)

- (a) In the event that the specified time for removal or return of Gas quantities is the next Day, the time frame for required removal or return shall begin from the time that Shipper receives notice from Transporter. Notices provided after business hours for the next Day will be provided to Shipper via Electronic Communication. In the event that Shipper makes a timely and valid nomination in response to notification by Transporter to remove parked quantities and/or return loaned quantities, Shipper shall be deemed to have complied with Transporter's notification; and
- (b) Unless otherwise agreed by Shipper and Transporter: (i) any parked quantity not nominated for removal within a time frame specified by Transporter's notice shall become the property of Transporter at no cost to Transporter free and clear of any adverse claims; (ii) any loaned quantity not returned within the time frame specified by Transporter's notice shall be sold to Shipper at Transporter's Cashout Price at the >25% Imbalance level for Imbalances Due Transporter, pursuant to Section 8.7(b) of the General Terms and Conditions.

Any penalty revenues received by Transporter as a result of the operation of Section 4.1(b) above will be credited pursuant to Section 23.1(a) of the General Terms and Conditions.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

 4.2 In the event parked quantities remain in Transporter's pipeline system and/or loaned quantities have not been returned to Transporter's pipeline system at the expiration of any Agreement executed by Shipper and Transporter, Transporter and Shipper may mutually agree to an extended time frame and/or modified terms, including the rate, of such Agreement. In the event that Shipper and Transporter are unable to come to such Agreement, Transporter shall notify Shipper, and Shipper shall nominate for removal of the parked quantities and/or return of the loaned quantities within the time frame specified in Transporter's notice, which in no instance shall be less than one (1) Day. Any parked quantity not nominated for removal within the time frame specified by Transporter's notice shall become the property of Transporter at no cost to Transporter, free and clear of any adverse claims. Any loaned quantities not nominated for Imbalances Due Transporter, pursuant to Section 8.7(b) of the General Terms and Conditions. Any penalty revenues received by Transporter as a result of the operation of Section 4.2 above will be credited pursuant to Section 23.1(b) of the General Terms and Conditions. COMMISSION AND OTHER REGULATORY FEES Shippers will reimburse Transporter for any separately stated fees required by the Commission or any other federal or any state regulatory body. GENERAL TERMS AND CONDITIONS All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule. 		RATE SCHEDULE PALS Parking and Lending Service (CONTINUED)
 Any penalty revenues received by Transporter as a result of the operation of Section 4.2 above will be credited pursuant to Section 23.1(b) of the General Terms and Conditions. 5. COMMISSION AND OTHER REGULATORY FEES Shippers will reimburse Transporter for any separately stated fees required by the Commission or any other federal or any state regulatory body. 6. GENERAL TERMS AND CONDITIONS All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule.		4.2 In the event parked quantities remain in Transporter's pipeline system and/or loaned quantities have not been returned to Transporter's pipeline system at the expiration of any Agreement executed by Shipper and Transporter, Transporter and Shipper may mutually agree to an extended time frame and/or modified terms, including the rate, of such Agreement. In the event that Shipper and Transporter are unable to come to such Agreement, Transporter shall notify Shipper, and Shipper shall nominate for removal of the parked quantities and/or return of the loaned quantities within the time frame specified in Transporter's notice, which in no instance shall be less than one (1) Day. Any parked quantity not nominated for removal within the time frame specified by Transporter at no cost to Transporter, free and clear of any adverse claims. Any loaned quantities not nominated to be returned within the time frame specified by Transporter's notice shall be sold to Shipper at Transporter's Cashout Price at the >25% Imbalance Level for Imbalances Due Transporter, pursuant to Section 8.7(b) of the General Terms and Conditions.
 COMMISSION AND OTHER REGULATORY FEES Shippers will reimburse Transporter for any separately stated fees required by the Commission or any other federal or any state regulatory body. GENERAL TERMS AND CONDITIONS All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule. 		Any penalty revenues received by Transporter as a result of the operation of Section 4.2 above will be credited pursuant to Section 23.1(b) of the General Terms and Conditions.
 Shippers will reimburse Transporter for any separately stated fees required by the Commission or any other federal or any state regulatory body. 6. GENERAL TERMS AND CONDITIONS All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule. 	5.	COMMISSION AND OTHER REGULATORY FEES
6. GENERAL TERMS AND CONDITIONS All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule.		Shippers will reimburse Transporter for any separately stated fees required by the Commission or any other federal or any state regulatory body.
All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule.	6.	GENERAL TERMS AND CONDITIONS
		All of the General Terms and Conditions of this Tariff are specifically incorporated into this Rate Schedule.

Issued by: Issued on:

	INDEX	
DES	CRIPTION	Sheet No
-		0.4
1.		• • 24
2.	Measurement and Measurement Equipment	30
3.		43
4.		46
5.	Priority of Service	53
6.	Scheduling and Curtailment	55
7.	Determination of Daily Allocated Receipts and Deliver	1es. 58
8.	Imbalance Resolution Procedures	62
9.	Billing	72
10.	Payments	74
11.	Possession of Gas	78
12.	Receipt and Delivery Point Pressure	78
13.	Operational Flow Orders	79
14.	Warranty of Title	85
15.	Force Majeure	86
16.	Notices	87
17.	Modification	87
18.	Non-Waiver and Future Default	87
19.	Schedules and Contracts Subject to Regulation	88
20.	Operational Balancing Agreements ("OBAs")	88
21.	New Facilities Policy	90
22.	Periodic Rate Adjustments	92
23.	Penalties and Penalty Crediting Mechanism	95
24.	Electronic Communication Service	98
25.	Capacity Release Provisions	109
26.	Requests For Service	135
27.	Creditworthiness	141
28.	Right of First Refusal	147
29.	Incorporation in Rate Schedules and Agreements	149
30.	Negotiated Rates	149
31.	North American Energy Standards Board Standards	151
32.	Default and Termination	153
33.	Standards of Conduct Compliance Procedures	154
34.	Limitation of Liability of Member and Administrator .	155

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

		GENERAL TERMS AND CONDITIONS (CONTINUED)
1.	DEFIN	VITIONS
	In so Scheo Agree	ome instances, definitions are set forth in the Rate dules, General Terms and Conditions and the Forms of Service ements.
	1.1	The term "Agreement" shall mean the agreement executed by the Shipper and Transporter and any applicable exhibits, attachments and/or amendments thereto.
	1.2	The term "Backhaul" shall mean the receipt and delivery of Gas, which is accomplished by the Transporter's delivery of Gas at point(s) which are upstream from the point(s) at which Gas is received.
	1.3	The term "Business Day" shall mean Monday through Friday, excluding Federal Banking Holidays for transactions in the United States.
	1.4	The term "BTU" shall mean one (1) British thermal unit, the amount of heat required to raise the temperature of one (1) pound of water from 58.5 degrees Fahrenheit to 59.5 degrees Fahrenheit, (BTU is measured and reported on a dry basis at 14.73 psia and 60 degrees Fahrenheit).
	1.5	The term "Cashout" shall mean the monetary settlement of quantities of Gas owed to or by Transporter or third parties, as further described in Section 8 of these General Terms and Conditions.
	1.6	The term "Cashout Party" shall mean any Shipper or other contractually liable entity who has an imbalance under any Agreement, which imbalance will be resolved in accordance with Section 8 of these General Terms and Conditions.
	1.7	The term "Cashout Price" shall mean the price determined pursuant to Section 8 of these General Terms and Conditions.

Original Sheet No. 25

GENERAL TERMS AND CONDITIONS (CONTINUED)		
1.8	The term "Central Clock Time" or "CCT" shall mean Central Standard Time ("CST") except when Daylight Savings Time is in effect, when it shall mean one hour in advance of CST. All times referenced in Transporter's Tariff shall be in CCT.	
1.9	The term "Commission" or "FERC" shall mean the Federal Energy Regulatory Commission or any successor regulatory authority.	
1.10	The term "Confirmed Price" shall mean the Transportation rate inclusive of all applicable fees and surcharges agreed upon, in writing and/or via LINK®, by Transporter and Shipper or as otherwise required in this Tariff.	
1.11	The term "Day" or "Gas Day" shall mean a period of 24 consecutive hours, beginning at 9:00 a.m. CCT, and ending on the following 9:00 a.m.	
1.12	The term "Dekatherm" (or "Dth") shall mean the quantity of heat energy which is equivalent to one (1) million (1,000,000) BTU; thus the term Mdth shall mean one (1) thousand (1,000) Dth.	
1.13	The term "Delivery Point" shall mean an interconnection point on Transporter's pipeline system that Shipper and Transporter shall agree upon, where Gas exits facilities owned by Transporter, and is metered.	
1.14	The term "Delivery Point MDQ" shall mean the greatest number of Dekatherms that Transporter is obligated to deliver, on a Priority Class Onebasis to or on behalf of Shipper on any Day at the applicable Primary Delivery Point. The aggregate of the Delivery Point MDQs may not exceed the MDQ set forth in the Agreement; provided, however, a Shipper that has the Gulfstream Delivery Point as a Primary Delivery Point shall also have the FGT Delivery Point with a Delivery Point MDQ equal to such Shipper's Delivery Point MDQ at the Gulfstream Delivery Point.	

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.15	The term "Delivery Point Operator" shall mean the party that is responsible for operating the facilities that are immediately downstream of the applicable Delivery Point.
1.16	The term "Elapsed Prorata Capacity" shall mean that portion of the capacity that would have theoretically been available for use prior to the effective time of the intraday recall based upon a cumulative uniform hourly use of the capacity.
1.17	The term "Elapsed Prorated Scheduled Quantity" shall mean that portion of the scheduled quantity that would have theoretically flowed up to the effective time of the intraday nomination being confirmed, based on a uniform hourly flow rate for each nomination period affected.
1.18	The term "Electronic Communication" shall mean the transmission of information via Transporter's LINK® system, Electronic Delivery Mechanism prescribed by NAESB or other mutually agreed communication methodologies used to transmit and receive information.
1.19	The term "Electronic Delivery Mechanism" or "EDM" shall mean the Electronic Communication methodology used to transmit and receive data related to gas transactions. Transporter and Shipper shall designate an electronic "site" at which Shippers and Transporter may exchange data electronically. All data provided at such site shall be considered as being delivered to the appropriate party. Transporter's use and implementation of EDM shall conform to all appropriate NAESB standards.

Issued by: Issued on:

Original Sheet No. 27

GENERAL TERMS AND CONDITIONS (CONTINUED)
1.20 The term "Equivalent Quantities" shall mean a quantity of Gas containing an amount of Dekatherms equal to the amount of Dekatherms received by Transporter for the account of Shipper at the Receipt Point(s) reduced, where applicable, by the Dekatherms removed for Transporter's Use.
1.21 The term "FGT Delivery Point" shall mean the interconnections between Transporter's system and Florida Gas Transmission Company's system near Mobile, Alabama; and the interconnections between Transporter's system and Gulf South's system and Transco's system in the Mobile, Alabama area.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.23	The term "Gas" shall mean natural gas, including cap gas, casinghead gas produced with crude oil, gas from gas wells, gas from condensate wells, synthetic natural gas, or any mixture of these gases meeting the quality standards under Section 3 of these General Terms and Conditions.
1.24	The term "Gas Delivered Hereunder" shall mean the quantities of Gas allocated to Shipper by Transporter, as determined in accordance with the provisions of Section 7 of these General Terms and Conditions.
1.25	The term "Gulfstream Delivery Point" shall mean the interconnection between Transporter's system and Gulfstream Natural Gas System, L.L.C.'s system near Coden, Alabama.
1.26	The term "High Common" shall mean the highest price listed under the heading "Common" in "Platts Gas Daily" "Daily Price Survey" for the applicable day.
1.27	The term "Imbalance Management Services" shall mean the options available to Shippers for resolution of imbalances including the application of the cashout mechanism set forth in Section 8 of the General Terms and Conditions. These options include service under PALS, Imbalance Netting and Trading and, as a final resolution, Cashout.
1.28	The term "Internet Website" shall mean Transporter's HTML site accessible via the Internet's World Wide Web located at http://www.link.duke-energy.com.
1.29	The term "LINK®" shall mean the LINK® Customer Interface System.
1.30	The term "Low Common" shall mean the lowest price listed under the heading "Common" in "Platts Gas Daily" "Daily Price Survey" for the applicable day.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.31	The term "Maximum Daily Quantity" ("MDQ") shall mean the greatest number of Dekatherms that Transporter is obligated to transport, on a firm basis, to or on behalf of Shipper on any day.
1.32	The term "Maximum PALS Quantity" ("MPQ") shall mean the greatest number of Dekatherms that shipper may have parked or loaned under its Rate Schedule PALS Agreement at any time.
1.33	The term "Mcf" shall mean one (1) thousand (1,000) cubic feet of Gas; the term MMcf shall mean one (1) million (1,000,000) cubic feet of Gas. (Mcf is measured on a dry basis at 14.73 psia. and 60 degrees Fahrenheit.)
1.34	The term "Month" shall mean the period beginning on the first Day of a calendar Month and ending at the same hour on the first Day of the next succeeding calendar Month.
1.35	The term "Monthly Imbalance" shall mean a Shipper's monthly quantity subject to resolution through the cashout mechanism described in Section 8 of the General Terms and Conditions, calculated as the difference between (i) allocated quantities received from a Cashout Party for the Month, as determined in accordance with Section 7 of the General Terms and Conditions, adjusted for Transporter's Use, and (ii) allocated quantities delivered to a Cashout Party for the Month, as determined in accordance with Section 7.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.36	The term "Negotiated Rate" shall mean a rate or rate formula for computing a rate for service under a single Agreement. For scheduling and curtailment purposes, a Shipper paying a Negotiated Rate in excess of the maximum Recourse Rate, will be considered to be paying the maximum Recourse Rate.
1.37	The term "Net Present Value" ("NPV") shall mean the discounted cash flow of expected revenues per Dekatherm of the applicable service for a term of up to twenty (20) years, using the interest rate set forth in Section 154.501 of the Commission's Regulations.
1.38	The term "Netting" shall be used to describe the process of resolving imbalances for a Shipper within an Operational Impact Area. There are two types of Netting:
	a. Summing is the accumulation of all imbalances above any applicable tolerances for a Shipper or agent.
	b. Offsetting is the combination of positive and negative imbalances above any applicable tolerances for a Shipper or agent.
1.39	The term "North American Energy Standards Board" or "NAESB" shall mean the accredited organization established to set standards for certain natural gas industry business practices and procedures.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.40	The term "Operational Impact Area" shall describe a Transportation Service Provider's (as defined by the NAESB Standards) designation of the largest possible area(s) on its system in which imbalances have a similar operational impact. For Transporter, the entire pipeline system shall comprise a single Operational Impact Area.
1.41	The term "Posted Point of Restriction" shall mean any point or segment on Transporter's pipeline system for which Transporter has posted on its web site a reduction of scheduled capacity notice, a notice that the point or segment is scheduled at its capacity, or a notice of expected restrictions due to weather, operating conditions or maintenance.
1.42	The term "Primary Delivery Point" shall mean the Delivery Point(s) as specified in the Agreement.
1.43	The term "Primary Receipt Point" shall mean the Receipt Point(s) as specified in Section 4.1 of Rate Schedule FTS.
1.44	The term "Receipt Point" shall mean an interconnection point on Transporter's pipeline system that Transporter and Shipper shall agree upon, where Gas enters facilities owned by Transporter, and is metered.
1.45	The term "Receipt Point MDQ" shall mean the greatest number of Dekatherms that Transporter is obligated to receive on a Priority Class Onebasis for or on behalf of Shipper on any Day at the applicable Primary Receipt Point(s). Receipt Point MDQs shall only be specified in Rate Schedule FTS Agreements that were executed by Transporter and Shipper on or before December 29, 2006.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.46	The term "Recourse Rate" shall mean the maximum rate for service set forth on the rate sheets in this Tariff for the applicable rate schedule.
1.47	The term "Receipt Point Operator" shall mean the party that is responsible for operating the facilities that are immediately upstream of the applicable Receipt Point.
1.48	The term "Reput" shall mean the reinstatement of a capacity release transaction that was recalled.
1.49	The term "Secondary Delivery Point" shall mean a Delivery Point that is not specified as a Primary Delivery Point.
1.50	The term "Service Day" shall mean the Day during which Shipper receives Transportation Service pursuant to a nomination in accordance with Section 4 of the General Terms and Conditions.
1.51	The term "Service Month" shall mean the Month during which Shipper receives services under this Tariff.
1.52	The term "Shipper" shall mean any person, corporation, limited liability company, partnership or any other legal entity who enters into an Agreement for service with Transporter.
1.53	The term "Tariff" shall mean Transporter's FERC Gas Tariff as effective from time to time.
1.54	The terms "Tender Gas" and "Tender of Gas" shall mean that the delivering party is able and willing, and offers, to deliver Gas to the receiving party at the appropriate Receipt Point or Delivery Point.

Original Sheet No. 34

GENERAL TERMS AND CONDITIONS (CONTINUED)

1.55 The term "Title Transfer" shall mean the change of title to Gas between parties at a location.

1.55A The term "Title Transfer Tracking" shall mean the process of accounting for the progression of title changes from party to party which process does not effect a physical movement of the Gas.

1.55B The term "Title Transfer Tracking Service Provider" or "TTTSP" shall mean a party conducting Title Transfer Tracking activities.

1.56 The terms "Transportation" and "Transportation Service" shall mean transportation of Gas by either forward haul, displacement or Backhaul or any combination thereof.

1.57 The term "Transportation Path" shall mean the transportation path from the Primary Receipt Point to the Primary Delivery Point.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

Γ

	GENERAL TERMS AND CONDITIONS (CONTINUED)
1.58	The term "Transporter" shall mean Southeast Supply Header, LLC.
1.59	The term "Transporter's Use" shall mean the quantity of Gas required by Transporter for compressor fuel, other company use and lost-and-unaccounted for Gas for service under each Agreement, and shall be equal to the Transporter's Use (%) under each such Agreement multiplied by Receipt Point quantities Tendered to Transporter.
1.60	The term "Transporter's Use (%)" shall mean the applicable percentage of Transporter's Use, which shall be an allocable amount of Transporter's Use, as calculated pursuant to Section 22.2, provided, however, that no Transporter's Use (%) shall be assessed on Backhaul transportation. The applicable percentage is shown in the Statement of Rates for Transportation of Natural Gas and shall be annually redetermined and filed to be made effective June 1 of each year in accordance with Section 22.2 of these General Terms and Conditions.
1.61	The "Usage-1 Rate" shall be stated in the Statement of Rates for Transportation of Natural Gas and shall be assessed as described in Section 3 of Rate Schedules FTS and ITS.
1.62	The "Usage-2 Rate" shall be stated in the Statement of Rates for Transportation of Natural Gas and shall be assessed as described in Section 3 of Rate Schedules FTS and ITS.
1.63	The term "Wire Transfer" shall mean payments made/effected by wire transfer (Fedwire, CHIPS, or Book Entry), or Automated Clearinghouse, or any other recognized electronic or automated payment mechanism that is agreed upon by Transporter in the future.

Original Sheet No. 36

2. MEASUREMENT AND MEASUREMENT EQUIPMENT	1
2.1 (a) The volume of Gas delivered at the Receipt Point(s) and at the Delivery Point(s) shall be measured by one of the following devices installed by Transporter at its election, or as agreed to by Transporter and the operator of the interconnecting facilities:	
(1) An orifice meter, designed and installed in accordance with the current edition of American National Standard ANSI/API 2530 (American Gas Association Report No. 3), entitled "Orifice Metering of Natural Gas and Other Related Hydrocarbon Fluids" (hereinafter referred to as "AGA Report No. 3"); or	
(2) A turbine meter, designed and installed in accordance with the current edition of American Gas Association Transmission Measurement Committee Report No. 7, entitled "Measurement of Gas by Turbine Meters", (hereinafter referred to as "AGA Report No. 7"); or	
(3) An ultrasonic meter, designed and installed in accordance with the current edition of American Gas Association Transmission Measurement Committee Report No. 9, entitled "Measurement of Gas by Multipath Ultrasonic Meters" (hereinafter referred to as "AGA Report No. 9"); or	
(4) A positive displacement meter, designed and installed in accordance with generally accepted industry practices.	
(b) Meters shall be maintained and operated, and auxiliary measuring equipment shall be installed, maintained and operated, in accordance with generally accepted industry practices.	

Issued by: Issued on:

Original Sheet No. 37

	GENERAL TERMS AND CONDITIONS (CONTINUED)		
2.2 (a)	The volume of Gas delivered at each Receipt Point and Delivery Point shall be calculated by means of an electronic flow computer located at each Receipt Point or each Delivery Point, in the following manner:		
	(1) The volume of Gas delivered through an orifice meter shall be computed in accordance with AGA Report No. 3, properly using all factors set forth therein.		
	(2) The volume of Gas delivered through a turbine meter shall be computed in accordance with AGA Report No. 7, properly using all factors set forth therein.		
	(3) The volume of Gas delivered through an ultrasonic meter shall be computed in accordance with AGA Report No. 9, properly using all factors set forth therein.		
	(4) The volume of Gas delivered through a positive displacement meter shall be computed by properly applying, to the volume delivered at flowing Gas pressures and temperatures, correction factors for (i) absolute static pressure, (ii) flowing Gas temperature, and (iii) compressibility ratio.		
(b)	The volume of Gas delivered shall be computed using the standards and factors determined as follows:		
	(1) The unit of volume for the purpose of measurement shall be one thousand cubic feet of Gas at a temperature of sixty (60) degrees Fahrenheit and a pressure of 14.73 pounds per square inch absolute. For the purpose of pricing hereunder, the Dekatherm equivalent of such unit of volume shall be determined by multiplying each such unit of volume by the total heating value per cubic foot of the Gas delivered hereunder (adjusted to a common temperature and pressure base) and by dividing the result by one thousand (1000).		

Issued by: Issued on:

Original Sheet No. 38

GENE	RAL TERMS AND CONDITIONS (CONTINUED)
(2)	The average absolute atmospheric (barometric) pressure at each Receipt Point and each Delivery Point shall be assumed to be 14.7, irrespective of the actual location or elevation above sea level of the Receipt Point or Delivery Point or of variations in actual atmospheric pressure from time to time.
(3)	The static pressure and temperature of the Gas at flowing conditions through a meter and, where applicable, the differential pressure across the orifice plate of an orifice meter shall be determined by means of instruments of standard manufacture accepted in the industry for these purposes.
(4)	The supercompressiblity factor used in computing the volume of Gas delivered through an orifice meter shall be determined using the procedures presented in American Gas Association Transmission Measurement Committee Report No. 8, entitled "Compressibility Factors of Natural Gas and Other Related Hydrocarbon Gases."
(5)	The specific gravity of the Gas used in computing the volume of Gas delivered through a meter shall be determined at each Receipt Point and at strategic locations determined by Transporter to be representative for each Delivery Point by standard methods accepted in the industry for this purpose.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(6) The compressibility ratio factor "s" used in computing the volume of Gas delivered through a turbine meter, an ultrasonic meter, or a positive displacement meter shall be determined by the equation s = (Fpv)2, in which "Fpv" is the supercompressiblity factor determined as described in subparagraph (4) of this subsection (b).
2	All flow measuring, testing and related equipment shall be of standard manufacture and type approved by Transporter. If applicable, Transporter or Shipper may install check measuring equipment and telemetering equipment, provided that such equipment shall be so installed as not to interfere with the operations of the operator. Transporter, or Shipper, in the presence of the other party, shall have access to measuring equipment at all reasonable times, but the reading, calibrating, and adjusting thereof shall be done by the operator of the facilities. Transporter or Shipper shall have the right to be present at the time of the installing, reading, cleaning, changing, repairing, inspecting, testing, calibrating or adjusting done by the operator of the measuring equipment. The records from such measuring equipment shall remain the property of the operator, but upon request the other party may request records including charts, together with calculations therefrom for inspection, subject to return within thirty (30) days after receipt thereof. Reasonable care shall be exercised in the installation, maintenance and operation of the measuring equipment so as to avoid any inaccuracy in the determination of the volume of Gas received and delivered.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)	
The accuracy of all measuring equipment shall operator at least once each year and if reques presence of representatives of the other party Transporter nor Shipper shall be required to v accuracy of such equipment more frequently tha thirty (30) Day period. If the operator agree verification and test of measuring equipment a perform such verification and testing, then th shall have the right to cease or temporarily of service relative to such measuring equipment. party at any time desires a special test of an equipment, it will promptly notify the other p parties shall then cooperate to secure a promp of the accuracy of such equipment. Transporta related expenses involved in the testing of me borne by the party incurring such expenses, pr however, that Shipper shall not be responsible Transportation and related expenses if the spe reveals that the meter(s) is (are) not operati required tolerance level of one percent (1%).	be verified by sted, in the r, but neither verify the an once in any es to and fails to any measuring barty and the out verification attion and eters shall be covided, a for such scial testing ang within the
The operator, for purposes of this section, sh owner of the equipment referenced herein, or t such owner, or such other person as the partie writing.	all be the the agent of es may agree in
Tagwad but	

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(c) by estimating the quantity of receipt or delivery based on receipts or deliveries during preceding periods under similar conditions when the meter was registering accurately.
2.5	If at any time during the term hereof, a new method or technique is developed with respect to Gas measurement or the determination of the factors used in such Gas measurement, such new method or technique may be substituted upon mutual agreement thereto by both parties.
2.6	The parties agree to preserve for a period of at least three (3) years or such longer period as may be required by public authority, all test data, if any, and other similar records.
2.7	Shipper or Transporter may install, maintain, and operate odorizing (at a Delivery Point only), regulating, telemetering, heating and fogging equipment at its own expense as it shall desire at each Receipt Point or Delivery Point, and the operator of such equipment at its own expense shall provide the other party a suitable site therefore and allow the other party free access to and use of the site; provided that such equipment shall be so installed, maintained and operated as not to interfere with the operation or maintenance of the operating party's measuring equipment at each Receipt Point or Delivery Point. All such equipment as Shipper or Transporter shall desire to install shall be constructed, installed and operated to conform to the other party's requirements.
2.6	substituted upon mutual agreement thereto by both partie The parties agree to preserve for a period of at least three (3) years or such longer period as may be required public authority, all test data, if any, and other simil records. Shipper or Transporter may install, maintain, and operat odorizing (at a Delivery Point only), regulating, telemetering, heating and fogging equipment at its own expense as it shall desire at each Receipt Point or Delivery Point, and the operator of such equipment at it own expense shall provide the other party a suitable sit therefore and allow the other party free access to and u of the site; provided that such equipment shall be so installed, maintained and operated as not to interfere w the operation or maintenance of the operating party's measuring equipment at each Receipt Point or Delivery Point. All such equipment as Shipper or Transporter sha desire to install shall be constructed, installed and operated to conform to the other party's requirements.

٦

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)			
3.	QUALITY OF GAS			
	3.1 Except as expressly set forth herein to the contrary, the gas received or delivered by Transporter shall be a combustible gas consisting of methane and such other hydrocarbon constituents or a mixture of two or more of them which, in any case, meets the following qualify specifications:			
	 (A) The gas shall have a total heating value not less than one thousand (1,000) Btu per cubic foot of gas nor greater than one thousand seventy-five (1,075) Btu per cubic foot of gas; 			
	(B)Transporter may not refuse to accept delivery of gas with a hydrocarbon dew point equal to or less than 15 degrees Fahrenheit, provided that such gas satisfies all other applicable provisions of Transporter's Tariff.			
	(1) To the extent operationally practicable through aggregation or other reasonable means, Transporter may accept gas with a higher hydrocarbon dew point than that established in 3.1(B), but not exceeding (.04) gallons per Mcf (GPM)of C6+.			
	(C) The gas shall be commercially free, under continuous gas flow conditions, from objectionable odors, solid matter, dust, gums, gum-forming constituents, water or any other solid or liquid matter which might cause damage to or interference with proper operations of the pipeline, compressor stations, meters, regulators or other appliances through which the gas flows;			
	(D) The gas shall not have uncombined oxygen content in excess of two- tenths (0.2) of one percent (1%) by volume;			
	(E) The gas shall not contain more than three (3.0%) by volume, of a combined total of carbon dioxide and nitrogen;			
	(F) The gas shall not contain more than one-quarter (0.25) grain of hydrogen sulfide per one-hundred (100) cubic feet;			
	(G) The gas shall not contain more than ten (10) grains of total sulphur, excluding any mercaptan sulphur, per one-hundred (100) cubic feet;			

Issued by: Issued on:

Γ

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(H) The flowing ga	as shall not have a temperature of more than one-
hundred twenty	y degrees (120) Fahrenheit or less than forty (40)
degrees Fahren	nheit.
(I) The gas shall the temperatum delivered.	be free of water and hydrocarbons in liquid form at re and pressure at which the gas is received and
(J) The gas shall	not contain in excess of seven (7) pounds of water
vapor per mill	lion cubic feet;
(K) The gas shall	not contain, either in the gas or in any liquids with
the gas, any r	microbiological organism, active bacteria or bacterial
agent capable	of contributing to or causing corrosion and/or
operational ar	md/or other problems. Microbiological organisms,
bacteria or ba	acterial agents include, but are not limited to,
sulfate reduci	ing bacteria (SRB) and acid producing bacteria (APB).
Tests for bact	teria or bacterial agents shall be conducted on samples
taken from the	e meter run or the appurtenant piping using American
Petroleum Inst	titute (API) test method API-RP38 or any other test
method accepta	able to Transporter and Shipper which is currently
available or n	may become available at any time.
3.2 The test equipmen	t and methodology utilized by Transporter to determine
whether gas meets	the qualify specifications set forth in Section 3.1
shall be posted o	on its LINK® System.
3.3 At Transporter's obtain and provid kept by upstream for Shipper's acc ensure and verify using appropriate specifications ap Section 3.	request, Shipper shall use all reasonable efforts to be to Transporter all records regarding gas quality pipelines transporting the gas received by Transporter count. Shipper shall use all reasonable efforts to for Transporter that such upstream pipelines are equipment to monitor compliance with the gas quality plicable on Transporter's system as stated in this
3.4 If the gas tender	ed for Shipper's account to Transporter shall fail at
any time to confo	orm to any of the specifications set forth in this
Section 3 or in T	transporter's reasonable judgment, may cause harm to
its facilities or	diminish the quality of gas in the system, then
Transporter shall	have the right, after either written, oral or
telephonic notice	to Shipper, to refuse to accept all or any portion of
such quality defi	cient gas. In the event Transporter refuses to accept

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

gas tendered by Shipper because such gas does not conform to the specifications set forth herein, Shipper shall not be relieved of its obligation to pay any Reservation Charge provided for in Shipper's service agreement. If the gas tendered by Transporter for Shipper's account shall fail at any time to conform to any of the specifications set forth in this Section 3 then Shipper shall notify Transporter of such deficiency and may, at its option, refuse to accept delivery pending correction by Transporter.

- 3.5 Transporter may waive the requirements set forth in Section 3.1 in order to allow Shipper to tender or cause to be tendered, gas which does not when injected into Transporter's pipeline meet the quality specifications set forth in Section 3.1; provided that Transporter's acceptance of such gas shall not adversely impact Transporter's system facilities or operations, and further provided that once such gas has been blended, to the extent blending occurs, the commingled gas stream at any delivery point on Transporter's system shall be compliant with the quality specifications set forth in Section 3.1. Transporter shall implement this Section 3.5 on a non-discriminatory basis and may cancel any waiver at any time if necessary to assure that the commingled gas stream is compliant with the quality specifications set forth in Section 3.1 at any delivery point on Transporter's system.
- 3.6 Odorization. Transporter shall have no obligation to odorize the gas tendered by Shipper other than to conform to the regulations of appropriate governmental authorities having jurisdiction. However, if Transporter odorizes the gas, such odorization shall be by use of a malodorant agent of such character as to indicate by a distinctive odor the presence of gas. Whenever odorized gas is delivered, the quality and specifications, as set forth in this Section 3, of such gas shall be determined prior to the addition of malodorant or with proper allowance for changes or additions due to such malodorant. Such odorization of the gas by Transporter shall be for the purpose of detection of the gas only during the time when the gas is in the possession of Transporter, prior to delivery to the Shipper.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)			
4. NOM	INATIONS		
4.1	Transporter shall accept nominations twenty-four (24) hours a day via the LINK® System or EDM. All nominations must contain the mandatory data elements included in the NAESB standards and any additional business-conditional or mutually agreeable data elements applicable to Transporter's services. All nominations shall include Shipper defined begin dates and end dates. All nominations excluding intra-day nominations should have roll-over options. Specifically, Shippers have the ability to nominate for several Days, Months, or years, provided the nomination begin and end dates are within the term of the Shipper's Agreement. Nominations under Rate Schedule FTS must be for a minimum period of one (1) Day, and all quantities must be stated in Dths. Nominations under Rates Schedule ITS or PALS must specify the daily scheduled quantity, and all quantities must be stated in Dekatherms. At the end of each Day, Transporter will provide the final scheduled quantities for the just completed Day. With respect to the implementation of this process via NAESB Standard No. 1.4.x scheduled quantity related standards, Transporter should send an end of Day Scheduled Quantity document. Receivers of the end of Day Scheduled Quantity document can waive the sender's sending of the end of Day Scheduled Quantity document.		
	(a) All nominations must be communicated via the LINK® System or EDM unless otherwise mutually agreed and must be submitted in accordance with the standard nomination timelines set forth below. A revised nomination supersedes the previous nomination in effect, but only for the Days specified in such revised nomination, after which the previous nomination once again takes effect until its end date or time or until superseded by another new or revised nomination, whichever is earlier.		

Issued by: Issued on:

GEN	ERAL TERMS AN (CONTIN)	ID CONDITIONS UED)
The s	standard nomi	nation timelines are as follows:
(1)	The Timely (All times Service Day	Nomination Cycle: are CCT on the day prior to the)
	11:30 a.m.	Latest time that nominations may leave control of the nominating party;
	11:45 a.m.	Receipt of nominations by Transporter (including from Title Transfer Tracking Service Providers (TTTSPs));
	12:00 p.m.	Transporter sends quick response;
	3:30 p.m.	Receipt of completed confirmations by Transporter from upstream and downstream connected parties;
	4:30 p.m.	Receipt of scheduled quantities by Shipper and point operator.
Sche Nomi on t	duled quantit nation Cycle he next Servi	ties resulting from the Timely shall be effective at 9:00 a.m. CCT .ce Day.
(2)	The Evening on the Day	y Nomination Cycle (All times are CCT prior to the Service Day.)
	6:00 p.m.	Latest time that nominations may leave control of the nominating party;
	6:15 p.m.	Receipt of nominations by Transporter (including from TTTSPs);

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)			
	6:30 p.m.	Transporter sends quick response;	
	9:00 p.m.	Receipt of completed confirmations by Transporter from upstream and downstream connected parties;	
	10:00 p.m.	Transporter to provide scheduled quantities to affected Shippers and point operators, and to provide scheduled quantities to bumped parties (notice to bumped parties).	
Sched Nomin Trans shoul Servi anoth it is be ef Day.	uled quantit ation that d porter to re d be effecti ce Day; and er Shipper o being bumpe fective at 9	ies resulting from an Evening oes not cause another Shipper on ceive notice that it is being bumped ve at 9:00 a.m. CCT on the next when an Evening Nomination causes n Transporter to receive notice that d, the scheduled quantities should :00 a.m. CCT on the next Service	
(3)	The Intra-d are CCT on	ay 1 Nomination Cycle: (All times the Service Day.)	
	10:00 a.m.	Latest time that nominations may leave control of the nominating party;	
	10:15 a.m.	Receipt of nominations by Transporter (including from TTTSPs);	
	10:30 a.m.	Transporter sends quick response;	
	1:00 p.m.	Receipt of completed confirmations by Transporter from upstream and downstream connected parties;	

Original Sheet No. 49

GENERAL TERMS A (CONTI	ND CONDITIONS NUED)
2:00 p.m.	Transporter to provide scheduled quantities to affected Shippers and point operators, and to provide scheduled quantities to bumped parties (notice to bumped parties).
Scheduled quanti Nomination Cycle on the same Serv	ties resulting from the Intra-day 1 shall be effective at 5:00 p.m. CCT ice Day.
(4) The Intra- are CCT on	Day 2 Nomination Cycle: (All times the Service Day.)
5:00 p.m.	Latest time that nominations may leave control of the nominating party;
5:15 p.m.	Receipt of nominations by Transporter (including from TTTSPs);
5:30 p.m.	Transporter sends quick response;
8:00 p.m.	Receipt of completed confirmations by Transporter from upstream and downstream connected parties;
9:00 p.m.	Transporter to provide scheduled quantities to affected Shippers and point operators.
Scheduled quanti Nomination Cycle on the same Serv during the Intra	ties resulting from the Intra-Day 2 shall be effective at 9:00 p.m. CCT ice Day. Bumping is not allowed -Day 2 Nomination Cycle.
For purposes of 4.1(a)(4) above, transmittals pur the designated s of transmittal,	Sections 4.1(a)(2), 4.1(a)(3), and "provide" shall mean, for suant to standards 1.4.x, receipt at ite, and for purposes of other forms it shall mean send or post.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(d)	Shipper shall include in its nominations the desired order of priority of receipts and deliveries under each Agreement which Transporter will use when taking action to change receipts and/or deliveries according to Section 5.2. The order of priority shall indicate that a priority of one (1) shall be the last to be affected by changes. Nominations with the same priority will be adjusted pro rata.
(c)	If Shipper completes and resubmits an otherwise incomplete nomination, then Transporter will process the nomination in the first nomination cycle that occurs where the Shipper's complete nomination meets the deadline for nominations.
(d)	Variations by Shipper of actual receipts and deliveries from the nominated receipts and deliveries shall be kept to a minimum. Receipts and deliveries shall be made at uniform hourly rates unless provisions to deliver the Gas at a non-uniform rate are confirmed by Transporter's Gas Control Department prior to Gas flowing. Under no circumstances shall Transporter be obligated to deliver to any Shipper, on any Day, a quantity of Gas under any Agreement greater than Transporter received at the Receipt Point(s) on behalf of such Shipper under such Agreement.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)
(e) Any shipper may designate an agent, which may be Transporter, to nominate and schedule transportation service on Shipper's behalf. Shipper shall notify Transporter, in writing or via the LINK® System, of the designated agent. Transporter is authorized to rely on nominations and scheduling information provided by Shipper's agent. By designating an agent, Shipper agrees to indemnify and save Transporter harmless from all suits, actions, debts, accounts, damages, costs, losses and expenses arising in any way from Shipper's agent's actions on behalf of Shipper, Shipper's agent's failure to act on behalf of Shipper, or Transporter's reliance upon the information provided to Transporter by Shipper's agent; provided, however that such indemnification will not excuse Transporter from liability for actions taken when Transporter is acting as agent.
4.2 Implementation of Intra-day Nominations.
 (a) Subject to the deadlines in Section 4.1(a)(2),(3) or (4), above, intra-day nominations may be nominated twenty-four (24) hours a Day and will be processed in the same manner as other nominations. However, the nomination deadline and effective time of intra-day nominations specified in Section 4.1 (a) will not apply to OFO related intra-day nominations.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)
(b) Subject to upstream and downstream operators' confirmations and Transporter's operating conditions, an intra-day nomination submitted pursuant to one of the deadlines set forth in Section 4.1(a) above can be used to request increases or decreases in total flow, changes to Receipt Points, or changes to Delivery Points of scheduled Gas.
(c) With respect to intraday nominations for reductions in previously scheduled quantities, Transporter will accept, subject to the limitations set forth in Section 4.1(a), any explicitly confirmed quantity, down to and including zero, for such intraday nomination; provided, however, if such intraday nomination requires confirmation from an upstream and/or downstream interconnected pipeline then any intraday nomination to reduce previously scheduled quantities will be subject to, and limited to, the reduced quantity confirmed by such upstream and/or downstream interconnected pipeline.
(d) Transporter shall allow Shipper to alter the order of priority of receipts and deliveries upon which Transporter shall rely in taking actions to adjust receipts and/or deliveries under Section 4.1 above, provided that such changes are submitted via the LINK® System or EDM in accordance with the nomination deadlines set forth in 4.1(a), above.
(e) Notice. For purposes of providing notice of any nomination changes (including where an interruptible Shipper's nomination is bumped by a firm Shipper's intra-day nomination) to a Shipper and/or Shipper's agent, Transporter shall use Electronic Communication.

Issued on:

Original Sheet No. 53

_____]

	GENERAL TERMS AND CONDITIONS (CONTINUED)
5. PRIOF	RITY OF SERVICE
5.1	Transporter shall have the right to curtail or discontinue services, in whole or in part, on all or a portion of its system at any time for reasons of Force Majeure or when capacity or operating conditions so require, or it is necessary to make modifications, repairs or operating changes to its system. Transporter shall provide Shipper notice of such curtailment as is reasonable under the circumstances. Notwithstanding anything to the contrary contained in this Section 5.1, Transporter will schedule routine repairs and maintenance in a manner that to the greatest extent possible will not disrupt the flow of quantities scheduled and confirmed in accordance with Section 4 of the General Terms and Conditions.
5.2	For each nomination cycle, Transporter shall allocate capacity on the basis of nominations made by Shippers, utilizing the priorities of service, from highest to lowest, as set forth below:
	(a) Priority Class One. Rate Schedule FTS Primary Receipt Point(s) and Primary Delivery Point(s) within MDQ.
	(b) Priority Class Two. Rate Schedule FTS. Secondary Delivery Point(s) within MDQ within the Transportation Path.
	(c) Priority Class Three. Rate Schedule FTS. Secondary Delivery Point(s) within MDQ and outside the Transportation Path.

Original Sheet No. 54

GENERAL	TERMS	AND	CONDITIONS
	(CONT	INUE	D)

- (d) Priority Class Four. Rate Schedule ITS, Rate Schedule PALS.
- (e) Priority Class Five. Make-up Gas scheduled at Transporter's discretion.

For purposes of determining whether points are located within the Transportation Path, Transporter shall consider a movement of gas from a Receipt Point to a Delivery Point which is counter to the gas flow contemplated by the Primary Receipt Point(s) and Primary Delivery Point(s) as outside the Shipper's Transportation Path. In addition, for any movement of gas that traverses a segment(s) in which the total nominated quantity for that contract exceeds the Transportation Path MDQ, the quantity in excess of the contractual entitlement shall be deemed to be outside of the Shipper's Transportation Path.

Issued by: Issued on:

Original Sheet No. 55

		GENERAL TERMS AND CONDITIONS (CONTINUED)
6.	SCHEI	DULING AND CURTAILMENT
	6.1	Scheduling Capacity during a Start of Day Nomination Cycle.
		(a) Transporter shall allocate its pipeline capacity as well as each Receipt Point and each Delivery Point capacity on the basis of the priority classes listed in Section 5 above as follows:
		(i) prorata for Priority Class One nominations; then
		 (ii) prorata for Priority Class Two; then (iii) prorata for Priority Class Three; then (iv) on the basis of Confirmed Price for Priority
		(v) make-up gas for FTS Agreements, then make-up gas for ITS Agreements.
		(b) Ties within any Priority Class shall be allocated pro rata based on nominations.
	6.2	Scheduling Available Capacity during an Intra-day Nomination Cycle. Transporter shall schedule available capacity during each of the Intra-day Nomination Cycles in accordance with Section 6.1 above. Bumping of service is not allowed during the Intra-Day 2 Nomination Cycle which is effective at 9:00 p.m. CCT on the same service Day and all cycles thereafter for the remainder of the Day.
Issued by: Issued on:		Effective on:
	GENERAL TERMS AND CONDITIONS (CONTINUED)	
-----	---	
6.3	Curtailment of Scheduled Volumes during a Day. If, at any time, Transporter determines that the capacity of its system, or portion(s) thereof, is insufficient to serve all scheduled service, or to accept the quantities of Gas tendered, capacity which requires curtailment shall be curtailed so as to provide the service which is feasible in the order prescribed for Scheduling in Section 6.1 above; provided, however, once scheduled, Priority Class Two and Priority Class Three will have the same curtailment priority as Priority Class One; and provided, further, if a capacity constraint occurs on the upstream or downstream system which results in a curtailment, the upstream or downstream operator shall determine the change in scheduled nominations of its Shippers. Such change in scheduled nominations shall be confirmed via the LINK® System or EDM. To enable prompt action in an emergency situation where capacity is insufficient, Transporter shall have the authority to take all necessary and appropriate actions, as then may appear necessary, to preserve the operational integrity of its system. Transporter shall notify Shippers of the existence of any such emergency situation by use of Electronic Communication, as scon as it is reasonably practicable.	
6.4	Segmentation of Capacity by Nomination. Any Shipper receiving Transportation Service under Rate Schedule FTS shall have the right to segment its firm capacity by utilizing multiple Receipt Points and Delivery Points. The right to segment is subject to the requirement that a Shipper's firm capacity utilization pursuant to its Rate Schedule FTS Agreement and, if such Agreement is the result of capacity release, the firm capacity utilization of all other Shippers of capacity rights derived from the initial Rate Schedule FTS Agreement, does not exceed, in the aggregate (based on all relevant Shipper firm capacity utilization), the contract entitlements of the initial Rate Schedule FTS Agreement in any segment or at any point (including, without limitation, the relevant MDQ) where the nominated segments overlap. For the purpose of determining whether there is an overlap of MDQ, a forward haul and a Backhaul nominated to the same	

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

Delivery Point at the same time shall not be deemed to be an overlap at that point. For the purpose of determining whether there is an overlap of MDQ on a segment, a forward haul and a Backhaul nominated on the same segment at the same time shall be deemed to be an overlap on the segment. As a general matter, Shipper will have the right to segment its capacity so long as it is utilizing its Primary Receipt Point(s) and Primary Delivery Point(s), as well as all Secondary Delivery Points, as long as such use does not impair Transporter's ability to render firm Transportation Service, does not adversely affect Shippers' firm Transportation Service rights, and/or does not adversely affect the safe and reliable operation of Transporter's pipeline system.

6.5 Segmentation of Capacity by Capacity Release. Releasing Shippers can also segment capacity through capacity release in accordance with Section 25 of the General Terms and Conditions of Transporter's Tariff, subject to the requirement that the release (or multiple releases) does not increase the total contract entitlements in any segment or at any point (including, without limitation, the relevant MDQ) above the contract entitlement of the initial Rate Schedule FTS Agreement. For the purpose of determining whether there is an overlap of MDQ, a forward haul and a Backhaul nominated to the same Delivery Point at the same time shall not be deemed to be an overlap at that point. As a general matter, Shipper will have the right to segment its capacity so long as it is utilizing its Primary Receipt Point(s) and Primary Delivery Point(s), as well as all Secondary Delivery Points, as long as such use does not impair Transporter's ability to render firm Transportation Service, does not adversely affect Shippers' firm Transportation Service rights, and/or does not adversely affect the safe and reliable operation of Transporter's pipeline system and/or does not result in quantities being nominated in any manner that is inconsistent with Section 25.1(a) of these General Terms and Conditions.

Issued by: Issued on:

	GENERAL TERMS AND CONDITIONS (CONTINUED)
7.	DETERMINATION OF DAILY ALLOCATED RECEIPTS AND DELIVERIES
	7.1 Allocation of Receipts/Deliveries. Unless Transporter and Operator mutually agree to allocate deliveries each Day using ranked, pro rata, percentage, swing or operator provided value methodologies, such deliveries will be allocated through the meter pro rata, to the extent applicable, based on confirmed nominations.
	Operator shall notify Transporter via LINK® after or durin confirmation and before start of the Day, that it desires to establish allocation priorities at Receipt and/or Delivery Points using any of the following methodologies: ranked, prorata, percentage, swing or Operator provided value provided, however, Transporter will not be required to agree to any of such allocation methodologies if they are operationally or administratively infeasible.
	Transporter shall advise such Operator of the confirmed nominations at such Receipt/Delivery Point.
	The Operator shall establish separate allocation prioritie for over and under production at the level of detail that the confirmed nominations are provided, and advise Transporter of such priorities via LINK® before the beginning of the Day. Any confirmed nominations that do not have established allocation priorities shall be prorated based upon confirmed nominations and shall be allocated after all confirmed nominations that have established allocation priorities.

GENERAL TERMS AND CONDITIONS (CONTINUED)
In the case of under production, such allocation priorities shall be used by Transporter to allocate Gas, such that Transporter shall allocate Gas to each Shipper, in order of priority designated by the Operator, up to the full nomination of that Shipper, until the entire gross measured volume at such Receipt/Delivery Point is allocated.
In the case of over production, such allocation priorities shall be used by Transporter to allocate Gas, such that Transporter shall allocate Gas to each Shipper, in order of priority designated by the Operator, equal to the full nomination of that Shipper, with any over produced quantities being allocated to the Shipper(s) with the lowest priority, until the entire gross measured volume at such Receipt/Delivery Point is allocated.
Simultaneous Receipts and Deliveries. To the extent that both receipts and deliveries have been nominated at the same meter for any day:
If the actual flow through the meter represents a delivery by Transporter, then the nominated receipts shall be allocated as nominated and the sum of such receipts shall be added to the metered quantity before any allocation is made in accordance with Section 7.1; or
If the actual flow through the meter represents a receipt by Transporter, then the nominated deliveries shall be allocated as nominated and the sum of such deliveries shall be added to the metered quantity before any allocation is made in accordance with Section 7.1.

Issued by: Issued on:

r-----

	GENERAL TERMS AND CONDITIONS (CONTINUED)
7.2	Prior Period Adjustments.
	In accordance with the provisions of Sections 2 and 11 of these General Terms and Conditions, Transporter shall use the best information available to close its allocation of quantities for a Service Month five (5) Business Days after such Service Month. To the extent that adjustments are made after the date of such close, such adjustments ("Prior Period Adjustments" or "PPA") shall be treated under this Section 7.2. If the PPA is due to the correction of measurement data or allocations, such adjustments shall be processed within six (6) Months of the applicable Service Month. If the affected party disputes the as-adjusted quantity, it is entitled to rebut the basis for the PPA, but only if it does so within three (3) Months of the processing of the PPA. Notwithstanding the above-specified deadlines for processing/rebutting PPAs, such deadlines shall not apply in the case of deliberate omission or misrepresentation or mutual mistake of fact. Parties' other statutory or contractual rights shall not be diminished by this standard.
Issued by: Issued on:	Effective on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
7.3	Trespass Gas. Gas that is received by Transporter during a Service Month at a Receipt Point for which there is no valid nomination shall be considered Trespass Gas. If Transporter receives Trespass Gas during a Service Month, it shall post such fact on LINK®, including the location and quantity of such Trespass Gas, for a period of thirty (30) Days after the end of the Service Month. The owner of such Trespass Gas may claim such Gas by informing Transporter in writing of such fact and by having the ownership verified by the Operator of the facilities upstream of the Receipt Point. Upon receiving a valid claim of ownership, Transporter shall first give the claimant the opportunity to move the Gas off of Transporter's system upon payment of the applicable Transportation and PALS charges. Alternatively, the claimant may request payment of an amount (as full consideration, inclusive of taxes and any other amounts) equal to the product of the quantity of Trespass Gas times the Low Common price (as determined pursuant to Section 8.7 of the General Terms and Conditions) for the Service Month in which the Trespass Gas was received. If there is no valid claim for such Trespass Gas within such thirty (30) day posting period, Transporter shall be allowed to retain such Trespass Gas.
7.4	Conversion of Gas. Any party that takes Gas without Transporter's authorization shall be liable for paying the High Common price (as determined pursuant to Section 8.7 of the General Terms and Conditions) for the Month in which the Gas was taken, in addition to any other costs, losses, and damages attributable to such taking, in addition to any legal remedies otherwise available. Any penalty revenues received by Transporter as a result of the operation of Sections 7.3 and 7.4 above will be credited pursuant to Sections 23.2 and 23.3 of the General Terms and Conditions.

Issued by: Issued on:

			GENERAL TERMS AND CONDITIONS (CONTINUED)
8.	IMBAL	ANCE F	ESOLUTION PROCEDURES
	8.1	For t shall 7 of Trans mean After rende betwe of Ca State provi	the purposes of this Section 8, "Receipt" or "Receipts" mean quantities of Gas allocated pursuant to Section these General Terms and Conditions, net of porter's Use, and "Delivery" or "Deliveries" shall quantities of Gas allocated pursuant to Section 7. The end of each Service Month, Transporter shall er to Cashout Party a statement detailing any imbalance een Monthly Receipts and Monthly Deliveries under all ashout Party's Transportation Agreements ("Imbalance ement"), subject to Transporter's Billing and Payment sions contained in Sections 9 and 10 of these General and Conditions.
	8.2	Cumul to th	ative Daily Transportation Imbalances shall be subject he following imbalance resolution procedures.
		(a)	Definition of Transportation Imbalance: "Transportation Imbalance" shall mean the difference between a Shipper's allocated Receipts and allocated Deliveries under any firm or interruptible Agreement. All imbalances will be calculated on a daily basis and designated to be at the applicable Receipt Point.
		(b)	Definition of an Imbalance Due Cashout Party: "Due Cashout Party" shall mean that Deliveries under an Agreement at the Delivery Point are less than Receipts at the Receipt Point, adjusted for Transporter's Use; such difference in quantity is "Due To" a Cashout Party (or its Agent).
		(c)	Definition of an Imbalance Due Transporter: "Due Transporter" shall mean that Deliveries under an Agreement at the Delivery Point exceed Receipts at the Receipt Point, adjusted for Transporter's Use; such difference in quantity is "Due From" a Cashout Party (or its Agent).

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

-

 8.3 Netting: For each Month, all cumulative Transportation Imbalances within an Operational Impact Area will be netted among each of Cashout Party's firm and interruptible Agreements. 8.4 Trading: Posting and trading of the previous Month's netted Transportation Imbalances will be allowed within each Operational Impact Area between imbalance agents (or the Cashout Party, if no imbalance agent exists) from the first calendar Day of the current Month until the end of the 17th Business Day of the current Month until the each of the 17th Business Day of the current Month until the each of the 17th Business Day of the current Month until the set posted for trading should be authorized by the Cashout Party. Authorizations to post imbalances that are received by Transporter by 11:45 A.M. should be effective by 8:00 A.M. the next Business Day (Central Clock Time). Imbalances previously authorized for posting should be posted on or before the ninth Business Day of the month. Transporter should provide the ability to view and, upon request, download posted imbalances. Transporter should not be required to post zero imbalances. When trading imbalances, a quantity should be specified. Trading will be allowed 	GENERAL TERMS AND CONDITIONS (CONTINUED)
8.4 Trading: Posting and trading of the previous Month's netted Transportation Imbalances will be allowed within each Operational Impact Area between imbalance agents (or the Cashout Party, if no imbalance agent exists) from the first calendar Day of the current Month until the end of the 17th Business Day of the current Month. Imbalances to be posted for trading should be authorized by the Cashout Party. Authorizations to post imbalances that are received by Transporter by 11:45 A.M. should be effective by 8:00 A.M. the next Business Day of the month. Transporter should provide the ability to view and, upon request, download posted imbalances. Transporter should not be required to post zero imbalances. When trading imbalances, a quantity should be specified. Trading will be allowed	8.3 Netting: For each Month, all cumulative Transportation Imbalances within an Operational Impact Area will be netted among each of Cashout Party's firm and interruptible Agreements.
	8.4 Trading: Posting and trading of the previous Month's netted Transportation Imbalances will be allowed within each Operational Impact Area between imbalance agents (or the Cashout Party, if no imbalance agent exists) from the first calendar Day of the current Month until the end of the 17th Business Day of the current Month. Imbalances to be posted for trading should be authorized by the Cashout Party. Authorizations to post imbalances that are received by Transporter by 11:45 A.M. should be effective by 8:00 A.M. the next Business Day (Central Clock Time). Imbalances previously authorized for posting should be posted on or before the ninth Business Day of the month. Transporter should provide the ability to view and, upon request, download posted imbalances. Transporter should not be required to post zero imbalances. Mhen trading imbalances, a quantity should be specified. Trading will be allowed

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Issued by: Issued on:

Original Sheet No. 65

GENERAL TERMS AND CONDITIONS (CONTINUED) 8.5 Final Resolution of Transportation Imbalances: If Cashout Party has a Transportation Imbalance remaining after the close of the trading period, such Transportation Imbalance will be cashed out in accordance with the cashout provisions set forth in Section 8 herein. 8.6 All balancing shall be based on the applicable Delivery Point within an Operational Impact Area. Cashout Party or its Agent(s) may nominate transactions (in accordance with Section 4 of the General Terms and Conditions) during the Month to correct Transportation Imbalances within an Operational Impact Area. Transporter's ability to receive or deliver imbalance quantities shall be dependent upon Transporter's physical operations, and Transporter is under no obligation to allow Receipt or Delivery of such quantities for resolution of Transportation Imbalances if it determines, such activity would jeopardize pipeline operations. Issued by:

Issued on:



Original Sheet No. 67

shall determine the relevant cashout prices using another similar publication selected by Transporter, in its reasonable judgment, that is broadly published and widely accepted within the natural gas industry as a reliable source for the quotation of Gas prices.

Issued by: Issued on:

	GENEI	RAL TERMS AND CONDITIONS
		(CONTINUED)
(۲	Imbala imbala Transp excess calcul the ap Party' 5%, th averac 8.7(a) calcul for th Monthl cashou the re and un accour	ance Due Transporter. In the event a monthly ince is an Imbalance Due Transporter, porter shall charge Cashout Party for such a Deliveries plus an allowance for fuel ated by multiplying such excess Deliveries by oplicable Transporter's Use %. If a Cashout s monthly imbalance is less than or equal to be monthly cashout bill will be based on the ge Midpoint price contemplated in Section . The Monthly System Imbalance will be sated at the same time the initial cashout bill at Month is generated. If a Cashout Party's by Imbalance is greater than 5%, the monthly at bill will be based on the accumulated sum of esults of the formulas listed below such that, atil, the total Monthly Imbalance is fully ated for:
Imbalance		
Level	Factor	Applicable Cashout Price
0% - = <5%	1.00	average Midpoint
> 5% - =<10%	1.10	(High Common x quantity > 5%) + level above
>10% - =<15%	1.20	(High Common x quantity >10%) + levels above
>15% - =<20%	1.30	(High Common x quantity >15%) + levels above
>20% - =<25%	1.40	(High Common x quantity >20%) + levels above
>25%	1.50	(High Common x quantity >25%) + levels above
	For pur Factor determ imbala the in at the Receip Delive For OF Section to exc Transp calcul by the Rates the Mo	arposes of determining the appropriate cashout c, Cashout Party's imbalance level shall be mined by taking the lower of (a) the level of ance supplied pursuant to Section 24.2, or (b) abalance computed by comparing (i) the Deliveries a Delivery Point and (ii) the Receipts at the point and by dividing the amount of the excess eries by the Receipts less the Transporter's Use. BA imbalances that are resolved pursuant to this on 8, the calculation of cashout charges relating cess Deliveries shall also include a portation imbalance charge, which shall be ated by multiplying the excess Delivery quantity a actual weighted average of all applicable Usage owed on all quantities of Gas delivered during onth to that OBA Party.

[____

Original Sheet No. 69

-

	GENERAL TERMS AND CONDITIONS (CONTINUED)	
(c)	Imbalance Due Cashout Party. Monthly Imbalance which is an Party, Transporter shall make Cashout Party reflecting such	In the event of a Imbalance Due Cashout a cashout payment to excess Receipts.

	If a or eq based Secti Imbal bill resul until for:	Cashout Party's Monthly Imbalance is less that ual to 5%, the monthly cashout bill will be on the average Midpoint price contemplated : on 8.7(a). If a Cashout Party's Monthly ance is greater than 5%, the monthly cashout will be based on the accumulated sum of the ts of the formulas listed below such that, an , the total Monthly Imbalance is fully accourt
Imbalance Level	Factor	Applicable Cashout Price
0% - = <5%	1.00	average Midpoint
>5% - =<10%	.90	(Low Common x quantity > 5%) + level above
>10% - =<15%	.80	(Low Common x quantity >10%) + levels above
>15% - =<20%	.70	(Low Common x quantity >15%) + levels above
>20% - =<25%	.60	(Low Common x quantity >20%) + levels above
	Fol p Facto deter imbal the i Deliv Recei exces Trans resol of th exces imbal multi weigh on al to th respo the i resol	r, Cashout Party's imbalance level shall be mined by taking the lower of (a) the level o ance supplied pursuant to Section 24.2, or (i mbalance computed by comparing (i) the eries at the Delivery Point and (ii) the pts at the Receipt Point and by dividing the s Receipts by the total Receipts less porter's Use. For OBA imbalances that are ved pursuant to this Section 8, the calculat e amount due the Cashout Party relating to s Receipts shall also include a Transportati ance credit, which shall be calculated by plying the excess Receipt quantity by the ac ted average of all applicable Usage Rates ow 1 quantities of Gas delivered during the Mon at OBA Party. Transporter shall have no nsibility for the distribution of funds beyo nitial distribution, in accordance with this utions procedure, to the Cashout Party.

Issued by: Issued on:

Original Sheet No. 71

GENERAL	TERMS	AND	CONDITIONS
	(CONT	INUE	D)

- (d) A Cashout of Transportation Imbalances at prices above or below the average Midpoint price shall not occur if it has been determined that such Transportation Imbalances are due to Transporter's negligence. Additionally, a Cashout of Transportation Imbalances due to Imbalance Due Transporter quantities or Imbalance Due Cashout Party quantities shall be limited to the average Midpoint price if such imbalances occurred during circumstances of Force Majeure that directly affect the Transporter's or upstream or downstream facilities over which Gas is transported under the applicable Agreement, or during circumstances of Force Majeure that directly affect Shipper's facilities for the period until Shipper has an opportunity to adjust its nominations (Shipper shall give written notice within forty-eight (48) hours of such Force Majeure event), or were the direct result of an OFO issued to the Shipper or its supplier.
- 8.8 Cashout of Transportation Imbalances at Agreement Expiration. At the time of expiration of an Agreement, all Transportation Imbalances shall be resolved pursuant to the provisions of Section 8.7 above.
- 8.9 Annual System Cashout Mechanism. Transporter shall establish an annual mechanism to determine the costs of implementing this Cashout provision. Such mechanism shall calculate, on a system-wide basis, the annual gross balance (positive or negative) derived from the Cashout program, which will be accounted for and disposed of in accordance with Section 22.3 of the General Terms and Conditions.

Issued by: Issued on:



Original Sheet No. 72

Original Sheet No. 73

GENERAL TERMS AND CONDITIONS (CONTINUED) 9.3 Both Transporter and Shipper shall have the right to examine at any reasonable time the applicable records of the other to the extent necessary to verify the accuracy of any statement made under or pursuant to the provisions of the Agreement. Upon receipt of a request, the recipient will either send the relevant information to the requestor or will provide the requestor the right to review such information in the recipient's offices.

Issued by: Issued on:



	GENERAL TERMS AND CONDITIONS (CONTINUED)
10.3	In the event an error is discovered in the amount billed in any statement rendered by Transporter, such error shall be adjusted within 30 Days of the determination of the error; provided that any claim therefore shall have been made within 60 Days of discovery of such error and, in any event, within 6 Months from the date of the statement claimed to be in error. Billing errors shall be corrected as follows:
	(a) Where Shipper has been overcharged and has paid the statement, in the event the overcharge is not the result of Transporter's negligence or bad faith, fraud or willful misconduct, the amount of the overpayment will be refunded to Shipper without interest provided the overpayment is refunded within 30 Days. Overpayments not refunded within 30 Days will be subject to interest charges at the interest rate determined in accordance with Section 154.501(d) of FERC's regulations from the date of the overpayment to the date of the refund. Where the refund is provided to Shipper by way of credit on a subsequent invoice rendered to Shipper by Transporter, the overpayment will be deemed to have been refunded on the date the credited invoice was received by Shipper.

٦

	GENERAL TERMS AND CONDITIONS (CONTINUED)	
) Where Shipper has been undercharged by Transport Shipper will pay the amount of the undercharge without interest provided the undercharge is paid within 30 Days. Undercharge amounts not paid wit 30 Days will be subject to interest charges at to interest rate determined in accordance with Sect 154.501(d) of FERC's regulations from the date of statement. Shipper shall have the right to revi- all records pertaining to its performance under Shipper's Agreement to verify the amount payable Shipper to Transporter under the Agreement in an Month, so long as such review shall be completed within two years following the end of the calend year in which such amount is payable. Such revie shall be conducted during normal business hours written request to Transporter and at Shippers' expense.	ter, id thin the tion of the iew e by ny d dar ew , upon own
10.4	an invoice is in dispute, Shipper shall pay the post of in dispute and provide documentation identifying the sis for the dispute. If Shipper in good faith:	rtion the
	a) disputes the amount of any such bill or part th	ereof;
	pays to Transporter such amounts as it concedes correct;	to be
	provides Transporter with a written notice incl a full description of the reasons for the dispu together with copies of supporting documents; a	uding te, nd

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(d) at any time thereafter within 30 Days of a demand made by Transporter furnishes good and sufficient surety bond, guaranteeing payment to Transporter of the amount ultimately found due upon such bill after a final determination which may be reached either by agreement or judgement of the courts, as may be the case, then Transporter shall not be entitled to suspend further services because of such non-payment pursuant to Section 10.2 unless and until default be made in the conditions of such bond.
10.5	In the event that Shipper does not pay the full amount due Transporter in accordance with this Section 10, Transporter, without prejudice to any other rights or remedies it may have, shall have the right to withhold and set off payment of any amounts of monies due or owing by Transporter to Shipper, against any and all amounts or monies due or owing by Shipper to Transporter for Transportation Services provided.
10.6	Any payments received under this Section 10 shall first be applied to accrued interest, then to additional charges due, then to the previously outstanding principle, and lastly, to the most current principle due.
Issued by: Issued on:	Effective on:

Original Sheet No. 78

GENERAL TERMS AND CONDITIONS (CONTINUED)

11. POSSESSION OF GAS

Unless otherwise provided in the Agreement or applicable Rate Schedule, as between Transporter and Shipper, Shipper shall be deemed to be in exclusive control and possession of the Gas (i) prior to receipt by Transporter at the Receipt Point(s) and (ii) after delivery by Transporter at the Delivery Point(s); otherwise, Transporter shall be in exclusive control and possession of the Gas. The party which shall be in exclusive control and possession of the Gas shall be responsible for all injury or damage caused thereby to any third party except any injury or damage caused by Gas provided by Shipper that fails to conform with the specifications set forth in Section 3. In the absence of bad faith or willful misconduct on the part of Transporter, Shipper waives any and all claims and demands against Transporter, its officers, employees or agents, arising out of or in any way connected with (i) the quality, use or condition of the Gas after delivery from Transporter for the account of such Shipper, (ii) any losses or shrinkage of Gas during or resulting from Transportation hereunder, and (iii) all other claims and demands arising out of Transporter's performance of its duties hereunder.

12. RECEIPT AND DELIVERY POINT PRESSURE

- 12.1 All gas tendered by or on behalf of Shipper to Transporter will be delivered at Receipt Points at a pressure sufficient to enter Transporter's system up to Transporter's Maximum Allowable Operating Pressure.
- 12.2 Unless otherwise agreed to, Transporter will redeliver Gas at the Delivery Points nominated by Shipper at Transporter's prevailing line pressure of no less than 250 pounds per square inch, gauge pressure ("Minimum Delivery Pressure"). If Transporter and Shipper otherwise agree on the Minimum Delivery Pressure at a Delivery Point(s), it will be set forth on Exhibit B of the Agreement.

Issued by: Issued on:

Original Sheet No. 79

	GENERAL TERMS AND CONDITIONS (CONTINUED)
13.	OPERATIONAL FLOW ORDERS (OFOs)
	13.1 Notification of Conditions that May Require the Issuance of an OFO or Action Alert: Transporter shall provide prior notice, via posting on LINK® and to affected Shippers and point operators through the affected party's choice of Electronic Delivery Mechanism(s), of upcoming events that may affect Transporter's pipeline system such as anticipat weather patterns or operational situations that may necessitate the issuance of an OFO pursuant to this Section 13.
	13.2 Circumstances Warranting Issuance of an Operational Flow Order: Transporter shall have the right to issue Operation Flow Orders as specified in this Section 13 that require actions by Shippers/point operators in order (1) to alleviate conditions that threaten to impair Transporter's ability to provide reliable service, (2) to maintain pipeline operations at the pressures required to provide efficient and reliable service, (3) to have adequate Gas supplies in Transporter's system to receive and deliver Gas consistent with its firm Transportation Service obligation (4) to maintain Transportation Services, and (5) to maintain Transporter's system in balance for the foregoing purposes. Transporter shall lift any effective Operations Flow Order, promptly upon the cessation of operating conditions that caused the relevant system problem(s). Routine repairs and maintenance will not be used as a bas for issuing OFOs. Transporter will plan routine repairs maintenance by scheduling such activities in advance.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)	
13.5 Notice: All Operational Flow Orders be issued via posting on LINK® to be or telephone notification to the aff operators and notification to the aff the affected party's choice of Elect: Mechanism(s). The Operational Flow G set forth (1) the time and date of is effectiveness, (2) the actions a Ship required to take, (3) the time by whi operator must be in compliance with Order/Action Alert, (4) the anticipa Operational Flow Order/Action Alert, that Transporter may reasonably requ effectiveness of the Operational Flo Each Shipper and point operator must persons, but not more than three per contact on operating matters at any Day, 365-Day a year basis. Such con adequate authority and expertise to matters. If Transporter cannot cont operator because that Shipper/point designate a contact person or Shippe contact person is unavailable, Trans responsible for any consequences tha subsequent actions taken to alleviat Transporter, however, will make reas efforts to notify the affected Shipp addition to the other information co section 13.5, such notice shall also about the status of operational vari when an Operational Flow Order or Ac and end, and Transporter shall post information, promptly upon occurrenc in the information. Transporter will informing the Shipper/point operator Flow Order or Action Alert in effect specifying the factors that caused t Order or Action Alert to be issued a extent such factors are known.	and Action Alerts will followed by facsimile acted Shippers and point fected parties through ronic Delivery Order/Action Alert will ssuance and pper/point operator is ich a Shipper/point the Operational Flow ted duration of the and (5) any other terms ire to ensure the w Order or Action Alert. designate one or more sons, for Transporter to time, on a 24-Hour a tact persons must have deal with such operating act any Shipper/point operator has failed to r's/point operator's porter shall not be t result from its e the system problem. onable continuing er/point operator. In ntemplated by this include information ables that determine tion Alert will begin periodic updates of such e of any material change l post a notice on LINK® when any Operational will be cancelled and he Operational Flow nd then lifted, to the

Issued by: Issued on:

Original Sheet No. 82

GET	NERAL TERMS AND CONDITIONS (CONTINUED)
13.6 Action Ale judgment, issue, Tra	erts: In the event that, in Transporter's action is required to avoid a system integrity insporter may issue Action Alerts.
(a) Issu noti Sect hour deem requ	Hance of Action Alerts: Action Alerts will be aced in accord with the procedures set forth in tion 13.5 and will be issued a minimum of four rs, or such shorter period of time as Transporter as reasonable under the circumstances, prior to the thired action by the Shipper/point operator.
(b) Requ effe	aired Actions: Action Alerts can be issued to ect any of the following:
(i)	curtailment of interruptible services;
(ii)	restrictions of receipts or deliveries at specific Receipt or Delivery Point(s) covered by an Operational Balancing Agreement to the aggregate MDQ under the firm Agreements whose Primary Receipt and/or Delivery Points are at the affected locations;
(iii	.) forced balancing such that point operators will be required to assure that nominations equal flows or that receipts and deliveries fall within the tolerance level designated in the Action Alert; and/or
(iv)	any action required to maintain the integrity of Transporter's System.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

- 13.7 Operational Flow Orders: In the event that (1) Shipper/point operator does not respond to an Action Alert, or (2) the actions taken thereunder are insufficient to correct the system problem for which the Action Alert was issued, or (3) there is insufficient time to carry out the procedures with respect to Action Alerts, Transporter may periodically take unilateral action, including the curtailment of firm Transportation Service, to maintain the operational integrity of Transporter's system (or any portion thereof). For purposes of this Section 13.7, the operational integrity of Transporter's system shall encompass the integrity of the physical system and the preservation of physical assets and their performance, the overall operating performance of the entire physical system (or any portion thereof), and the maintenance (on a reliable and operationally sound basis) of total system deliverability and the quality of Gas delivered. Notice of an Operational Flow Order will be provided pursuant to and in accordance with Section 13.5 above.
- 13.8 Penalties: If a Shipper/point operator fails to comply with an Action Alert or Operational Flow Order, the Shipper/point operator shall be subject to a penalty as follows:

Action Alert penalty for each Dekatherm of Gas by which Shipper/point operator deviated from the requirements of the Action Alert equal to the product of 200% times the average Cashout price as determined pursuant to Section 8.7(a) of these General Terms and Conditions, for each Day that said Action Alert is in effect.

Operational Flow Order penalty for each Dekatherm of Gas by which Shipper/point operator deviated from the requirements of the Operational Flow Order equal to the product of 500% times the average Cashout price as determined pursuant to Section 8.7(a) of these General Terms and Conditions, for each Day that said Operational Flow Order is in effect.

Any penalty revenues received by Transporter as a result of the operation of Section 13.8 above will be credited pursuant to Section 23.4 of the General Terms and Conditions.



Original Sheet No. 85

GENERAL TERMS AND CONDITIONS (CONTINUED)					
14. WARR	ANTY OF TITLE				
14.1	This Article shall apply to all service unless otherwise provided in the applicable Rate Schedule or Agreement.				
14.2	Shipper warrants for itself, its successors and assigns, that it will have, at the time of delivery of Gas hereunder, good title to the Gas it delivers, that the Gas it delivers hereunder shall be free and clear of all liens, encumbrances and claims whatsoever, that it will indemnify the Transporter and save it harmless from all suits, actions, debts, accounts, damages, costs, losses, and expenses arising from or out of any adverse claims of any and all persons to said Gas and/or to royalties, taxes, license fees, or charges thereon which are applicable for such delivery of Gas and that it will indemnify the Transporter and save it harmless from all taxes or assessments which may be levied and assessed upon such delivery and which are by law payable by and the obligation of the party making such delivery.				
14.3	If Shipper's title or right to deliver Gas to be transported is questioned or involved in any action, Shipper shall not qualify for or shall be ineligible to continue to receive service until such time as Shipper's title or right to deliver is free from question; provided, however, Transporter shall allow Shipper to qualify for or continue receiving service under this Tariff if Shipper furnishes a bond satisfactory to Transporter.				
14.4	Title to the Gas received by Transporter at the Receipt Point(s) shall not pass to Transporter, except that title to Gas delivered for Transporter's system fuel and uses and Gas lost and unaccounted for shall pass to Transporter upon delivery at the Receipt Point(s).				

Original Sheet No. 86

			GENERAL TERMS AND CONDITIONS (CONTINUED)		
15.	FORCE	MAJEUR	RE		
	15.1	If either Transporter or Shipper fails to perform any obligations under an Agreement due to an event of Force Majeure, such failure shall be deemed not to be a breach of such obligations and neither party shall be liable in damages or otherwise as a result of an event of Force Majeure. A party that fails to perform any obligations under an Agreement where such failure is caused by an event of Force Majeure shall promptly remedy the cause of the Force Majeure insofar as it is reasonably able to do so.			
	15.2	Notwit Majeu:	thstanding the above provisions, no event of Force re shall:		
		(a)	relieve any party from any obligation or obligations pursuant to an Agreement unless such party gives notice with reasonable promptness of such event to the other party;		
		(b)	relieve any party from any obligation or obligations pursuant to an Agreement after the expiration of a reasonable period of time within which, by the use of its due diligence, such party could have remedied or overcome the consequences of such event of Force Majeure; or		
		(c)	relieve either party from its obligations to make payments of amounts as provided in the applicable Rate Schedule, subject to any credit provided for in the applicable Rate Schedule.		

Issued by: Issued on:

Original Sheet No. 87

	GENERAL TERMS AND CONDITIONS (CONTINUED)
16.	NOTICES
	Except when the terms of this Tariff require or allow for communication via LINK® or EDM, any communication, notice, request, demand, statement, or bill provided for in the Tariff or in an Agreement or OBA, or any notice which either Transporter or Shipper may desire to give to the other, shall be in writing and shall be considered as duly presented, rendered, or delivered when mailed by either post-paid registered or ordinary mail or when sent by telecopy, telex, express mail service, or such other method mutually agreed upon between the parties. The material so sent shall be addressed to the pertinent party at its last known post office address, or at such other address as either party may designate.
17.	MODIFICATION
	No modification of the terms and provisions of an Agreement shall be made except by the execution of written contracts.
18.	NON-WAIVER AND FUTURE DEFAULT
	No waiver by either Transporter or Shipper of any one or more defaults by the other in the performance of any provisions of the Agreement shall operate or be construed as a waiver of any future default or defaults, whether of a like or of a different character.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

Issued on:

	GENERAL TERMS AND CONDITIONS (CONTINUED)
19.	SCHEDULES AND CONTRACTS SUBJECT TO REGULATION
	This Tariff, including these General Terms and Conditions and the respective obligations of the parties under an Agreement, are subject to valid laws, orders, rules, and regulations of duly constituted authorities having jurisdiction and are subject to change from time to time by addition, amendment, or substitution as provided by law.
20.	OPERATIONAL BALANCING AGREEMENTS ("OBAs")
	20.1 For the purposes of minimizing operational conflicts between various natural gas facilities with respect to the delivery of gas to and from Transporter's facilities, Transporter may negotiate and execute on a not-unduly discriminatory basis mutually agreeable OBAs with appropriate parties that operate natural gas facilities interconnecting with Transporter's system (any such party will be referred to herein as the "OBA Party"). Transporter must enter into OBAs at all points of interconnection between its system and the system of another interstate or intrastate pipeline. Such OBAs shall specify the Gas custody transfer procedures to be followed by Transporter and the OBA Party for the confirmation of scheduled quantities to be received by Transporter at Receipt Point(s) and delivered by Transporter at Delivery Point(s). Such OBA will provide that any variance between actual quantities and scheduled quantities at the point where the OBA is in place for any Day shall be resolved pursuant to the terms of the OBA.
	To facilitate such determination of variances on a timely basis, Transporter and the OBA Party will agree in the OBA on necessary measurement and accounting procedures. Transporter shall post on LINK® those Receipt Point(s) and Delivery Point(s) at which an OBA is in effect.
Issued by:	Effortive on .

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL	TERMS	AND	CONDITIONS			
(CONTINUED)						

20.2 Transporter shall have no obligation to negotiate and execute OBAs with any OBA Party that:

- (a) is not creditworthy as determined pursuant to Section 27 of the GT&C; for purposes of such provision, references to Shipper shall refer to the OBA Party;
- (b) does not maintain dispatching operations which are staffed on a continuous around-the-clock basis every day of the year;
- (c) would cause the level of regulation which Transporter is subject to prior to the execution of the applicable OBA to increase; or
- (d) does not commit to timely determination of variances based on reasonable available measurement technology; or
- (e) has not demonstrated operational consistency commensurate with the OBA relationship over a minimum period of three years.
- 20.3 If Receipt or Delivery Point Operators have not executed an OBA with Transporter as described in Section 20.1, then any variance between actual quantities and scheduled quantities for any Day for that Receipt or Delivery Point shall be cumulated for the month for the Shipper(s) responsible for the imbalance, and such monthly imbalances will be subject to the cashout of monthly imbalances as set forth in Section 8 herein.
- 20.4 Resolution of OBA Imbalance: Transporter and the OBA Party shall resolve any imbalances in accordance with the procedures set forth in the OBA. Unless otherwise agreed, OBA imbalances shall be resolved on a monthly basis by cashout mechanism.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

20.5 Nothing in this Section 20 nor any executed OBA shall limit Transporter's rights to take action as may be required to adjust receipts and deliveries under any Agreement to reflect actual experience or to alleviate conditions which threaten the integrity of Transporter's system, including maintenance of service to higher priority Shippers and/or services.

20.6 Form of OBA Agreement. A Form of OBA Agreement is displayed on LINK® for informational purposes.

21. NEW FACILITIES POLICY

21.1 Unless otherwise mutually agreed to by the parties, Transporter shall not be required to own, construct and install any facilities to perform any service requested by a Shipper under this Tariff. In the event Transporter agrees to own, construct and install facilities to perform services requested including, but not limited to, hot tap, side valve, measurement, Gas supply lateral lines, looping and/or compression facilities, Transporter shall do so on a not unduly discriminatory basis. Shipper shall reimburse Transporter (a) for the costs of such facilities installed by Transporter to receive, measure, transport or deliver natural gas for Shipper's account and (b) for any and all filings and approval fees required in connection with such construction that Transporter is obligated to pay to the Commission or any other governmental authority having jurisdiction. Nothing in this policy statement shall require Transporter to file an application for a certificate of public convenience and necessity under Section 7 (c) of the Natural Gas Act. Nothing in this policy statement, further, shall prevent Transporter from contesting an application for service filed pursuant to Section 7 (a) of the Natural Gas Act. Transporter reserves the right to seek a waiver of the policy set forth herein, for good cause shown.

Issued by: Issued on:


Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	21.4 In order to maintain and expand service and utilization of Transporter's system, Transporter may negotiate Agreements with Shippers in connection with which Transporter could make a contribution in aid of construction (CIAC) to the Shipper. The Shipper would use such funds to assist in the development of its natural gas related facilities. For any newly agreed to CIAC, Transporter will post on LINK® for a period of thirty (30) days (1) the amount of the CIAC, (2) the name of the Shipper receiving the CIAC, and (3) the economic feasibility of the CIAC. Such CIACs are includible in Transporter's jurisdictional rate base and amortizable. All CIACs entered into pursuant to this provision shall be subject to review and challenge by the Commission and all parties in a general rate case requesting inclusion of such costs.
22.	PERIODIC RATE ADJUSTMENTS
	Transporter and Shipper recognize that Transporter will from time to time experience changes in costs related to providing service under this Tariff, including, but not limited to, changes in the cost of labor, benefits, materials and supplies, taxes, required rate of return, costs associated with the resolution of past disputes or outstanding uncertainties concerning amounts owed by Transporter or Shipper or attributable to Transporter or Shipper, and costs generated by decisions of the Commission, the courts or by an arbitration panel or other body having jurisdiction over Transporter. Transporter and Shipper further recognize that it may be appropriate, equitable and consistent with cost responsibility to allocate such costs among Shippers based on or taking into account past period factors, such as contract demand levels, throughput or other factors related to a prior period of time. Shipper agrees that Transporter shall have the right from time to time to make rate change filings which may include such costs and utilize an allocation methodology based in whole or in part on factors related to past periods. Shipper shall have the right to intervene and protest any such filing.

Issued on:

Original Sheet No. 93

٦

	GENERAL TERMS AND CONDITIONS (CONTINUED)
22.1	Federal Energy Regulatory Commission Annual Charge Adjustment.
	 (a) The purpose of this Section 22.1 is to establish an Annual Charge Adjustment ("ACA") as permitted by Section 154.402 of the Commission's Regulations to permit Transporter to recover from its Shippers all annual charges assessed it by the Commission under Part 382 of the Commission's Regulations.
	(b) Applicable Rate Schedules: The ACA as set forth in the Statement of Rates for Transportation of Natural Gas of this Tariff, is applicable to Transporter's Rate Schedules FTS and ITS.
	(c) Filing Procedure. Proposed changes in the ACA shall be filed by Transporter at least thirty (30) Days prior to the proposed effective date unless, for good cause shown, lesser periods are allowed by valid Commission Order. The proposed effective date of the filings shall be October 1 of each calendar year. Any such filing shall not become effective until it becomes effective without suspension or refund obligation.
	(d) Remittance to the Commission. Transporter shall remit to the Commission, not later than forty-five (45) Days after receipt of the Annual Charges Billing, the Total Annual Charge stated on such billing.
	(e) Basics of the Annual Charge Adjustment. The Rate Schedules specified in Section 22.1(b) hereof shall include an increment for an Annual Charge Adjustment for costs specified in Section 22.1(a), above. Such adjustment shall be the billable charge factor from the Commission, adjusted to the Company's pressure base and heating value, if required, which is stated in the Commission's Annual Charges Billing. The Annual Charge Adjustment shall be reflected in the Statement of Rates for Transportation of Natural Gas of this Tariff.

Issued by: Issued on:

Original Sheet No. 94

GENERAL TERMS AND CONDITIONS (CONTINUED)
22.2 Transporter's Use.
(a) The initial Transporter's Use (%) will be calculated based upon appropriate engineering principles. After one year of operation and each June 1 thereafter commencing in 2009, Transporter's Use (%) will be redetermined by dividing Transporter's projection for the next 12 Months beginning June 1 of fuel usage and any lost and unaccounted-for gas by Transporter's projection of applicable deliveries for the account of Shippers for the next 12 Months beginning June 1. This percentage will go into effect on June 1. Transporter may file interim proposals between annual filings subject to approval by the Commission.
(b) Pursuant to Section 22.3, Transporter shall maintain a separate System Balancing Adjustment account. This account shall be credited for all sales of excess fuel collected under Transporter's Use, debited for all purchases for Transporter's Use and further adjusted for the operational activities enumerated in Section 22.3(a).
22.3. System Balancing Adjustment. In order to maintain an operational system balance on its system, Transporter will calculate a system balancing adjustment ("SBA") charge.
(a) Transporter's SBA balance shall be the sum of:
(1) The net annual system cashout balance determined in accordance with Section 8 of the General Terms and Conditions and OBA cashouts;
(2) The net Transporter's Use Adjustment balance, determined in accordance with Section 22.2 of the General Terms and Conditions;
(3) Penalty revenues credited pursuant to Sections 23.1(a),23.1(b), 23.2, and 23.3 of the General Terms andConditions; and
(4) Any other account balance as may be approved by the FERC.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

- (b) The net SBA balance determined in Section 22.3(a), through January 31 of the year in which the filing pursuant to Section 22.3(c) is made will be refunded or recovered from Shipper pursuant to the procedures in this Section 22.3. Upon determining the net SBA balance at the end of the accumulation period, Transporter shall calculate surcharges or refunds designed to allocate such balance to Shippers based upon each Shipper's actual throughput during the twelve-month accumulation period. A Shipper's net debit or credit for the accumulation period shall be due and payable sixty (60) Days after the Commission's acceptance of the filing pursuant to Section 22.3(c). Notwithstanding the immediately preceding sentence, if the net SBA balance results in a surcharge/debit, each Shipper who is allocated a surcharge/debit shall have the right by providing notice to Transporter within the sixty (60)-Day period to elect to pay the surcharge/debit ratably over the twelve (12)-Month period, commencing with the first Day of the first calendar month following the last Day of the sixty (60)-Day period, with interest calculated for each payment from the end of the sixty (60)-Day period until the payment is made (at the rate set forth in Section 154.501(d) of the Commission's regulations).
- (c) Transporter shall file on May 1 of each year and each year thereafter, to establish the SBA refund or surcharge determined pursuant to the procedures in this Section 22.3.

23. PENALTIES AND PENALTY CREDITING MECHANISM

- 23.1 Rate Schedule PALS penalties.
 - (a) Penalty for PALS Non-compliance In the event that a Shipper incurs a penalty pursuant to Section 4.1(b) of Rate Schedule PALS, which section is applicable if a Shipper does not comply with Transporter's notice given pursuant to Section 4.1(a) of Rate Schedule PALS to either remove Park service quantities or to return Loan service quantities, Transporter shall credit the penalty revenue, net of costs, to the System Balancing Adjustment, Section 22.3 of the General Terms and Conditions. Any penalty revenue credited to the System Balancing Adjustment

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	pursuant to this section shall include interest calculated in accordance with Section 154.501 of the Commission's regulations.
(b) Balances Remaining Upon PALS Contract Termination In the event that Transporter receives penalty revenue from a PALS Shipper as the result of the application of Section 4.2 of Rate Schedule PALS to such PALS Shipper's unresolved balance, Transporter shall credit the penalty revenue received, net of costs, to the System Balancing Adjustment, Section 22.3 of the General Terms and Conditions. Any penalty revenue credited to the System Balancing Adjustment pursuant to this section shall include interest calculated in accordance with Section 154.501 of the Commission's regulations.
Issued by:	

Original Sheet No. 97

	
	GENERAL TERMS AND CONDITIONS (CONTINUED)
23.2	Trespass Gas
	In the event that Transporter receives penalty revenue from a Shipper as the result of the application of Section 7.3 (Trespass Gas) of the General Terms and Conditions, Transporter shall credit the penalty revenue received, net of costs, to the System Balancing Adjustment, Section 22.3 of the General Terms and Conditions. Any penalty revenue credited to the System Balancing Adjustment pursuant to this section shall include interest calculated in accordance with Section 154.501 of the Commission's regulations.
23.3	Conversion of Gas
	In the event that Transporter receives penalty revenue from a Shipper as the result of the application of Section 7.4 (Conversion of Gas) of the General Terms and Conditions, Transporter shall credit the penalty revenue received, net of costs, to the System Balancing Adjustment, Section 22.3 of the General Terms and Conditions. Any penalty revenue credited to the System Balancing Adjustment pursuant to this section shall include interest calculated in accordance with Section 154.501 of the Commission's regulations.
23.4	Action Alert/Operational Flow Order Penalties
	Any penalty revenue collected by Transporter pursuant to Section 13.8 of the General Terms and Conditions will be credited, net of costs, to any firm or interruptible Shipper that did not incur penalties pursuant to Section 13.8 of the General Terms and Conditions in the Month for which penalty revenues were received ("Non-Offending Shipper"), based on the ratio of the actual quantities taken by the Non- Offending Shipper to the actual quantities taken by all Non- Offending Shippers in such Month. Such credits shall be made within 90 days following each anniversary of the initial in-service date of Transporter's system and shall include interest at the rate determined in accordance with Section 154.501 of FERC's regulations.
1	

Original Sheet No. 98

GENERAL TERMS AND CONDITIONS (CONTINUED)

24. ELECTRONIC COMMUNICATION

24.1. System Description

(a) Transporter provides for interactive electronic communications with its Shippers and other parties through the LINK® Customer Interface System (hereinafter called the "LINK $\ensuremath{\mathbb{S}}$ System"). The LINK® System shall be available on a nondiscriminatory basis to any party (such party is referred to herein as the "LINK® System Subscriber"), provided that such party (i) has a Valid Service Agreement or has executed a LINK® System Agreement, and (ii) has designated a Local Security Administrator, who shall have the responsibilities set forth in Section 24.3 below, by completing the Local Security Administrator Form, as such form is amended from time to time and posted on the $\texttt{LINK} \ensuremath{\mathbb{R}}$ System. For purposes of this Section 24 and the form of LINK® System Agreement only, a "Valid Service Agreement" includes any service agreement pursuant to any of Transporter's Rate Schedules and/or a capacity release umbrella agreement. Parties with a Valid Service Agreement are not required to execute a LINK® System Agreement in conjunction therewith. Transporter will allow a Shipper that does not have a Valid Service Agreement or an executed LINK® System Agreement to submit a request for service pursuant to one of Transporter's Rate Schedules online via the LINK $\ensuremath{\mathbb{R}}$ System, and, if applicable, to execute the resulting Service Agreement online via the LINK® System. There is no requirement that a LINK® System Subscriber must be a party that has executed a Valid Service Agreement. Any party desiring access to the LINK® System for purposes other than submitting a request for service or executing a resulting Service Agreement that has not executed a Valid Service Agreement, however, must execute a LINK® System Agreement or have on file a valid LINK® System Agreement in accordance with the terms of this tariff and the form of LINK® System Agreement contained herein. By accessing the LINK® System, all LINK® System Subscribers agree to comply

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

with the procedures for access to and use of the LINK $\ensuremath{\mathbb{S}}$ System as set forth in this Section 24. All LINK® System Subscribers must complete and submit to Transporter the Local Security Administrator Designation Form, the Contact Information Form, and any other applicable LINK® forms, as updated from time to time and posted on the LINK $\ensuremath{\mathbb{S}}$ System. In addition, LINK $\ensuremath{\mathbb{S}}$ System Subscriber must complete and submit to Transporter the Designation of Agency Form and/or the Designation of Affiliated Companies Form, if applicable, as updated from time to time and posted on the LINK® System. In the event Shipper desires to change any of the information provided by Shipper on such form(s), Shipper shall be required to promptly provide Transporter with the updated information by submitting revised form(s) to Transporter. Such revised form(s) shall supersede in its entirety any form(s) previously submitted to Transporter. Transporter reserves the right to implement, to contract for or obtain a license for enhancements to the $\ensuremath{\texttt{LINK}}\xspace$ System at its sole discretion; provided however, all such enhancements when fully operational shall be available to all LINK® System Subscribers. Transporter will exercise due diligence to ensure the LINK® System operates correctly and will provide timely and non-discriminatory access to on-line LINK® System help features and to any information available on the LINK® System that LINK® System Subscriber is entitled to access.

(b) The LINK® System provides on-line help, a search function that permits a LINK® System Subscriber to locate information concerning a specific transaction, and menus that permit LINK® System Subscribers to separately access notices of available capacity, records in the transportation request log, and standards of conduct information. The LINK® System will permit a LINK® System Subscriber to electronically download information on transactions from the LINK® System and to separate extremely large documents into smaller files prior to such download. Transporter shall maintain and retain daily back-up records of the information displayed on the LINK® System and the web site and through electronic data interchange for three years and shall permit LINK® System Subscriber to review those records upon request. Completed

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	transactions will remain on the LINK® System for at least ninety days after completion and will then be archived. Archived information will be made available by Transporter if possible within two weeks after receipt of a Shipper's request for such information. Information on the most recent entries will appear ahead of older information.
(c)	Shippers' Notices pursuant to Section 25 of the General Terms and Conditions shall be submitted electronically and, in addition, posted electronically by the Shipper via the LINK® System. Electronic communications may also be transmitted, where applicable, via electronic data interchange, which will be available on a nondiscriminatory basis to any LINK® System Subscriber, provided such LINK® System Subscriber has entered into a trading partner agreement with Transporter, in addition to the agreements specified in Section 24.1(a) above. Specifically, a LINK® System for purposes of: (a) requesting service under Transporter's Rate Schedules set forth in Transporter's FERC Gas Tariff; (b) executing, tracking and amending certain Service Agreements under Transporter's rate schedules set forth in Transporter's FERC Gas Tariff; (c) providing nominations and viewing allocations and operational imbalances under all rate schedules as a Shipper of Transporter pursuant to the applicable rate schedule and the General Terms and Conditions; (d) exercising its rights as a Shipper of Transporter pursuant to Section 25 of the General Terms and Conditions or submitting a bid as a Replacement Shipper of Transporter under such section; (e) exercising its rights as a Shipper of Transporter pursuant to Section 25 of the General Terms and Conditions (which, if submitted utilizing the LINK® System, will be posted at that time) or submitting a bid as a Replacement or Prearranged Shipper of Transporter pursuant to such section; (f) viewing and downloading operational data for any gas flow day on the second subsequent gas flow day; (g) viewing Transporter's notice of an OFO as contemplated by Section 13 of the General Terms and Conditions; (h) effectuating Imbalance Netting and Trading

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	pursuant to Sections 8.3 and 8.4 of the General Terms and Conditions; (i) requesting a discount of the maximum recourse rate(s) for service under Transporter's Open-access Rate Schedules or viewing such discounts previously granted; and (j) such other functions as may be available on the LINK® System from time to time.
24.2	Information. Transporter shall post at least four times a day on the LINK® System and the web site information relevant to the availability of firm and interruptible capacity at Receipt Points, on the mainline, and at Delivery Points. The LINK® System and the web site will indicate whether the capacity is available from Transporter directly or through Transporter's capacity release mechanism as set forth in Section 25 of the General Terms and Conditions. The LINK® System and the web site shall provide the best available information about imbalances on an hourly and a dai basis. The LINK® System and the web site also include information allowed or required to be posted thereon by other provisions of th tariff including Section 25, information that Transporter is required to post pursuant to the Commission's regulations, or othe information Transporter chooses to post in furtherance of the operation of its system. Transporter shall maintain both in writt form and on the LINK® System a Master Receipt/Delivery Point List containing the following information for each point. Such information shall be updated promptly whenever Receipt Point(s) or Delivery Point(s) are added to Transporter's system.
	 Name of the Point; Meter Number of the Point; Location (legal description) of the Point; Operator name and phone number to the extent available; and Whether there is an operational balancing agreement in effect at the Point.
24.3.	Local Security Administrators

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

		GENERAL TERMS AND CONDITIONS (CONTINUED)
		Security Administrator Designation Form, as such form is amended from time to time and posted on the LINK® System. The Local Security Administrator shall, via the LINK® System, be responsible for (1) identifying those persons who are duly authorized by LINK® System Subscriber to use the LINK® System to perform one or more of the functions available on the LINK® System ("LINK® System User"); (2) providing LINK® System Users with individualized USERIDS and passwords; (3) maintaining LINK® System Users' account information; (4) adding and terminating LINK® System Users immediately upon a change in status requiring such addition or termination; (5) creating and modifying security rights for LINK® System Users; and (6) ensuring that USERIDS are used only as appropriate and as contemplated by these General Terms and Conditions and the LINK® System Agreement.
	(b)	Transporter shall be entitled to rely upon the representation of LINK® System Subscriber's Local Security Administrator that the LINK® System User(s) identified by the Local Security Administrator may (i) transmit information to Transporter; (ii) view information posted on the LINK® System; and/or (iii) perform the LINK® System contracting function in accordance with the rights granted by Local Security Administrator.
24.4	Auth	norized Use of LINK® System; Confidentiality
	(a)	LINK® System Subscriber shall complete and submit a Contact Information Form as such form is amended from time to time and posted on the LINK® System. LINK® System Subscriber shall be required to keep such form current by submitting a new form as such information changes. Such revised form shall supersede in its entirety any form previously submitted to Transporter.
	(b)	LINK® System Subscriber shall not disclose to persons other than Local Security Administrator and LINK® System Users that are employed by LINK® System Subscriber, or properly designated affiliates or agents of LINK® System Subscriber, and shall otherwise keep confidential, all USERIDs and passwords issued by Local Security Administrator. In addition, LINK® System Subscriber shall cause Local Security Administrator and LINK®

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	System User(s) to refrain from disclosing to any other person, whether or not employed by LINK® System Subscriber, and shall otherwise keep confidential, the individualized USERID and password issued to each such LINK® System User.
(c)	LINK® System Subscriber shall be solely responsible for any unauthorized or otherwise improper use of USERIDs and passwords issued by or for its Local Security Administrator, including, but not limited to, the use of such USERIDs and passwords by LINK® System Users who are not within LINK® System Subscriber's employment or control.
(d)	Transporter reserves the right to disable for due cause any USERID issued to any LINK® System User. Transporter shall give prior notice to LINK® System User and Local Security Administrator and allow such Local Security Administrator a reasonable amount of time to respond before disabling any such USERID. In the event Local Security Administrator fails to adequately respond within the time period established in Transporter's notice, Transporter shall have the right to disable such USERID without further notice. In addition, upon thirty (30) days prior notice to the LINK® System User and the Local Security Administrator, Transporter will disable any USERID that has not been used to access the LINK® System for six (6) consecutive months.
(e)	LINK® System Subscriber shall immediately notify Transporter, by e-mail to link-help@duke-energy.com, of the need or desire to change or delete a Local Security Administrator of LINK® System Subscriber. In addition, LINK® System Subscriber must complete a Local Security Administrator Designation Form as required in Section 24.3(a) above in order to document and allow such change or deletion to take place. LINK® System Subscriber shall be solely responsible for any unauthorized actions of Local Security Administrator due to LINK® System Subscriber's failure to so notify Transporter of the need to change or delete such Local Security Administrator.
(f)	Transporter warrants that, without the express consent of LINK® System Subscriber or as otherwise provided in this FERC Gas Tariff, no Transporter employee or agent will disclose to any

Issued by: Issued on:

Original Sheet No. 104

GENERAL TERMS AND CONDITIONS (CONTINUED)

third party any non-public information regarding research performed through the use of the LINK® System by LINK® System Subscriber.

- 24.5 LINK® System Subscriber; Affiliated Companies
 - (a) If LINK® System Subscriber belongs to a group of affiliated companies and requires LINK® System access on behalf of one or more of said affiliates, LINK® System Subscriber shall, or shall cause one of the affiliates of LINK® System Subscriber to, complete and submit to Transporter a Designation of Affiliated Companies Form, as such form is amended from time to time and posted on the LINK® System. LINK® System Subscriber shall, or shall cause one of its properly designated affiliates to keep such form current by submitting to Transporter an updated form when such information changes. Such revised form shall supersede in its entirety any form(s) previously submitted to Transporter. LINK® System Subscriber warrants that access consistent with any Designation of Affiliated Companies Form submitted by LINK® System Subscriber or one of its affiliates is appropriate and authorized. Determining the propriety of such access is the responsibility of LINK® System Subscriber and/or its affiliates, but Transporter reserves the right to reject such Designation of Affiliated Companies Form if it determines that granting such designation would violate any contractual, legal, or regulatory responsibility of Transporter.
 - (b) In order for LINK® System Users of LINK® System Subscriber to access the LINK® System on behalf of LINK® System Subscriber's affiliates identified on the Designation of Affiliated Companies Form, LINK® System Subscriber and each affiliate of LINK® System Subscriber identified on the Designation of Affiliated Companies Form must either be: (1) a Shipper with a Valid Service Agreement as defined in Section 24.1(a) of these GT&C, or (2) a LINK® System Subscriber under a LINK® System Agreement.
- 24.6. LINK® System Subscriber; Agency
 - (a) If LINK® System Subscriber desires to designate one or more persons or entities to act as an agent on behalf of LINK®

r

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	System Subscriber ("Agent"), LINK® System Subscriber must complete and submit to Transporter the Designation of Agency Form, as such form is amended from time to time and posted on the LINK® System, for each such Agent, specifying which rights are granted to each Agent. Transporter may require that LINK® System Subscriber provide additional documentation to confirm that LINK® System Subscriber desires Agent to act on its behalf.
(b)	In order for LINK® System Users of Agent to access the LINK® System on behalf of LINK® System Subscriber, such Agent must either be: (1) a Shipper with a Valid Service Agreement as defined in Section 24.1(a) of these GT&C, or (2) a LINK® System Subscriber under a LINK® System Agreement.
(c)	Transporter may fully rely upon all communications received from and direction given by Agent with respect to all actions indicated in the Designation of Agency Form for which Agent is authorized to act on behalf of LINK® System Subscriber. Transporter may grant Agent access to LINK® System Subscriber's data contained in the LINK® System as necessary to perform the functions identified in the Designation of Agency Form. Transporter may accept and fully rely upon a Designation of Agency Form that has been properly executed by LINK® System Subscriber and Agent. LINK® System Subscriber will defend, indemnify and hold harmless Transporter from and against any and all claims, demands, liabilities and/or actions, and/or any and all resulting loss, costs, damages, and/or expenses (including court costs and reasonable attorney's fees) of any nature whatsoever, that may be asserted against or imposed upon Transporter by any party associated with Transporter's reliance on a Designation of Agency Form provided pursuant to this Section 24.6.
(d)	The rights specified on the Designation of Agency Form having the latest commencement date shall supersede all prior rights granted by LINK® System Subscriber to Agent identified on such Designation of Agency Form. In no event can an agency right granted to one Agent be simultaneously granted to another Agent. It is the obligation of the LINK® System Subscriber to notify Transporter when an Agency relationship changes or

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

terminates, either by specifying a termination date on the Designation of Agency Form or by providing Transporter with a properly executed superseding Designation of Agency Form. LINK® System Subscriber must update and re-submit to Transporter the Designation of Agency Form on an annual basis. Transporter shall disable the USERIDS of all LINK® System Users of Agent pertaining to access granted by LINK® System Subscriber pursuant to the Designation of Agency Form if LINK® System Subscriber does not update and re-submit the Designation of Agency Form on an annual basis.

(e) Agent is authorized to act on behalf of LINK® System Subscriber under any or all of LINK® System Subscriber's contracts with Transporter as effective from time to time, or with respect to any or all meter locations as available from time to time, respectively, as specified in the Designation of Agency Form, until LINK® System Subscriber properly notifies Transporter that the agency relationship is terminated or superseded in accordance with Section 24.6(d). The Designation of Agency Form does not provide for an assignment of the rights and obligations of any contract between Transporter and LINK® System Subscriber.

24.7. Liability

(a) Transporter shall not be liable to LINK® System Subscriber nor any other party in damages for any act, omission or circumstance related to the LINK® System occasioned by or in consequence of an event of Force Majeure as defined in Section 15 of these General Terms and Conditions, that is not within the control of Transporter and which by the exercise of due diligence Transporter is unable to prevent or overcome. To the extent the information displayed on the LINK® System is originated solely by Transporter and such information is subsequently determined to be inaccurate, LINK® System Subscriber shall not be subject to any penalties otherwise collectable by Transporter based on Shipper conduct attributable to such inaccuracy during the period the inaccurate information was displayed on the LINK® System.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

24.9. Rights to LINK® System. Transporter or an affiliate of Transporter is the exclusive proprietor of the programming that generates the LINK® System and of all the copyrights and proprietary interests therein, except insofar as any third party (whose materials are made available in the files of the LINK® System under license to Transporter or an affiliate of Transporter) possesses a copyright or proprietary interest in such materials, but not of the files of and the information displayed on the LINK® System. A LINK® System Subscriber will not by virtue of this Section 24 or the executed LINK® System Agreement acquire any proprietary interests in the programming that generates the LINK® System.

Issued by: Issued on:

Original Sheet No. 109

i	GENERAL TERMS AND CONDITIONS (CONTINUED)
25.	CAPACITY RELEASE PROVISIONS
	This section sets forth the terms and conditions that are applicable to the release of firm entitlements under various services that are provided pursuant to this Tariff.
	25.1 Procedure. Capacity released shall be subject to the terms and conditions of this Section 25.1.
	 (a) Eligibility. Any Shipper ("Releasing Shipper") under Rate Schedule FTS of this Tariff, shall be entitled, subject to the terms and conditions of this Section 25.1, to release any or all of its firm Transportation entitlements held under an Agreement but only to the extent that the capacity so released is acquired by another Shipper ("Replacement Shipper") pursuant to the provisions of this Section 25.1. Any such release shall result in a temporary suspension of the Releasing Shipper's right to use the released firm entitlements.
Taguad but	

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

·····		
		GENERAL TERMS AND CONDITIONS (CONTINUED)
	(b)	Released Capacity shall be made available on a basis that is not unduly discriminatory, and any Replacement Shipper shall be entitled to acquire Releasing Shipper's capacity subject to the terms and conditions under this Section 25.1, provided the Replacement Shipper meets all provisions governing eligibility under this Tariff in a timely manner. A Replacement Shipper shall be entitled to release acquired capacity to another Replacement Shipper, subject to the requirement that the original Replacement Shipper satisfies all of the provisions of this Section 25.1 as if such Replacement Shipper were a Releasing Shipper, and the new Replacement Shipper meets all provisions governing eligibility under this Tariff in a timely manner, provided, however, that a Replacement Shipper that acquired released capacity through a volumetric bid shall not be entitled to re-release that capacity.
	(c)	Term. Any release under this Section 25 shall not extend beyond the expiration of the initial primary term of the Agreement that is released.
Issued by: Issued on:		Effective on:

۰,

Original Sheet No. 111

·····	
GENERAL TERMS AND CONDITIONS (CONTINUED)	
(d) Recall	/ Reput Rights.
(1) F	Recall Provisions.
F a c a a E E c	Releasing Shipper's rights to recall capacity on a full day or partial day basis shall be stated clearly in Shipper's Notice. Purchase of gas by a Releasing Shipper from a Replacement Shipper at the Releasing Shipper's Primary Delivery Point(s) shall not be deemed to be the exercise of a recall by the Releasing Shipper.
] r s t t i c	The Releasing Shipper shall provide capacity secall notification to Transporter via the LINK® System. The recall notification shall specify the recall notification period for the specified effective Gas Day, as well as any other information needed to uniquely identify the capacity being recalled.
T r s	Transporter shall support the following recall notification periods for all released capacity subject to recall rights:
1	Timely Recall Notification:
-	 A Releasing Shipper recalling capacity should provide notice of such recall to Transporter and the first Replacement Shipper no later than 8:00 A.M. CCT on the day that Timely Nominations are due; Transporter shall provide notification of such recall to all affected Replacement Shippers no later than 9:00 A.M. CCT on the day that Timely Nominations are due;
I	Early Evening Recall Notification:
-	- A Releasing Shipper recalling capacity should provide notice of such recall to Transporter and the first Replacement Shipper no later than 3:00 P.M. CCT on the day that Evening Nominations are due;

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)	
- Transporter shall provide notification such recall to all affected Replacemen Shippers no later than 4:00 P.M. CCT o day that Evening Nominations are due;	of t n the
Evening Recall Notification:	
 A Releasing Shipper recalling capacity provide notice of such recall to Trans and the first Replacement Shipper no 1 than 5:00 P.M. CCT on the day that Even Nominations are due; Transporter shall provide notification such recall to all affected Replacement Shippers no later than 6:00 P.M. CCT or day that Evening Nominations are due; 	should porter ater ning of t n the
Intraday 1 Recall Notification:	
 A Releasing Shipper recalling capacity provide notice of such recall to Trans and the first Replacement Shipper no 1 than 7:00 A.M. CCT on the day that Int Nominations are due; Transporter shall provide notification such recall to all affected Replacement Shippers no later than 8:00 A.M. CCT of day that Intraday 1 Nominations are due 	r should porter ater raday 1 . of .t .t .t .t .e; and
Intraday 2 Recall Notification:	
 A Releasing Shipper recalling capacity provide notice of such recall to Trans and the first Replacement Shipper no 1 than 2:30 P.M. CCT on the day that Int Nominations are due; Transporter should provide notification such recall to all affected Replacement Shippers no later than 3:30 P.M. CCT on day that Intraday 2 Nominations are due. 	y should sporter ater graday 2 on of at on the se.
prior to the recall notification deadline	2

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)		
specified above and received between 7:00 A.M. CCT and 5:00 P.M. CCT, Transporter shall provide notification to all affected Replacement Shippers no later than one hour after receipt of such recall notification. For recall notification provided to Transporter after 5:00 P.M. CCT and prior to 7:00 A.M. CCT, Transporter shall provide notification to all affected Replacement Shippers no later than 8:00 A.M. CCT after receipt of such recall notification.		
Transporter's notices of recalled capacity to all affected Replacement Shippers shall be provided via the LINK® System, along with written notice via e-mail communication to the individual the Replacement Shipper identified in the Replacement Shipper's bid submitted pursuant to Section 25.1(h) of these General Terms and Conditions. Such notices shall contain the information required to uniquely identify the capacity being recalled, and shall indicate whether penalties will apply for the Gas Day for which quantities are reduced due to a capacity recall. Upon receipt of notification of the recall from Transporter, each affected Replacement Shipper shall revise its nominations within the applicable nomination cycle in order to implement the recall. Each affected Replacement Shipper will be solely responsible for adjusting its supply and transportation arrangements, which may be necessary as a result of such recall. Replacement Shippers involved in re-release transactions may receive notice slightly after the first Replacement Shipper receives notice. The recalling Releasing Shipper may nominate the recalled capacity consistent with the applicable nomination cycle, pursuant to Section 4 of these General Terms and Conditions.		

Issued by: Issued on:

Original Sheet No. 114

-

GENE	RAL TERMS AND CONDITIONS (CONTINUED)
	If, on the day of a partial day recall, the quantity of gas delivered to the Replacement Shipper is in excess of the MDQ remaining on the replacement contract after the partial day recall and/or the quantity of gas delivered to the Releasing Shipper that recalled the capacity is in excess of the MDQ recalled by the Releasing Shipper, then the Shipper(s) to whom such excess gas is delivered will be charged the applicable Usage-2 Rate pursuant to Section 3.2(b) of Rate Schedule FTS on such excess quantites of gas in addition to all other applicable charges.
(2)	Partial Day Recall Quantity.
	The daily contractual entitlement that can be recalled by a Releasing Shipper for a partial day recall is a quantity equal to the lesser of: The quantity specified in the Releasing Shipper's notice to recall capacity; or The difference between the quantity released by the Releasing Shipper and the Elapsed Prorata Capacity; or The difference between the quantity released by the Releasing Shipper and the quantity actually delivered to the Replacement Shipper within the limitations of the MDQ.
	In the recall notification provided to Transporter by the Releasing Shipper, the quantity to be recalled shall be expressed in terms of the adjusted total released capacity entitlements based upon the Elapsed Prorata Capacity. In the event of an intra-day capacity recall, Transporter shall determine the allocation of capacity between the Releasing Shipper and the Replacement Shipper(s) based upon the Elapsed Prorata Capacity only in the case of (ii) above.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	The amount of capacity allocated to the Replacement Shipper(s) shall equal the original released quantity less the recalled capacity. This allocated daily contractual quantity shall be used for purposes of nominations, billing, and if applicable, for overrun calculations. As a result of the allocation of capacity described in this section, Transporter shall not be obligated to deliver a combined quantity to the Releasing Shipper and the Replacement Shipper(s) that is in excess of the total daily contract quantity of the release.
	(3) Reput Provisions.
	Transporter shall support the function of reputting by the Releasing Shipper. The Releasing Shipper may reput previously recalled capacity to the Replacement Shipper pursuant to the reput rights and methods identified in the Releasing Shipper's notice to release capacity, as required by Section 25.1(g)(10) below. When capacity is recalled, such capacity may not be reput for the same Gas Day. The deadline for the Releasing Shipper to notify Transporter of a reput of capacity is 8:00 A.M. CCT to allow the Replacement Shipper to submit timely nominations for gas to flow on the next Gas Day.
(e)	Bidding Period. Releasing Shipper may specify the date and time that the Bidding Period starts and the date that the Bidding Period ends, provided, however, that the Bidding Period shall not commence or end any later than the times set forth in Section 25.1(f) below. Releasing Shipper's offer shall be posted for the Bidding Period; provided, however, that the Releasing Shipper will have the right to withdraw its Releasing Shipper's Notice any time prior to the close of the Bid Period associated with such Releasing Shipper's Notice where unanticipated circumstances justify the withdrawal and no bids meeting the minimum conditions of Releasing Shipper's Notice have been made.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

Γ

GENERAL TERMS AND CONDITIONS (CONTINUED) Offers should be legally binding until written or electronic notice of withdrawal is received by Transporter. Transporter should post offers and bids, including prearranged deals, upon receipt. A releasing Shipper may request a later posting time for posting of such offer, and Transporter should support such request insofar as it comports with the standard Capacity Release timeline specified in Section 25.1(f) below.

Issued by: Issued on:

-

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(f)	The following capacity release timeline is applicable to all parties involved in the capacity release process; however, it is only applicable if (1) all information provided by the parties to the transaction is valid and (2) the Replacement Shipper has been determined to be creditworthy before the capacity release bid is tendered, and (3) there are no special terms or conditions of the release:
	 (1) For biddable releases (less than one (1) year): Offers should be tendered by 12:00 P.M. on a Business Day; Open season ends no later than 1:00 P.M. on a Business Day (evaluation period begins at 1:00 P.M. during which contingency is eliminated, determination of best bid is made, and ties are broken); Evaluation period ends and award posting if no match required at 2:00 P.M.; Match or award is communicated by 2:00 P.M.; Match required, award posting by 3:00 P.M.; and Contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract. (Central Clock time)
Issued by: Issued on:	Effective on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

ENERAL TERMS AND CONDITIONS (CONTINUED)
For biddable releases (one (1) year or more):
 Offers should be tendered by 12:00 P.M. four Business Days before award; Open season ends no later than 1:00 P.M. on the Business Day before timely nominations are due (open season is three Business Days); Evaluation period begins at 1:00 P.M. during which contingency is eliminated, determination of Best Bid is made, and ties are broken; Evaluation period ends and award posting if no match required at 2:00 P.M.; Match or award is communicated by 2:00 P.M.; Match response by 2:30 P.M.; Where match required, award posting by 3:00 P.M.; and Contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract. (Central Clock Time)
For non-biddable releases:
Timely Cycle:
 Posting of prearranged deals not subject to bid are due by 10:30 A.M.; Contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract. (Central Clock Time)

Original Sheet No. 119

GENERAL TERMS AND CONDITIONS (CONTINUED)
Evening Cycle:
 Posting of prearranged deals not subject to bid are due by 5:00 P.M.; Contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract. (Central Clock Time)
Intraday 1 Cycle:
 Posting of prearranged deals not subject to bid are due by 9:00 A.M.; Contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract. (Central Clock Time)
Intraday 2 Cycle:
 Posting of prearranged deals not subject to bid are due by 4:00 P.M.; Contract issued within one hour of award posting (with a new contract number, when applicable); nomination possible beginning at the next available nomination cycle for the effective date of the contract. (Central Clock Time)
(g) Required Information for the Release of Capacity. The Releasing Shipper shall submit the following information, objectively stated and applicable to all potential Shippers on a non-discriminatory basis, to Transporter via the LINK® System:
(1) The Releasing Shipper's legal name, contract number, and the name, title, address, e-mail address and phone and fax number of the individual who will authorize the release of capacity for the Releasing Shipper.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GEI	NERAL TERMS AND CONDITIONS (CONTINUED)
(2)	Whether the capacity is biddable.
(3)	The level of daily firm entitlements that the Releasing Shipper elects to release, expressed as a numeric quantity per Day for transportation, which will be displayed in the LINK® System posting for prospective Replacement Shippers as the available MDQ.
(4)	The Transportation Path(s) or segment within such Transportation Path(s), and quantity to be released.
(5)	The requested effective date and the term of the release.
(6)	The minimum acceptable period of release and minimum acceptable quantities (if any).
(7)	The Releasing Shipper's maximum reservation rates (including any demand type surcharges, direct bills, or similar mechanisms), any minimum rate requirement, whether bids are to be submitted on a reservation or volumetric basis, and whether the bids should be stated in
	dollars and cents or percent of the maximum tariff rate. The maximum and minimum rates may separately identify surcharges and direct bills, or such amounts can be included in the total rate. For purposes of this Section 25, the maximum reservation rate(s) for Shipper paying a negotiated rate will be deemed to be the maximum rate(s) as set forth on the Statement of Rates for Transportation of Natural Gas.
(8)	Whether the Releasing Shipper is requesting that Transporter actively market the capacity to be released.

Issued by: Issued on:

Original Sheet No. 121

٦

GENE	ERAL TERMS AND CONDITIONS (CONTINUED)
(9)	The legal name of the Replacement Shipper that is designated in any pre-arranged release ("Prearranged Shipper").
(10)	Whether the capacity is to be released on a recallable basis, and, if so, (i) the terms and conditions of such recall, including whether it is recallable on a full day or a partial day basis, (ii) whether the Releasing Shipper's recall notification must be provided exclusively on a Business Day, (iii) which recall notification period(s), as identified in Section 25.1(d) above, will be available for use by the parties, and (iv) any reput methods and rights associated with returning the previously recalled capacity to the Replacement Shipper.
. (11)	Whether the capacity to be released is contingent on the release of other capacity, or on certain terms and conditions, and if so, the capacity, terms and/or conditions upon which the release is contingent.
(12)	The terms and conditions under which Releasing Shipper will accept contingent bids, including bids that are contingent upon the Replacement Shipper acquiring transportation on a pipeline interconnected to Transporter, the method for evaluating contingent bids, what level of proof is required by the contingent bidder to demonstrate that the contingency did not occur, and for what time period the next highest bidder will be obligated to acquire the capacity if the next winning contingent bidder declines the release.

Original Sheet No. 122

٦

GENER	AL TERMS AND CONDITIONS (CONTINUED)
(13) W R p p c c s s s s r r r r r r r r r r r r r r	Thether the Releasing Shipper will require the Replacement Shipper to post a deposit, not to exceed the amount required by Transporter bursuant to Section 25.2, to guard against bayment defaults if Transporter waives the Reposit requirement contained in Section 25.2. Such deposit will be paid by the Replacement Shipper to Transporter at the time specified in Section 25.2, and will be credited against the Replacement Shipper's invoices until fully stillized.
(14) I t m v 2 1 t t s s s	The bid evaluation method which shall be, at the Releasing Shipper's option, either one of the following three standard evaluation methods: highest rate, net revenue or present value; or alternative Releasing Shipper defined bid evaluation methods pursuant to Section 25.1(g)(15) below; provided, however, that Cransporter shall not be required to process the capacity release transaction using the standard process timeline should the Releasing Shipper elect an alternative method of bid evaluation.
(15) 2 c F n c F 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	At the Releasing Shipper's option and in lieu of Transporter implementing the Best Bid determination stated in Section 25.1(k), the Releasing Shipper may state the bid evaluation method. Such bid evaluation method shall be objectively stated, applicable to all Replacement or Prearranged Shippers and not unduly discriminatory and shall enable Fransporter to rank the bids received by utilizing the weight assigned by the Releasing Shipper to each element of the Releasing Shipper's Notice.
(16) # 5 1 1 2 1	Any restriction on the use of higher rate Secondary Delivery Points, or any requirement that the Replacement Shipper reimburse the Releasing Shipper for any incremental charges assessed by Transporter for use of Secondary Delivery Points by the Replacement Shipper.

Issued by: Issued on:

Original Sheet No. 123

(17) Any other additional information that Transporter deems necessary, from time to time, to effectuate releases hereunder.
Transporter shall not be liable for information provided by Shipper to Transporter, including any such information that is posted on the LINK® System.

Issued by: Issued on:

٦

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(h)	Open Bidding Process. Prospective Shippers wishing to acquire capacity available for release ("Bidding Shipper"), shall place a bid on the LINK® System for the available capacity during the Posting Period. If such bid is not expressly labeled as a contingent bid, such bid shall be binding. The bid shall contain the following information:
	(1) The Bidding Shipper's legal name and the name, title, address, phone number and e-mail address of the individual who will authorize the acquisition of the available capacity.
	(2) The level of daily firm entitlements that the Bidding Shipper requests and the minimum quantity it will accept.
	(3) The requested effective date and the term of the acquisition.
	(4) The Bidding Shipper's bid, addressing all criteria required by the Releasing Shipper. The Bidding Shipper shall be entitled to withdraw its bid either via the LINK® System or EDM, prior to the end of the bidding period. Bidding Shipper cannot withdraw its bid after the Bidding Period ends. If Bidding Shipper withdraws its bid, it may not resubmit a lower bid. If Bidding Shipper submits a higher bid, lower bids previously submitted by Bidding Shipper will be automatically eliminated. A Bidding Shipper may have only one valid bid posted. Transporter shall post all information provided by Bidding Shippers, except the information provided in Section 25.1(h)(1), above.

Issued by: Issued on:

٦

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	No bid shall exceed the maximum applicable Recourse Rates, in addition to any and all applicable fees and surcharges, as specified in this Tariff. The quantity or the requested term of the release of such bid shall not exceed the maximum quantity or primary term specified in the executed Agreement.
(i)	Pre-Arranged Release. Releasing Shipper shall have the right to elect not to post a release for bidding if (1) the proposed capacity release has a duration of thirty-one (31) days or less and Releasing Shipper has obtained a Prearranged Shipper, or (2) for proposed capacity releases of any duration for which Releasing Shipper has obtained a Prearranged Shipper and the Prearranged Shipper is paying the maximum Recourse Rate and all other terms and conditions of the release are met. If Releasing Shipper exercises such right, Releasing Shipper must notify Transporter prior to the nomination of the released entitlements, and the Replacement Shipper shall adhere to the requirements set forth in Section 25.2. Releasing Shipper will post the information on the LINK® System by 9:00 a.m. the Day before the release transaction begins. The Replacement Shipper shall confirm the prearranged release by 9:30 a.m. and meet any eligibility requirements under this Section 25. Transporter will support the electronic upload of prearranged releases.

Original Sheet No. 126

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(j)	(CONTINUED) Matching Rights. A Prearranged Replacement Shipper shall have matching rights for a period of thirty (30) minutes following the time the Prearranged Shipper has been notified of the winning bid ("Matching Period"). In the event a higher bid is received, Transporter shall provide the Prearranged Shipper an opportunity during the Matching Period to match such higher bid. No later than 2:00 p.m. CT of the Day prior to the Day nominations are due, the Prearranged Shipper shall be notified via the LINK® System of the terms and conditions of the higher bid, and shall have the Matching Period to respond via the LINK® System. Absent a response from the Prearranged Shipper by 2:30 p.m. CT of the Day prior to the Day nominations are due, the capacity shall be awarded to the higher Bidding Shipper no later than 3:00 p.m. CT of the Day prior to the Day nominations are due.

_ _

Original Sheet No. 127

GENERAL TERMS AND CONDITIONS (CONTINUED)

Awarding of Capacity Available for Release. Capacity (k) will be awarded no later than 3:00 p.m. CT of the Day prior to the Day nominations are due. The capacity available for release shall be awarded to the Bidding Shipper with the highest bid ("Best Bid") matching all terms and conditions provided by the Releasing Shipper. If multiple bids meet the minimum conditions stated in the Releasing Shipper's Notice, Transporter shall award the capacity, best bid first, until all offered capacity has been awarded. If bids are received that do not match all the terms and conditions provided by the Releasing Shipper, bids will be evaluated by the criteria provided by the Releasing Shipper. If no criteria are provided by the Releasing Shipper, the Bidding Shipper bidding the highest present value shall be awarded the capacity. Present value shall be determined based on a 10% discount rate. The ultimate awarding of capacity will be posted on the LINK® System by 3:00 p.m. CT on the Day prior to the Day nominations are due. Unless the bidder was a contingent bidder and the contingency did not occur, Transporter will tender an Addendum, as described in Article 1 of the Capacity Release Umbrella Agreement, to the winning bidder by 10:00 a.m. of the Day nominations are due.

Transporter shall not award capacity release offers to the Replacement Shipper until and unless the Replacement Shipper meets Transporter's creditworthiness requirements applicable to all services that it receives from Transporter, including the service represented by the capacity release.

Issued by: Issued on:
Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(1)	Remaining Capacity. In the event that a Releasing Shipper does not release all of its firm entitlements, the Releasing Shipper shall remain responsible for the remaining entitlements and is entitled to utilize the remaining entitlements with the MDQ reduced accordingly by the released capacity quantities.
(m)	No Rollover. The Releasing Shipper shall not release firm entitlements that were previously released pursuant to Section 25.1(i) to the same Prearranged Shipper on a prearranged basis, until twenty-eight (28) Days after the end of the first release period, unless the Prearranged Replacement Shipper agrees to pay the maximum rate and meet all other terms and conditions of the release.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)
Obligations of Replacement or Prearranged Shippers. The Replacement or Prearranged Shipper must satisfy all other provisions of this Tariff governing Shipper eligibility and must execute all required agreements and acknowledgements before it may contract with Transporter for the released capacity. In addition, as a pre-requisite to becoming a Replacement or Prearranged Shipper, a party must have been placed by Transporter on Transporter's pre-approved bidder list that is posted on the LINX® System. To be placed on such list, a party must have been accepted by Transporter as satisfying the credit standards of Section 27 of these General Terms and Conditions, must have executed a Capacity Release Umbrella Agreement and must continue to satisfy the credit standards of Section 27 when its bid is made and accepted or it is offered as a Prearranged Shipper, as applicable. Transporter shall process requests for credit approval with diligence. Any previously listed party that fails to continue to satisfy the standards of Section 27 shall be deleted from the list. Transporter will waive the credit requirements of Section 27 on a non- discriminatory basis for Replacement or Prearranged Shipper and permit such Replacement or Prearranged Shipper and permit bids, if the Releasing Shipper provides Transporter with a guarantee or other form of credit assurance in form and substance satisfactory to Transporter of all financial obligations of the Replacement of Service to the Replacement or Prearranged Shipper if the release is pre-arranged and not subject to bidding or prior to the close of the bid period if the release is subject to bidding requirements of this Section 25. Any bid submitted will legally bind the Replacement or Prearranged Shipper to the terms of the bid if Transporter chooses such bid as the Best Bid until written or electronic notice of withdrawa lis received by Transporter. Bids cannot be withdrawa fiter the bid period ends. Once the Replacement on preareared of diverser de
capacity, the Replacement or Prearranged Shipper is awarded capacity, the Replacement or Prearranged Shipper becomes an existing Shipper like any other Shipper and is subject to the

Issued by: Issued on:

Original Sheet No. 130

GENERAL TERMS AND CONDITIONS (CONTINUED)

applicable provisions of Transporter's Tariff, including, but not limited to, Transporter's billing and payment and operational provisions. In addition, the Replacement or Prearranged Shipper as an existing Shipper may also release its capacity pursuant to this Section 25. Nominations may be submitted upon the award of capacity, and such nominations will be processed in accordance with the nomination and scheduling requirements of Sections 4 and 6 of these General Terms and Conditions; provided, however, in no circumstances will gas flow prior to the effective date of the release as specified in the Releasing Shipper's Notice.

- (o) Capacity Release Umbrella Agreement. All nominations, scheduling and billing will be done under the contract number provided in the Addendum.
- 25.2 Obligations of the Parties.
 - (a) Contractual Obligations. All Replacement Shippers shall be required to comply with the provisions of Rate Schedule FTS and these General Terms and Conditions and to accept by a release all Transportation rights and obligations of the Releasing Shipper with respect to the capacity released, including, but not limited to, nominations and Transportation Paths. Furthermore, the Releasing Shipper shall remain fully liable to Transporter for all reservation rates, including reservation type surcharges and direct bills that were due under the Releasing Shipper's Agreement. In the event that the Replacement Shipper invoiced amounts for reservation rates are in arrears by 60 days or more, the Releasing Shipper shall be responsible for paying all such amounts with the next invoice rendered to the Releasing Shipper by Transporter.
 - (b) Billing. Pursuant to Sections 9 and 10, Replacement Shipper shall be billed for all reservation type charges contained within its bid and all usage charges according to Section 3 of Rate Schedule FTS.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(c) Credits. Except as otherwise agreed to between Transporter and Releasing Shipper, Releasing Shipper shall receive a credit against its Monthly Reservation Charges equal to the amount of reservation rates contained within the Replacement Shipper's bid subject to the obligations of Releasing Shipper under Section 25.2(a).
	(d) Refunds. Releasing Shipper and any Replacement Shipper must track any changes in Transporter's rates approved by the Commission. In the event the Commission orders refunds of any such rates charged by Transporter and previously approved, Transporter and/or Releasing Shipper, as the case may be, must make corresponding refunds to such Releasing Shipper or any Replacement Shipper, to the extent that Releasing Shipper or Replacement Shipper(s) has paid a rate in excess of Transporter's just and reasonable, applicable maximum rates. Transporter shall assume no liability or responsibility whatsoever for the failure of the Releasing Shipper to comply with its obligations under this Section 25.2(d).
25.3	Posting of Purchase Offers. Transporter shall allow a potential Replacement Shipper to post for at least thirty (30) Days its offers to acquire released firm entitlements. The offer must contain the following information:
	(a) The potential Replacement Shipper's legal name and the name, title, address, phone number and e-mail address of the individual who will authorize the acquisition of the available capacity.

Original Sheet No. 132

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(b) The daily quantities of capacity which the potential Replacement Shipper requests.
	(c) The Receipt Points and/or Delivery Points where capacity is requested, as applicable.
	(d) The requested effective date and the term of the acquisition.
25.4	Marketing Fee. Transporter shall be entitled, upon Releasing Shipper's request, to actively market the capacity available for release on Releasing Shipper's behalf. Transporter and Releasing Shipper will negotiate the terms and conditions upon which Transporter will market the Releasing Shipper's capacity.
25.5	Permanent Releases. A Shipper which has a currently effective executed Agreement with Transporter under Transporter's Rate Schedule FTS may release its capacity to a Replacement Shipper for the remaining primary term of the contract and be relieved of all liability under its Agreement prospectively from the effective date of such release, provided that the following conditions are satisfied:
	(a) The Replacement Shipper executes a new Agreement under the applicable Rate Schedule;
	(b) The Replacement Shipper agrees that the minimum bid acceptable to Transporter shall be a bid for the remainder of the term of Releasing Shipper's service agreement at the rate(s) Releasing Shipper is obligated to pay Transporter for the capacity to be permanently released and accepts all obligations of the Releasing Shipper;
	(c) The Replacement Shipper meets all of the credit- worthiness requirements contained in Section 27 of the General Terms and Conditions of Transporter's Tariff.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(d) Transporter may refuse to allow a permanent capacity release if it has a reasonable basis to conclude that it will not be financially indifferent to the release. If Shipper's request to permanently release capacity is denied by Transporter, Transporter shall notify Shipper via e-mail and shall include in the notification the reasons for such denial.
25.6	Transporter's Rights to suspend and/or Terminate Temporary Capacity Release Transactions.
	(a) In the event of a temporary release for which: (i) the Releasing Shipper no longer maintains creditworthiness as outlined in Section 27 of Transporter's General Terms and Conditions and Transporter has terminated Releasing Shipper's Service Agreement; and(ii) the reservation charge specified in the applicable Addendum is less than the level of the reservation charge which the Releasing Shipper was obligated to pay Transporter (or, if the Releasing Shipper is paying a negotiated rate, the sum of all reservation-type and commodity-type charges), then Transporter shall be entitled to terminate the service described in the Addendum, upon 30 Days' written notice to the Replacement or Prearranged Shipper agrees, at its sole election, prior to the end of said 30-Day notice period to pay for the remainder of the term specified in the Addendum one of the following: (i) the reservation and commodity charges at levels which the Releasing Shipper was obligated to pay Transporter, (ii) the applicable maximum tariff rate, or (iii) such rate as mutually agreed to by Transporter and Replacement or Prearranged Shipper.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	(b) In the event of a temporary release for which the Replacement Shipper no longer satisfies Transporter's credit requirements as set forth in Section 27 of the General Terms and Conditions: (i) Transporter may notify the Releasing Shipper, without any liability or prior notice to Replacement Shipper, that the Replacement Shipper no longer meets the credit requirements of Transporter's Tariff; and (ii) subject to Transporter exercising its rights under Section 27 of the General Terms and Conditions to suspend and/or terminate such capacity release transaction, the firm capacity subject to the release transaction shall revert to Releasing Shipper immediately upon the effectiveness, and for the duration, of such suspension or permanently if the release transaction is terminated.
25.7	Notices to Releasing Shippers. Transporter shall provide the original Releasing Shipper with Internet E-mail notification reasonably proximate in time with any of the following formal notices given by Transporter to the Releasing Shipper's replacement Shipper(s), of the following:
	 (a) Notice to the Replacement Shipper regarding the Replacement Shipper's past due, deficiency, or default status pursuant to Transporter's tariff; (b) Notice to the Replacement Shipper regarding the Replacement Shipper's suspension of service notice; (c) Notice to the Replacement Shipper regarding the Replacement Shipper's contract termination notice due to default or credit-related issues; and (d) Notice to the Replacement Shipper that the Replacement Shipper(s) is no longer creditworthy and has not provided credit alternative(s) pursuant to Transporter's tariff.

Original Sheet No. 135

	GENERAL TERMS AND CONDITIONS (CONTINUED)
26. REQUESTS Specific directed	FOR SERVICE requests for information concerning service(s) should be to:
South Atter P.O. Houst Teler	heast Supply Header, LLC htion: Marketing Department Box 1642 con, Texas 77251-1642 phone: 1-800-827-LINK, or in Houston (713) 989-LINK
Transporte in writing	er shall provide the requested information orally, or g, as appropriate.
26.1 Reque	ests for Service.
(a)	Persons desiring a new service or an amendment to existing service under one of Transporter's Rate Schedules set forth in Volume No. 1 of Transporter's FERC Gas Tariff must be a LINK® System User pursuant to Section 24 of these General Terms and Conditions and must submit a request for service electronically via the LINK® System. Persons submitting a bid for firm service under one of Transporter's Open Access Rate Schedules pursuant to Section 27 of the General Terms and Conditions must submit the bid electronically via the LINK® System.
(b)	A request for a new service or an amendment to an existing service shall contain the information identified on the Request for Service Information List posted on Transporter's public web site, as such list may be amended from time to time. Requests to amend existing service that will affect a Shipper's financial obligations to Transporter, without regard to the impact of any applicable discount or negotiated rates, are referred to as Billing Amendments. Requests to amend existing service that will not affect a Shipper's financial obligations to Transporter, without regard to the impact of any applicable discount or negotiated rates, are referred

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	to as Non-Billing Amendments. A Shipper requesting a new service or an amendment to existing service shall also provide the following to Transporter:
	(1) Either at the time of the request for new service or an amendment to existing service is submitted to Transporter or at the time of execution of the Service Agreement, such other information (if any), in writing, as may be required to comply with regulatory reporting or filing requirements; and
	(2) Within ten (10) Business Days of the submittal of the request for new service or a request for a Billing Amendment, any credit information required to be provided pursuant to Section 27 of the General Terms and Conditions.
(c)	Neither a request for new service nor a request that would result in a Billing Amendment shall be deemed to have been received by Transporter until Shipper has submitted such request online via the LINK® System and Transporter has received the information required or requested pursuant to this Section 26.1 and Section 26.3 of the General Terms and Conditions. A request that would result in a Non-Billing Amendment shall be deemed to have been received on the date such request is submitted in the LINK® System. If Transporter requests additional information or assurance in accordance with this Section 26.1, and such additional information or assurance is received within ten (10) Business Days of Transporter's request, Shipper's request for service shall be deemed to have been received on the date on which Shipper's additional financial information is received by Transporter; otherwise, Shipper's request for service shall be deemed to be null and void.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	the request for service shall be considered to be null and void. In addition, Transporter shall reject any request for service created in the LINK® System by Shipper, but not submitted to Transporter within ninety (90) days of Shipper's creation of such request.
26.2 All foll	firm transportation requests shall be subject to the owing conditions:
(a)	No request for transportation from a Primary Receipt Point or to a Primary Delivery Point shall be considered valid or be granted if to do so would impair Transporter's ability to render existing services pursuant to Transporter's firm service rate schedule(s).
	Subject to the provisions of (a) above, amendments to any firm Service Agreement or exhibit to add additional Primary Delivery Point(s) pursuant to an applicable firm rate schedule will not be considered a new transaction for purposes of complying with this Section 26. Any Shipper receiving permission from Transporter to use any new Primary Receipt Point(s) or new Primary Delivery Point(s) shall be deemed to have complied with the requirements of this Section 26 for purposes of receiving priority in contracting for such new Primary Receipt Point(s) or new Primary Delivery Point(s) for a firm Receipt Point MDQ or Delivery Point MDQ over any third party subsequently requesting firm transportation under a firm rate schedule at that Primary Receipt Point(s) or Primary Delivery Point(s) if, at the time of Shipper's request, said third party's request has not been accepted by Transporter. The priority for such new Primary Receipt Point(s) or Primary Delivery Point(s) shall be determined in accordance with this Section 26.

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

(c) In certain situations, Transporter may use an accounting meter number to represent a physical location on its pipeline system. A Delivery Point identified on Shipper's executed Service Agreement(s) may be designated in the LINK® System by means of an accounting meter number and description that differs from the physical meter number and description specified on the Service Agreement. The same rights and obligations exist for both Transporter and Shipper regardless of whether a location is identified in Shipper's executed Service Agreement by means of a physical meter number or an accounting meter number.

26.3 Execution of Service Agreement and Amendments. A Service Agreement and/or an amendment to an existing Service Agreement shall be executed, as specified in this Section 26.3, by Shipper and Transporter following the completion of the approval process.

(a) All interruptible Service Agreements, all interruptible Service Agreement amendments, firm Service Agreements with a term of one (1) year or less, and all amendments for firm Service Agreements with a term of one (1) year or less shall be executed electronically via the LINK® System by Shipper and Transporter. All firm Service Agreements with a term of more than one (1) year and all amendments to firm Service Agreements with a term of more than one (1) year shall be executed in writing. A Service Agreement shall be executed and, if executed in writing returned to Transporter, within fifteen (15) days of the tender of a Service Agreement by Transporter. In the event Shipper fails to submit a valid nomination for transportation pursuant to an interruptible Service Agreement within ninety (90) days after the later of (i) the date service is to commence, (ii) the date the Service Agreement is fully executed by Shipper and Transporter, or (iii) the date that the facilities, if any, to be

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
	constructed are ready for service, the Service Agreement and the corresponding transportation request for service shall be considered null and void.
	(b) For each of Transporter's firm Rate Schedule(s), the applicable Executable Contract Summary, in the form contained in this FERC Gas Tariff, reflects contract specific information that appears in the Form of Service Agreement associated with each of Transporter's firm Rate Schedule(s) and will serve as the executable version of the Service Agreement for Service Agreements with a term of one (1) year or less, when execution is performed electronically via the LINK® System. The Executable Contract Summary, the Form of Service Agreement, the Exhibit(s) executed by Shipper and Transporter, the applicable rate schedule, the General Terms and Conditions of this FERC Gas Tariff, and any applicable negotiated rate or discount agreement will comprise the entire agreement between Shipper and Transporter.
	(c) For each of Transporter's interruptible Rate Schedule(s), the applicable Form of Service Agreement, in the form contained in this FERC Gas Tariff, will serve as the executable version of the Service Agreement. The Form of Service Agreement, the Exhibit(s) executed by Shipper and Transporter, the applicable rate schedule, the General Terms and Conditions of this FERC Gas Tariff, and any applicable negotiated rate or discount agreement will comprise the entire agreement between Shipper and Transporter.
26.4	Extension of Service Agreements. Prior to the expiration of the term of a Part 284 Service Agreement and prior to Transporter's posting the availability of capacity under Transporter's Right of First Refusal provisions, if applicable, Transporter and Shipper may mutually agree to an extension of the term of the Service Agreement (the exact length of which is to be negotiated on a case-by- case basis, in a not unduly discriminatory manner).

Original Sheet No. 140

GENERAL TERMS AND CONDITIONS (CONTINUED)

26.5 It is a condition precedent to project financing that Transporter enter into firm service agreements for a minimum quantity of capacity with Shippers (or assignees or replacement Shippers) that meet the credit criteria in Section 27 below.

Issued by: Issued on:

Original Sheet No. 141

27. CREDITWORTHINESS

- 27.1 (a) Transporter shall not be required to (i) execute an Agreement providing for service under the applicable Rate Schedule for any Shipper who fails to meet Transporter's standards for creditworthiness, or (ii) initiate service for a Shipper who subsequently fails to meet Transporter's standards for creditworthiness, or (iii) continue service for any Shipper who is or has become insolvent or who, at Transporter's request, fails within a reasonable period to demonstrate creditworthiness pursuant to Transporter's standards.
 - (b) To permit Transporter to conduct a creditworthiness review, a Shipper shall, upon request by Transporter, render to Transporter: (i) complete and current financial statements, including annual reports, 10K reports or other filings with regulatory agencies, prepared in accordance with generally accepted accounting principles, or for non U.S.-based Shippers, prepared in accordance with equivalent principles; (ii) a list of corporate affiliates, parent companies and subsidiaries; and (iii) any credit reports available from credit reporting agencies. In addition to the establishment of creditworthiness: (i) Shipper must not be operating under any chapter of the bankruptcy laws and must not be subject to liquidation or debt reduction procedures under state laws such as an assignment for the benefit of creditors, or any informal creditors' committee agreement; (ii) Shipper should not be subject to the uncertainty of pending liquidation or regulatory proceedings which could cause a substantial deterioration in its financial condition, a condition of insolvency, or the inability of Shipper to exist as an ongoing business entity; (iii) if Shipper has an ongoing business relationship with Transporter, no undisputed delinquent balances should be consistently outstanding for any services performed previously by Transporter, and Shipper must have paid its account

7

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

		GENERAL TERMS AND CONDITIONS (CONTINUED)
		in the past according to the credit terms and contract provisions and not made deductions or withheld payment for claims not authorized by contract; and (iv) no significant collection, lawsuits or judgments are outstanding which would adversely impact the ability of Shipper to remain solvent.
	(c)	For purposes of Section 27.1(b) above, the insolvency of a Shipper shall be presumed by the filing by such Shipper or any parent entity thereof of a voluntary petition in bankruptcy or the entry of a decree or order by a court having proper jurisdiction adjudging the Shipper or any parent entity thereof bankrupt or insolvent. The insolvency of a Shipper shall also be presumed by the filing by the Shipper or its parent entity of a voluntary or involuntary proceeding, reorganization, receivership, liquidation, a debt reduction procedure, assignment for the benefit of creditors, formal or informal creditor restructuring agreement, or the filing of any case under the United States Bankruptcy Code, or any other applicable federal or state law.
	(d)	If any of the events or actions described in Section 27.1(c) above shall be initiated or imposed during the term of service hereunder, Shipper shall provide notification thereof to Transporter within two (2) Business Days of any such initiated or imposed event or action.
27.2	Credi great with	t Requirements for long-term Shippers (contracts er than 1 year). Shipper shall at all times comply one of the following creditworthiness requirements:
	(a)	obligations under the Agreement) has an investment grade credit rating for its long term senior unsecured debt from Moody's Investor Service of Baa3 or better or from Standard & Poor's of BBB- or better. A Shipper who qualifies under this category initially but is later downgraded below such investment grade will be required to qualify pursuant to Section 27.2(b)below.

.

Original Sheet No. 143

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(d)	A Shipper whose long term senior unsecured debt does not have an acceptable rating as set forth in Section 27.2(a) above will be accepted as creditworthy if (i) Transporter determines that, notwithstanding the absence of an acceptable rating, the financial position of Shipper (or an affiliate who guarantees Shipper's obligations under the Agreement) is acceptable to Transporter and its lenders; or (ii) the Shipper provides an irrevocable letter of credit in an amount equal to twelve (12) months of estimated reservation charges under the Agreement; provided that such amount shall be adjusted annually to reflect any change in the estimated reservation charges under the Agreement for the succeeding twelve (12) months or (iii) Shipper provides other security acceptable to Transporter and its lenders, each acting reasonably.
	Transporter shall provide such Shipper with a written statement supporting Transporter's request for the security amount requested at the time such security is requested. If Transporter rejects the security provided by Shipper in accordance with Section 27.2(b)(i)-(iii) above, Transporter shall re-issue its request for the security and include a written explanation for the rejection of the security previously provided by Transporter.
(c)	Nothing herein shall be read to preclude Transporter from requiring, and enforcing for the term of the contracts, a greater amount of security in agreements supporting an application for a certificate to construct new or expanded facilities, including any replacement contract entered into upon a permanent release of capacity under such contract, any assignment of such contract or any resale of capacity subject to such contract in the event of a default.

Original Sheet No. 144

GENERAL TERMS AND CONDITIONS (CONTINUED)

- 27.3 Credit Requirements for short-term Shippers (contracts less than 1 year). Shipper shall establish credit in accordance with Section 27.2.
 - If a Shipper fails to establish creditworthiness as (a) provided in Section 27.2, Shipper may still receive Transportation Service if, and only if, Shipper furnishes and maintains in effect one of the following at Shipper's discretion and acceptable to Transporter: (i) a written guarantee for unconditional payment from a third party which is creditworthy as determined above; or (ii) an irrevocable standby letter of credit; or (iii) a prepayment amount equal to the amount which would be charged to Shipper for six (6) month's service or the term of service, whichever is less, plus an amount equal to the three highest cashout payments, if any incurred during the previous twelve months, plus an amount equal to the cost of gas associated with any lending requirements requested under Rate Schedule PALS (if no prior history exists between the parties, Transporter shall determine the amount of advance payment hereunder to be deposited with Transporter) or (iv) other security acceptable to Transporter.

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1



Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
(c)	Transporter's credit appraisal procedures involve the establishment of dollar credit limits on a standardized nondiscriminatory basis which appraisal shall consider a number of relevant factors including but not limited to the cost of constructing any applicable facilities. To the extent that a Shipper's account(s) with Transporter do not exceed such limits and/or provided no new information regarding Shipper's financial or business position becomes known to Transporter, no new credit approval shall be necessary for Shipper's existing Agreement(s) unless subsequently amended; provided however, that Transporter shall have the right, with Shipper's assistance and cooperation, to update Shipper's credit file at any time.

· · · · · · · · · · · · · · · ·	
	GENERAL TERMS AND CONDITIONS (CONTINUED)
28. RIGHI	OF FIRST REFUSAL
28.1	Purpose. The purpose of this Section 28 is to provide the necessary information pertaining to the right of long-term firm Shippers to continue firm Transportation Service at the expiration of their Agreements by exercising a right of first refusal.
28.2	Eligibility. Any Shipper with a firm Agreement under a Part 284 Rate Schedule with an initial term of greater than two (2) years must give notice to the Transporter that Shipper desires to continue its Agreement at least two years in advance of the end of the primary term of the Agreement, and any Shipper with a firm Agreement under a Part 284 Rate Schedule with a primary term of (i) at least twelve (12) months of consecutive Transportation Service, or (ii) firm Transportation Service Agreements with a primary term of more than one (1) year for service which is not available for twelve (12) consecutive months ("seasonal contracts") must give notice to Transporter that Shipper desires to continue its Agreement at least six (6) months in advance of the end of the primary term of the Agreement. Shipper also must agree that it will match (a) the longest term, up to the maximum term allowed by the Commission, and (b) the highest rate for such Service, up to the maximum Recourse Rate, that is offered by any other person desiring such capacity; provided, however, that Transporter shall not be obligated to provide service at less than the maximum Recourse Rate(s). A Shipper paying a negotiated rate which exceeds the maximum Recourse Rate will be considered for purposes of this Section 28.2 to be paying the maximum rate as set forth in the Statement of Rates for Transportation of Natural Gas. Failure of the Shipper to give the notice specified will constitute a waiver of the Shipper's right of first refusal.

Issued by: Issued on:

Original Sheet No. 148

	GENERAL TERMS AND CONDITIONS (CONTINUED)
28.3	Procedure.
	(a) Transporter shall notify Shipper no later than three (3) Months prior to the expiration of the Agreement whether any outstanding bona fide offers exist for Transporter's capacity at a higher rate and/or for a longer term which could be satisfied by the relinquishment of Shipper's capacity. Offers will be deemed bona fide if made in compliance with Section 26 of these General Terms and Conditions. Any party that has an outstanding request for firm service under Section 26 of these General Terms and Conditions shall be notified and given the opportunity to specify the rate and term it is willing to offer for Shipper's capacity. If Transporter has received any such offers, Transporter shall inform Shipper of the rate, up to the maximum rate, and the term, up to a maximum time allowable by the Commission, that has been offered for Shipper's capacity. Shipper shall notify Transporter within ten (10) Business Days after notification whether it desires to match the rate and term offered, and, if so, to provide a binding commitment in writing to Transporter to execute a contract containing said terms within the next thirty (30) Business Days.
	(b) If Transporter does not notify Shipper of the existence of any offers for Shipper's capacity under Section 28.3(a), Transporter and Shipper may negotiate the terms and conditions of a new Agreement; provided, however, that in no event shall Shipper have any automatic right to renew service at a negotiated or discounted rate; provided further, however, Shipper may select the term of the Agreement after agreeing to pay maximum rates, and all applicable surcharges.

Issued by: Issued on:

Original Sheet No. 149

_

	GENERAL TERMS AND CONDITIONS (CONTINUED)
29.	INCORPORATION IN RATE SCHEDULES AND AGREEMENTS
	These General Terms and Conditions are incorporated in and are a part of Transporter's Rate Schedules and Agreements. To the extent there is any inconsistency between terms in these General Terms and Conditions and terms in Transporter's Rate Schedules or Agreements, these General Terms and Conditions shall govern.
30.	NEGOTIATED RATES
	30.1 Availability. Notwithstanding anything to the contrary contained in this Tariff, Transporter and Shipper may mutually agree to a negotiated rate and contract term for all or any portion of the capacity under any Part 284 Agreement, provided that Shipper has not acquired its capacity under the capacity release provisions of Section 25. If only a portion of the capacity under any Agreement will be priced at negotiated rates, the original Agreement must first be bifurcated, and the existing maximum or discounted recourse rates will continue to apply to the Agreement not subject to the negotiated rates. If Transporter and Shipper fail to agree to a negotiated rate, Shipper may receive service at the applicable maximum tariff rates, including surcharges, for service under the Rate Schedule applicable to the service.
	30.2 Filing Requirement. Transporter will submit to the Commission a tariff sheet stating the exact legal name of the Shipper, the negotiated rate, the rate schedule, the contract term, the Receipt Point(s), Delivery Point(s), the MDQ, and where applicable, the exact formula underlying a negotiated rate for any negotiated rate agreement. Unless Transporter executes and files a non-conforming Agreement, such tariff sheet will contain a statement that the negotiated rate agreement does not deviate in any material respect from the Form of Service Agreement for the applicable Rate Schedule.
	such tariff sheet will contain a statement that the negotiated rate agreement does not deviate in any material respect from the Form of Service Agreement for the applicable Rate Schedule.

Issued by: Issued on:

Effective on:

-

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

	GENERAL TERMS AND CONDITIONS (CONTINUED)
30.3	Limitations. This Section 30 does not authorize Transporter to negotiate terms and conditions of service.
30.4	Right of First Refusal. For purposes of exercising rights to continue service pursuant to Section 28 of these General Terms and Conditions, the highest rate that a Shipper must match if it desires to retain all or a portion of its capacity, and continue to receive firm service under the same rate schedule beyond the expiration date of such long- term firm Agreement, is the recourse rate for such service.
30.5	Accounting Treatment. Transporter shall maintain a separate account within Account 489.2, Revenues from transportation of gas of others through transmission facilities, for recording all revenues associated with charging negotiated rates. Transporter shall record each volume transported, billing determinant, rate component, surcharge, and the revenue associated with its Negotiated Rates so that this information can be filed, separately identified, and separately totaled, as part of and in the format of Statements G, I, and J in Transporter's next Section 4 rate case.

Issued by: Issued on:

Original Sheet No. 151



Issued by: Issued on:

Electr	12223333333333333333333333333333333333	C 1101 1101 1234556789	D,08,7753198652		Y, 119, 12844220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884220 10884000000000000000000000000000000000	Me 4 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3 3		h 322 333333333333333333333333333333333	ni, 121223456778	(sm.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		an, 23.1 33.3 33.3 33.3 33.3 33.3 33.3 33.3	NU da.3, 22086421088	E = 12) 554	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	4.1,3233456788	24,319754219	6 43 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4			214,4234865320 .54,420865320	7,.34	4.444444444		2.145234456789	86.531976431	4	447.44444444	22	91.623455678	
Capaci	ty23335555555555555555555555555555555555	Re: .1 .2 .3 .4 .1	lea ,9,,9,,7,,5	s 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	Re] 2, 30 .30 .52	at 5),), , , , , , , , , , ,	e355545	d .533414	St ,2331,0,1	an., 1,, 5, 8,	r94	ds,3332114.	5.22 34 , 19	3 , , , 55 ,	10555445	,333314	52.34,22	3,,,,50,	4.45 5.55 5.55 5.55 5.55 5.55 5.55 5.55	1	3.	5.44 342. 21	3,,,45,	1255 54 ai	2,3 .3 .1 .1		3,75445.	3.6.	17.55,422	·33 ·51	28	, 67

Issued by: Issued on:

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1





Transporter shall post on its Internet Website its procedures for implementation of and compliance with the Commission's Standards of Conduct regulations. All information required to be posted pursuant to such regulations, including, but not limited to, organizational charts, information on shared facilities and shared operating personnel, discounts granted, and notices of waivers and/or exercises of discretion in the application of tariff provisions, will be provided on Transporter's Internet Website under Informational Postings. Such information will be updated as required by applicable regulation(s) issued by the Commission.

Southeast

Issued by: Issued on:

Effective on:

Original Sheet No. 154

Original Sheet No. 155

	GENERAL TERMS AND CONDITIONS (CONTINUED)
34.	GENERAL TERMS AND CONDITIONS (CONTINUED) LIMITATION OF LIABILITY OF MEMBERS AND OPERATOR Shipper acknowledges and agrees that (a) Transporter is a pelaware limited liability company; (b) Shipper shall have no reacourse galist any member of Transporter with respect to fransporter's obligations under any Agreement and its sole recourse shall be against the assets of Transporter, irrespective of any failure to comply with applicable law or any provision of any Agreement; (c) no claim shall be made against the company operating the business and physical operations of Transporter or its members or the officers, employees, and agents of operator or its members (collectively "Operator"), under or in connection with any Agreement and the performance by Operator of its duties as Operator (provided that this provision shall not bar claims resulting from the gross negligence or willful misconduct of the Operator) and Shipper's insurance company for all such claims; and (b) this representation is made expressly for the benefit of the members in Transporter and the Operator.

Original Sheet No. 156

GENERAL	TERMS	AND	CONDITIONS
	(CONT	INUE	D)

35. DISCOUNT POLICY

- 35.1 Any Shipper desiring a discount of the maximum recourse rates for service under Transporter's open-access rate schedules must submit a valid request for such discount pursuant to the procedures of this Section 35. To be considered a valid request, Shipper must complete and submit a request for discount via the LINK® System, specifically including the information for all mandatory fields. Upon receipt of a valid request for a discount, Transporter will log such request and either deny or grant such request.
- 35.2 If and when Transporter discounts the rates applicable for service under any Service Agreement under Rate Schedules included in Transporter's FERC Gas Tariff, the amount of any such discount shall be accounted for as a reduction of maximum rates in the following sequence to the extent any of the following components are included in the maximum rate; the first item discounted shall be trackable rate components (if any), to the extent not otherwise agreed to in approved settlements, followed by the base rate (maximum less minimum rate and excluding all other components specified here).
- 35.3 In the event that Transporter agrees to discount its maximum recourse rates under any of its open-access rate schedules, Transporter and Shipper may agree to the types of discounts specified herein without such discounts constituting a material deviation from Transporter's pro forma service agreement. Transporter and Shipper may agree that a specified discounted rate will apply: (i) only to specified quantities under the Service Agreement; (ii) only if specified quantities are achieved or only with respect to quantities below a specified level; (iii) only during specified periods of the year or for a specifically defined period; (iv) only to specified points, combination of points, markets, transportation paths or other defined geographic area(s); (v) only to reserves committed by Shipper; (vi) only in a specified relationship to the quantities actually delivered (i.e., that the reservation charge will be adjusted in a specified relationship to quantities actually delivered);

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

GENERAL TERMS AND CONDITIONS (CONTINUED)

(vii) so that the applicable rate may be adjusted in the following manner: when one rate component, which was equal to or within the applicable maximum and minimum recourse rates at the time Shipper received the Discount Confirmation pursuant to Section 35.5 below specifying the terms of the discount, subsequently exceeds the applicable maximum recourse rate or is below the applicable minimum recourse rate, so that such rate component must be adjusted downward or upward to equal the new applicable maximum or minimum recourse rate, then other rate components may be adjusted upward or downward to achieve the agreed-upon overall rate, so long as none of the resulting rate components exceed the maximum recourse rate or are below the minimum recourse rate applicable to the rate component (such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets; however, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable); and/or (viii) based upon published index prices for specific receipt and/or delivery points or other agreed-upon published pricing reference points for price determination (such discounted rate may be based upon a single published index price or the differential between published index prices or arrived at by formula; provided that the discounted rate shall not change the underlying rate design, shall not include any minimum bill or minimum take obligation, and shall define the rate component to be discounted). Notwithstanding the foregoing, no discounted rate shall be less than the applicable minimum rate.

- 35.4 In the event that Transporter rejects Shipper's request for a discounted rate, Transporter shall notify Shipper via e-mail of the reason for such rejection.
- 35.5 The terms of any discount request granted by Transporter pursuant to this Section 35 shall be transmitted by e-mail to Shipper in the form of a Discount Confirmation. The Discount Confirmation shall identify the applicable Shipper's name,

Issued by: Issued on:

GENERAL TERMS AND CONDITIONS (CONTINUED)

contract number, rate schedule, term of the discount, discount rate, applicable quantities, Receipt Point(s) and delivery, and/or the pipeline path being discounted. The Discount Confirmation may also include other information required for posting under the Commission's regulations and other conditions consistent with Section 35.3. No particular discount transaction shall be contractually binding on either Transporter or Shipper until Transporter has confirmed the terms of the discount upon Transporter's e-mail to Shipper of the Discount Confirmation for the transaction, subject to the underlying Service Agreement being fully executed. All discounts granted shall be effective no sooner than the beginning of the next gas day following the Gas Day on which the request is granted by Transporter. Once the discount is contractually binding, the Discount Confirmation will constitute an addendum to the underlying Service Agreement. Each such addendum is an integral part of the underlying Service Agreement as if executed by both parties and fully copied and set forth at length therein.

35.6 If Transporter's recourse rates are subject to refund at any time during the effectiveness of a Discount Confirmation, with respect to the applicable discounted rate, Shipper shall be entitled to refunds of payments made by Shipper only in the event that the final, non-appealable maximum recourse rate, whether usage-based or reservation-based, as determined by the Commission for a given time period is lower than the rate actually paid by Shipper during such time period. Subject to the condition precedent set forth in the immediately preceding sentence, Shipper's principal refund amount shall be equal to (i) with respect to usage-based rates, the product of (aa) the positive difference between the final, non-appealable maximum recourse rate and the discounted rate, and (bb) the quantities of gas delivered to Shipper, or for Shipper's account, during the refund period; and (ii) with respect to reservation-based rates, the product of (cc) the positive difference between the final, non-appealable maximum recourse rate and the discounted rate, (dd) the MDQ covered by the discounted rate, and (ee) the number of Months in the refund period (partial Months shall be prorated for the number of Days in the Month that fall within the refund period and a discounted rate that is not a Monthly rate shall be adjusted for purposes of this calculation to reflect the Monthly equivalent of the rate).

Original Sheet No. 159

GENERAL TERMS AND CONDITIONS (CONTINUED) OFF-SYSTEM PIPELINE CAPACITY 36. From time to time, Transporter may enter into transportation and/or storage agreements with other interstate or intrastate pipeline companies (individually, an "off-system pipeline"). In the event that Transporter acquires capacity on an off-system pipeline, Transporter will use such capacity for operational reasons and will only render Transportation Service to Shippers on the acquired capacity pursuant to Transporter's Tariff and subject to Transporter's approved rates, as such tariff and rates may change from time to time. For purposes of Transportation Service on an off-system pipeline, the "shipper must have title" requirement is waived, permitting a Shipper utilizing such Service to have title to the Gas on such off-system pipeline.

Original Sheet No. 160

DESCRIPTION				-	Sheet No.	
FTS Service PALS Service Capacity Rel LINK® System ITS Service	Agreement e Agreement .ease Umbrella .Agreement	 Agreemen 	 nt 	· · · · · ·	. 161 . 167 . 170 . 180 . 183	
						•
	DESCRIPTION FTS Service PALS Service Capacity Rel LINK® System ITS Service	DESCRIPTION FTS Service Agreement PALS Service Agreement . Capacity Release Umbrella LINK® System Agreement ITS Service Agreement	DESCRIPTION FTS Service Agreement	DESCRIPTION FTS Service Agreement	DESCRIPTION	DESCRIPTION Sheet No.

Issued by: Issued on:

Original Sheet No. 161

Southeast Supply Header, LLC Executable Contract Summary for Service Agreement under Form of Service Agreement in Original Volume No. 1
DATE (1): CONTRACT NO.(1A) RATE SCHED: FTS MLL:
PRIMARY TERM: Effective on (5A) Ending on (5B) and From (5C) To (5D) with Written Notice of (5E)
SERVICE REQUESTER NAME (2):
EXHIBIT C DATED (14): SUPERCEDES EXHIBIT C DATED (15):
MDQ (3) Effective From: Effective To:
SUPERCEDING CONTRACT (7): SUPERCEDED BY:
EXHIBIT A DATED (8): SUPERCEDES EXHIBIT A DATED (9):
FIRM POINTS OF RECEIPT (10):
Maximum Receipt Location: RMDQ: Effective From: Effective To: Pressure
EXHIBIT B DATED (11): SUPERCEDES EXHIBIT B DATED (12):
SPECIFIC FIRM POINTS OF DELIVERY (13):
Location: DMDQ: Effective From: Effective To: Pressure
This Service Agreement, executed, pursuant to Transporter's effective tariff between Transporter and Service Requester is heretofore made a part of and subject to the aforementioned Form of Service.

Issued by: Issued on:

Original Sheet No. 162

FORM OF SERVICE AGREEMENT
(APPLICABLE TO RATE SCHEDULE FTS)
Date: (1), Contract No. (1A)
SERVICE AGREEMENT
This AGREEMENT is entered into by and between Southeast Supply Header, LLC, ("Transporter") and
(2) ("Shipper").
In consideration of the premises and of the mutual covenants herein contained, the parties do
agree as follows:
1. Transporter agrees to provide and Shipper agrees to take and pay for service under this
Agreement pursuant to Transporter's Rate Schedule FTS and the General Terms and Conditions of
Transporter's Tariff, which are incorporated herein by reference and made a part hereof.
2. The Manimum Daily Quantity (MDQ) for corrigo under this Agreement and any right to increase
2. The Maximum Daily Quantity (MDQ) for service under this Agreement and any right to increase
bereto The Point(s) of Delivery are listed on Exhibit B attached hereto. Exhibit(s) A B
and C are incorporated herein by reference and made a part hereof
and o are incorporated nertin by relationed and made a part nerter.
3. This Agreement shall be effective on (5A) and shall continue until (5B)
("Primary Term") and from (5C) to (5D) thereafter (not less than
year to year for the secondary term for Agreements with a primary term of more than 1 year)
until terminated by Transporter or Shipper upon at least (5E) [not less than 2 years
for agreements with a primary term of 2 years or more and not less than 1 year for agreements
with a primary term of more than 1 year but less than 2 years) prior written notice (if
Transporter and Shipper agree on a fixed term, the evergreen and notice of termination
language shall be deleted). Any portions of this Service Agreement necessary to correct or
cash-out imbalances under this Service Agreement as required by the General Terms and
Conditions of Transporter's FERC Gas Tariff shall survive the other parts of this Service
Agreement until such time as such balancing has been accomplished.
[Only with respect to an Igreement that was executed by Transporter and Shipper on or before
Confy with respect to an Agreement that was executed by framsporter and shipper on of before
becentber 257 20007 and the rotrowing ranguage.
Notwithstanding any other provision in this Agreement, after service has commenced hereunder
if as a result of an event of Force Majeure Transporter is not able to deliver Shipper's
scheduled quantities for a period of one hundred eighty five (185) consecutive days during any
three hundred sixty five (365) consecutive day period and at any minimum delivery pressure
specified on Exhibit B of this Agreement, then Shipper shall have the right to terminate this
Agreement or reduce the MDQ (with an associated reduction in the Delivery Point MDQs specified
on Exhibit B of this Agreement) of this Agreement upon sixty (60) days prior written notice.]

i

Issued by: Issued on:

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE FTS)

- 4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted or negotiated rate to which Shipper and Transporter have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Service Agreement. Provisions governing such negotiated rate and term shall be as specified on an appropriate rate sheet filed, with the consent of Shipper, as part of Transporter's Tariff. It is further agreed that Transporter may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Transporter's Tariff, and Transporter shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Shipper any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
- 5. Unless otherwise required in the Tariff, all notices shall be in writing and mailed to the applicable address below or transmitted via facsimile. Shipper or Transporter may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:

Transporter:

Shipper:(6)

- 6. The interpretation and performance of this Agreement shall be in accordance with the laws of the State of ______, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.
- This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable,
 - (7) [None or an appropriate description]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above.

|--|

SOUTHEAST SUPPLY HEADER, LLC

By:	:	
_		

Ву:_____

Title:____

Title:_____

Issued by: Issued on:
Original Sheet No. 164

	FORM OF SERVICE :	AGREEMENT
	(APPLICABLE TO RATE :	SCHEDULE FTS)
	EXHIBIT :	A
	Point(s) of R	eceipt
	Dated: (8)	
To the service agree (Transporter) and (2	ment under Rate Schedule FTS bet)	<pre>ween Southeast Supply Header, LLC (Shipper) concerning Point(s) of</pre>
Receipt.		
The receipt points a the following, and a	vailable to Shipper pursuant to	Section 4.1 of Rate Schedule FTS includes
Agreement:	ny dadicional receipt points con	
Receipt	Receipt Point	Maximum
Point	MDQ	Receipt Pressure
(10)		
Signed for Identific	ation	
Transporter:		
Shipper:		
Supercedes Exhibit A	. Dated (9)	
·	· · · · · · · · · · · · · · · · · · ·	
		, , , , , , , , , , , , , , , , , , ,
Issued by:		

Issued on:

Original Sheet No. 165

	FORM OF SERVICE AG	REEMENT
	ALLUTCHDIE IO KALE 201	101 E 101
	Exhibit B	
	Point(s) of Deli	very
Ι	Pated: (11)	
To the service agre (Transporter) and (Delivery.	ement under Rate Schedule FTS b 2)	etween Southeast Supply Header, Li (Shipper) concerning Point(:
Primary	- • •	
Point of	Delivery Point	Minimum Delivery Pressure
Defivery	MUQ	Derryety Lieszate
(13)		
Signed for Identifi	cation	
Transporter:		
Shinner.		
authhet:	<u></u>	

Issued by: Issued on:

Original Sheet No. 166

	(APPLICABLE TO RATE SCREDULE FTS)	
	Exhibit C	
	Transportation Quantities	
	- ···	
	Dates. (14)	
To the s (Transpo quantiti	ervice agreement under Rate Schedule FTS between Southeast Supply Header, LLC rter) and (2) (Shipper) concerning transporta es.	tic
MAX IMUM Dt	DAILY QUANTITY (MDQ): (3) Dth h Period	
Signed f	for Identification	
Transpor	ter:	
Shipper	· · · · · · · · · · · · · · · · · · ·	
Superse	des Exhibit C Dated (15)	

Issued by: Issued on:

Original Sheet No. 167

FORM OF SERVICE AGREEMENT (APPLICABLE TO RATE SCHEDULE PALS)
Date:, Contract No
SERVICE AGREEMENT
This AGREEMENT is entered into by and between Southeast Supply Header, LLC, ("Transporter") and ("Shipper").
In consideration of the premises and of the mutual covenants herein contained, the parties do agree as follows:
 Transporter agrees to provide and Shipper agrees to take and pay for service under this Agreement pursuant to Transporter's Rate Schedule PALS and the General Terms and Conditions of Transporter's Tariff, which are incorporated herein by reference and made a part hereof.
2. The Maximum Park Quantity or Maximum Loan Quantity, as appropriate, and the PALS Point(s) of Transaction are set forth in the Exhibit(s) A to this agreement. Shipper shall initiate a request for each park or loan service transaction by executing and delivering to Transporter one or more Exhibit(s) A. Upon execution by Transporter, Shipper's Exhibit(s) A shall be incorporated in and made a part hereof.
3. This Agreement shall be effective on and shall continue until and including ("Primary Term") and from to thereafter until terminated by Transporter or Shipper upon at least prior written notice. Any portions of this Service Agreement necessary to correct or cash-out imbalances under this Service Agreement as required by Rate Schedule PALS and the General Terms and Conditions of Transporter's FERC Gas Tariff shall survive the other parts of this Service Agreement until such time as such balancing has been accomplished.
4. Maximum rates, charges, and fees shall be applicable to service pursuant to this Agreement except during the specified term of a discounted or negotiated rate to which Shipper and Transporter have agreed. Provisions governing such discounted rate shall be as specified in the Discount Confirmation to this Service Agreement. Provisions governing such negotiated rate and term shall be as specified on an appropriate rate sheet filed, with the consent of Shipper, as part of Transporter's Tariff. It is further agreed that Transporter may seek authorization from the Commission and/or other appropriate body at any time and from time to time to change any rates, charges or other provisions in the applicable Rate Schedule and General Terms and Conditions of Transporter's Tariff, and Transporter shall have the right to place such changes in effect in accordance with the Natural Gas Act. Nothing contained herein shall be construed to deny Shipper any rights it may have under the Natural Gas Act, including the right to participate fully in rate or other proceedings by intervention or otherwise to contest increased rates in whole or in part.
5. Unless otherwise required in the Tariff, all notices shall be in writing and mailed to the applicable address below or transmitted via facsimile. Shipper or Transporter may change the addresses or other information below by written notice to the other without the necessity of amending this Agreement:
Transporter:
Shipper:

Original Sheet No. 168

FORM OF SERVI	CE AGREEMENT			
(APPLICABLE TO RAT	TE SCHEDULE PALS)			
6. The interpretation and performance of this Agreement shall be in accordance with the laws of the State of, excluding conflicts of law principles that would require the application of the laws of a different jurisdiction.				
 This Agreement supersedes and cancels, as of t contract(s) between the parties hereto as desc 	he effective date of this Agreement, the ribed below, if applicable:			
[None or an appropriate description]				
IN WITNESS WHEREOF, the parties hereto have cause respective Officers and/or Representatives thereu date stated above.	ed this Agreement to be signed by their anto duly authorized to be effective as of the			
SHIPPER:	SOUTHEAST SUPPLY HEADER, LLC			
Ву:	Ву:			
Title:	Title:			

Original Sheet No. 169

	SOUT PARK DA'	HEAST SUPPLY AND LOAN (PA TED	HEADER, LLC LS) AGREEMENT		
	EXHIBIT	A DATED _			
TRANSPORTER:	Southeast Supply 5400 Westheimer (Houston, Texas	Header, LLC Court 77056-5310			
	Attention: Duke	Energy Gas	Iransmission Ma	rketing Depart	ment
SHIPPER:					
· · · · · · · · · · · · · · · · · · ·	······				
	Commencem Service Date	Nent	Termination of Service Date	Maximum Park/Loan Quantity	Specific Points
Park Service				<u></u>	
Loan Service					
		SOUTH	ידע עוססווס שהיסעיני	ADER LIC	
		50011			
		Ву			
ATTEST:					
		[NAME	OF SHIPPER]		
		Bv			
			<u></u>		
ATTEST:					

Issued by: Issued on:

Original Sheet No. 170

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULE FTS

This Umbrella Service Agreement, made and entered into this _____ day of ______, by and between ______ (Replacement Shipper), and Southeast Supply Header, LLC (Transporter),

WITNESSETH:

WHEREAS,

NOW, THEREFORE, for and in consideration of the mutual covenants and promises herein contained, the Replacement Shipper and Transporter hereby agree as follows:

ARTICLE I SCOPE OF AGREEMENT

Subject to the terms, conditions and limitations hereof, so long as the financial evaluation and credit appraisal requirements are met in order for Replacement Shipper to be on Transporter's approved bidder list for capacity releases and execute this Umbrella Service Agreement pursuant to Section 25 of Transporter's GT&C, and this Umbrella Service Agreement is effective, Replacement Shipper may bid from time to time on proposed capacity releases under Rate Schedule FTS pursuant to the procedure set forth in Section 25 of Transporter's GT&C. If at anytime a bid submitted by Replacement Shipper is accepted by Transporter with respect to a given capacity release, Transporter will promptly finalize by means of Transporter's LINK® System the appropriate Addendum to this Umbrella Service Agreement, in the format attached hereto, depending upon the rate schedule under which the capacity is being released. An Addendum shall be deemed to be an executed Service Agreement under the rate schedule designated therein, subject to the terms and conditions of the rate schedule, the form of service agreement applicable to such rate schedule, and the General Terms and Conditions of Transporter's Tariff. The parties agree that each Addendum is an integral part of this Umbrella Service Agreement as if executed by the parties hereto and fully copied and set forth herein at length and is binding on the parties hereto. Upon finalization of such Addendum, Replacement Shipper and Transporter agree that Replacement Shipper shall be considered for all purposes as a Shipper with respect to the released service.

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULE FTS (CONTINUED)

Upon the finalization of an Addendum, subject to the terms, conditions and limitations hereof and of Transporter's Rate Schedule FTS, Transporter agrees to provide the applicable released service for Replacement Shipper under the applicable rate schedule, provided however, the Replacement Shipper qualified under the financial evaluation and credit appraisal requirements set forth in Section 27 of Transporter's GT&C at the time it submitted the bid Transporter accepted with respect to such release.

Replacement Shipper hereby agrees to promptly provide any information necessary for Transporter to reevaluate Transporter's credit appraisal as contemplated by Section 27 of Transporter's GT&C and to advise Transporter of any material change in the information previously provided by the Replacement Shipper to Transporter.

ARTICLE II TERM OF AGREEMENT

The term of this Agreement shall commence on ______ and shall continue in force and effect until ______ and ______ to ______ thereafter unless this Umbrella Service Agreement is terminated as hereinafter provided. If Transporter determines at anytime that Replacement Shipper fails to meet the financial standards or credit criteria of Section 27 of the GT&C, Transporter may terminate this agreement and all Addenda attached hereto prospectively in accordance with Section 27 of the GT&C.

ARTICLE III RATE SCHEDULE

This Umbrella Service Agreement does not have separate terms and conditions for particular services, but only provides a means for a Replacement Shipper to utilize a service subject to the applicable provisions of the relevant Service Agreement and the terms and conditions for Rate Schedule FTS, by finalization of a copy of an Addendum attached hereto and fully incorporated herein as a part of this Umbrella Service Agreement.

Issued by: Issued on:

Original Sheet No. 172

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULE FTS (CONTINUED)

If Replacement Shipper utilizes an agent, it will so indicate on the appropriate Addendum, along with any terms and conditions relevant to such agency relationship. Transporter will act in accordance with the Addendum and in so acting will be fully protected in relying upon such agent.

Replacement Shipper agrees that Transporter shall have the unilateral right to file with the appropriate regulatory authority and make changes effective in (a) the rates and charges applicable to service pursuant to this Umbrella Service Agreement (b) the terms and conditions of this Umbrella Service Agreement, pursuant to which service hereunder is rendered or (c) any provision of the GT&C applicable to this Umbrella Service Agreement. Transporter agrees that the Replacement Shipper may protest or contest the aforementioned filings, unless the Replacement Shipper has otherwise agreed not to protest or contest any or all of the aforementioned filings, and the Replacement Shipper does not waive any rights it may have with respect to such filings.

ARTICLE IV ADDRESSES

Except as herein otherwise provided or as provided in the GT&C of this Tariff, any notice, request, demand, statement, invoice or payment provided for in this Umbrella Service Agreement, or any notice which any party may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered, certified, or regular mail to the post office address of the parties hereto, as the case may be, as follows:

(a) Transporter:

(b) Replacement Shipper:

or such other address as either party shall designate by formal written notice.

Issued by: Issued on:

Original Sheet No. 173

Southeast Supply Header, LLC Pro Forma FERC Gas Tariff Original Volume No. 1

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULE FTS (CONTINUED)

ARTICLE V INTERPRETATION

The interpretation and performance of this Umbrella Service Agreement shall be in accordance with the laws of the State of ______, without recourse to the law governing conflict of laws.

This Umbrella Service Agreement and the obligations of the parties are subject to all present and future valid laws with respect to the subject matter, either State or Federal, and to all valid present and future orders, rules, and regulations of duly constituted authorities having jurisdiction.

ARTICLE VI RELATIONSHIP BETWEEN REPLACEMENT SHIPPER AND RELEASING SHIPPER

The parties recognize that, pursuant to Commission orders, Releasing Shipper may require that the Replacement Shipper agree that a breach of this Agreement, including a failure to pay, or to pay timely, by Replacement Shipper under this Agreement, constitutes a breach of contract as between Replacement Shipper and Releasing Shipper. The existence of such an agreement will be indicated on the appropriate Addendum to this Capacity Release Umbrella Agreement. If Replacement Shipper fails to pay Transporter, fails to timely pay Transporter, or otherwise breaches this Agreement with Transporter: (a) both Replacement Shipper and Releasing Shipper (except to the extent otherwise provided in Section 25.2(a) of the GT&C and except with respect to penalties attributable to Replacement Shipper's conduct) shall be liable to Transporter for such failure to pay or breach (it being understood that nothing in this Article VI relieves Releasing Shipper from responsibility to pay Transporter in accordance with its service agreements with Transporter) and (b) if, as a result of such breach by Replacement Shipper, Releasing Shipper is accordingly required to pay Transporter or otherwise perform, Releasing Shipper may have a cause of action for breach against Replacement Shipper.

Issued by: Issued on:

.

Original Sheet No. 174

IN WITNESS Service Agreeme	WHEREOF, the parties hereto have caused this Umbrel ant to be signed by their respective Presidents, Vice
Presidents or c seals to be her or Assistant Se	other duly authorized agents and their respective correctore affixed and attested by their respective Secreta accretaries, the day and year first above written.
	SOUTHEAST SUPPLY HEADER, LLC
ATTEST:	Ву
	By
ATTEST:	

Issued by: Issued on:

Original	Sheet	No.	175
----------	-------	-----	-----

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDEF RATE SCHEDULE FTS (continued)	3
Deal No.: Southeast Supply Header, LLC Addendum Contract No.: Capacity Release Umbrella Agreement No.:	
Addendum No Capacity Release Rate Schedule	
Replacement Shipper: Releasing Shipper:	
Releasing Shipper's Contract No.:	
Begin Date of Release:	
End Date of Release:	
U.S. \$	
Surcharges:	
Description	Rate
	\$ \$
	₹

Issued by: Issued on:

CAPA	FORM OF SERVI CITY RELEASE UM RATE SC. (con	CE AGREEMENT FOR BRELLA AGREEMENT HEDULE FTS tinued)	UNDER
Addendum No Capacity Release Rate Schedule	(Con't)		
Volume Commitment	(Dt)	n/Monthly Billing	Period)
Maximum Daily Qua	ntity (MDQ):	(Dth)	
Billable Quantiti	es:		
Service:			
From		To	Quantity

Original Sheet No. 177

	FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULE FTS (continued)
Addendum No Capacity Re Rate Schedu	(Con't) Lease Le
Specific Fi	<pre>cm Delivery Point(s):</pre>
Delivery Po:	Int Delivery Point MDQ Effective From Effective To
Is this capa	acity subject to right of recall? YesNo
Recall Cond:	tions (if applicable):
<u> </u>	
. <u> </u>	
Are there an Restrictions	<pre>ny restrictions on released capacity? Yes No s (if applicable):</pre>
Are there an Restriction:	ny restrictions on released capacity? Yes No s (if applicable):
Are there an Restrictions	<pre>hy restrictions on released capacity? Yes No s (if applicable): </pre>
Are there an Restrictions	<pre>hy restrictions on released capacity? Yes No s (if applicable): </pre>
Are there an Restriction:	<pre>hy restrictions on released capacity? Yes No s (if applicable): </pre>
Are there an Restrictions	<pre>hy restrictions on released capacity? Yes No s (if applicable): </pre>

Issued by: Issued on:

Original Sheet No. 178

FORM OF SERVICE AGREEMENT FOR CAPACITY RELEASE UMBRELLA AGREEMENT UNDER RATE SCHEDULE FTS (continued)
Addendum No (Con't) Capacity Release Rate Schedule
Was Southeasat Supply Header System, LLC's default bid evaluation criteria used? Yes No
Evaluation Criteria (if applicable):
Were contingent bids accepted? Yes No
Contingency comments (if applicable):
Other Terms and Conditions of Release: [e.g., restrictions on release, third party agent and terms of third party agency relationship, and agreements between Replacement Shipper and Releasing Shipper]

Issued by: Issued on:

Original Sheet No. 181

FORM OF SERVICE AGREEMENT FOR THE LINK® SYSTEM (CONTINUED)

ARTICLE III ADDRESSES

Except as provided in the General Terms and Conditions of Transporter's Tariff, any notice, request, demand, statement, bill or payment pursuant to this LINK® System Agreement shall be in writing and shall be considered as duly delivered when received on-line via the LINK® System, or when received as registered, certified, or regular mail at the address of the parties hereto, as the case may be, as follows:

(a) Transporter:

(b) LINK® System Subscriber:

or such other address as either party shall designate by completing and submitting the information required by the Contact Information Form, as such form is amended from time to time and posted on the LINK® System.

ARTICLE IV INTERPRETATION

The interpretation and performance of this LINK® System Agreement shall be in accordance with the laws of the State of Texas without recourse to the law governing conflicts of law.

This LINK® System Agreement and the obligations of the parties are subject to all present and future valid laws with respect to the subject matter hereof, either State or Federal, and to all valid present and future orders, rules, and regulations of duly constituted authorities having jurisdiction.

Issued by: Issued on:

Original Sheet No. 182

	and the second
FORM OF SEE	RVICE AGREEMENT
FOR THE	LINK® SYSTEM
(CO)	NTINUED)
1	ARTICLE V
AGREEMENTS	5 BEING SUPERSEDED
When this LINK® System supersede any LINK® System Ag an earlier execution date.	Agreement becomes effective, it shall reement(s) between the parties hereto with
IN WITNESS WHEREOF, the System Agreement to be signed authorized, the day and year :	parties hereto have caused this LINK® by their respective agents thereunto duly first above written.
	SOUTHEAST SUPPLY HEADER, LLC
	By:
	Title:
	Signature
	LINK® System Subscriber
	By:
	Title:
	Signature
Issued by:	

Issued by: Issued on:

Original Sheet No. 184

<pre>District EXPERIENT LIPERATION AND CONTRIBUTION 1. On interpretation of the laws of a different jurisdiction. 2. One or an appropriate description EXPERIENT EXPER</pre>		······································
6. The interpretation and performance of this Agreement shall be in accordance with the laws of the state of	FORM (APPLIC)	M OF SERVICE AGREEMENT ABLE TO RATE SCHEDULE ITS)
7. This Agreement supersedes and cancels, as of the effective date of this Agreement, the contract(s) between the parties hereto as described below, if applicable: [None or an appropriate description] IN WINDESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives thereunto duly authorized to be effective as of the date stated above. SNUTHERST SUPPLY HEADER, LLC By:	 The interpretation and performance of the State of, excluding con application of the laws of a different 	f this Agreement shall be in accordance with the laws of nflicts of law principles that would require the nt jurisdiction.
<pre>[None or an appropriate description] TM WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective officers and/or Representatives thereunto duly authorised to be effective as of the attes stated above. SHIPPER:</pre>	 This Agreement supersedes and cancels contract(s) between the parties here 	s, as of the effective date of this Agreement, the to as described below, if applicable:
<pre>IN WIENESS WHEEREF, the parties hereto have caused this Agreement to be signed by their respective Officers and/or Representatives therewith duly authorized to be effective as of the aite stated above. SHIPPER:</pre>	[None or an appropriate descripti	ion]
SRIPPER:	IN WITNESS WHEREOF, the parties hereto be respective Officers and/or Representative date stated above.	have caused this Agreement to be signed by their ves thereunto duly authorized to be effective as of the
By:	SHIPPER:	SOUTHEAST SUPPLY HEADER, LLC
itle:	Ву:	By:
	Title:	Title:

Issued by: Issued on:

REDACTED

West/East Pipeline Transportation Purchase Business Analysis Package

Sponsoring Business Unit: Regulated Fuels (RFD) of behalf of Progress Energy Florida

Funding Legal Entity: Progress Energy Florida (PEF)

Date Prepared: January 16, 2007

Contacts to discuss project

	<u>Kole/Name</u>	Extension
Progress Fuels (RFD)	Bob Bazemore, VP and Project Sponsor	770-4083
Progress Fuels (RFD)	Kent Fonvielle, Manager	770-3257
Progress Fuels (RFD)	Rick Rhodes, Gas Supply Rep.	770-7613

CONFIDENTIAL EXHIBIT C REDACTED VERSION



Exhibit ____ Estimated Total Annual Pipeline Costs 80% Utilization



(2) Estimated variable costs based on 80% utilization of pipeline capacity.

CONFIDENTIAL EXHIBIT D

REDACTED





The projected in-service date of the Southeast Supply Header pipeline is June 1, 2008.
 Estimated variable costs based on 90% utilization of pipeline capacity.

CONFIDENTIAL EXHIBIT E

20061228-3043 Issued by FERC OSEC 12/28/2006 in Docket#: CP07-45-000

UNITED STATES OF AMERICA FEDERAL ENERGY REGULATORY COMMISSION

Southeast Supply Header, LLC Southeast Supply Header, LLC Southern Natural Gas Company Southeast Supply Header, LLC Southeast Supply Header, LLC Docket Nos. CP07-45-000 Docket Nos. CP07-44-000

Docket Nos. CP07-46-000 Docket Nos. CP07-47-000

NOTICE OF APPLICATION

(December 28, 2006)

Take notice that on December 18, 2006 Southeast Supply Header, LLC (SESH) filed an application in Docket No. CP07-45-000, requesting a certificate of public convenience and necessity pursuant to section 7 of the Natural Gas Act (NGA) and Subpart A of Part 157 of the Commission's regulations authorizing the Applicants to construct, own, operate, and maintain a new 269-mile natural gas pipeline system commencing in the area of the Perryville Hub near Delhi, Louisiana, continuing in a southeasterly direction through Mississippi and Alabama, and terminating near Coden, Alabama (SESH Project). In addition, SESH requests that the Commission issue to SESH: (i) a blanket certificate in Docket No. CP07-46-000 authorizing SESH to construct, operate, and abandon certain facilities under Part 157, Subpart F of the Commission's regulations; (ii) a blanket certificate in Docket No. CP07-47-000 authorizing SESH to transport natural gas, on an open access and self-implementing basis, under Part 284, Subpart G of the Commission's regulations; and (iii) authorizations necessary to charge initial recourse rates for certain services to be rendered by SESH.

Also take notice that on December 18, 2006, SESH and Southern Natural Gas Company (Southern Natural) filed with the Commission, in Docket No. CP07-44-000, for authorization under section 7 of the NGA and Part 157, Subpart A of the Commission's regulations to construct, own and operate certain jointly owned facilities which comprise a portion of the SESH Project (Joint Segment).

The application for SESH's and Southern Natural's proposals are more fully described as set forth in the applications that are on file with the Commission and open to public inspection. The instant filings may be also viewed on the web at http://www.ferc.gov using the "eLibrary" link. Enter the docket number excluding the last three digits in the docket number field to access the document. For assistance, call (866) 208-3676 or TTY, (202) 502-8659

	EXHIBIT
ibbles'	\square
3	· _

Docket Nos. CP07-44-000, et al. 2

Any initial questions regarding these applications should be directed to Brian D. O'Neill, LeBoeuf, Lamb, Greene & MacRae LLP. Telephone: (202) 986-8000.

On May 30, 2006, the Commission staff granted SESH's request to utilize the Pre-Filing process and assigned Docket No. PF06-28-000 to staff activities involving the SESH Project. Now, as of the filing of this application on December 18, 2006, the Pre-Filing Process for this project has ended. From this time forward, these proceedings will be conducted in Dockets No. CP07-44-000, CP07-45-000, CP07-46-000, and CP07-47-000 as noted in the caption of this Notice.

Pursuant to Section 157.9 of the Commission's rules, 18 C.F.R. §157.9 and to ensure compliance with the Energy Policy Act of 2005, the Commission staff will issue a Notice of Schedule for Environmental Review within 90 days of the date of this Notice. The Notice of Schedule for Environmental Review will indicate, among other milestones, the anticipated date for the Commission staff's issuance of the final environmental impact statement (FEIS) for SESH's and Southern Natural's proposal. The Notice will also alert other agencies of the requirement to complete necessary reviews and authorizations within 90 days of the date of issuance of the Commission staff's FEIS.

There are two ways to become involved in the Commission's review of this project. First, any person wishing to obtain legal status by becoming a party to the proceedings for this project should, on or before the below listed comment date, file with the Federal Energy Regulatory Commission, 888 First Street, NE, Washington, D.C. 20426, a motion to intervene in accordance with the requirements of the Commission's Rules of Practice and Procedure (18 CFR 385.214 or 385.211) and the Regulations under the NGA (18 CFR 157.10). A person obtaining party status will be placed on the service list maintained by the Secretary of the Commission and will receive copies of all documents filed by the applicant and by all other parties. A party must submit 14 copies of filings made with the Commission and must mail a copy to the applicant and to every other party in the proceeding. Only parties to the proceeding can ask for court review of Commission orders in the proceeding.

However, a person does not have to intervene in order to have comments considered. The second way to participate is by filing with the Secretary of the Commission, as soon as possible, an original and two copies of comments in support of or in opposition to this project. The Commission will consider these comments in determining the appropriate action to be taken, but the filing of a comment alone will not serve to make the filer a party to the proceeding. The Commission's rules require that persons filing comments in opposition to the project provide copies of their protests only to the party or parties directly involved in the protest.

Persons who wish to comment only on the environmental review of this project should submit an original and two copies of their comments to the Secretary of the 20061228-3043 Issued by FERC OSEC 12/28/2006 in Docket#: CP07-45-000

Docket Nos. CP07-44-000, et al. 3

ş

Commission. Environmental commenters will be placed on the Commission's environmental mailing list, will receive copies of the environmental documents, and will be notified of meetings associated with the Commission's environmental review process. Environmental commenters will not be required to serve copies of filed documents on all other parties. However, the non-party commenters will not receive copies of all documents filed by other parties or issued by the Commission (except for the mailing of environmental documents issued by the Commission) and will not have the right to seek court review of the Commission's final order.

Motions to intervene, protests and comments may be filed electronically via the internet in lieu of paper; see, 18 CFR 385.2001(a)(1)(iii) and the instructions on the Commission's web site under the "e-Filing" link. The Commission strongly encourages electronic filings.

Comment Date: January 18, 2007

Magalie R. Salas Secretary



Table of Contents



Table of Contents

Secretary's Message
Executive Order
l. Executive Summary
II. Florida's Ability to Generate, Transmit,
and Distribute Electric Power
a. Current and Projected Electric Generating Supply and Demand \ldots 13
b. Current and Projected Infrastructure Needs for Production and Supply . 21
c. Current and Projected Consumer Costs
d. Current Regulatory Oversight of Electric Utilities
e. Florida's Energy Efficiency and Alternative Energy Initiatives
III Flavida's Ability to Draduce Stave
III. Florida's Ability to Produce, Store
and Distribute Transportation Fuel
a. Current and Projected Supply and Demand for Transportation Fuel \ldots .30
b. Current and Projected Infrastructure Needs for Production and Supply . 31
c. Current and Projected Consumer Costs
d. Current Regulatory Oversight
e. Fuel Efficiency and Alternative Fuel Initiatives in Florida
IV. Overview of Florida Statutes
V. Recommendations for Electric Power Generation
VI. Recommendations for Transportation Fuels

1



"An adequate, reliable, diverse, efficient and affordable energy supply, coupled with a long-term commitment to energy conservation, is vital for maintaining Florida's growing economy and quality of life."

- Colleen M. Castille Secretary Department of Environmental Protection Governor Jeb Bush The Capitol 402 South Monroe Street Tallahassee, Florida 32399

Dear Governor Bush:

Since your issuance of Executive Order #05-241 on November 10, 2005, the Department of Environmental Protection has worked diligently to gather information toward a comprehensive energy plan for the State of Florida. As directed in your Order, the Department explored options for diversifying Florida's electric generation capacity, increasing and diversifying transportation fuel supply and increasing the State's conservation and efficiency initiatives.

Nationwide, demand for energy and transportation fuel is outpacing supply. At times over the last 18 months the margin of spare capacity in the oil market has fallen from the historic norms. In addition, the storms and hurricanes of 2004 and 2005 impacted lives and properties in a way we have never experienced before, severely disrupting petroleum and oil production and the nation's fuel supply systems. Production platforms in the Gulf of Mexico were shut down, pipelines were inoperable, and refining systems were off line for months – exposing our vulnerability and reliance on natural gas.

By evaluating Florida's current and future energy supply and demand, the Department has developed a suite of recommendations, built on the principals of conservation and efficiency, which provide the foundation for a far-reaching energy plan. In developing its proposals, the Department adopted two guiding principles: reliance on markets and no new taxes. Instead of mandates, recommendations rely on the power of the marketplace, using targeted incentives and government's purchasing power to stimulate the free market. Consumers have already been impacted with rising fuel costs – government should not add to that burden.

The Department is recommending that legislation be introduced during the 2006 Regular Legislative Session to increase capacity and diversify Florida's electric generation and transportation fuel supply by:

- Amending the Power Plant Siting Act and the Florida Electrical Transmission Line Siting Act to reduce regulatory barriers and streamline permitting.
- Amending Chapter 403.519, Florida Statues, to allow the Florida Public Service Commission to consider fuel diversity and fuel reliability as factors when determining the need for new electric generation.

January 17, 2006

- Increasing grant funding for research and demonstration projects associated with renewable energy systems and alternative fuel vehicles.
- Providing consumer and corporate rebates to encourage investments in solar technologies and ENERGY STAR[™] appliances.
- Providing sales and corporate income tax incentives to encourage the production of clean fuels in Florida and for pollution-free hydrogen fuel cells, vehicles and fueling infrastructure.
- Establishing an energy council to provide policy advice and counsel to the Governor, Speaker of the House and President of the Senate.

Administratively, the Department will immediately begin working with other state agencies and entities to improve energy diversity, sustainability, efficiency and conservation statewide by:

- Adopting updated interconnection standards to include all distributed generation technologies, such as wind, solar and biomass.
- Requiring all new State government buildings to meet U.S. Green Building Council's Leadership in Environmental Design standards.
- Expediting State performance contracting with Energy Service Companies.
- Promoting awareness of energy conservation, alternative energy technologies and alternative fuel vehicles.
- Facilitating redundant and diverse petroleum supply and distribution mechanisms into and within Florida.
- Encouraging fueling stations to adopt a generator sharing program before the upcoming hurricane season as a cooperative method for allocating generators and reestablishing temporary power service after storm events.
- Fostering a state-level partnership with local planning boards to encourage well-designed transportation and transit systems within new community development.

An adequate, reliable, diverse, efficient and affordable energy supply, coupled with a long-term commitment to energy conservation, is vital for maintaining Florida's growing economy and quality of life. Florida must act now to respond to growing demand and to overcome the vulnerabilities highlighted by the hurricanes. We look forward to finalizing a comprehensive energy plan that will provide long-term energy security for the State of Florida.

Sincerely,

Coller M Cestill

Colleen M Castille Secretary Florida Department of Environmental Protection

STATE OF FLORIDA OFFICE OF THE GOVERNOR EXECUTIVE ORDER NUMBER 05-241

WHEREAS, according to a 2001 study by the United States Energy Information Administration, Florida ranks third in total energy consumption; and

WHEREAS, Florida's need for electrical generation is expected to grow by approximately 58 percent between 2002 and 2020; and

WHEREAS, Florida uses 8.6 billion gallons of gasoline per year, and consumption is growing by 300 million gallons per year; and

WHEREAS, less than one percent of Floridians own automobiles that use alternative fuels; and

WHEREAS, Florida has one of the nation's fastest growing populations with an average of 980 new residents arriving per day and approximately 80 million visitors arriving per year, thereby increasing the demand on Florida's energy supply; and

WHEREAS, current trends indicate Florida's dependence on natural gas to generate electricity will continue to increase; and

WHEREAS, Florida annually produces less than one percent of crude oil production and depends almost exclusively on other states and countries for supplies of oil; and

WHEREAS, catastrophic hurricane seasons in 2004 and 2005 have underscored Florida's vulnerability to disruptions in energy supply and the resulting impacts to Florida's economy, environment, and quality of life; and

WHEREAS, a long-term commitment to energy conservation in conjunction with an adequate, reliable, diverse, efficient, and affordable energy supply is vital to Florida's population growth, economic expansion and security; and

WHEREAS, the Governor's Office and the Governor's executive agencies are leading Florida's conservation efforts by adopting multi-phased, event based, cost-effective, efficient practices, which include, but are not limited to, replacing some traditional motor vehicles with hybrid vehicles, eliminating the use of non-essential equipment and appliances, turning off all lights, computers, and office equipment while not in use, and adjusting thermostats in state buildings;

NOW, THEREFORE, I, JEB BUSH, Governor of Florida, by virtue of the authority vested in me by the constitution and laws of the State of Florida, do hereby promulgate the following executive order:

Section 1. Energy Conservation.

The Governor's Office and the Governor's executive agencies are hereby directed to continue their energy conservation efforts to reduce the demand for energy in Florida and are further encouraged to develop innovative conservation initiatives to serve as a model for all Floridians. In addition, all other departments and agencies of state government, as well as all local governments, are hereby encouraged to develop and implement long-term conservation initiatives. For example, state and local governments should invest in energy efficient equipment and hybrid electric or alternative fuel vehicles.

Section 2. Energy Supply.

The State of Florida, through the Secretary of the Department of Environmental Protection shall develop a comprehensive energy plan by evaluating Florida's current and future energy supply and demand. This evaluation shall include an analysis of the following sectors: utility providers; petroleum companies; automobile manufacturers; fuel suppliers; technology companies; environmental organizations; researchers; the United States Department of Energy; members of the Florida Public Service Commission; members of the Florida Energy 2020 Study Commission; and consumers.

To assist with developing the State's energy plan, the Secretary of the Department of Environmental Protection shall host the 2005 Florida Energy Forum before December 31, 2005, in Tallahassee, Florida and serve as chairperson for the Forum. Forum participants shall address the diversification of Florida's energy supplies, energy generation, transmission, distribution, conservation and energy security, as well as discuss the barriers presented by government and potential incentives that may be offered to help Florida's future energy needs.

The State's energy plan shall consider all relevant topics, including, but not limited to the following:

- A. Florida's current and projected energy needs.
- B. A review of Florida's efforts to meet its current energy needs, including, but not limited to, laws, regulations, executive orders, Florida's Building Code, alternative energy investments through the Office of Tourism Trade and Economic Development, Florida's Energy 2020 Study Commission, Florida's Energy Future: Opportunities for Our Economy, Environment and Security Report, and conservation plans implemented by the state.

C. Florida's ability to generate, transmit and distribute electric power.

- 1. Florida's current and projected electric generating capacity for natural gas, liquefied natural gas, oil, coal, nuclear power, alternative and renewable energy sources (hydrogen, solar, biomass, wind and landfill methane), and other emerging energy technologies.
- 2. Florida's current and projected infrastructure needs for the production and supply of natural gas, liquefied natural gas, oil, coal, nuclear power, alternative and renewable energy sources (hydrogen, solar, biomass, wind and landfill methane), and other emerging energy technologies.
- 3. Florida's current and projected consumer costs of natural gas, liquefied natural gas, coal, nuclear power, alternative and renewable energy sources (hydrogen, solar, biomass, wind and landfill methane), and other emerging energy technologies.
- 4. Current regulatory oversight, both state and federal, of natural gas, liquefied natural gas, coal, nuclear power, alternative and renewable energy sources (hydrogen, solar, biomass, wind and

landfill methane), and other emerging energy technologies.

- 5. A review of Florida's successes in achieving energy efficiency.
- 6. Goals, both public and private, for the diversification of Florida's electric power supply.
- D. Florida's ability to generate, store and distribute fuel.
 - 1. Florida's current and projected capacity for gasoline, diesel fuel, ethanol, biodiesel, hydrogen and natural gas.
 - 2. Florida's current and projected infrastructure needs for the production and supply of gasoline, diesel fuel, ethanol, biodiesel, hydrogen and natural gas.
 - 3. Florida's current and projected consumer costs of gasoline, diesel fuel, ethanol, biodiesel, hydrogen and natural gas.
 - 4. Current regulatory oversight, both state and federal, of gasoline, diesel fuel, ethanol, biodiesel, hydrogen and natural gas.
 - 5. A review of Florida's successes in achieving energy efficiency.
 - 6. Goals, both public and private, for the diversification of Florida's fuel supply.
- E. Traditional and alternative fuel vehicles, consumer access to alternative fuels, the current and projected costs to consumers for traditional and alternative fuels, and the current and projected infrastructure needs for the production and supply of alternative fuel vehicles and the relative costs and benefits of any said alternatives.
- F. Methods by which Florida can protect its energy supplies during an emergency.
- G. Methods by which the State can reduce barriers and provide incentives to increase energy efficiency in power and fuel consumption.
- At the conclusion of the 2005 Florida Energy Forum, the Governor's Office and the Department of Environmental Protection will issue an updated energy strategy by no later than January 17, 2006.
- H. All agencies under the control of the Governor are directed, and all other agencies are requested, to render assistance and cooperation to the 2005 Florida Energy Forum.
- J. The Department of Environmental Protection shall provide all funds and administrative support necessary to implement the provisions of this Executive Order.

IN TESTIMONY WHEREOF, I have hereunto set my hand and have caused the Great Seal of the State of Florida to be affixed at Tallahassee, The Capitol, this 10th day of November, 2005.

. Jeb Bush

Executive Summary



I. Executive Summary

Florida's economy and quality of life depends on a secure, adequate and reliable supply of energy. As the fourth most populous state, Florida ranks third nationally in total energy consumption.

With more than 17 million citizens and nearly 1,000 new residents arriving daily, Florida is one of the fastest growing states in the nation. Because of its expanding economy, current forecasts indicate that Florida's electricity consumption will increase by close to 30 percent over the next ten years. But electricity demand presents just one challenge. Next to buildings, transportation is Florida's second largest energy use sector. The demand for motor vehicle transportation fuel currently tops 28 million gallons per day and is expected to increase to more than 32.3 million gallons per day during the next decade, assuming a 15 percent population increase.

Since the last review of Florida's energy policy in 2000, several unpredictable events have heightened concern over energy reliability, security and supply. The 2003 blackout in the northeast, along with tremendous back-to-back hurricane seasons in 2004 and 2005, demonstrated the impact power outages and fuel interruptions have on the nation's economic welfare.

Producing less than one percent of the energy it consumes and limited by its geography, Florida is more susceptible to interruptions in energy supply than any other state. Unlike other states that rely on petroleum pipelines for fuel delivery, more than 98 percent of Florida's transportation fuel arrives by sea. The state's reliance on imported petroleum products, in addition to its anticipated growth in consumption, underscores its vulnerability to fluctuations in the market and interruptions in fuel production, supply and delivery.

The need to pursue a comprehensive energy plan for Florida cannot be understated. Energy demand is increasing at a significant rate. In turn, such increases have the potential to contribute to a continued limited market and sustained higher oil and natural gas prices.

ENERGY PRODUCTION AND A GROWING ECONOMY

Florida depends almost exclusively on other states and nations for supplies of oil and gasoline, generating less than one percent of the nation's crude oil production annually.

To generate electricity, Florida primarily relies on natural gas, coal and oil imports.

Together, fossil fuels represent 86 percent of Florida's total generating capacity. Less than 10 percent of its generating capacity is derived from cleaner nuclear and renewable fuels. In fact, no new nuclear plants have entered service in Florida since 1983.

Current forecasts indicate that new generation capacity will be 80 percent natural gas-fired and 19 percent coal-fired. Meeting these projections could prove expensive at today's prices and lead to an over-reliance on one fuel type, affecting the reliability of electric utility generation supply in Florida. While expansions for natural gas capacity are needed and already underway, improving generation fuel diversity would enhance reliability over the long-term. Too great a reliance on a single fuel source leaves Floridians subject to the risks of price volatility and supply interruption.

A New Class of Energy

Although the nation's reliance on traditional fossil fuels is currently high, Florida is investing in alternative fuels and developing "next generation" energy technologies. In 2003, Governor Jeb Bush launched "H2 Florida" to accelerate the commercialization of hydrogen technologies and spur economic investment in Florida's economy. With a four to one return on investment, Florida and its federal partners have invested \$9 million to date in hydrogen infrastructure. Construction of a "hydrogen highway" is underway, 28 hydrogen demonstration projects are in progress and more than 100 hydrogen research and development projects are taking place at Florida's universities.

Utilization of biofuels is in its infancy with the cost of renewable fuels relatively high compared with traditional hydrocarbon fuels. Currently, Florida has just one biodiesel facility and, absent a manufacturing plant, imports ethanol from refineries outside of the state. Increasing production, supply and infrastructure of biofuels through financial incentives would provide both economic and environmental returns for the state. Likewise, a stronger investment both residentially and commercially in solar technology would not only reduce utility costs but generate pollution-free power for Floridians. To date, solar technology has remained largely inefficient and expensive, however, costs are gradually decreasing as system quality and reliability increases. To encourage continued investment in solar energy, systems received a permanent exemption from Florida sales and use tax in 2005.

STORM IMPACTS AND HURRICANE RECOVERY

The unprecedented level of storm activity during the 2004 and 2005 hurricane seasons spotlighted Florida's vulnerability to energy supply disruptions both in terms of power generation and transportation fuel supply. The impacts of the 2005 summer storms were particularly widespread. During the peak of the hurricane season, Panhandle ports remained closed to ocean traffic for several days, leaving residents and businesses without fuel. As Hurricane Rita followed Hurricane Katrina through the Gulf of Mexico, the Florida Reliability Coordinating Council declared a generating Capacity Advisory, appealing for conservation to prevent potential brownouts. Across the nation, production shortages led to a jump in natural gas prices and an increase in oil prices; gasoline prices rose to more than \$3 per gallon and winter heating expenses were expected to increase by up to 35 percent over the previous year.

The Gulf of Mexico is of critical importance to the nation, supplying 29 percent of the nation's domestic oil production and 19 percent of the domestic gas production. When Hurricane Katrina hit on August 30, 2005, 95 percent of daily oil production and 88 percent of daily gas production was shut down. The storm left offshore production facilities, coastal refineries, pipelines and other energy infrastructure severely damaged. By January 2006, nearly five months after the storm, more than a quarter of the oil production and 19 percent of natural gas production from federal leases in the Gulf

8

of Mexico remained shut down. Approximately 100 oil and gas platforms in federal waters remain out of service today. The United States Mineral Management Service estimates that cumulative production losses from Hurricanes Katrina and Rita so far amount to 547 million barrels of oil, equivalent to more than 20 percent of yearly oil production in the Gulf of Mexico. Full recovery of oil and natural gas production along the Gulf Coast is not expected until the summer of 2006, keeping pressure on energy prices.

A VISION FOR FLORIDA'S FUTURE

Florida's energy policy was last reviewed by the Florida Energy 2020 Study Commission in 2000. The Commission addressed extended forecasts of energy supply and demand, reliability of the electric and natural gas supply within the state, emerging electric generation technologies, the potential impact of electrical restructuring and deregulation, and the environmental impact of electricity supply production, generation, and transmission in the state. Since then, considerable advances have been made in "next generation" energy technologies. In addition, Florida's experiences during the past two hurricane seasons have highlighted the critical need to reduce the state's susceptibility to energy disruptions.

To begin laying the foundation for Florida's energy future, Governor Bush signed Executive Order #05-241 on November 10, 2005, directing State government, through the Secretary of the Department of Environmental Protection (DEP), to develop recommendations for a new comprehensive statewide energy plan. The Order also directed State executive agencies to continue their energy conservation efforts and encouraged the development of additional innovative conservation initiatives to serve as a model for all Floridians.

To assist with developing new energy policy, the Order directed the DEP to convene a public energy forum. On December 14, 2005, DEP Secretary Colleen Castille hosted the 2005 Florida Energy Forum in Tallahassee, inviting representatives from Florida's Public Service Commission, the U.S. Department of Energy, utilities, fuel companies, alternative and renewable energy experts, businesses, the Florida Legislature and the environmental community to provide input and guidance on achieving energy security.

Forum participants examined a breadth of issues included in the charge from the Order, addressing the diversification of Florida's energy supplies, energy generation, transmission, distribution, conservation and efficiency, along with existing regulatory barriers and potential incentives for achieving energy security.

As a result, this report contains recommendations for achieving a diverse and reliable energy future that is built on the underlying principles of conservation and efficiency:

- The state's energy should be derived from a wide mix of available fuel sources and technologies, incorporating home-grown energy from renewable resources and alternative fuels and avoiding reliance on any one fuel type.
- The state's energy supply should be reliable, with redundancy in fuel delivery mechanisms and expanded capacity.
- Government, businesses and consumers should maximize the use of energy consumed,

employing sustainable practices, eliminating waste and adopting conservation measures.

- Government, businesses and consumers should utilize energy efficient products and practices that use less energy, conserve resources and reduce utility costs.
- Florida should support its energy strategy by leveraging the incentives provided in the 2005 Federal Energy Policy Act

To prevent duplication where other official entities are acting, this report touches briefly on improving electric infrastructure. Following damage to electric transmission service lines and subsequent extended power outages during the 2005 hurricane season, the Public Service Commission is hosting an Electric Infrastructure workshop with electric utilities, cooperatives and local governments on January 23, 2006 to focus on actions needed for minimizing future infrastructure damage.

The recommendations outlined in this report describe administrative actions for immediate implementation, proposals for legislative action during the 2006 Regular Legislative Session and policy improvements that will enhance electric power generation and transportation fuel supply to help provide energy stability over the long-term.
Recommendations for Electricity Generation

- Streamline and expedite the siting and permitting of generation resources by revising the provisions of the Florida Electrical Power Plant Siting Act.
- Streamline and expedite the siting and permitting of electric transmission and distribution resources by revising the provisions of the Florida Electrical Transmission Line Siting Act. Incorporate the siting of substations into the Transmission Line Siting Act.
- Promote fuel diversity, fuel supply reliability and energy security.
- Facilitate additional fuel delivery mechanisms in Florida for power generation. Expedite all State permits required for the redundancies and increased capacity.
- Adopt updated interconnection standards to include all distributed generation technologies.
- Establish an energy council to provide policy advice and counsel to the Governor, Speaker of the House and Senate President.
- Expedite State performance contracting with Energy Service Companies.
- Promote awareness of energy conservation and alternative energy technologies.
- Use discretionary enforcement authority to allow approved alternative energy projects that provide a greater public benefit in lieu of civil monetary penalties.
- Require all new State government building construction to meet the U.S. Green Building Council's Leadership in Environmental Design standards. Encourage local governments and community developers to adopt high performance green building practices.
- Provide grant funding for research and demonstration projects associated with the development and implementation of renewable energy systems. Expand solar, hydrogen, biomass, wind, ocean current and other emerging technologies.
- Identify alternative energy production and distribution industries as Qualified Target Industries.
- Provide consumer and commercial rebates to assist with initial cost of photovoltaic and solar thermal technology installations on residential and commercial buildings.
- Provide consumer rebates for purchases of energy efficient ENERGY STAR™ appliances.
- Provide sales and corporate tax incentives for the manufacture, purchase and use of fuel cells for supplemental and backup power.

11

Recommendations for Transportation Fuels

- Facilitate additional and diverse petroleum supply and distribution mechanisms into and within Florida. Expedite all State permits required for the redundancies and increased capacity.
- Encourage fueling stations to cooperatively adopt a system modeled after the Florida WARN System to facilitate the relocation and use of generators to reestablish service.
- Foster state-local partnerships to encourage well-designed transportation and transit systems between established communities and within new community development.
- Raise public awareness for alternative fuel vehicles through public programs. Encourage public entities, including school districts and local governments, to use biofuels in fleets.
- Provide grant funding for applied research and demonstration projects associated with the development and implementation of alternative fuel vehicles and other emerging technologies.
- Provide sales and corporate income tax credits for hydrogen vehicles and fueling infrastructure.
- Provide corporate sales and income tax incentives to improve production, develop distribution infrastructure and increase availability of clean fuels, including biodiesel and ethanol.

Florida's Ability to Generate, Transmit and Distribute Electric Power



II. Florida's Ability to Generate, Transmit and Distribute Electric Power

A. CURRENT AND PROJECTED ELECTRIC GENERATING SUPPLY AND DEMAND

According to a 2001 study by the United States Energy Information Administration, Florida ranks third nationally in total energy consumption. Florida's demand for electric generation is expected to grow by approximately 58 percent between 2002 and 2020.

Florida currently has 57 electric utilities, consisting of five investor-owned (IOU), 18 cooperatively owned, and 34 municipally owned utilities. The largest investor owned utilities include Florida Power & Light Company (FPL), Progress Energy Florida, Tampa Electric Company (TECO), and Gulf Power Company (Gulf). The largest cooperatively owned utility is Seminole Electric Cooperative, and the largest municipally owned utilities are Jacksonville Electric Authority, Orlando Utilities Commission, City of Tallahassee, City of Lakeland, and Gainesville Regional Utilities.

Based on 2003 data, 51 percent of all electric energy was consumed by residential customers, 32 percent by commercial customers, and 12 percent by industrial customers, with approximately two percent being used for other purposes. The following graphic displays the actual megawatt hours consumed by each customer class in 2003.

As illustrated above, the load profile of Florida is heavily influenced by the residential customer class. Due to high residential consumption, Florida must have adequate energy generation capacity to satisfy the changing needs of consumers throughout the day.





A representation of the typical summer and winter peak load profiles is shown below.¹

It is Florida's fast growing population combined with the unique operational constraints of satisfying these peak summer and winter load profiles which dictates the need for continued growth in electric generation capacity.

Currently, Florida's electric utilities have 51,377 megawatts² (winter ratings³) of installed capacity⁴ to meet the needs of customers. To meet Florida's growing demand for electricity, an acceleration of power plant construction is occurring. Over the next ten years, Florida's electric utility industry is constructing or planning to add approximately 19,390 megawatts (winter ratings) of new generating capacity. Future electric needs will also continue to be supplemented by utility demandside management, energy efficiency programs and electricity production from renewable resources. Electricity consumption statewide, currently at 241,514 gigawatt-hours⁵, is also expected to significantly increase during the ten-year planning horizon. Current forecasts indicate that electricity consumption will increase more than 67,276 gigawatt-hours (27.9 percent) over the next ten years.

To meet the changing load on Florida's electric systems, utilities must construct and operate various types of units. Baseload generation, primarily made up of large coal-fired and nuclear units, meets the load that is continuously on the system. These units have high capital costs but lower fuel costs associated with their operation. As the load rises during the day, intermediate units, primarily oil and natural gas-fired, are brought on line and then ramped down as the load decreases at night. These units may run for between 50 and 70 percent of the day and generally have lower capital costs to construct, but higher fuel costs. To meet peak demand combustion turbine units, primarily oil and natural gas-fired, are cycled on for shorter periods. These units may only run between five and 20 percent of the day depending upon weather conditions. Combustion turbine units generally

¹ Florida Energy 2020 Study Commission Report, 2001

² A megawatt (MW = 1.000 kilowatts) is a measure of real power at any instant in time, that is, a measure of demand on the grid at any moment in time. A typical home might have a demand that ranges from 1 to 10 kilowatts or more depending on what electrical appliances were being used.

³ Electric generating facilities have different megawatt ratings for the summer and winter seasons as cooler temperatures typically allow for higher output.

⁴ Includes existing generation and purchases from outside the state.

⁵ A gigawatt-hour ((GWH) = 1000 megawatt hours (MWH), and a MWH = 1000 kilowatt hours (KWH)) is a measure of the kilowatt demand aggregated over some time interval and represents the amount of electric energy consumed. A typical Florida residential customer will consume about 1,150 KWH per month.

have the lowest capital costs but the highest fuel costs due to lower efficiencies in converting fuel to electricity.

Electric utilities regularly conduct short and long-term planning exercises to ensure sufficient resources are in place to meet increasing demand. These studies determine the type of generation that is needed on the system. Baseload generation would generally take seven to ten years to receive regulatory approval and construction. Intermediate units, specifically combined cycle units, require approximately five years for approval and construction. Therefore, the ability of a utility in the short term to modify its overall fuel requirements is severely limited due to the time required to bring new generation online.

Presently, Florida's electric generating capacity is based on a variety of fuels: natural gas represents 39 percent of installed capacity, coal represents 23 percent, and oil represents 24 percent. Cumulatively, these petroleum based fuels represent approximately 86 percent of Florida's total generating capacity. Additionally, Florida's installed generating capacity includes nuclear generation (eight percent), Non Utility Generators (four percent), and renewables (one percent). In the future, new generation capacity additions are forecasted to be primarily 80 percent natural gas-fired and 19 percent coal-fired.



The charts above show the current installed generating capacity by primary fuel source and the projected new generation capacity additions for 2005-2014. One area of concern, which these charts highlight, is the projected growth of natural gas as the primary fuel source. While these projected new generating capacity additions are typically selected for construction based on the most cost-effective combination of capital and operating (fuel) costs, maintaining generation fuel diversity is quickly becoming a concern as well.

The following section specifically examines the expected trends in installed capacity and total fuel consumption for each major source of electrical power generation: natural gas, oil, coal, nuclear, interchange purchases, purchases from Non-Utility Generators (NUGs), and renewable energy sources.

Natural Gas

Florida's utilities continue to project a substantial increase in natural gas-fired generation. Natural gas-fired generation, currently at 29.9 percent of total statewide energy consumption, is expected to increase to 44.4 percent over the next ten years. Of the approximately 19,100 megawatts in gross capacity additions⁶ projected in the state over the 2014 planning horizon, nearly 15,300 megawatts is anticipated to come from gas-fired capacity in the form of new combined cycle and combustion turbine (CT) units. Natural gas consumption forecasts do not include usage from proposed new Independent Power Producer (IPP) generating units.

Oil

Florida ranks first among all states in the amount of electricity it produces from oil. Forecasts indicate that oil-fired energy will decrease from 12.2 percent to 7.0 percent of total statewide energy



production over the next ten years. Oil-fired generation decreased substantially during the 1980s in response to rising oil prices in the 1970s. Many utilities, however, still use oil in peaking combustion turbine units as a primary and secondary fuel. Due to escalating natural gas prices, utilities have recently burned oil more frequently in baseload and intermediate combined cycle units for economic



16

6 Gross additions exclude capacity decreases due to unit derating, retirement, or terminating contracts.

reasons. In addition, recent experience with Hurricanes Katrina and Rita illustrate the need for electric utilities to have oil available as a backup fuel when interruptions to natural gas supplies occur.

Coal

Coal generation increased substantially during the 1980s in response to the oil price increases of the 1970s. Coal plants have traditionally been operated based on low forecasts of coal prices relative to oil or natural gas. However, coal plants are capital-intensive. Stricter environmental regulations may lead to increased capital investments at coal plants. In 2004, the state's utilities forecasted increases in coal-fired capacity of approximately 1,100 megawatts from the previous year's forecast of new coal capacity to be added. The 2005 forecast shows an expected 3,786 megawatts of new coal capacity over the next ten years.

Nuclear

Nuclear generation increased considerably during the 1970s and early 1980s. Nuclear plants were



built based on low fuel price forecasts relative to oil or natural gas. However, nuclear plants are capital-intensive, take as much as ten years to certify and build and present concerns surrounding the storage and disposal of spent fuel rods. No new nuclear plants have entered service in Florida since 1983. While no utility's Ten-Year Site Plan contains proposed nuclear units, Progress Energy Florida recently announced its intention to pursue a new nuclear generating unit in Florida within the next ten years. Any nuclear additions would be operational past the 2014 planning horizon.

Interchange Purchases

Florida's utilities continue to rely on capacity and energy purchases from out-of-state utilities. Interchange purchases are typically short-term purchases of excess capacity and energy between utilities. Florida can safely import around 3,600 megawatts over the Southern Company-Florida interconnection. Approximately 2,500 megawatts of the interface is currently reserved for firm sales and for delivery of capacity from generating units owned by Florida utilities located in Southern Company's region. Approximately 1,100 megawatts remains available for non-firm, economy transactions.

Purchases from Non-Utility Generators

Non-utility generators (NUGs) build and operate power plants to satisfy contractual requirements with retail-serving electric utilities. NUGs sell firm capacity to many of Florida's utilities under long-term and short-term purchased power contracts. Forecasts indicate that the amount of NUG electricity purchased by Florida's utilities will decrease from three percent to 1.5 percent of statewide energy consumption during the planning horizon. The forecasted decrease is due to the expiration of 377 megawatts of firm purchase contracts with qualifying facilities and 339 megawatts of firm purchase contracts. However, once the current contracts expire, these generators will remain in place in Florida and should remain available to provide capacity and energy under new purchased power contracts with utilities.

Renewables

The renewable resources presently in the state are derived from a small amount of hydroelectric, landfill gas, biomass and waste-to-energy sources. Electric utilities and NUGs produce renewable energy in Florida. Non-utility producers of renewable energy use some of their output onsite, selling the remainder to electric utilities either under firm contracts or on an as-available basis.

Hydroelectric units at two sites in northwest Florida, one utility-owned and one operated by the Federal government, supply approximately 50 megawatts of renewable capacity. Hydroelectric generation accounts for less than 0.1 percent of Florida's generation mix. There are no planned new units due to the absence of a feasible location, as Florida's flat terrain does not lend itself to hydroelectric power.

Landfill gas provides a combined 2.5 megawatts of capacity to Gainesville Regional Utility (GRU) and Jacksonville Electric Authority (JEA). When factoring in direct use and co-generation applications, landfill gas provides nearly 40 megawatts of power statewide. Further development of landfill gas energy resources could result in an additional 68 megawatts.⁷

Florida's utilities purchase 506 megawatts of non-utility generator capacity fired by municipal solid waste, wood and wood waste, and waste heat. The scheduled expiration of contracts during the planning horizon will reduce the amount of firm renewable capacity to 167 megawatts by 2014, a decrease of 339 megawatts.

Wind turbine efficiencies have increased and costs have dropped due to research and development advances and manufacturing improvements. A recent study funded by the U.S. Department of Energy's Wind Powering America program concluded that Florida's onshore wind resources traditionally considered "marginal to good" could now be "fair to excellent." In addition, utility scale wind power generation appears to be economically viable at certain offshore and at direct coastal sites within view of the Atlantic Ocean and Gulf of Mexico.

The study further suggests that large utility-scale wind power generation is unlikely to be economically viable at inland sites more than a few hundred yards from the coastline anywhere within Florida today, with the possible exception of outer Cape Canaveral, the Panama City Beach region and the lower Florida Keys. However, there are several locations where small, distributed wind power appears feasible. Steep hills that reach 100 to 300 feet above the surrounding terrain

18

⁷ USEPA Landfill Methane Outreach Program

near electric loads and distribution lines were identified in the highlands northwest of Orlando and northeast of Tampa. In addition, coastal sites in view of open Ocean or Gulf waters have good wind resources. Mounting small wind turbines on existing cell phone towers could avoid the cost of tower construction – about half the cost of a small-wind project. To improve the economics of Florida wind energy, turbine blade design can be engineered and optimized to better match Florida's low-speed wind resources.⁸

Solar technology has been in use for years with a gradual decrease in cost and an increase in system quality and reliability. Most are building owner purchased systems and the majority of those are solar thermal (water heating) systems. Solar swimming pool heaters are the leading solar product installed in Florida.

Additionally, utility-based solar programs exist. The City of Lakeland Electric and Water Department has a solar thermal program where the utility installs and owns the solar water heater and sells the produced hot water to the customer. This program recently passed a rate impact measures test (RIM test) conducted by a consulting firm hired by the utility to evaluate conservation and demand-side management measures.

Under both state and federal law, any generator who uses renewable resources is allowed to sell any or all of its electric output to a utility for "full avoided cost." Full avoided cost is defined as the cost of the next increment of power that the utility would have incurred if it had produced the power. Thus, as electric production cost increases, renewable generators have an economic incentive to sell to the grid.

[&]quot;Florida Wind Initiative", USDOE Wind Powering America, 2005.

Proposed New Generation Capacity⁹

Courses and a second	Muite			Thoma Bullard	itas (Önethese
		Planet val	25	l or u <u>u</u> us L'unmad	
Hillsborough Co	Hills Co. Resource Recovery Facility	17	Garbage	Resource Recovery Facility	N/A
Florida Power & Light	SW St. Lucie Coal – 2 Units	1700	Coal	Conventional	2012, 2014
Southern Co.	Demonstration Project at Stanton	285	Coal	Integrated Gasification Combined Cycle	2010
Seminole Electric	Unit 3 at Palatka	750	Coal	Pulverized/ Conventional	2012
JEA/FMPA	Coal Project	800	Coal	Conventional	2112
Gainesville Regional Util.	Deerhaven expansion	220	Coal	Cola Fluidized Bed/Biomass/ Other	2010
Progress Energy	Hines Unit 5	540	Gas	Combined Cycle	2009
Seminole Electric	Unknown – 2 Units	364	Gas	CC	2008, 2009
Pasco Co	Pasco Co. Resource Recovery Facility	20	Garbage	RRF	N/A
Palm Beach Co	Palm Beach Co Resource Recovery Facility	28	Garbage	RRF	2010
JEA	Circulating Fluidized Bed	250	Coal	CFB	2013
Progress Energy	Hines Unit 6	540	Gas	CC	2010
Progress Energy	Central Florida Nuclear	N/A	Nuclear	Nuclear	2015
Progress Energy	Unknown CC	536	Gas	CC	2012
Tampa Electric Co	Undetermined	502	Gas	CC	2013
Seminole Electric	Unknown – 3 Units	546	Gas	CC	2013, 2014
Progress Energy	Unknown CCs – 2 Units	1,072	Gas	СС	2013, 2014
JEA/Biomass Industries Group	Unknown – 2 Units	240	E-grass	Biomass	N/A

20

⁹ Projected electrical capacity under the Power Plant Siting Act

B. CURRENT AND PROJECTED INFRASTRUCTURE NEEDS FOR PRODUCTION AND SUPPLY

Florida has limited native fuel resources. Energy sources native to Florida include solar, biomass and hydroelectric power in northwest Florida. All other fuels used by Florida's utilities are fossil (natural gas, oil and coal) or nuclear, which must be brought into the state by various transport systems. Coal is delivered by rail or barge, oil is delivered by tanker and nuclear fuel is delivered by rail and truck.

Natural Gas Pipeline Adequacy

Florida currently relies primarily on two major gas pipeline companies, Florida Gas Transmission (FGT) and Gulfstream Natural Gas (Gulfstream), to supply natural gas to electric utilities, industrial customers, local distribution companies and two smaller pipelines serving customers in the Panhandle (Southern Natural Gas and Gulf South). Florida Gas Transmission currently has a system pipeline capacity of 2.2 billion cubic feet per day (BCF/day), while Gulfstream has a system pipeline capacity of 1.1 BCF/day. More than 85 percent of the state's natural gas consumption is for electricity generation by utilities and non-utility generators.



Natural Gas Consumption by End-User – 2004

Forecasts indicate that electric utility generation will cause a 92 percent increase in natural gas requirements over the next ten years. Increased dependency on natural gas could affect the reliability of electric utility generation supply in Florida. The primary threat to reliability is the possibility of natural gas supply disruption. The Florida Reliability Coordinating Council (FRCC) has formed a Gas/Electricity Interdependency Task Force to determine reliability impacts and to recommend mitigating measures should reliability risks arise. The North American Electric Reliability Council (NERC) has established a Gas/Electricity Interdependency Task Force whose scope is almost identical to that of the FRCC task force. The NERC task force completed a study in May 2004, concluding in part that gas pipeline reliability can substantially affect electric generation, and that electric system reliability can have an impact on gas pipeline operations. The FRCC continues to review the recommendations made by the NERC task force to determine where to focus future analyses. The FRCC task force stated that the region has adequate pipeline capacity for reliability purposes for both current and future natural gas demand. However, the FRCC task force's conclusion assumes that the generating units that have the capability to burn oil will do so at times of peak demand. Therefore, economics may be the driving factor for any future expansion of gas pipelines.

21

Based on the forecasted requirements of electric utilities and other sectors, the Public Service Commission (PSC) estimates that total pipeline demand will require an average of 3.15 BCF/day by 2014. Given the current pipeline capacity of FGT and Gulfstream, sufficient capacity currently exists to serve forecasted 2014 requirements. The estimate could be understated because it relies on average daily demand rather than on maximum delivery levels specified in gas transportation contracts.

Because the 2014 forecasted pipeline capacity requirement might be underestimated based on average demand, the PSC also conducted a forecast based on the projected peak demand for gas capacity. Based on this methodology, it is estimated that by 2014, incremental pipeline capacity requirements could increase up to 1.34 BCF/day.

Florida Gas Transmission

FGT operates 5,000 miles of pipeline nationwide, 3,300 miles of which are in Florida. Six expansions have occurred since its inception in 1959, which have increased the pipeline's capacity from its original 0.278 BCF/day to its current 2.2 BCF/day. In October 2005, FGT filed an application with the Federal Energy Regulatory Commission (FERC) seeking authority to construct its Phase VII Expansion Project. This project involves the construction of 33 miles of 36-inch diameter pipeline looping and the installation of 9,800 horsepower of compression. The expansion will provide approximately 0.16 BCF/day of additional capacity to transport natural gas from a connection with Southern Natural Gas Company's proposed Cypress Pipeline project.

Gulfstream

Gulfstream placed Phase I of its two-phase natural gas transmission system into service in 2002. Phase I, with a capacity of 1.1 BCF/day, crosses the Gulf of Mexico between Pascagoula, Mississippi and Manatee County, Florida with more than 430 miles of 36-inch pipe. In February 2005, Phase II, a 110-mile extension to Florida's east coast, entered service to serve Florida Power and Light's gasfired generating units at the Martin and Manatee plant sites.

Cypress

Southern Natural Gas has proposed expanding its existing interstate natural gas pipeline system between Port Wentworth, Georgia and an interconnection with FGT's system near Jacksonville, Florida. Project construction will occur in three phases. Phase I includes the initial pipeline of 165 miles of 24-inch diameter pipe. Phase II and III will consist of additional compression and looping. The source of natural gas will be Southern Natural Gas's Elba Island liquefied natural gas (LNG) facility near Savannah, Georgia. The Cypress pipeline will have the ability to flow gas between Florida and Georgia in both directions. In addition, the pipeline will provide Florida with a new, geographically diverse source of gas that will help mitigate supply disruptions caused by natural disasters such as hurricanes. In 2005, the PSC approved Progress Energy Florida's long-term contract to purchase gas supply on the Cypress pipeline. Phase I of the pipeline is expected to be in service in 2007, with Phase II and III becoming operational in 2009 and 2010, respectively.

Bahamas LNG Projects

Two consortia have proposed pipeline projects to transport LNG from the Bahamas to Florida. Two of the projects, the Tractabel Calypso Project and the AES Ocean Express Project, have received FERC approval, granting both projects a Presidential Permit to construct. The third project, the El Paso Seafarer Pipeline System, filed its certificate application with the FERC in November 2004, and the application is pending. These projects must first receive approval from the Bahamian government before construction can begin. The applicants have not yet contracted with an anchor utility customer to purchase the gas.

C. CURRENT AND PROJECTED CONSUMER COSTS

Electricity rates include base rates that include costs for electric generating capacity, transmission and distribution facilities, operation and maintenance expenses and administrative expenses such as billing. Rates are a function of the type of generating units on a utility's system, the costs to operate those units, including fuel, and the cost to transmit and distribute power to consumers.

Company Contraction Collins		i Foutts≞‼(sin	South Plage is	D-057 ige
	i Braiter Brizan Efficient venant	nachaitail bhin Seacath 2006		
	Florida Power and Light Company	Progress Energy Florida, Inc.	Tampa Electric Company	Gulf Power Company
Base Rate	\$38.12	\$41.18	\$51.92	\$49.30
Fuel Cost Recovery	\$58.41	\$49.79	\$54.35	\$30.92
Energy Conservation Cost Recovery	\$1.42	\$0.76	\$1.69	\$0.88
Environmental Cost Recovery	\$0.26	-\$3.72	\$0.62	\$3.64
Capacity Cost Recovery	\$6.03	\$3.56	\$9.93	\$2.72
Storm Damage Cost Surcharge	\$1.65	n/a	\$3.61	\$2.71
Gross Receipts Tax	\$2.72	\$2.74	\$2.74	\$2.31
Total Monthly Bill	\$108.61	\$109.56	\$109.61	\$92.48

Fuel costs incurred by the electric utilities to generate needed power are directly passed through to customers. The cost of fuel does not have any "markup" or generate earnings for the utilities.





The cost of renewable fuels remains relatively high compared with traditional hydrocarbon fuels, but advancements in renewable technologies have made some of these fuels more cost-competitive. The following chart compiled from several sources provides the estimated consumer cost per kilowatt hour of select renewable fuels.

Construction Construction	niner Cosis (S/KW/h) ¹⁰ - Alexandra (S/KW/h)
Renewable Energy Source	Current Price
Hydrogen	0.26
Solar Thermal (for water heating)	0.054
Solar Photovoltaic (PV)	0.18 - 0.32
Biomass	0.08 - 0.12
Wind	0.04 - 0.06
Landfill Methane	0.02 - 0.06
Waste-to-Energy (WTE)	0.04 - 0.15

D. CURRENT REGULATORY OVERSIGHT OF ELECTRIC UTILITIES

Florida Public Service Commission (PSC)

The PSC has authority to ensure the provision of adequate, reliable, reasonable cost electricity to consumers. The PSC has specific authority under Chapter 366, Florida Statues, to regulate the rates and service of investor-owned electric utilities in the state. It also has authority to oversee the reliability of the electric grid, to determine the need for new electric generating facilities (Section 403.519, F.S.), to establish utility conservation goals (Sections 366.80-.82, F.S.) and oversight of the safety of electric facilities (Section 366.04, F.S.).

The PSC has authority to set the rates of natural gas utilities that provide service to end-use customers. Proposed intrastate natural gas pipelines must receive an affirmative determination of need from the PSC prior to construction (Section 403.9422, F.S.) Inspection of natural gas facilities to help ensure safe operations is also conducted by the PSC.

10 Hydrogen figure derived from actual operating data reported to the Florida Energy Office. Solar figures provided by the Florida Solar Energy Center. Wind figure provided by National Renewable Energy Laboratory. Remaining figures derived from the 2003 report titled "An Assessment of Renewable Electric Generating Technologies for Florida".

Department of Environmental Protection (DEP)

Power Plants

The Power Plant Siting Act is a centralized, coordinated licensing process encompassing the permitting and other authorizations, including proprietary interests, of all state, regional, and local agencies in the jurisdiction of which an electric power plant is proposed to be located. The Act provides for a single certification (license) for those electric power plants, as defined by the Act, which are steam or solar powered, 75 megawatts or greater and are constructed after October 1, 1973. Only electric utilities as defined by the Act may apply for certification. The process includes mandatory land use and certification hearings by an administrative law judge, a Determination of Need by the PSC, with ultimate approval/denial authority vested in the Siting Board (Governor & Cabinet). The DEP coordinates the process.

Transmission Lines

The Transmission Line Siting Act is a centralized, coordinated licensing process encompassing permitting and other authorizations, including proprietary interests, of all state, regional, and local agencies in the jurisdiction of which a transmission line is proposed to be located. The Act provides for a single certification (license) for transmission lines. Transmission lines subject to the Act are those that are 230 kilovolts or greater, 15 miles or more in length and cross a county line. The process includes a mandatory hearing by an administrative law judge, a Determination of Need by the PSC, with ultimate approval/denial authority vested in the Siting Board (Governor & Cabinet). The DEP coordinates the process.

Natural Gas

The Natural Gas Pipeline Siting Act is a centralized, coordinated licensing process encompassing permitting and other authorizations, including proprietary interests, of all state, regional, and local agencies in the jurisdiction of which a natural gas pipeline is proposed to be located. Natural gas pipelines subject to the Act are those that are 15 miles or more in length and cross a county line. There are exemptions for pipelines that are designated for local distribution only and for pipelines that have received a certificate of public convenience and necessity from FERC. Most pipelines do receive a certificate of public convenience and necessity. As a result, no pipeline has been certified yet under the Act. The process includes a mandatory hearing by an administrative law judge, a Determination of Need by the PSC, with ultimate approval/denial authority vested in the Siting Board (Governor and Cabinet). The DEP coordinates the process.

Environmental Impacts

The DEP administers resource protection and pollution control permitting processes for all federal environmental permits that are delegated to or approved by the State. Federally delegated or approved permit programs include air permits such as Title V and Prevention of Significant Deterioration permits, water permits including National Pollutant Discharge Elimination System permits and Underground Injection Control. The DEP also administers state environmental permits for facilities that are not subject to the siting acts. The DEP also regulates any impacts to State owned lands.

Other State Regulatory Authorities

Power generation associated with some sources of renewable power is regulated through a variety of delegated authorities. The following section is intended to provide highlights of regulatory oversight but may not be all-inclusive:

Solar

Regulatory standards are tied to the type of buildings that consider solar technology, and there are few restrictions on how the technology is utilized or requirements for its use. All solar equipment sold in Florida must be certified by the Florida Solar Energy Center (FSEC) pursuant to Section 377.705, F.S. Certifications are issued by FSEC and the Solar Rating and Certification Corporation. Contractors are trained and tested by the FSEC and must obtain licensure through the Florida Department of Business and Professional Regulation pursuant to Section 489.105, F.S. As established by the PSC, interconnection of small photovoltaic systems (up to ten kilowatts) is allowed.¹¹ The interconnection requests are processed by the utility company, who will accept all requests meeting the standards of the application process. Building codes for installation of solar systems are regulated by the Department of Community Affairs.

Biomass

Biomass covers a broad array of fuel sources and different permitting may be applicable depending on the source utilized. Federal regulatory standards for biomass are administered by the United States Environmental Protection Agency through the authority of the Biomass Research and Development Act of 2000. In Florida, the DEP regulates biomass energy under the authority of the Clean Air Act.

Waste-to-Energy

Waste-to-energy is regulated by the DEP, which also permits waste-to-energy facilities.

Wind Energy

Florida Statues regarding the Renewable Energy Access Laws require that ordinances, deed restrictions, covenants or similar binding agreements cannot prohibit solar equipment use (and other renewable energy technologies). As such, regulatory oversight for wind energy is largely a local government issue.

E. FLORIDA'S ENERGY EFFICIENCY AND ALTERNATIVE ENERGY INITIATIVES

Demand-side management (DSM) reduces customer peak demand and energy requirements, resulting in the deferral of need for new generating units. Utilities have offered DSM programs since 1980 under the requirements of the Florida Energy Efficiency and Conservation Act. The Act emphasizes reducing the growth rate of weather-sensitive peak demand, reducing and controlling the growth rate of electricity consumption and reducing the consumption of expensive resources such as petroleum fuels. To meet these objectives, the PSC sets numeric conservation goals and utilities develop and implement DSM programs to meet these goals.

Overall, Florida's utilities have been successful in meeting the overall objectives of the Act. Since 1980, utility conservation programs have reduced statewide summer peak demand by 4,951

26

¹¹ Section 25-6.065, Florida Administrative Code

megawatts, winter peak demand by 5,511 megawatts and annual energy consumption by 5,488 gigawatt-hours. By 2014, DSM programs are forecasted to reduce summer peak demand by 5,563 megawatts, winter peak demand by 6,068 megawatts, and annual energy consumption by 6,883 gigawatt-hours.

SHUC HURSHING SUPPLY AND A STRUGG			
DSM Savings	2005	By 2014	
Summer Peak Demand	4,951 MW	5,563 MW	
Winter Peak Demand	5,511 MW	6,068 MW	
Annual Energy Consumption	5,488 GWh	6,883 GWh	

Numeric Conservation Goals and DSM Plans

The Florida Energy Efficiency and Conservation Act requires that all investor-owned utilities and any municipal or cooperative utility with annual energy sales of at least 2,000 gigawatt-hours as of July 1, 1993 meet numeric conservation goals set by the PSC. Seven Florida utilities are subject to this requirement: Progress Energy – Florida, Florida Power & Light, Gulf Power, TECO, Florida Public Utilities Company, JEA, and Orlando Utilities Commission.

The PSC set new numeric demand and energy goals for these seven utilities in July 2004. The new numeric goals were generally lower than the previous goals set by the PSC in 1999 for three primary reasons: (1) the Florida Building Code contains increased minimum energy efficiency levels, limiting the amount of incremental savings from utility sponsored programs; (2) many utility DSM programs have reached a saturation in participation levels; and (3) the relatively low cost of new generating units has reduced the cost-effectiveness of several DSM programs.

Energy Conservation Cost Recovery

Investor-owned utilities have the opportunity to recover prudently incurred expenditures associated with PSC-approved DSM programs through the Energy Conservation Cost Recovery Clause (ECCR). Since 1981, Florida's investor-owned utilities have collected approximately \$4.15 billion through the ECCR, with nearly \$2.54 billion recovered in the last ten years. Annual ECCR expenditures have stabilized at just under \$250 million per year over the past six years for two primary reasons: 1) DSM programs have reached saturation in participation levels, and 2) DSM program cost-effectiveness continues to decline due to the lower cost of new generating units. However, as utility plans include more solid fuel generation options, DSM program cost-effectiveness will improve.

Federal and State Solar Energy Initiatives

To conserve energy and reduce electricity bills, the State is providing 150 solar water heaters to residents in 20 underserved communities throughout the state. Named *Front Porch Sunshine*, Florida is the first in the nation to install solar energy technology in weatherized, low-income homes.

Florida schools are using the state's free supply of sunshine to light classrooms and the imaginations of students. Florida's *SunSmart Schools Program* is installing 29 solar electric systems in schools

¹² Demand savings are cumulative from 1980. Energy savings are on an annual basis.

throughout the state. The program combines State funding with private partnerships to provide clean energy and science education. The electric power generated by the system is used to power the school's classrooms, with excess energy returned to the local power grid. The system also provides an on site classroom for students to learn more about solar power and the benefits of energy conservation.

SunBuilt is the latest program to expand solar energy technology in communities throughout Florida. A partnership between DEP, the Florida Home Builders Association and the Florida Solar Energy Research and Education Foundation, *SunBuilt* provides rebate checks to home builders who install solar hot water heaters in newly constructed homes. *SunBuilt* Builders are eligible to receive rebates for each solar equipped home constructed. The reliable, low-maintenance solar systems use the sun's energy, instead of electricity, to heat water. A solar collector installed on the roof holds water that is heated by the sun. A traditional water heater serves as a backup supply of hot water on overcast or rainy days. Benefiting both the economy and the environment, the amount of energy generated annually by a single solar water heater is equal to two barrels of oil, lessening the state's dependence on petroleum imports.

Federal and State Energy Efficiency Initiatives

Florida's 35 million hotel guests can now opt to take a break at a *Green Lodge* – a new State designation that recognizes environmentally-friendly hotels and motels. The voluntary program establishes environmental guidelines for hotels and motels to conserve natural resources and prevent pollution. Hotels and motels reduce costs and earn designation by investing in simple and innovative 'green' practices that conserve water, save energy and reduce waste.

Rebuild America is a growing network of community-driven voluntary partnerships that foster energy efficiency and renewable energy in commercial, government and public-housing buildings. For example, the University of Central Florida is outfitting three buildings on campus with energy efficient technologies. The project will dramatically reduce energy consumption in these buildings, with energy savings reinvested in the project for wider application across campus. In addition, school districts in Florida, North Carolina and Tennessee were selected by the National Energy Foundation to receive energy education materials and teacher training. In school year 2003-04, four school districts with more than 19 million square feet of building space reported saving 18,346,495 kilowatt hours and nearly \$1.2 million from energy conservation practices.

In 2004-05, the *Florida Energy Conservation Assistance Program* provided energy surveys of more than 9,717,000 square feet of building space for state/municipal governments, small businesses, agricultural and hospitality in Florida. The surveys identified more than \$11.8 million of possible energy savings for the clients involved.

Federal and State Hydrogen Initiatives

In July 2003, Governor Jeb Bush launched "H2 Florida," a statewide initiative to accelerate the commercialization of hydrogen technologies, spur investment and economic opportunity and safeguard the nation's natural resources. H2 Florida partners the State of Florida with industry, governments and academia to showcase hydrogen technologies and stimulate a consumer market for cleaner, sustainable sources of energy. The State's commitment supports that of the White House's proposed investment of \$1.7 billion over five years to develop hydrogen-powered fuel cells, hydrogen

28

infrastructure and advanced automotive technologies.

Florida has 28 hydrogen demonstration projects underway and seven state universities are conducting more than 100 hydrogen research and development projects. In 2003, the Florida Energy Office and Florida Power and Light installed a fuel cell at Hugh Taylor Birch State Park. Since that installation, the Florida Energy Office has installed fuel cells at North Port High School and Homosassa Springs State Park. Additionally, the DEP has purchased 12 fuel cells to provide backup power at their District and Branch Offices statewide.

Florida's Ability to Produce, Store and Distribute Transportation Fuel



III. Florida's Ability to Produce, Store and Distribute Transportation Fuel

A. CURRENT AND PROJECTED SUPPLY AND DEMAND FOR TRANSPORTATION FUEL

Transportation is Florida's second largest energy use sector comprising more than a third of the total energy used. Florida depends almost exclusively on other states and nations for supplies of oil and gasoline, producing less than one percent of the nation's crude oil production annually. Florida consumes 8.6 billion gallons of gasoline per year, and consumption is growing by 300 million gallons per year.¹³

Current transportation fuel needs are approximately 28.1 million gallons per day¹⁴, not including aviation fuels. Motor gasoline and diesel fuel make up more than 87 percent of Florida's transportation energy costs, with aviation fuel accounting for less than 10 percent.¹⁵

In ten years, Florida's transportation fuel demand is expected to increase to 32.3 million gallons per day, assuming a 15 percent increase in the state's population.¹⁶

Refined Petroleum Products

Florida's reliance on imported petroleum products, in addition to its anticipated growth in consumption during the next decade, underscores a potential vulnerability to interruptions in fuel supply delivery during natural or man-made disasters.

Ethanol

The demand for ethanol is driven largely by the federal Energy Policy Act of 1992 (EPAct 1992), which required that public and private vehicle fleets operated within selected Metropolitan Statistical Areas (MSA) acquire and operate Alternative Fuel Vehicles (AFV). Florida contains nine designated MSAs in which the EPAct 1992 AFV standards apply. There are seven ethanol fueling stations in Florida, all of which are restricted for private fleet usage for EPAct 1992 compliance. The demand for ethanol-based fuels is expected to grow in coming years as at least one major automobile manufacturer (Ford Motor Company) has announced plans to increase production of AFVs by 2010.

Florida currently has no operational ethanol plants in the state. Florida meets this demand for ethanol by imports from refineries outside of the state. Ethanol stations are subject to fire and building code requirements, regulated by the National Fire Protection Association and the Florida Department of Community Affairs respectively. Depending upon the size of the facility, additional

¹³ Florida Department of Revenue

¹⁴ Daily consumption based on data from Florida Department of Revenue.

¹⁵ Florida's Energy Future, January 2003.

¹⁶ Assumes an average of 1.5% growth annually for the next ten years, based on U.S. Census Bureau population trend from 2000-2004.

permits for storm and wastewater and air emission permits may be required from DEP.

Biodiesel 17

The total "dedicated" biodiesel production capacity in the United States is approximately 180 million gallons per year. Another 100 million gallons is projected to be online by May 2006 and an additional 470 million gallons are planned nationwide.

In Florida, a multi-feedstock biodiesel plant in Lakeland has the capacity to produce 18 million gallons annually and will eventually produce up to 30 million gallons. Two additional companies have set up biodiesel distribution facilities at the Port of Tampa and Port Everglades.

Hydrogen

Florida's current hydrogen capacity is less than 20 gallons of gasoline equivalent per day (gge/day). Florida currently has one hydrogen energy station in operation, two stations in the construction and permitting stages and two stations in the planning phase. Once permitted and constructed, the two additional hydrogen fueling stations will provide more than 130 additional gallons of gas equivalent per day.

Once all five hydrogen stations are installed and operating, they will meet the demand of 15 vehicles - approximately 290 gallons of gas equivalent per day.

B. CURRENT AND PROJECTED INFRASTRUCTURE NEEDS FOR PRODUCTION AND SUPPLY

As the demand for transportation fuels increases over the coming years, Florida's infrastructure for producing, storing and transporting that fuel to market will also increase. The following table summarizes Florida's current and projected transportation fuels capacity.

Banyonetton Fuel Capacity	(Churgene 2010)	Protected
(and the gallons)		
Gasoline – Bulk Storage	34218	
Gasoline – Retail Storage	18719	
Diesel – Bulk Storage	141	
Diesel – Retail Storage	24	
Ethanol	0	80 MMG/Yr
Biodiesel	18 MMG/Yr	66 MMG/Yr
Hydrogen	3000 gge ²⁰	

20 gge = gallons of gasoline equivalent.

¹⁷ National Biodiesel Board

¹⁸ Represents bulk storage capacity at the ports. Volume derived from data reported to DEP, Bureau of Petroleum Storage Systems, adjusted down by 20% to account for unused min/max levels kept in the tanks.

¹⁹ Represents storage capacity at retail stations. Volume derived from data reported to DEP, Bureau of Petroleum Storage Systems, adjusted down by 10%.

In-state Oil Production

Florida's oil production capability is approximately 10,000 barrels per day, or 120 million gallons of gasoline per year. Production has been steadily declining for the past few decades due to the declining economic efficiency of the state's oil fields. Crude oil production within Florida is negligible in meeting the state's current or future demand for motor fuel as Florida does not contain any native refinery capacity. Crude oil is moved via pipeline to refineries outside the state for processing.²¹ Additional oil production within Florida is not expected to occur within the planning horizon.

Refined Petroleum Products

Florida receives 98 percent of its fuel by sea via barge and tanker ship into seven ports. Ninety percent is received into three ports: the Port of Tampa, the Port of Jacksonville and Port Everglades. The remaining two percent is supplied from out-of-state via tanker truck. Fuel is supplied by domestic and international refineries as well as the pipeline spur in Bainbridge, Georgia.



From bulk storage, fuel is distributed by approximately 1,000 tanker trucks under various commercial arrangements to more than 9,200 retail gas stations in the state.²² Fuel distribution fluctuates based on the supply and demand for fuel and the business objectives of petroleum companies and suppliers.

²¹ Sources: Oil and Gas Annual Production Reports, Florida Geological Survey, 2004, and the Florida Petroleum Council, Fuel Fact Sheet

²² Florida Petroleum Council

Ethanol

While there are currently no ethanol production plants in Florida, there is the potential for two or three plants in southwest Florida becoming operational over the next few years with a combined annual production capacity of 80 million gallons.

Florida's seven ethanol fueling stations are not open to the public and serve organizational fleets such as the Department of Transportation.

Biodiesel

Annual biodiesel production capacity in Florida is 18 million gallons and is anticipated to grow to 30 million gallons in the near future. Storage capacity for imported biodiesel is expected to grow to over 36 million gallons within the next year.

The National Biodiesel Board currently lists nine companies as distributors of biodiesel throughout Florida. Biodiesel is not readily available through retail fueling stations.

Hydrogen

Nationally, approximately 20 hydrogen stations providing fuel to controlled fleets are in operation. Florida currently has one station in operation, two stations in the construction and permitting stages and two stations in the planning phase.

- Existing Hydrogen Station in Operation
 - o BP facility on State Road 426 in Oviedo
 - Designed to support controlled fleet of five fuel cell vehicles
 - Generation capacity: <20 gge/day
- Hydrogen Stations Under Construction/Permitting
 - o Chevron Hydrogen Company facility at Boggy Creek Road and Tradeport Drive in Orlando
 - Designed to support controlled fleet of four shuttle buses
 - Generation capacity: <120 gge/day
 - o Hydrogenics facility at the Orlando International Airport
 - Designed to support controlled fleet of two baggage carriers
 - Generation capacity: <16 gge/day
- Planned Hydrogen Stations
 - o Supplier To Be Determined on International Drive in Orlando
 - Designed to support controlled fleet of 4 shuttle buses
 - Generation Capacity: <120 gge/day
 - o Supplier To Be Determined at NASA/Kennedy Space Center in Cape Canaveral
 - Designed to support controlled fleet of two shuttle buses
 - Generation Capacity: <60 gge/day

C. CURRENT AND PROJECTED CONSUMER COSTS

 Fransport from Equil Consumer Consts (Signification) 	Currence 2005
Diesel	2.30 - 2.60
Ethanol	1.22 - 1.60
Biodiesel	2.40 - 3.70
Hydrogen	\$8.56/gge

Ethanol is sold into the gasoline blending market where it competes with other oxygenates, octane components and gasoline. As such, ethanol prices are highly correlated with the price of gasoline and gasoline blending components. Additionally, biodiesel is sold into the diesel blending market and correlates with the price of diesel. Availability of hydrogen is limited to specific fleet applications.

D. CURRENT REGULATORY OVERSIGHT

The following section is intended to provide highlights of regulatory oversight but may not be all-inclusive.

Refined Petroleum Products

- The U.S. Environmental Protection Agency regulates emissions associated with the combustion of petroleum products and regulates the sulfur content in the fuel.
- The DEP certifies the installation of above ground and underground storage tanks.
- The Florida Department of Agriculture and Consumer Services regulates fuel quality, maintains blending requirements, sets the Reid Vapor Pressure requirements and is responsible for the weights and measures associated with dispensers at retail stations.
- The Florida Department of Revenue applies taxes to petroleum products.
- The Florida Department of Transportation is responsible for all regulation associated with on road distribution including truck weights, tanker truck storage requirements, operational hours, etc.
- The Florida Department of Community Affairs is responsible for all building codes associated with retail facilities and the construction of any structures.
- The Florida Department of Financial Services is responsible for all hazardous material requirements, including fire and safety codes.

Hydrogen Refueling Stations

Codes for permitting both stationary fuel stations and hydrogen motor fuel dispensing stations are currently being developed. The report "The Regulator's Guide to Hydrogen Development" details the methodology to be used in creating these guidelines and was developed through a collaborative effort involving the National Fire Protection Association (NFPA), the International Code Council (ICC), Pacific Northwest National Laboratory and the National Renewable Energy Laboratory. Stationary hydrogen refueling facilities must meet NFPA codes and building codes. Building codes are regulated by the Florida Department of Community Affairs, while the fire codes are created through the NFPA. The Vehicular Fuel Systems Code of 2006, Chapters 12.1

-14.13.9, is the specific reference for the NFPA rules regarding hydrogen fueling stations.

Ethanol Refueling Stations

Ethanol stations are subject to fire and building code requirements.

Owners or operators of vehicles powered by alternative fuel are required to obtain a valid Alternative Fuel Use Permit from the Florida Department of Revenue. This decal is in lieu of the excise tax on gasoline. Refueling stations are not allowed to fuel an alternative fuel vehicle that does not display the appropriate decal. State and local government alternative fuel vehicle fleets are exempt from paying the decal fee.

Biodiesel Refueling Stations

Pursuant to Chapter 206, F.S., biodiesel is not categorized as an alternative fuel, but as a diesel fuel and is subject to all regulatory issues cited for refined petroleum products. The Florida Department of Revenue alternative fuel decals do not apply to biodiesel.

E. FUEL EFFICIENCY AND ALTERNATIVE FUEL INITIATIVES IN FLORIDA

Florida has the fourth largest number of registered hybrids in the nation. Florida's state government agencies have committed to purchase alternative fuel and clean energy vehicles. More than 22 percent of DEP's fleet is comprised of "green" transportation with more than 90 hybrid vehicles and more than 290 alternative fuel vehicles. The State of Florida's entire vehicle fleet comprises just over 19,000 cars and light trucks. Of that amount, 107 are hybrid and 1,491 are alternative fueled vehicles.²³ Since its inception in 1993, the U.S. Department of Energy's Clean Cities Program -- with 88 coalitions nationwide, including two in Florida -- has helped conserve enough petroleum to fuel two million cars for a year.

The U.S. Department of Energy's Clean Cities Program is advancing the economic, environmental and energy security of the United States by adopting practices that reduce petroleum consumption in the transportation sector. Clean Cities carries out this mission through a network of volunteer coalitions, which develop public/ private partnerships to promote alternative fuels and vehicles, fuel blends, fuel economy and hybrid vehicles.

Florida has 28 hydrogen demonstration projects underway and seven state universities are conducting more than 100 hydrogen research and development projects. By 2006, Florida aims to establish metro-Orlando as a "hydrogen hub," building four hydrogen energy stations to supply hydrogen vehicle fleets in Central Florida.

In September 2005, Florida received its first fleet of hydrogen vehicles. Five Ford Focus Fuel Cell vehicles were delivered to the DEP and Progress Energy Florida. The vehicles are in use at Wekiva Springs State Park, DEP's Central District and at Progress Energy's Longwood facility and will also be showcased in regional alternative and clean energy conferences, summits and events.

In January 2006, the DEP, Ford, Hydrogenics, Tug Technologies, and Delta Airlines expect to begin demonstrating hydrogen baggage carriers at the Orlando International Airport. Additionally, in the fall of 2006, DEP, ChevronTexaco and Ford will provide eight hydrogen shuttle busses to the Orlando-area for use in the tourism industry.

Florida's Energy Plan

²³ Florida Department of Management Services

Overview of Florida Statutes



IV. Overview of Florida Statutes

Currently, Florida law addresses energy concerns with several distinct statutory chapters. These statutes provide the institutional structure for the regulation and implementation of energy policy in the State of Florida. Responsibilities for energy regulation and policy implementation are divided between a legislative commission, the PSC and two executive level agencies: the Florida Department of Community Affairs and the DEP.

Chapter 186, F.S.

Chapter 186, F.S., promotes intergovernmental coordination and the effective allocation of resources by outlining a coordinated planning process. Section 186.801, F.S., highlights the requirement that all regulated utilities submit a Ten-Year Site Plan to the PSC. This document estimates a utilities power-generating needs and the general location of its proposed power plant sites.

Chapter 350, F.S.

Chapter 350 F.S. provides the enabling legislation for the utility regulatory body, the PSC. This chapter contains the statutory provisions under which the PSC operates including how commissioners are chosen and confirmed, the purview and statutory authority of the commission, and the funding mechanism through the Florida Public Service Regulatory Service Trust Fund.

Chapter 366, F.S.

Chapter 366, F.S., governs the PSC's jurisdiction over electric and gas utilities.

This statute gives the PSC authority over five broad areas for all electric utilities in the state, including municipals and rural cooperatives. The chapter provides the source for the Commission's rate making authority, including cost recovery. It also specifically addresses cogeneration and small power production, the requirements of the Florida Energy Efficiency and Conservation Act, ensures the development and maintenance of a reliable and coordinated power grid, as well as electric safety requirements.

This chapter directs the PSC to establish and maintain continuous liaison with all other appropriate state and federal agencies whose policy decisions and rulemaking authority affect those utilities over which the commission has primary regulatory jurisdiction.

Chapter 368, F.S.

Chapter 368, F.S., authorizes the establishment of rules and regulations covering the design, fabrication, installation, inspection, testing and safety standards for installation, operation and maintenance of gas transmission and distribution systems, including gas pipelines, gas compressor

stations, gas metering and regulating stations, gas mains, and gas services up to the outlet of the customer's meter set assembly, gas-storage equipment of the closed-pipe type fabricated or forged from pipe or fabricated from pipe and fittings and gas-storage lines.

This Chapter also creates the rate making authority of the PSC for natural gas transmission companies for any service relating to the transmission or sale of natural gas in the state. The section applies only to facilities located wholly within this state for the transmission or delivery for sale of natural gas. Local distribution pipelines are exempt.

Chapter 377, F.S.

Chapter 377, F.S., addresses the regulation, planning and development of the energy resources of the state. It is the policy of the State of Florida to conserve and control the natural resources of oil and gas to prevent waste of natural resources and to safeguard the health, property and public welfare of the citizens of the state.

Chapter 377, F.S., specifically addresses the funding by electric utilities of local governmental solid waste facilities that generate electricity, the Solar Energy Standards Act of 1976, functions of the Florida Department of Community Affair's Energy Emergency Contingency Plan and Federal and State Conservation Programs.

Finally, Chapter 377, F.S., provides the language for the duties and activities of the Florida Energy Office.

Chapter 403, F.S.

Chapter 403, F.S., recognizes that the predicted growth in electric power demand requires the development of a procedure for the selection and utilization of sites for electrical generating facilities and the identification of a state position with respect to each proposed site. Thus, Chapter 403, F.S., outlines the requirements of the Florida Electrical Power Plant Siting Act which applies to any steam or solar electrical generation facility except those that are less than 75 megawatts.

This statute also provides an exclusive forum for determination of need for an electrical power plant. Chapter 403, F.S., addresses electric power transmission concerns by outlining the Transmission Line Siting Act, which applies only to a transmission lines that operates at 230 kilovolts or more, while transmission lines which are less than 15 miles in length or which do not cross a county line are exempt and by providing a process for the determination of need for transmission lines. Chapter 403, F.S., also creates the Natural Gas Pipeline Siting Act. The DEP also regulates electric and magnetic fields from electrical transmission lines under the provisions of sections 403.061(30) and 403.523(14), F.S.

Chapter 553, F.S.

Chapter 553, F.S., addresses thermal efficiency standards by instructing the Florida Department of Community Affiars to provide a thermal efficiency code to provide for a statewide uniform standard for energy efficiency in the thermal design and operation of all buildings statewide, consistent with energy conservation goals, and to best provide for public safety, health, and general welfare.

Chapter 553, F.S., also specifies that the Florida Building Commission shall adopt the Florida Energy Efficiency Code for Building Construction within the Florida Building Code, and shall

modify, revise, update, and maintain the code to implement the provisions of this thermal efficiency code and amendments thereto, in accordance with the procedures of Chapter 120, F.S. The DCA shall, at least triennially, determine the most cost-effective energy-saving equipment and techniques available and report its determinations to the commission, which shall update the code to incorporate such equipment and techniques.

Chapter 553, F.S., addresses energy efficiency by providing statewide minimum standards for energy efficiency in certain products, consistent with energy conservation goals. The standards are required to be based on feasible and attainable efficiencies which will reduce Florida's energy consumption growth rate and the growth rate of energy demand.

Finally, Chapter 553, F.S., outlines the Building Energy Efficiency Rating System to provide for a statewide uniform system for rating the energy efficiency of buildings.

Recommendations for Electric Power Generation



V. Recommendations for Electric Power Generation

Diversification Recommendations

Keenmenderen SI Somming und opendere Provinsionen erdipermeter autometer endipermeter autometer erstellender Provinsionen Sounderen Stellender Provinsion Sounderen Stellenderen denne verschen Jerrenbiet meter erstelpermeter Provinsion ander Withouw reaminer provinsion source of the Provinsion ander with provinsion integration of the adometer Barter Barter and adometer Barter Provinsion of the adometer Provinsion Barter Provinsion Barter Provinsion of the adometer Provinsion Barter Provinsion Barter Provinsion of the adometer Provinsion Barter The Power Plant Siting Act (PPSA) is a centralized, coordinated licensing process encompassing the permitting, land use and zoning, and proprietary interests of all state, regional, and local agencies in the jurisdiction of which an electric power plant is proposed for location. The PPSA provides for a single certification (license) for those electric power plants, as defined by the PPSA, which are steam or solar powered, 75 megawatts or greater, and are constructed after October 1, 1973. The provisions apply to nuclear power, although regulation of nuclear radiation is preempted by the federal government. Only electric utilities as defined by the PPSA may apply for certification. The process includes mandatory land use and certification hearings by an administrative law judge, a Determination of Need by the Public Service Commission (PSC), with ultimate approval/denial authority vested in the Siting Board, which consists of the Governor and Cabinet. The DEP coordinates the process.

While the PPSA has proved an effective mechanism for the licensing of new generation, recommended changes for simplifying and expediting the review process, ensuring protection for the environment and maintaining public participation include:

- Simplifying and streamlining completeness and sufficiency procedures;
- Reducing mandatory hearings;
- Revising time limits;
- Clarifying applicability; and
- Clarifying the statutes defining the authority of the Siting Board to review local government determinations on land use consistency to ensure the successful development of projects critical to the State's welfare.

Improving the efficiency of the PPSA and clarifying provisions will expedite power plant licensing, ensure quicker development of diverse generation infrastructure and enhance reliability.

Recommendation #2.

Streambline and expedite whe stringsend permitting of electric fitnesmission and disenfond or destinates by revising the provisions of the Filondia Electrical Transmission Unite String Act, ES 20006 sectore and average processing sinte for new contentions of resources by the Sectorities while mechanismission essources by the Sectorities while mechanismission

envrorment of Jepfrement offEnvirormental Bookerion Silter Board)

Sulfar Board

Similar to the PPSA, the Transmission Line Siting Act (TLSA) is a centralized, coordinated licensing process encompassing permitting, land use and zoning, and proprietary interests of all state, regional, and local agencies in the jurisdiction of which a transmission line is proposed for location. The TLSA provides for a single certification (license) for transmission lines subject to the TLSA. Transmission lines subject to the TLSA are those which are 230 kilovolts or greater, 15 miles or more in length and cross a county line. The process includes a mandatory hearing by an administrative law judge, a Determination of Need by the PSC, with ultimate approval/ denial authority vested in the Siting Board. The DEP coordinates the process.

Recommended changes for simplifying the TLSA, expediting the review process, improving efficiency and enhancing reliability, while protecting the environment and maintaining public participation include:

- Simplifying and streamlining completeness and sufficiency procedures;
- Reducing mandatory hearings;
- Revising time limits;
- Clarifying who may be an applicant; and
- Clarifying comprehensive planning and zoning issues.

Improving the efficiency of the TLSA and clarifying provisions will expedite transmission line licensing and ensure quicker development of reliable transmission infrastructure.

Recommendation #3

Promore fuel diversity fuel supply reliability end energy security

 Provide the Public Service Comparison while ambients to an include diversion that supply establish pointer are as some the superprise generation syntem cyclication the superprise public of table service comparison before public of table service comparison before public of table service

Acted Charlen Silver and Charlen Silver and Charlen a

Smildy subpertainents
 placoutent of distribution
 line and the number of substitution
 line and the number of substitution
 ansing mean the OOF and 2
 2005 function of acceptible
 legisteriae should provide the
 Richer Service commission()
 subtract of acceptible

nation de adhershiftwo Enclas power distribution Intersectation de huitcanes (Pable Steue Commission) The Florida PSC is charged with ensuring the development of reliable electric system resources in a manner that is economically competitive and protective of consumers. While a diverse fuel base enhances system reliability and energy security, it is currently not a major factor in the PSC's decision making. Empowering the PSC to include fuel diversity, fuel supply reliability, and energy security as high priorities in its decision making processes will enable the development of a diverse infrastructure in the state.

Chapter 366, Florida Statutes, provides the PSC with the authority and mechanisms for ensuring the provision of adequate, reliable, reasonable cost electricity to consumers. The PSC should support an update of Chapter 366, Florida Statutes, to include fuel diversity and fuel supply reliability as criteria for the installation of new electric generating infrastructure.

Chapter 403.519, Florida Statues, provides the guidelines for a determination of need for an electric generating facility. Prior to building a facility, Rule 25-22.082, Florida Administrative Code, requires that an applicant apply for a determination of need. A clause should be added to Chapter 403.519, Florida Statutes, to ensure the PSC considers fuel diversity and fuel reliability as factors when determining the need for new generation.

With a growing economy, additional transmission capacity will likely become necessary. There also may be a need to improve the efficiency and reliability of transferring power within certain regions of the state. It is therefore recommended that the PSC conduct a study of transmission grid reliability and emergency contingency conditions. The utilities in the state are encouraged to provide assistance to the PSC in conducting the study.

Distribution lines are essential to the delivery of generated power to residential and commercial consumers. These lines are particularly vulnerable to disruption when impacted by a hurricane. The PSC should work with local and state officials to review the feasibility of subterranean placement of distribution lines and other infrastructure hardening issues.

Recommendation #4

Baerffears additional fiel delively moblements in Flouch for power generation. Dispetitic all state potentis required for rodundancies and moreased capters. disparement of baismonic and Protection. Reard of heusters of the factorial balance on the factorial balance on the factorial balance.

Recommente Ritori #5

Adopt induced Interconnection standards to include allidismbined generation technologies (Bublic Service Contrassion

The interruption in the supply of natural gas and petroleum resulting from Hurricane Katrina in 2005 left Florida in an extended state of emergency. Source refineries for petroleum and a pipeline delivering a major portion of Florida's natural gas supply were damaged and offline. Because of these major interruptions, Florida's utilities began implementing emergency procedures using back up fuels and urging consumers to reduce electricity consumption and conserve fuel.

Florida can limit interruptions to electric generation fuel supply by diversifying its sources of fuel. Two natural gas pipelines have been approved to deliver liquefied natural gas, and there is interest in establishing additional natural gas and petroleum pipelines. The DEP recommends the State continue to explore mechanisms for creating redundancy in fuel supply and fuel delivery.

Distributed generation refers to self-generated, modular electricity generators sited close to the customer load. Distributed generation technologies include wind, solar, biomass, fuel cells, gas microturbines, hydrogen, combined heat and power, and hybrid power systems. Distributed generation systems can be integrated with electricity provided from a utility, enabling utilities to defer or eliminate costly investments in transmission and distribution system upgrades, and provide customers with better quality, more reliable energy supplies.

PSC Rule 25-6.065, which references the Institute of Electrical and Electronics Engineers' standard IEEE 929, currently allows for streamlined standards for the interconnection of "small photovoltaic systems" of 10 kilowatts or less to the local power grid.

It is recommended that PSC Rule 25-6.065 be expanded to incorporate the updated national standard IEEE 1547, which considers uniform connection standards associated with all distributed generation technologies including larger systems greater than 10 kilowatts. Uniform connection standards will minimize barriers to distributed generation interconnection.



Energy is essential to the economic health of Florida. With the increase in population and demand for energy, and the threat of supply disruptions it is imperative that policy makers are provided with objective recommendations on current and projected energy issues. The council should be comprised of a diverse group of stakeholders, including utility providers, researchers, fuel suppliers, technology manufacturers, environmental interests and others, and should advise the State on ideas and solutions to address energy needs and concerns.

Conservation Recommendations

Recommendation #7

Expedite start performance entricement with Energy Service comparates EV 2006. Surger entries Savuige of order the information of a sublitter kitowan hours anomality as areally of these constants Department of Managements Subjects Department of Managements Informations In 1997, the Department of Corrections executed an energy savings contract with Florida Power and Light involving 16 institutions covering 4.6 million square feet. Over four years, the energy service contract has achieved a savings in electric, water and operating costs of more than \$1.3 million annually.

In 2003, at the direction of Governor Bush, the State began to initiate contracts with Energy Service Companies (ESCOs) to evaluate State facilities for energy efficiency improvements. The ESCOs operate under performance contracts, receiving payment based on the savings generated for the State.

The DEP recommends the Department of Financial Services, Department of Management Services and State agencies expedite contract negotiations with ESCOs and implement performance contracts to allow earlier construction and energy savings.

Recommendation#8

Protectic siver stress of straig constant threat stress of straig constant the straight strai

Kateorprovate entensis Control metalonic efficientes Cateful elementiste ententists Cateful elementiste ententists Cateful elementiste ententists

an K-12 popular schools and ontwarstand die parameters of Records and the Rectord of the

Pricochinace in controls, endatable, brokkling commacross, developens and coronations, provinces are priciliterial analytic analytic endersiterial analytic on building analytic on building stategress (Department), or Framommontal Proceeding)

 Encourage maniferences and independent power providers aoreale advantage of vasie anongy and use cogenet arous resintologies (Dependences) - Moredione

Regelonium antakistorin #495

Use discretionary antered manufacturity to ellow approved alternative anong protects this provide regener public benefic in the of out monoless parentics (POperturien of Bastronanenset Borredson) Fostering energy conservation and efficiency begins with awareness. Through a network of existing partners, the State is building understanding about energy conservation, energy efficiency and alterative and renewable energy technologies for the home and workplace.

As advancements in clean energy technologies and energy efficient practices evolve, continued outreach and education is needed to make these opportunities mainstream. For example, Combined Heat and Power (CHP) is an efficient, clean, and reliable approach to generating power and thermal energy from a single fuel source. By better understanding the potential benefit of CHP, Florida's industrial sector can increase operational efficiency and decrease energy costs, while reducing emissions of greenhouse gases.

Other education efforts can have the ability to reach a variety of audiences including the online Utility Report Card for public schools energy officials, training for home builders on energy efficient and hurricane resistant construction practices, case studies on biogas generation for dairy farmers, and energy-related classroom activities for elementary school students.

The DEP is also currently pursuing nearly \$380,000 in funding from the State Technologies Advancement Collaborative to lead a five-state initiative to promote ENERGY STAR[™] benchmarking tools and technical assistance for K-12 public schools and local governments throughout the Southeast United States. This outreach project builds on Florida's existing SunSmart Schools Program, which has installed 29 solar electric systems in public schools throughout the state. The program combines State funding with private partnerships to generate clean energy used by the school and create a hands-on science project for students.

The DEP has the authority to collect cash penalties for violations of environmental laws. The agency currently manages an enforcement program called "Pollution Prevention Projects in Enforcement" that allows DEP to settle enforcement actions by requiring the completion of an environmental project in lieu of civil penalties. The program requires the company to invest 1.5 times the value of the civil penalty in the pollution prevention project.

Where less serious violations occur, the DEP recommends allowing options for implementing alternative energy or energy efficiency projects in lieu of cash penalties, where the project provides a greater environmental and community benefit than civil enforcement. The DEP would recommend appropriate projects on a case by case basis.

Recommendation #10

Require allows: State government shtilding construction to meet the U.S. Fatam Building Countills Londershiptin Environmental Design standards: Encourag local governments and continuing: developers to adopt high profotments group high profotments. Generatives Dependents of Communication (Managements State government is leading by example with a strategic goal to reduce energy consumption by 25 percent below 2002 levels at all State government facilities by 2007.

To improve building efficiencies within State government and achieve the 2007 energy reduction goals, all new State buildings should meet the U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED) standards. Developed by all sectors of the building industry, the LEED program is a voluntary, consensus-based national standard for developing high-performance, sustainable buildings.

To improve building efficiencies outside of State government, the Department of Community Affairs should encourage local governments and community developers to adopt the Florida Green Building Coalition's recommendations for municipalities, such as expediting residential building permits for homes built to high energy efficient standards. Utilities should be encouraged to offer rebates for new and renovated ENERGY STAR[™] certified homes.

Economic Incentive Recommendations

Recommendation #11 Reevide gestershundling for stescarcin antel dectors from the the weather and the second states of the second iene cieverophivene serdiin make a construction of Remember Onedrew Systems: Byennergi stoller, farvolgergeren BUTCHERENSE AVENAGE OR FRANK feinter den anneinennen erenternannen Nerelassiation Hexas (197 2000) addice generation in confidence of too it is NEWLEY TO MEL ... REAL DESCRIPTION a (alter te experience and search alternation of the search of the searc เสียงอาเหตุ: เต่มสาย, ระบัติไปต่อมรถศาสติมไรเต่ะ เต่มี. Lenature and the second of the second (cont.)

The DEP is currently administering approximately \$5 million in grant funding to advance renewable and emerging alternative energy technologies for electricity generation. Additional funding is needed to build on these efforts and continue vital research and demonstration of these next generation technologies.

Industry and the State's universities contain the experience and expertise necessary to improve alternative energy technologies. By sponsoring further research, development and demonstration projects, the State can continue fostering these technologies in the marketplace and spurring economic investment in Florida.

Recommendation #12

Identify alternative energy groche than and distribution and its button the contract of the energy the state of the state of the arising trade and theoremic the algorithm. Dependent of Reconces

Recommendation #13 Provide consumer and commencial reforces to assess with installedge of otherworking and solar phonovoltate phonovoltate



Administered through the Governor's Office of Tourism, Trade, and Economic Development, Qualified Target Industries (QTI) incentives are available to companies that create high wage jobs in targeted industries. Economic incentives include refunds on insurance premiums and corporate income, intangible personal property, sales and ad valorem taxes.

QTI designation will authorize tax refunds for companies that invest in Florida's alternative energy production and distribution industries.

Photovoltaic systems convert sunlight into electrical current but have high initial costs associated with installation. Similarly, while solar water heaters are considered cost effective, the thermal technology has a higher initial cost than traditional fossil-fuel based water heaters. When installed, however, solar water heaters can provide as much as 80 percent of the hot water demand of the typical Florida residence.

By reducing installation costs, a rebate program would create an incentive for utilizing photovoltaic systems and solar water heaters both residentially and commercially. Promoting these technologies would also provide an economic benefit for Florida's native solar industry, reduce consumer energy costs and reduce grid demand at peak times. In addition, photovoltaic systems of two kilowatts or higher power output can supplement grid supplied power.

Subject to appropriation, the DEP will establish a solar rebate program for solar thermal and solar photovoltaic technologies, providing:

- Rebates per installed watt for photovoltaic systems in residential, commercial, public and non-profit applications.
- Rebates for the installation of solar thermal systems, excluding swimming pool heaters. A fixed rebate will be offered for residential applications. For commercial, public and non-profit applications, the rebate will be on a system performance basis.
Recommendation

Baty also dony united relation in introducional diseates of a finite Physical Contest of a straight supplicities and a supplicities of a straight description of the straight of the description of the straight of the interference of int

Roccional analysis

Convertes siller molfactorio entre anome denoitées fériculation anome denoitées fériculation anome as calébral calls for.
Supplier normanique d'undergit produce an entre variable for anome son antiger and herelogites by 4000 portes de avelable in exe avec sons of Department of fiteory constants of Programment of the songle constants of Programment of the songle constants of Programment of the songle constants of Programment of ENERGY STAR[™] is a government-backed program that promotes the use of industrial and household energy efficient appliances. According to ENERGY STAR[™], the national average for household expense for energy bills is \$1,500 a year. With ENERGY STAR[™] products and home improvements, energy savings can reach 30 percent or more than \$450 per year.

Subject to appropriation, the DEP will establish a rebate and incentive program for Florida residents on the purchase of eligible ENERGY STAR[™] appliances. Specified funds will be directed toward an incentive program to support residents in low income communities.

Under the 2005 Federal Energy Policy Act, additional federal financial incentives may be available to consumers that purchase ENERGY STAR[™] appliances.

Hydrogen fuel cells are unique in terms of the variety of their potential applications, providing energy for systems from laptop computers to utility power stations. Offering several benefits over conventional combustionbased technologies, fuel cells are currently used in many power plants and generators and are virtually pollution free.

Additional investments in the fuel cell industry will continue the advancement of fuel cells for use in backup and supplemental electricity generation and lead to lower cost, higher efficiency products.

Sales and corporate income tax incentives will spur investment in the fuel cell industry promoting advancements and improvements in the technology. Tax incentives will encourage Florida businesses to invest in alternative energy for backup and supplemental power. Recommendations for Transportation Fuels



V. Recommendations for Transportation Fuels

Diversification Recommendations

Recommendation #1 In these additional and diverse particularm supply and director particularm supply and director and the second for additional particles equired for additional particles equired for additional particles addition additional particles are additioned applying a displaying barrel and the transformation of the senses of Frusteestor the data that the provenuent Toust.



Florida imports 98 percent of its transportation fuel into three major ports. Although Florida did not experience extended closures at its major ports during the 2004 and 2005 hurricane seasons, the state experienced long lines and limited fuel supplies at retail stations following each storm. In addition, shortages were experienced by emergency responders and electricity generating facilities. The interruption in the supply of petroleum resulting from Hurricane Katrina in 2005 left Florida in an extended state of emergency.

Several factors can impact Florida's fuel supply and the 2004 and 2005 hurricane seasons caused several of them to surface. Reoccurring problems in the current fuel supply and distribution networks can be linked to the following:

• Storage capacity limitations in the Panhandle: In 2004, Hurricane Ivan severely damaged a Transmontaigne fuel terminal in Pensacola. The facility is still closed.

• Seaside re-supply disruptions: Since Florida is dependant upon re-supply via sea, long term port closures impact fuel availability.

• Electricity outages at supply terminals and retail stations: Widespread electricity outages plagued fuel distribution in Florida during the 2004 and 2005 hurricane seasons.

• Inadequate ground transportation of fuel: The 2004 and 2005 hurricane seasons proved that pre and post-storm activities caused demand for fuel to exceed supply at retail stations. A lack of fuel delivery trucks resulted in rapid draw down and retail facility outages.

To meet Florida's increasing demand for fuel, the DEP recommends facilitating the expansion of fuel infrastructure, exploring the feasibility of developing petroleum pipelines to create redundancies in fuel delivery mechanisms and increasing petroleum storage capacity at Florida's ports.

Recommendation #2

Encourage faishing stations ... to cooperatively adopted system moduled later the Florida WAROS system robic dagate the religention and give of generators to astrability service allopted encourt of the robic stations as preferipting on the network. Touch Statistication by three 1 2002 - Opparing on of Excommendal Place allows (hierdel Paroleum Enclosed) During the 2004 and 2005 hurricane seasons some fueling stations used generators to provide electric power to the station and the fueling pumps. This power enabled the stations to quickly return to operation and provide fuel. Fuel stations without generators were reliant on the restoration of local power.

Utilizing a system similar to the Florida WARN (Wastewater/water Agency Response Network) System will facilitate the installation of temporary power to return fuel to hurricane-affected communities without mandating the use of generators at gas stations or requiring government intervention in the market place. A web-based system would enable private entities to share resources. Generators in unaffected areas can be moved to impacted areas by private parties.

Conservation Recommendations

Recommendation #3 Foster state local parateliships to siteotrage well-designed music evolution and annels witered between a sublishiph community is individuent low community developments. Operationers of Environmental Photeeinon; Deparement of Community, Africe, Departmental?

herbacopereneteder

Smart growth is development that serves the economy, community and the environment. The features that distinguish smart growth in a community vary from place to place. In general, sustainable growth invests time, attention, and resources in restoring community and vitality to center cities and older suburbs. This growth tends to be more town-centered, is transit and pedestrian oriented, and has a greater mix of housing, commercial and retail uses. It also preserves open space and many other environmental amenities.

The DEP recommends exploring the potential for partnerships with local planning boards that foster smart growth, with emphasis on improved transportation and transit systems. The partnerships are not intended to create regulations or process but rather to facilitate information sharing about emerging technologies and existing infrastructure that reduce a community's dependence on fossil fuels.

The State should also facilitate continued discussions between the Florida Department of Transportation, local governments and the CSX Corporation to utilize existing freight tracks for commuter rail service in Central Florida. Mass transit projects like this aid in reducing congestion on Florida roads and provide a viable transportation alternative for commuters. Recommendation #4 Raise public awaren ess for alicenative hel vehicles through public programs. Encourage

schöch disences and tocal governments in aschörnel In deas ADepartmentor Enveronmental Protection)

Economic Incentives

-Rezonmendation#5

Provide grantheoding floorappheatresearch and demonstration proters associated with the development and inplemontation of alternative merging account option by 2007. The grant position of should realize an aggregate course on investment granted ball two recourse the premetted two recourse the premetted the articlemental Protections.

Provide saits and somprime income divide distant hydrogen whitely sand huding infersionicity, 5, 2007 function related for mobile hydrogen related by solution 50 percent. (Departmention Environmental Procession Displacement of Revenue) Numerous alternative fuel vehicle options exist, including hybrid electric, biodiesel, ethanol blends, electric, compressed natural gas, liquefied natural gas, propane and hydrogen.

Increased awareness about vehicle options will help foster demand for these alternative fuels and vehicles. Improved market penetration will help economies of scale. Government can lead by example and create market demand to help drive unit costs down.

The DEP is currently administering approximately \$7 million in grant funding to support transportation-related research, development and demonstrations in alternative fuels and advanced energy technologies in Florida. Additional funding is needed to build on these efforts and continue vital research and demonstration of these next generation technologies.

Industry and the State's universities contain the experience and expertise necessary to improve alternative fuels and energy technologies. By sponsoring further research and demonstration projects, the State can continue to bring these technologies closer to market.

Worldwide, energy companies, automakers and petroleum companies are investing more than \$2 billion annually to expand the hydrogen technology industry. Nationally, President George W. Bush proposed a \$1.7 billion, five-year investment to develop hydrogen-powered fuel cells, hydrogen infrastructure and advanced automotive technologies.

Since July 2003, the State has invested \$3.8 million for hydrogen research, development and demonstration and has experienced an overall return on investment of nearly four to one. Continued investment from the State will stimulate Florida's hydrogen economy, create incentives for corporate investment, diversify the economy and create high-wage jobs.



While biodiesel production capacity in Florida is growing, a large percentage of product is railed out-of-state.

Bulk storage capacity for biodiesel currently exists at the Port of Tampa and Port Everglades, although there are no existing plans for additional storage elsewhere in the state. To reduce costs and increase demand within the state, the DEP recommends expanding bulk storage at main petroleum terminals in Central and North Florida to allow fuel companies to blend biodiesel at the rack. By increasing availability and access, the state can enhance the market for biodiesel.

Fuel-grade ethanol production is growing in Florida and across the nation, although retail access to ethanol-blended fuel is limited. The DEP recommends using E85 (a blend of 85 percent ethanol and 15 percent gasoline) to fuel government fleets with flex-fuel vehicles, creating an immediate market for E85 and spurring private investment in infrastructure. The DEP also recommends exploring the potential for lower blends of ethanol, such as E10 (a blend of 10 percent ethanol with 90 percent gasoline), which may be viable for mainstream use without the need for flex-fuel vehicles.

Corporate sales and income tax credits will be offered to the production, storage and distribution of clean fuels. Investment from the State will stimulate investment in Florida, create incentives for corporate investment, diversify the economy and create high-wage jobs.

In addition to the economic incentives proposed by the State, the U.S. Department of Agriculture (USDA) has a comprehensive energy strategy to help farmers and ranchers deal with high energy costs. As part of its support for energy projects, the USDA awarded \$1.2 million in Small Business Innovation Research grants to eight projects involving biomass fuels, byproducts, and power. In addition, the USDA's Rural Development program plans to maximize its use of approximately \$1.4 billion in various business and electric loans and loan guarantees to help farmers, ranchers, and rural communities create renewable energy systems and businesses. This program provides guarantees up to eighty percent of a loan made by a commercial lender. The DEP recommends utilizing federal loans by Florida businesses to expand and enhance ethanol production in the state.





July 7, 2006

Gas Production Outlook: U.S. Onshore Mature Group

BCF/D	12-02	12-03	12-04	12-05	12-06
U.S. Dry Production	51.2	51.5	50.4	49.4	53.0
Onshore Total	37.9	39.1	39.9	41.2	43.7
Onshore Mature Group	22.6	23.5	23.7	24.4	26.3
Rockies/Other	15.3	15.5	16.1	16.8	17.4
Offshore GOM	13.3	12.4	10.6	8.2	9.4
May not sum to totals owing to roundi	ng	······			• • • • • • • • • • • • • • • • • • •

G as producers' unprecedented efforts to revitalize U.S. production have turned the corner. Despite the Onshore Mature Group's (MG) dramatic post-'02 rig count escalation, the region had little to show for its efforts outside a one-time rebound from price-depressed '02 levels that is, until the second half of last year. By 12-05, the Onshore MG's Y/Y growth had reached ~0.7 BCF/D (~0.9 BCF/D excluding lingering effects of last year's hurricanes).

Barring the curtailment of gas-oriented drilling from further summer '06 price erosion, we see no slowdown in sight thanks to the upward momentum of the Group's drilling intensity and its support from the NYMEX forward price curve. Indeed, PIRA's *Vintage Models* foresee Onshore MG Y/Y growth of 1.5-2.0 BCF/D by 12-06, sending a powerful message to cynics who gave up on the region's upside potential some time ago. Elsewhere, growth underpinned by the Rockies is forecast to push total onshore production higher Y/Y by 2.5 BCF/D.

Until the current quarter, massive offshore GOM losses largely attributable to hurricanes Ivan, Katrina and Rita had been surpassing onshore production gains. But we anticipate that offshore GOM output will turn positive Y/Y in 2H06, even in the face of another active hurricane season. If so, U.S. dry gas production would top the year-earlier level by a staggering 3-4 BCF/D in 12-06.

Since producers are not known for making speculative pipeline investments, the bevy of supply-driven pipeline projects moving forward adds credibility to our production outlook. The unmistakable fact that producers as opposed to LNG importers anchor the vast majority of projects, which overlay the Onshore MG, calls attention to producers' conviction that more supply-area pipeline capacity will be needed.

At the end of the day, U.S. gas demand will be hard-pressed to keep pace with such indigenous production growth. Ultimately, therefore, PIRA foresees the NYMEX "strip" giving domestic producers a more cautious price signal to bring the market into better balance.

PIRA employs multiple procedures to forecast near-term U.S. gas production. For Majors and Independent producers, we use drilling/output models that separately forecast gas output of each group. For the Onshore Mature Group, we use the *Vintage Models* discussed in this report that reflect correlations between rig counts and new gas together with decline rates of older wells.



PIRA Energy Group, 3 Park Avenue, 26th Floor, Ne For authorized users of PIRA Client S -686-6808 (Fax) 212-686-6628 www.pira.com ¹ Energy Group. All rights reserved.

July 7, 2006

Dry Gas Production (BCF/D)	12-02	12-03	12-04	12-05	12-06
Onshore Mature Group	22.58	23.49	23.66	24.38	26.28
Gulf Coast Area	10.39	10.75	10.66	10.91	. 11.63
Midcontinent	8,10	8.64	8.90	9.51	10,42
Permian Basin	4.10	4.10	4.10	3.96	4.23

May not sum to totals owing to rounding.

OVERVIEW

Vintage Model Area: Following substantial growth since '02, conventional wisdom would argue that onshore Mature Group gas producers are in danger of becoming victims of their own success, given the dynamics of the proverbial gas "production treadmill" (a.k.a. the *increasing* supply from new wells needed to replace annual declines from older wells). Yet, current year-to-date results suggest just the opposite. Namely, the region's current year output growth appears on track to surpass all prior annual gains since the '02 trough.

	Table 1:	Vintage I	Model D	rilling é	& Produc	tion
Dec.	Drill Inten.	New Wells	Decl. Rate	New Gas	Other Gas	Total Prod.
	Rigs	Thous.	%		BCFD	
02	428	6.9	41	4.05	13.07	17.12
03	500	8.3	23	4.87	13.12	18.00
04	606	10.2	26	4.99	13.27	18.26
05	697	11.2	28	5.89	13.18	19.07
06	794	12.1	26	6.43	14.16	20.59

In Vintage Model Areas, drilling intensity almost breached the 700 mark in 12-05, surpassing the prior pace by 15%. Now in its fourth year and on track to post at least comparable growth in '06, the ongoing drilling boom has become truly imposing in both strength and duration. Model Area new gas wells are also on track to reach yet another all-time high.

Domestic producers have left none of the onshore MG's sub-regions behind. While new plays are enjoying a flurry of activity, traditional basins are also benefiting from a second wind. Indeed, the acceleration in drilling activity at work since 2003 has been observed in all Model Areas. Those "sub-regions" have been put together largely on the basis of homogeneous gas supply and location. They include the Texas Gulf Coast (railroad districts 1-4, and 6), north and south Louisiana, the Barnett Shale Area (RRD 5, 7B, 9), Oklahoma and RRD 10 in the Mid-Continent and RRD 7C in the Permian Basin¹.

Table 2: Vintage Model Area Drilling Efficiency								
Dec.	02	03	04	05	06			
MMCFD/Well	0.59	0.59	0.49	0.53	0.53			
Wells/Rig/Yr	16.0	16.6	16.9	16.1	15.3			
MMCFD/Rig	9.5	9.8	8.2	8.5	8.1			

In '05, overall Model Area drilling efficiency increased in terms of new gas delivered per active rig. Although the number of new wells per rig declined (16.9 to 16.1), that reduction was more than offset by improved first-year well productivity (490 to 530 MCF/D per well). Rig efficiency gains stemmed from the dominance of incremental drilling in high well productivity sub-regions (e.g. Texas Gulf, N. Louisiana) and little or no growth in low well productivity sub-regions (e.g. Oklahoma, RRD 7).

The combination of improved rig efficiency and more intensive drilling intensity pushed "new gas" in 12-05 up to 5.9 BCF/D, a stunning 0.9 BCF/D jump from the year-earlier level. However, "other gas" (largely from older wells) stagnated at 13.2-13.3 BCF/D, reflecting only modest Y/Y baseline expansion (18.0 to 18.3 BCF/D) and a faster decline rate (26 to 28%).

¹ Please refer to the Glossary of Terms on page 12 that defines "new gas" and other terms used in our vintage model analysis and map on page 11 that locates each sub-region.

July 7, 2006

For 2006, the Model Area's performance is shaping up to be a dramatically different story. We foresee "new gas" growth to reach only ~0.5 BCF/D but "other gas" expand by 1.0 BCF/D, causing total Model Area supply to jump 1.5 BCF/D. Slower "new gas" growth stems from a further contraction in the annual number of wells per rig without offsetting growth of output per well. Thus, annual rig efficiency falls from 8.5 to 8.1 MMCF/D per rig.

"Other gas" growth in '06 of 1.0 BCF/D provides the critical underpinning for robust overall output expansion. From 12-05 to 12-06, "other gas" will be the beneficiary of both higher drilling intensity and a *slower* annual decline rate (28 to 26%). The slower decline rate stems from the anomaly of depressed Louisiana output in 12-05 caused by the lingering effects of last year's hurricanes.

Other Mature Group Areas: In some sub-regions, available data are inadequate to capture the structural dynamics behind gas production trends (Alabama/Florida and Mississippi in the Gulf Coast, as well as Arkansas and Kansas in the Midcontinent). In S.E. New Mexico and RRD 8/8A in the Permian Basin gas-oriented drilling efficiency cannot be measured realistically owing to the dominance of associated gas.

Table 3: Mature Group Outside Model Areas (BCFD)									
Prod. Dec.	Gulf	Midcon	Permian	Total					
02	.66	1.53	3.27	5.46					
03	.76	1.48	.3.25	5.49					
04	.74	1.48	3.18	5.40					
05	.80	1.46	3.05	5.31					
06	.81	1.55	3.33	5.71					

In '06, we expect total production in these regions to expand by 0.4 BCF/D. Standouts for expected growth within this group are Arkansas (the Arkoma Basin's unconventional shale play) and Southeastern New Mexico (sub-normal 12-05 output).

GULF COAST AREA

<u>**Texas Gulf Area</u>**: Albeit in a less spectacular way than some other regions, the Texas onshore GOM has shown a major revival in drilling intensity over</u>



the past several years. In 12-05, 12-month drilling intensity (261) surpassed 12-04 (208) and 12-03 (192) by a 25% and 36%, respectively².

However, high "other gas" decline rates have speeded the region's production "treadmill," curbing the impact of new gas on total production growth. In addition, the systematic fall in rig efficiency have mitigated the positive impact of more intensive gasoriented drilling. Those declines of course can be anticipated when rising drilling intensity reduces prospect quality.

Table 4: Texas Gulf Area (RRD 1-4, 6)									
DEC.	02	03	04	05	06				
BCFD Sum	6.89	7.11	6.94	7.12	7.39				
% Decl. Yr.	32	29	33	35	34				
BCFD New	1.81	2.23	2.15	2.62	2.69				
Dril. Inten.	173	192	208	261	269				
MMCFD/Rig	10.4	11.6	10.4	10.0	10.0				
Wells/Rig/Yr	14	15	16	16	16				
MCFD/Well	772	749	658	628	625				

In '06, output is projected to reach \sim 7.4 BCF/D, a Y/Y increase of \sim 250 MMCF/D, or slightly more than last year's growth. The forecast acceleration can be attributed to continued aggressive drilling expansion with 12-06 drilling intensity reaching only 269 rigs, coupled with a rising baseline and fairly stable decline rates.

Southern Louisiana: Thanks to record-high gas prices, ready access to pipelines and a strong oil and gas industry culture, Louisiana has become, once again, an aggressively targeted E&P area. Within the State, however, major differences are evident in recent trends as well as in the outlook for gas production.

Output in southern Louisiana (slightly over one-half of the State's total) has been generally on the decline over the past five years. Until '05, drilling intensity as well as gas rig efficiency had remained fairly stable. In 12-05, however, the positive impact on

 $^{^2\,}$ Please refer to the tables on pages 8-9 for additional details on Vintage Model Area drilling and production from 12-02.

July 7, 2006

production of expanded drilling intensity was undermined by artificially high decline rates caused by earlier hurricane activity. Although a drilling slowdown seems in the making for the current year, overall gas supply in 12-06 should easily top the 12-05 level as output "normalizes" from those hurricane-distorted year-earlier levels.

Table 5: Southern Louisiana									
DEC.	02	03	04	05	06				
BCFD Sum	1.98	1.95	1.82	- 1 .63 -	1.80				
% Decl. Yr.	34	28	31	40	20				
BCFD New	.48	.52	.47	.53	.50				
Dril. Inten.	44	42	43	51	47				
MMCFD/Rig	10.9	12.4	10.9	10.5	10.5				
Wells/Rig/Yr	6	7	8	7	7				
MMCFD/Well	1.86	1.79	1.44	1.50	1.50				

Northern Louisiana: In sharp contrast to the southern part of the State, this area has achieved consistent growth over the past three years. Here, more intensive drilling has not overtly influenced rig efficiency. With current rig counts leading their 12month moving averages, the region's drilling intensity is set to escalate thanks to prospects in the expanded Cotton Valley play and the prolific Vernon field areas.

Table 6: Northern Louisiana									
DEC.	02	03	04	05	06				
BCFD Sum	0.86	0.93	1.15	1.36	. 1.63				
% Decl. Yr.	29	26	30	30	27				
BCFD New	.22	.30	.50	.55	.65				
Dril. Inten.	23	26	36	46	53				
MMCFD/Rig	9.9	11.6	13.8	12.0	12.2				
Wells/Rig/Yr	15	16	18	17	17				
MCFD/Well	677	731	763	706	718				

Moreover, output "normalization" from hurricanedistorted 12-05 levels should further boost current year growth, pushing overall supply up to \sim 1.60 BCF/D, up almost 0.3 BCF/D Y/Y.

<u>Gulf Coast Outside Model Areas</u>: In <u>Alabama</u> production has been remarkably stable of late owing



NATURAL GAS

to the counter-balancing impacts of modestly rising coalbed methane (CBM) from Black Warrior Basin and declining conventional gas output. CBM accounts for about 70% of the State's total production and should continue to expand given rising proven reserves. Although the opposite is true for conventional gas, future declines will reflect a baseline only slightly above 0.1 BCF/D. Dry gas output in **Florida** is only ~5 MMCF/D.

Table 7: Gulf Coast Outside Model Areas (BCF/D)									
DEC.	02	03	04	05	06				
Sum	.66	.76	.74	.80	.81 5				
Alabama/Fla.	.38	.43	.40	.40	.41				
Miss.	.28	.33	.34	.40	.40				

In recent years, <u>Mississippi</u> dry gas production has grown steadily. In 2005, output reached its highest annual average (386 MMCF/D) since the early 1980s and topped the prior year (361) by a comfortable margin. However, so far in 2006 the State's rig counts have fallen from year-earlier highs making sustained growth over the near-term problematic.

MIDCONTINENT

Barnett Shale Area: Located in northern Texas (RRD 5, 7B and 9), the Barnett Shale Area is the principal driver of production growth in both the State of Texas and the onshore MG as a whole. In contrast with most other plays, this expansion is being achieved through more intensive drilling without a major offset from reduced rig efficiency.

The quasi-constant flow of corporate announcements coming from the region confirms the sustainability of this upward trend. *And the numerous planned pipeline additions surrounding this play bolster our belief in the upside potential, too*. Pipeline capacity additions on the drawing board are rapidly becoming almost too many to list.

More than 100 companies are active in the Barnett Shale area. Devon, the largest player, has \sim 2,200 wells producing \sim 0.6 BCF/D. The company completed \sim 210 wells in 2005, of which \sim 150 were horizontal. Besides the 280 new wells planned for

July 7, 2006

2006, Devon plans to drill another 800 wells within five years in acreage recently acquired from Chief Oil and Gas. Other major players with active development plans include XTO Energy, ConocoPhillips, EnCana, EOG and Chesapeake. Such intense competition will play an important role behind the area's future growth.

Table 8: Barnett Shale Area (TX RRD 5, 7b, 9)								
DEC.	02	03	04	05	06			
BCFD Sum	1.74	2.14	2.16	- 2.63	23.17			
% Decl. Yr.	25	22	34	25	32			
BCFD New	.61	.78	.75	1.01	1.38			
Dril. Inten.	68	82	100	106	184			
MMCFD/Rig	8.9	9.6	7.5	9.5	7.5			
Wells/Rig/Yr	21	21	17	17	14			
MCFD/Well	425	464	443	559	536			

Improving technology and geological knowledge keep pushing the Shale's potential higher, but not all acreage holds equal potential. In 2005, ~60% of the Shale's production came out of District 9. The most actively drilled area is the Newark East field, listed as the largest in Texas by the *American Association* of *Petroleum Geologists*, and located across RRD 5's Tarrant and RRD 9's Denton counties. Tarrant County was producing ~0.3 BCF/D by end-2005, with neighboring Johnson County becoming the next "sweet spot" in the play.

Outside of these "core areas," important unknowns remain with respect to where the gas window ends in relation to future recovery factors and decline rates. Yet, large players are accelerating drilling intensity in part to hold their leases and increase their acreage. In such an intensely competitive environment, PIRA foresees new well startups in the Area soaring from ~1,800 in '05 to 2,400 in the current year, contributing to Y/Y overall production growth of ~0.5 BCF/D by 12-06. Alternatively, more stable drilling efficiency would translate into even higher production.

Oklahoma: In 2006, Oklahoma rig counts and gas production are on an upswing. The EIA's new Form-914 survey indicates that production was up

Y/Y by 4% in April, a visible improvement over relatively stagnant output during the prior year. May '06 rig counts were up to a record post-90s high of 165, an 11% Y/Y gain.

Table 9: Oklahoma									
DEC.	02	03	04	05	06				
BCFD Sum	4.03	4.14	4.34	4.36	4.52				
% Decl. Yr.	22	15	13	16	14				
BCFD New	.72	.73	.72	.70	.76				
Dril. Inten.	90	116	153	150	159				
MMCFD/Rig	8.1	6.3	4.7	4.6	4.8				
Wells/Rig/Yr	18	16	17	15	15				
MCFD/Well	446	386	276	310	320				

Although still in their infancy, the Woodford Shale and Caney plays on the Oklahoma (western) side of the Arkoma Basin have become a focus of attention for several producers. Similarities to Barnett Shale are partly responsible. Newfield has launched a Woodford Shale horizontal drilling program with some suggesting that output could reach 0.2 BCF/D by 2008. Devon also is getting active in the area.

RRD 10: During the past two years, RRD 10 production has trended upward from 0.9 BCF/D to slightly over 1.0 BCF/D by 12-05. From 2H05 into early '06, the area's production appears to have reached a plateau, but its drilling began a second round of escalation in late '05 leading to a May '06 monthly rig count of 67. Given this drilling resurgence, we foresee the district's gas output reaching 1.2 BCF/D in 12-06.

Table 10: RRD 10									
DEC.	02	03	04	05	06				
BCFD Sum -		.87	.92	1.06	1.18×				
% Decl. Yr.	19	12	18	18	19				
BCFD New	.07	.17	.21	.31	.32				
Dril. Inten.	13	22	40	57	65				
MMCFD/Rig	5.7	7.7	5.2	5.4	5.0				
Wells/Rig/Yr	23	20	16	14	14				
MCFD/Well	248	395	327	386	357				

PIRA Energy Group, 3 Park Avenue, 26th Floor, New York, NY 10016-5989 (Tel) 212-686-6808 (Fax) 212-686-6628 www.pira.com 5 For authorized users of PIRA Client Services. Copyright © 2006 PIRA Energy Group. All rights reserved.



PIRA

July 7, 2006

A particularly active area within the District has been the Granite Wash field, with wells reaching depths of 11,000 to 14,000 feet. Owing to the depths and complexity of zones being drilled, this play needed the higher gas price thresholds of recent years to become viably economic despite relatively high initial production rates.

<u>Midcontinent Outside Model Areas</u>: The two states included in this subdivision could not be more contradictory. In <u>Kansas</u>, production remains on a long-term downtrend largely reflecting the maturity of Hugoton Gas Area that extends into Oklahoma and Texas. In 2004, the Area produced ~ 0.9 BCF/D, down from ~ 1.1 BCF/D only two years earlier. Nevertheless, declines have moderated over the past year or so.

Table 1	1: Midco	ntinent C	utside 1	Model Ar	eas
DEC.	02	03	04	05	06
Sum	1.53	1.48	1.48	1.46	1.55
Kansas	1.08	1.00	0.94	0.93	0.93
Arkansas	0.45	0.48	0.53	0.53	0.62

At the other extreme, <u>Arkansas</u> gas rigs have leaped from under 5 to over 20 within the past year and further escalation is in the making. The impetus in this case is the Fayetteville Shale play on the Arkansas (eastern) side of the Arkoma Basin with key companies that include Southwestern Energy, Chesapeake, and Shell.

At this juncture, Southwestern Energy looks to be in the lead position to expand gas output with a massive leasehold position of 860,000 acres. Chesapeake holds the largest leasehold position (1.1 million acres) but has just started E&P activities in the area.

In May, the Arkansas Oil and Gas Commission projected that statewide gas output could jump from 0.5 to 0.8 BCF/D within two years assuming the aggressive development of Fayetteville Shale. Earlier this month, Arkansas adopted a new set of oil and gas leasing rules in response to the play's growing prominence.



NATURAL GAS

Despite the Shale's similar geological characteristics to Barnett Shale, Southwestern has indicated that successful development requires significantly more than mimicking Barnett frac technology. PIRA will closely monitor this play given its upside potential and lead-time uncertainties.

PERMIAN BASIN

<u>RRD 7C</u>: This Permian Basin Texas District is somewhat unique in that its production is heavily gas-oriented. Three of its top four producing counties (including top-producing Crockett County), in fact, derive close to 90% of total gas output from gas wells. Four contiguous counties of Crockett, Sutton, Terrell, and Upton produce over 80% of the District's gas.

	Tal	ble 12: R	RD 7C		
DEC.	02	03	04	05	06
BCFD Sum			.92	.91	.90
% Decl. Yr.	16	15	14	19	16
BCFD New	.14	.15	.19	.17	.14
Dril. Inten.	18	20	25	26	18
MMCFD/Rig	7.7	7.4	7.5	6.5	7.5
Wells/Rig/Yr	33	31	41	40	40
MCFD/Well	231	238	183	163	188

High gas prices have had a substantial impact on this area's highly gas-oriented E&P activities. However, production has not been commensurate with the aggressive drilling. Escalating "other gas" decline rates combined with falling rig productivity have limited output growth and should continue to dampen growth prospects in the months ahead.

After peaking ~ 0.9 BCF/D in '04, the District's gas production held steady into early '06. During the remainder of the year, output is projected to trend downward in response to weaker drilling intensity.

Permian Outside Model Areas: A high proportion of Permian Basin gas is associated with oil production; so high crude prices encourage oil drilling and its associated gas production. Oil's dominance is especially evident in RRD 8/8A and south eastern New Mexico.

July 7, 2006

<u>RRD 8 and 8A</u> show a precipitous decline in gas production over the past year due to an accounting

Table 13: I	Permian 1	Basin O	utside M	odel Ar	eas
DEC.	02	03	04	05	06
Sum	3.27	3.25	3.18	3.05	3.33
RRD 8/8A	1.82	1.79	1.75	1.71	1.83
S.E. New Mex.	1.46	1.45	1.43	1.34	1.50

inconsistency in District 8A regarding CO_2 injection. Our data, however, after adjustments for this inconsistency, reveal the Districts' gas production resilience in the 1.7-1.8 BCF/D area in recent years,



NATURAL GAS

with an almost even split between oil and gas wells. In this case, the oil *and* gas booms seem to be providing an equal dose of E&P incentives.

In <u>Southeastern New Mexico</u> gas production appears to have started '05 at almost 1.5 BCF/D before tailing off to 1.3-1.4 BCF/D by the fourth quarter. In 3Q05, however, the sub-region's rig counts began to move aggressively higher, starting from the mid-40s and reaching the mid-60s by May '06. Production also appears to have moved back into the 1.4-1.5 BCF/D range in 2Q06.



N 6 6 5 6		VINTA	GE MODE	EL DRILLING	& PROD	UCTION	en de la sec		
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Rig	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	428	6,856	16.0	4.05	591	9.5	38	13.07	17.12
Jun-03	439	7,229	16.5	4.68	647	10.7	35	13.27	17.95
Dec-03	500	8,300	16.6	4.87	587	9.8	23	13.12	18.00
Jun-04	559	9,439	16.9	5.00	530	8.9	25	13.45	18.45
Dec-04	606	10,224	16.9	4.99	488	8.2	26	13.27	18.26
Jun-05	653	10,385	15.9	5.27	507	8.1	26	13.68	18.95
Dec-05	697	11,197	16.1	5.89	526	8.5	28	13.18	19.07
Jun-06	745	11,347	15.2	6.18	544	8.3	27	13.93	20.10
Dec-06	794	12,120	15.3	6.43	530	8.1	26	14.16	20.59

- 学校会			GULF CO	AST TEXAS	RRD 1-4	, 6)			
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Rig	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	173	2,345	14	1.81	772	10.4	32	5.08	6.89
Jun-03	169	2,571	15	2.23	867	13.2	29	4.94	7.17
Dec-03	192	2,976	15	2.23	749	11.6	29	4.89	7.11
Jun-04	206	3,153	15	2.28	724	11.1	32	4.91	7.19
Dec-04	208	3,270	16	2.15	658	10.4	33	4.79	6.94
Jun-05	228	3,486	15	2.21	634	9.7	33	4.84	7.05
Dec-05	261	4,173	16	2.62	628	10.0	35	4.51	7.12
Jun-06	271	4,067	15	2.71	667	10.0	34	4.65	7.36
Dec-06	269	4,296	16	2.69	625	10.0	34	4.70	7.39

			NOR	THERN LOUI	SIANA			S. (1998)	1990 - M. 1980 - 1
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Rig	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	23	329	15	0.22	677	9,9	29	0.63	0.86
Jun-03	23	360	16	0.31	855	13.4	25	0.65	0.96
Dec-03	26	406	16	0.30	731	11.6	26	0.64	0.93
Jun-04	31	518	16	0.39	753	12.4	31	0.66	1.05
Dec-04	36	653	18	0.50	763	13.8	30	0.65	1.15
Jun-05	41	686	17	0.49	715	12.1	29	0.74	1.23
Dec-05	46	785	17	0.55	706	12.0	30	0.80	1.36
Jun-06	50	854	17	0.61	718	12.2	27	0.90	1.51
Dec-06	53	900	17	0.65	718	12.2	27	0.98	1.63

an an tha tha	1		SOU	THERN LOUI	SIANA	an ta an an	an a		요즘 말을 했다.
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Weil	MMCF/D Per Ria	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	44	258	6	0.48	1863	10.9	34	1.50	1.98
Jun-03	45	261	6	0.46	1752	10.2	25	1.64	2.10
Dec-03	42	293	7	0.52	1785	12.4	28	1.42	1.95
Jun-04	40	305	8	0.48	1577	11.9	29	1.49	1.98
Dec-04	43	328	8	0.47	1441	10.9	31	1.35	1.82
Jun-05	48	317	7	0.50	1574	10.4	29	1.40	1.90
Dec-05	51	356	7	0.53	1500	10.5	40	1.10	1.63
Jun-06	50	350	7	0.52	1500	10.5	30	1.33	1.85
Dec-06	47	332	7	0.50	1500	10.5	20	1.30	1.80

PIRA Energy Group, 3 Park Avenue, 26th Floor, New York, NY 10016-5989 (Tel) 212-686-6808 (Fax) 212-686-6628 www.pira.com ⁸ For authorized users of PIRA Client Services. Copyright © 2006 PIRA Energy Group. All rights reserved.



(PIRA



NATURAL GAS

July 7, 2006

	ТХ	RRD 9, 7B &	5 (Inclus	ive of entire	Barnett S	Shale pro	duction		
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Rig	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	68	1,426	21	0.61	425	8.9	25	1.13	1.74
Jun-03	67	1,460	22	0.68	463	10.1	21	1.31	1.99
Dec-03	82	1,689	21	0.78	464	9.6	22	1.36	2.14
Jun-04	93	1,759	19	0.73	417	7.9	25	1.48	2.22
Dec-04	100	1,690	17	0.75	443	7.5	34	1.41	2.16
Jun-05	102	1,691	17	0.90	530	8.8	25	1.67	2.56
Dec-05	106	1,810	17	1.01	559	9.5	25	1.62	2.63
Jun-06	139	2,089	15	1.11	533	8.0	32	1.75	2.86
Dec-06	184	2,570	14	1.38	536	7.5	32	1.79	3.17

4,35	 The sequence 		KLAHON	IA (Part of M	dcontine	nt)			
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Ria	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	90	1,620	18	0.72	446	8.1	22	3.31	4.03
Jun-03	100	1,653	17	0.75	454	7.5	19	3.31	4.06
Dec-03	116	1,881	16	0.73	386	6.3	15	3.42	4.14
Jun-04	136	2,450	18	0.75	305	5.5	15	3.47	4.21
Dec-04	153	2,621	17	0.72	276	4.7	13	3.62	4.34
Jun-05	158	2,421	15	0.75	309	4.7	14	3.63	4.38
Dec-05	150	2,252	15	0.70	310	4.6	16	3.66	4.36
Jun-06	151	2,268	15	0.73	320	4.8	14	3.78	4.50
Dec-06	159	2,384	15	0.76	320	4.8	14	3.76	4.52

		an an an Arr	X RRD 10) (Part of Mid	-Contine	nt)			
	Drilling Intensity	New Wells	Wells Per Ria	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Ria	Decline Rate %	Other Gas BCF/D	Total Prod. BCE/D
Dec-02	13	292	23	0.07	248	5.7	19	0.73	0.80
Jun-03	16	354	21	0.12	331	7.1	16	0.72	0.84
Dec-03	22	423	20	0.17	395	7.7	12	0.70	0.87
Jun-04	30	508	17	0.21	410	6.9	14	0.73	0.93
Dec-04	40	634	16	0.21	327	5.2	18	0.71	0.92
Jun-05	50	680	14	0.23	342	4.7	25	0.70	0.93
Dec-05	57	800	14	0.31	386	5.4	18	0.75	1.06
Jun-06	61	856	14	0.33	386	5.4	17	0.77	1.10
Dec-06	65	905	14	0.32	357	5.0	19	0.86	1.18

		1. 1993 - 199 7	X RRD 70	C (Part of Per	mian Bas	sin)			$C = C^{(1)}$
	Drilling Intensity	New Wells	Wells Per Rig	New Non-Asso. Gas BCF/D	MCF/D Per Well	MMCF/D Per Rig	Decline Rate %	Other Gas BCF/D	Total Prod. BCF/D
Dec-02	18	586	33	0.14	231	7.7	16	0.69	0.82
Jun-03	19	570	31	0.14	245	7.5	14	0.70	0,84
Dec-03	20	632	31	0.15	238	7.4	15	0.70	0.85
Jun-04	22	746	34	0.16	208	7.0	15	0.71	0.86
Dec-04	25	1,028	41	0.19	183	7.5	14	0.73	0.92
Jun-05	28	1,104	40	0.19	175	7.0	18	0.71	0.90
Dec-05	26	1,020	40	0.17	163	6.5	19	0.75	0.91
Jun-06	22	862	40	0.16	184	7.4	16	0.76	0.92
Dec-06	18	732	40	0.14	188	7.5	16	0.76	0.90

July 7, 2006

.

	ONSH	ORE MATURE	GROUP GAS	RIGS	
12 Mo. Avg.	Sum	Texas Gulf	North	South	Barnett
		Coast	Louisiana	Louisiana	Area
Dec-02	588	165	23	44	65
Dec-03	/ 09	203	28	41	100
Dec-04	932	215	30	40	102
Dec-03	10-1-1	209	40	52	121
By Month				10	
Jan-05	967	240	39	46	97
Feb-US	994	245	45	56	101
Mar-05	1009	201	49	0/	111
May-05	1035	277	44	47	112
Jun-05	1020	202	44	47	107
Jul-05	1046	280	48	52	112
Aug-05	1069	280	53	56	116
Sep-05	1070	283	53	53	114
Oct-05	1098	274	50	48	150
Nov-05	1094	266	50	52	161
Dec-05	1098	263	48	49	160
Jan-06	1090	248	54	47	166
Feb-06	1142	259	56	48	178
Mar-06	1123	261	55	46	184
Apr-06	1170	273	55	44	198
May-06	1197	276	57	44	201
<u>12 Mo. Avg.</u>	Midcon.	Permian	Oklahoma	TX RRD 10	TX RRD 7C
<u>12 Mo. Avg.</u> Dec-02	Midcon.	Permian 59	Oklahoma 91	TX RRD 10 14	TX RRD 7C 18
<u>12 Mo. Avg.</u> Dec-02 Dec-03	Midcon. 111 159	Permian 59 77	Oklahoma 91 127	TX RRD 10 14 25	TX RRD 7C 18 21
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04	Midcon. 111 159 212	Permian 59 77 91	Oklahoma 91 127 157	TX RRD 10 14 25 45	TX RRD 7C 18 21 27
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05	Midcon. 111 159 212 223	Permian 59 77 91 98	Oklahoma 91 127 157 150	TX RRD 10 14 25 45 60	TX RRD 7C 18 21 27 23
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u>	Midcon. 111 159 212 223	Permian 59 77 91 98	Oklahoma 91 127 157 150	TX RRD 10 14 25 45 60	TX RRD 7C 18 21 27 23
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05	Midcon. 111 159 212 223 218	Permian 59 77 91 98 88	Oklahoma 91 127 157 150 149	TX RRD 10 14 25 45 60 60	TX RRD 7C 18 21 27 23 30
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05	Midcon. 111 159 212 223 218 218 218	Permian 59 77 91 98 88 88	Oklahoma 91 127 157 150 149 154	TX RRD 10 14 25 45 60 60 59	TX RRD 7C 18 21 27 23 30 28
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05	Midcon. 111 159 212 223 218 218 218 218	Permian 59 77 91 98 88 89 89	Oklahoma 91 127 157 150 149 154 149	TX RRD 10 14 25 45 60 59 55	TX RRD 7C 18 21 27 23 30 28 26
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 Mar-05	Midcon. 111 159 212 223 218 218 218 211 219	Permian 59 77 91 98 88 89 89 92 92	Oklahoma 91 127 157 150 149 154 149 152	TX RRD 10 14 25 45 60 59 55 58	TX RRD 7C 18 21 27 23 30 28 26 27 30 28 26 27 23 30 28 26 27 23 30 28 26 27 27 23 20 20 20 20 20 20 20 20 20 20
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 May-05	Midcon. 111 159 212 223 218 218 218 211 219 215 225	Permian 59 77 91 98 88 89 89 92 92 92	Oklahoma 91 127 157 150 149 154 149 152 148	TX RRD 10 14 25 45 60 59 55 58 58 58 58	TX RRD 7C 18 21 27 23 30 28 26 27 22 22 22 22 22 22 22 22 22
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 May-05 Jun-05 Jun-05	Midcon. 111 159 212 223 218 218 218 211 219 215 225 225	Permian 59 77 91 98 88 89 89 92 92 92 90 90	Oklahoma 91 127 157 150 149 154 149 152 148 148	TX RRD 10 14 25 45 60 59 55 58 58 58 58 58 58	TX RRD 7C 18 21 27 23 30 28 26 27 22 22 22 22 24
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 May-05 Jun-05 Jun-05 Apr-05	Midcon. 1111 159 212 223 218 218 218 211 219 215 225 225 225 225 225	Permian 59 77 91 98 88 89 89 92 92 92 92 92 91 98	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 148	TX RRD 10 14 25 45 60 59 55 58 58 62 62 62 57	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 24 21 21 21 22 24 21 22 24 21 21 23 24 24 27 23 24 25 25 25 25 25 25 25 25 25 25
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 Jun-05 Jun-05 Aug-05 Sen-05	Midcon. 1111 159 212 223 218 218 211 219 215 225 225 225 225 225 225 228 228	Permian 59 77 91 98 88 89 89 92 92 92 92 90 98 104	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 148	TX RRD 10 14 25 45 60 59 55 58 58 58 62 62 62 62 57 61	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 21 23 30 28 26 27 22 24 21 23 30 28 26 27 22 22 24 21 27 23 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 27 23 26 26 27 27 23 26 26 27 27 23 26 27 27 23 26 26 27 27 27 27 28 26 27 27 27 27 27 28 26 27 27 27 22 22 22 22 24 22 22 22 22 24 23 23 26 27 22 22 22 22 23 23 23 26 27 22 22 22 24 22 22 24 23 24 23 26 27 22 22 24 23 24 23 24 23 24 23 24 21 22 22 24 23 23 23 24 23 24 23 24 24 21 22 22 24 23 23 23 24 23 24 23 24 24 23 23 24 24 23 23 24 23 24 23 24 23 24 23 23 24 23 23 24 23 23 23 24 23 23 23 23 24 23 23 24 23 24 23 23 24 23 24 23 23 24 23 24 23 24 23 24 24 25 25 25 25 25 25 25 25 25 25
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Mar-05 Mar-05 May-05 Jun-05 Jun-05 Sep-05 Oct-05	Midcon. 1111 159 212 223 218 218 211 219 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 89 92 92 92 90 98 104 114 104	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 146 154	TX RRD 10 14 25 45 60 59 55 58 58 58 62 62 62 57 61 62	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 21 23 21
<u>12 Mo. Avq.</u> Dec-02 Dec-03 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 Jun-05 Jun-05 Jul-05 Sep-05 Oct-05 Nov-05	Midcon. 111 159 212 223 218 218 218 218 218 218 213 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 89 92 92 92 92 90 98 104 114 109 106	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 146 154 151 149	TX RRD 10 14 25 45 60 59 55 58 58 58 62 62 57 61 62 57 61 62 63	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 21 23 21 18
<u>12 Mo. Avq.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 Jun-05 Jun-05 Jul-05 Sep-05 Oct-05 Nov-05 Dec-05	Midcon. 111 159 212 223 218 218 218 218 218 218 213 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 89 92 92 92 92 90 98 104 114 109	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 146 154 151 149 150	TX RRD 10 14 25 45 60 59 55 58 62 62 62 57 61 62 63 63 65	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 21 23 21 18 20
<u>12 Mo. Avg.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 May-05 Jun-05 Jul-05 Sep-05 Oct-05 Nov-05 Dec-05 Jan-06	Midcon. 111 159 212 223 218 218 218 218 211 219 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 92 92 92 92 92 92 92 91 104 114 109 106 109	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 148 154 154 151 149 150 149	TX RRD 10 14 25 45 60 59 55 58 62 62 62 62 57 61 62 63 65 63	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 21 23 24 21 23 24 21 23 20 20 20 20 20 20 20 20 20 20
12 Mo. Avg. Dec-02 Dec-03 Dec-04 Dec-05 By Month Jan-05 Feb-05 Mar-05 Apr-05 Jun-05 Jun-05 Sep-05 Oct-05 Nov-05 Dec-05 Jan-06 Feb-06	Midcon. 111 159 212 223 218 218 218 218 211 219 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 89 92 92 92 92 90 98 104 114 109 106 109 112	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 146 151 149 150 150 149 150	TX RRD 10 14 25 45 60 59 55 58 62 62 62 62 57 61 62 63 62 62 62 62 57 61 62 63 63 65 63 63	TX RRD 7C 18 21 27 23 30 28 26 27 22 22 24 21 23 21 18 21 23 21 23 21 22 24 21 23 21 23 20 20 20 20 20 20 20 20 20 20
<u>12 Mo. Avq.</u> Dec-02 Dec-03 Dec-04 Dec-05 <u>By Month</u> Jan-05 Feb-05 Mar-05 Apr-05 Jun-05 Jun-05 Jul-05 Sep-05 Oct-05 Nov-05 Dec-05 Jan-06 Feb-06 Mar-06	Midcon. 111 159 212 223 218 211 219 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 92 92 92 90 92 90 98 104 114 106 109 112 107 95	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 148 146 154 146 151 149 150 149 163 158	TX RRD 10 14 25 45 60 59 55 58 62 62 62 57 61 62 62 62 62 62 62 62 62 62 62 62 62 62	TX RRD 7C 18 21 27 23 30 28 26 27 22 22 24 21 23 21 18 20 20 20 23 21
12 Mo. Avg. Dec-02 Dec-03 Dec-04 Dec-05 By Month Jan-05 Feb-05 Mar-05 Apr-05 Jun-05 Jun-05 Jul-05 Sep-05 Oct-05 Dec-05 Dec-05 Jan-06 Feb-06 Mar-06 Apr-06	Midcon. 111 159 212 223 218 211 219 215 225 225 225 225 225 225 225	Permian 59 77 91 98 88 89 92 92 92 90 92 90 92 90 91 14 109 109 109 112 107 95 111	Oklahoma 91 127 157 150 149 154 149 152 148 148 148 146 154 146 154 149 150 149 150 149 163 158 160	TX RRD 10 14 25 45 60 59 55 58 62 62 62 57 61 62 63 63 63 63 63 63 63 62 63 63 63 63 63 63 63 63 63 63 63 63 63	TX RRD 7C 18 21 27 23 30 28 26 27 22 24 21 18 20 20 23 21 17

PIRA Energy Group, 3 Park Avenue, 26th Floor, New York, NY 10016-5989 (Tel) 212-686-6808 (Fax) 212-686-6628 www.pira.com ¹⁰ For authorized users of PIRA Client Services. Copyright © 2006 PIRA Energy Group. All rights reserved.



PIRA

July 7, 2006

PIRA NATURAL GAS



PIRA Energy Group, 3 Park Avenue, 26th Floor, New York, NY 10016-5989 (Tel) 212-686-6808 (Fax) 212-686-6628 www.pira.com ¹¹ For authorized users of PIRA Client Services. Copyright © 2006 PIRA Energy Group. All rights reserved.

July 7, 2006



Glossary Of Terms

Decline rates are measures of total production losses over one year from producing wells. Decline rates vary over time and across regions. For example, in the shallow water GOM, gas wells beyond their first year of production collectively exhibit decline rates in excess of 40%. By comparison, onshore gas wells collectively reflect decline rates in the vicinity of 30%, excluding wells in their first year of production.

Drilling intensity refers to 12-month moving average gas-oriented rig counts behind new gas deliverability. PIRA's models typically lag drilling intensity relative to output from first-year wells by three months to account for the time lapse between drilling and initial production.

Dry gas production measures the amount of produced consumer-grade natural gas and equals marketed gas production less extraction loss.

Marketed production refers to total reservoir withdrawals less gas used for repressuring, quantities vented and flared, and non-hyrdocarbon gases removed in the treating or processing operations. Includes all quantities of gas used in field and processing plant operations.

New gas is a measure of the average production from wells started within the past 12-month period. New gas production is a function of drilling intensity multiplied by rig efficiency.

Old gas is a measure of production from wells older than one year and is a function of year-ago total gas and annual decline rates.

Onshore Mature Group (MG) regions include the Onshore Gulf of Mexico, Midcontinent and Permian Basin. Together, the Group represents the backbone of U.S. gas production, with about 50% of total U.S. output.

"Other" Onshore Mature Group refers to specific areas of the Onshore MG that are not modeled by vintage, namely Alabama, Florida and Mississippi.

PIRA's U.S. rig counts differ from Baker Hughes counts, owing to the substitution of Offshore Data Services' rig counts for the Offshore Gulf of Mexico. Also, Smith International gas/oil splits are applied to PIRA's regional rig counts to determine regional gas rig counts.

Production treadmill relates to the need for new gas deliverability to offset old gas production losses. The larger the production base, the larger the old gas well losses, given a constant decline rate. Similarly, the faster the decline rate, the larger the old gas well losses (and visa versa), given a constant production base.

Rig efficiency is a measure of average production from first year well per active gas rig. It is calculated by dividing the volume of first-year gas output by drilling intensity and is measured in MMCF/D per rig. Generally, there is an inverse relationship between drilling intensity and rig efficiency that reflects the deterioration in drilling prospect quality when drilling intensity increases and visa versa.

Vintage analysis relates to measuring and assessing production based on the dynamic inter-relationships between new gas output from gas-oriented drilling and older gas output when taking into account annual declines from post-first-year wells. PIRA's North American regional models and Producer Group Simulation models are built around "vintage analysis" that systematically tracks correlations between gas-oriented rig counts and gas produced from first-year gas wells.

PIRA Energy Group, 3 Park Avenue, 26th Floor, New York, NY 10016-5989 (Tel) 212-686-6808 (Fax) 212-686-6628 www.pira.com ¹² For authorized users of PIRA Client Services. Copyright © 2006 PIRA Energy Group. All rights reserved.



CERA Advisory Service

Does Florida Remain Vulnerable to Fuel Disruptions from Hurricanes?

DECISION BRIEF





KENNETH L. YEASTING, CERA Director, is an expert on the interstate natural gas pipeline and storage industries and also specializes in eastern North America gas and power markets. Mr. Yeasting focuses on issues related to gas prices and basis; pipeline flows and development; storage usage and development; and the impacts of liquefied natural gas on gas prices, gas flows, and the need for new infrastructure. He is the lead author of CERA's *Eastern Energy Monthly Briefing* and has authored numerous CERA reports on supply and demand impacts, market strategies, new development effects, and market trends. Before joining CERA, Mr. Yeasting had a long career at ANR Pipeline Company, where he held a number of executive positions. Most recently he was Director of Energy Forecasting, and also held the positions of Director of Revenue Forecasting, Director of Market Research and Planning, and Director of Strategic Planning, among others. With the ANR Storage Company he was Director of Planning and Regulatory Affairs and Director of Financial Planning and Control, among other positions. He also worked for the Michigan Consolidated Gas Company. Mr. Yeasting holds a BBA and an MBA from the University of Michigan.

For clients with access to www.cera.com, additional features related to this report may be available online:

• Downloadable data (Excel file format)

Author biographies

• Downloadable, full-color graphics

Adobe PDF version of complete report

© 2006, All rights reserved, Cambridge Energy Research Associates, Inc., 55 Cambridge Parkway, Cambridge, Massachusetts 02142. No portion of this report may be reproduced, reused, or otherwise distributed in any form without prior written consent.

DISCLAIMER. The accompanying materials were prepared by Cambridge Energy Research Associates, Inc. (CERA), and are not to be redistributed or reused in any manner, with the exception of client internal distribution as described below. CERA strives to be supportive of client internal distribution of CERA content but requires that (a) CERA content and information, including but not limited to graphs, charts, tables, figures, and data, are not to be disseminated outside of a client organization to any third party, including a client's customers, financial institutions, consultants, or the public; and (b) content distributed within the client organization must display CERA's legal notices and attributions of authorship. Some information supplied by CERA may be obtained from sources that CERA believes to be reliable but are in no way warranted by CERA as to accuracy or completeness. Absent a specific agreement to the contrary, CERA has no obligation to update any content or information provided to a client.

DOES FLORIDA REMAIN VULNERABLE TO FUEL DISRUPTIONS FROM HURRICANES?

by Kenneth L. Yeasting

LINGERING VULNERABILITY

Florida relies on gas-fired generation to meet over 38 percent of the state's electricity requirements. Last year, gas deliveries into Florida were significantly disrupted by Hurricane Katrina. Gas service to Florida has been restored, even bolstered, with flows on Destin Pipeline Company from offshore at pre-Hurricane Katrina levels and all four gas processing plants in the Mobile, Alabama/Pascagoula, Mississippi, area operating. An increase in capacity at the Petal Gas Storage facility provides greater gas supply backup than last summer. Further, higher on-site power plant coal and oil product inventories in Florida have reduced the risk of disruption for coal- and oil-fired generation. However, petroleum coke inventories are running well below year-ago levels.

Despite these efforts to improve power plant fuel reliability, Florida remains vulnerable. A number of projects have been proposed that would by 2010 bring additional gas supplies from other areas to Florida. Several of these projects would improve the reliability of gas supplies into the Gulfstream Natural Gas System (Gulfstream).

HURRICANE KATRINA EXPOSES RISKS

Florida's power sector relies heavily on natural gas, oil, and coal (see Figure 1). Last August, Hurricane Katrina significantly disrupted both natural gas flows into Florida and the ability to replenish oil and coal supplies used by Florida power generators.* Significant impacts included

- Throughput on Florida Gas Transmission (FGT) into Florida declined by about 200 million cubic feet (MMcf) per day to about 1,600 MMcf per day.
- Throughput on Gulfstream into Florida declined from about 950 MMcf per day to a low of 150 MMcf per day on August 29, 2005, the day after Hurricane Katrina hit.
- The Florida Reliability Coordinating Council's (FRCC) State Capacity Emergency Coordinator issued an alert expressing concern that residual fuel oil inventories could not be husbanded, as they were likely needed to cover the shortfall in gas deliveries. Florida, which does not have any oil product pipelines, relies on waterborne deliveries of oil, primarily barges from the central Gulf Coast. With many Gulf Coast ports damaged and several refineries offline, the FRCC was concerned about replenishing residual fuel oil inventories used to cover a natural gas shortfall.

^{*}See the CERA Alert Gulf Coast Damage from Katrina Threatens Reliability of Florida Electric System.



• Florida also relies on waterborne deliveries of coal. TECO Energy reported that its coal terminal facility located in Davant, Louisiana, south of New Orleans, was flooded and without power and would have to rely on its 30-day supply of coal on the ground at its power stations.

Throughput on FGT was not as severely affected as Gulfstream by Hurricane Katrina, as FGT has more robust access through pipeline interconnections to other gas supply sources and storage from South Texas to Mobile Bay in Alabama and Petal by Destin. In contrast, Gulfstream has access to gas supply sources in the Mobile Bay area only. Receipts from the four processing plants in Mobile Bay and from offshore production from Destin were at one point reduced to zero. Delays in restoring power delayed reactivation of at least two of the processing plants. Flows on Gulfstream partially recovered to 414 MMcf per day on September 1, 2005, with over 90 percent of the flow into Gulfstream represented by withdrawals from Petal with delivery to Gulfstream by Destin.

NATURAL GAS SUPPLY REMAINS VULNERABLE

Florida's natural gas supply remains vulnerable to disruption from hurricanes. Gulfstream remains especially vulnerable, as its gas supply currently comes almost exclusively from Mobile, Alabama/Pascagoula, Mississippi, area gas processing plants and offshore production from Destin. Since February, pipeline flows on FGT and Gulfstream into Florida are running significantly above year-ago levels, increasing Florida's reliance on at-risk gas supplies (see Figure 2).



Figure 2 Flows into Florida

Pipelines flows on FGT and Gulfstream into Florida for May 2006 were 2,829 MMcf per day, an increase of 723 MMcf per day, or 34 percent, from May 2005. Over 70 percent of the increase occurred on the more vulnerable Gulfstream, although it has half the capacity of FGT (see Figure 3). The May increase in gas flows into Florida was likely the result of two factors:

- an increase in electric demand, as May 2006 in Florida had 2.5 percent fewer cooling degree-days than normal, whereas May 2005 had 8.3 percent fewer cooling degree-days than normal
- a switch by dual-fuel steam plants to natural gas from residual fuel oil as the Henry Hub gas price went from \$0.15 per million British thermal units (MMBtu) above 1% Gulf Coast residual fuel oil for May 2005 to \$1.93 per MMBtu below for May 2006

For this summer, CERA expects electric demand to be somewhat higher owing to economic growth, reduced hurricane outages, and a return to normal weather. Last summer Florida had 0.9 percent more cooling degree-days than normal. CERA also expects that natural gas prices will remain well below residual fuel oil prices this summer.

As noted above, last year withdrawals from Petal were the primary source of gas supply for Gulfstream immediately following Hurricane Katrina. In December 2005, Petal placed a third storage cavern into operation, increasing working capacity by 2,400 MMcf to 11,900



Figure 3 Gulfstream Flows into Florida

MMcf and increasing deliverability by 300 MMcf per day to almost 1,200 MMcf per day. The increase in deliverability improves the ability of Petal withdrawals to backstop a loss of gas supply going into Gulfstream.

OIL AND COAL INVENTORY SITUATION HAS IMPROVED

Inventories of liquid oil products and coal held by Florida generators are higher than yearago levels, improving Florida's ability to absorb a fuel supply disruption, but petroleum coke inventories are well below year-ago levels.

Oil Inventories Up Substantially

Residual fuel oil is an important source of fuel for Florida generation. For 2005 almost 13 percent of Florida's total generation was from residual fuel oil-fired generation. Further, Florida dual-fuel steam plants play an important balancing role, as pipeline capacity into Florida can be constrained during peak demand periods and Florida has no natural gas storage. Florida generators generally maintain substantial oil product inventories to facilitate this balancing role and to minimize the impact of oil supply disruptions on the availability of generation.

For January and February, the latest months for which data are available, liquid oil product inventories of Florida generators were running well above year-ago levels (see Figure 4). Current inventories represent 80 days of supply based on annual average usage. This level of oil inventories is probably more than adequate. One factor that should contribute to the maintenance of higher oil inventories is that since February natural gas prices have been at or below residual fuel oil prices.

Coal Inventories Up Slightly

For January and February, the latest months for which data are available, coal inventories of Florida generators were running above year-ago levels but well below 2004 levels (see Figure 5). Florida receives coal by rail, by barge across the Gulf of Mexico from New Orleans, and by oceangoing vessels. The shipping schedules of barges and oceangoing vessels can be disrupted by hurricanes. What was unique last year was that the New Orleans port facilities were heavily damaged by Hurricane Katrina. This disrupted the transshipment to Florida of domestic coal received in New Orleans from the Illinois Basin. Current inventories represent 47 days of supply based on annual average usage. This level of coal inventories is probably adequate.

Petroleum Coke Inventories Are Down

Florida meets 3.5 percent of its generation requirement with petroleum coke. Year over year, petroleum coke inventories are well below last year's levels but about even with 2004 levels (see Figure 6). Current inventories represent 35 days of supply based on annual average usage. This level of petroleum coke inventories is probably adequate.



CERA Decision Brief



Figure 5 Florida Generator Coal Inventories

Figure 6 Florida Generator Petroleum Coke Inventories



DIVERSIFICATION KEY TO LONG-TERM RELIABILITY OF FUEL SUPPLY

A major key to improving the reliability of Florida's fuel supply for generation is to diversify its gas supply options, especially in the near term, as diversifying generation technologies can take several years. Three groups of gas supply projects have been proposed to do just that, comprising new pipelines to Gulfstream from other gas supply areas, liquefied natural gas (LNG) terminals, and storage.

Several pipeline projects have been proposed to bring gas to Gulfstream from other gas supply areas, including competing projects by Gulfstream's partners:

- Duke Energy Gas Transmission, which owns half of Gulfstream, has filed a request to begin the prefiling process for a 1,000 MMcf per day Southeast Supply Header project from the Perryville Hub in northern Louisiana to interconnections with FGT and Gulfstream. The Southeast Supply Header would be a joint venture with CenterPoint Energy, with an expected online date in 2008. CenterPoint has proposed a 1,200 MMcf per day project from East Texas to the Perryville Hub, to be operational this year. These combined projects would give Florida direct access to growing East Texas production.
- Transcontinental Gas Pipe Line Corporation (Transco) recently held an open season for a 750 MMcf per day Production Area Mainline expansion from Station 45 in Louisiana to Station 85 in Alabama. Transco also held an open season for a 700 MMcf per day Mobile Bay South expansion of its Mobile Bay lateral from Station 85 to Gulfstream. Both projects are scheduled for start-up in 2008. Williams, Transco's parent, owns half of Gulfstream.
- Columbia Gulf Transmission recently held an open season for an expansion and extension of its east lateral from Venice, Louisiana, to Gulfstream at Pascagoula, Mississippi. This project would begin operation in 2009.
- Enbridge Energy Partners recently held an open season for a 1,000 MMcf per day Southeast Texas Expansion project from southeast Texas to Clarke County, Mississippi, where it would interconnect with Destin, which could back-haul gas to Gulfstream. This project would start operation in 2009.

At present there are still four active LNG projects primarily targeting Florida:

- Southern Natural Gas's 500 MMcf per day Cyprus lateral has been filed with the Federal Energy Regulatory Commission. This expansion would bring gas from the Elba Island LNG terminal in Georgia to an interconnection with FGT near Jacksonville, Florida, and would be phased in from 2007 through 2010.
- Calypso US Pipeline (Calypso), a subsidiary of SUEZ Energy North America, has filed to amend its approved pipeline that was intended to bring LNG to southern Florida from the Bahamas. Calypso is now proposing a buoy terminal about 10 miles offshore Florida with LNG deliveries by ships fitted with onboard regasification. The 832 MMcf per day pipeline is proposed to be in service in 2010.

- Chevron has filed to build the Casotte Landing Natural Gas Import Terminal in Pascagoula, Mississippi. This LNG terminal would have base-load send-out of 1,300 MMcf per day with a proposed in-service date of 2010. This terminal would be connected to Gulfstream and four other pipelines.
- Gulf LNG Energy has filed to build the Clean LNG Energy terminal in Pascagoula, Mississippi. This LNG terminal would have base-load send-out of 1,300 MMcf per day with a proposed in-service date of 2009. This terminal would be connected to Gulfstream and at least one other pipeline.

Petal has begun construction of a 5,000 MMcf working capacity expansion with 500 MMcf per day expansion of deliverability and an expected in-service date of 2008.

CONCLUSION: IMPLICATIONS FOR RELIABILITY OF FUEL SUPPLY

Florida's natural gas supply remains vulnerable to hurricane disruptions this summer, especially for Gulfstream, whose gas supply comes primarily from the Mobile, Alabama/Pascagoula, Mississippi, area. However, an increase in capacity at the Petal facility provides greater gas supply backup than last summer. Further, higher generator coal and oil product inventories have reduced the risk of disruption for coal- and oil-fired generation. Nevertheless, until greater gas supply diversity is achieved through one or more of the projects noted above, Florida remains at risk for significant gas supply disruptions.

CERA Advisory North American Natural Gas Service

North American Gas Capacity Race

Can Unconventional Gas Offset the Increased Decline in the Gulf of Mexico?

PRIVATE REPORT®

Disclaimer

The accompanying materials were prepared by Cambridge Energy Research Associates, Inc. (CERA), and are not to be redistributed or reused in any manner, with the exception of client internal distribution as described below.

CERA strives to be supportive of client internal distribution of CERA content but requires that

- CERA content and information, including but not limited to graphs, charts, tables, figures, and data, are not to be disseminated outside of a client organization to any third party, including a client's customers, financial institutions, consultants, or the public.
- Content distributed within the client organization must display CERA's legal notices and attributions of authorship.

Some information supplied by CERA may be obtained from sources that CERA believes to be reliable but are in no way warranted by CERA as to accuracy or completeness. Absent a specific agreement to the contrary, CERA has no obligation to update any content or information provided to a client.

For clients with access to www.cera.com, additional features related to this report may be available online:

- Downloadable data (Excel file format)
 Author biographies
- Downloadable, full-color graphics
 Adobe PDF versi
- Adobe PDF version of complete report
 - © 2005, Cambridge Energy Research Associates, Inc., 55 Cambridge Parkway, Cambridge, Massachusetts 02142 No portion of this report may be reproduced in any form without prior written consent.

NORTH AMERICAN GAS CAPACITY RACE: CAN UNCONVENTIONAL GAS OFFSET THE INCREASED DECLINE IN THE GULF OF MEXICO?

OVERVIEW

This Private Report, which captures CERA's outlook for North American gas productive capacity to 2012, chronicles the key issue for the industry in North America: despite record gas-related drilling activity, supply is struggling to avoid a decline.

Capacity in North America is projected to remain on an "undulating plateau," as CERA author Robert Esser describes it, with little change through 2010 before a moderate decline sets in through 2012 and beyond. A 2.6 billion cubic feet (Bcf) per day decline from 2004 in the United States by 2012 is largely offset by a 2.5 Bcf per day increase in Canada. This outlook is close to and directionally identical to CERA's previous outlooks for North American natural gas supply. While there have been both supply disappointments (e.g., the shelf and deepwater in the Gulf of Mexico) and upside surprises (e.g., a surge in Canadian coal seam methane [CSM]), these have tended to net for an outlook that places total supply quite close to CERA's more recent outlooks.

"The Gulf of Mexico is a major determinant of US capacity," writes Esser. The projected near-term peak in deepwater capacity, when combined with the ongoing decline on the shelf, has increased the overall decline in the Gulf of Mexico in CERA's view to 4.4 Bcf per day between 2004 and 2012.

Esser highlights the potential of unconventional gas—tight sands, shale gas, and CSM to increase capacity. Currently representing over 75 percent of rig activity, unconventional capacity is projected to increase by 5.1 Bcf per day between 2004 and 2012, thus offsetting the decline in the Gulf of Mexico. This resource remains one of the few potential upsides in the outlook.

For the near term, dry gas capacity in the US Lower 48 is projected to continue to decline slightly into 2007, losing around 0.3 Bcf per day. The increase in unconventional gas in the Gulf Coast and Rockies is offset by the declining Gulf of Mexico and Permian and San Juan Basins. A continued high rate of drilling could ultimately result in a slight increase in capacity in the near term.

Capacity in Canada is projected to increase slowly to 2010 with growth in the Western Canadian Sedimentary Basin (WCSB), blending with the arrival of the Mackenzie Delta gas in 2010. Unconventional gas—CSM and tight sands—is the primary source of the growth in the WCSB.

This outlook reconfirms CERA's repeated message about the North American gas industry: it has transitioned permanently from a market characterized by surplus supply and gas-on-gas competition to one marked by constrained continental supply, and thereby constrained growth wrought from competition for available supply. CERA's outlook points

CERA PRIVATE REPOR	~• ••	and the	and the second	A	Р	R	MILES.	V	A		E	R	Е	P	0	R	1
--------------------	----------	---------	----------------	---	---	---	--------	---	---	--	---	---	---	---	---	---	---

to a significant opportunity for new supplies, one that has captured the attention of the global energy industry. Indeed, CERA expects liquefied natural gas to play a significant and accelerating role in North America, from serving 3 percent of demand in 2004 to 13 percent in 2010 to about 25 percent by 2020.

—Daniel Yergin May 2005

NORTH AMERICAN GAS CAPACITY RACE: CAN UNCONVENTIONAL GAS OFFSET THE INCREASED DECLINE IN THE GULF OF MEXICO?

by Robert Esser

INTRODUCTION

Although drilling activity related to natural gas in both the United States and Canada has been at record levels, little effect has been seen on productive capacity. With increased drilling activity directed to unconventional gas—tight sands, shale gas, and coal seam methane (CSM)—this component has responded to record drilling levels in the Rockies, the Bossier and Cotton Valley plays in East Texas and north Louisiana, and the Barnett shale in north central Texas. In Canada a reversal of the wellhead capacity decline in the Western Canadian Sedimentary Basin (WCSB) can be attributed to tight sands and CSM plus the Monkman play in British Columbia. High drilling levels have also tended to reduce decline rates in the Permian and San Juan Basins and reversed the decline in Oklahoma and Canada. CERA has consistently felt that the consensus outlook for a permanent decline in the WCSB was unlikely to occur.

A key signpost for North American capacity, and a focus of this Private Report, is the ability of unconventional gas to offset the projected increased decline in the Gulf of Mexico.

Because the rig count is currently so close to full utilization, any increase in overall North American productive capacity will be limited; the only sources of new rigs will be from drilling companies ordering new rigs or from rigs being switched to gas from oil-related drilling (24 rigs have done so since February 2005). In Canada, 34 new rigs have been added to the fleet during the past 12 months. Most of these are coiled tubing rigs for shallow and mid-depth drilling.

NORTH AMERICAN DRY GAS OUTLOOK

After peaking at 70.8 billion cubic feet (Bcf) per day in 2001, North American dry gas productive capacity is projected to fluctuate on an undulating plateau between 68.4 and 69.2 Bcf per day to 2012 (see Figure 1 and Table 1). Beyond 2010, accelerated declines are projected to overtake supply additions. A 2.6 Bcf per day decline in wellhead capacity in the United States (from 51.5 Bcf per day in 2004 to 48.9 Bcf per day in 2012) is largely offset by a 2.5 Bcf per day increase in Canada (from 17.0 Bcf per day in 2004 to 19.5 Bcf per day in 2012).

The slow decline in the United States, following the 53.7 Bcf per day peak in 2001, begins to increase after 2010 as declines spread to all areas except the Rockies. Canadian capacity reached a low point of 16.6 Bcf per day in 2003, due mainly to the decline in the Ladyfern field, and from this point is projected to increase steadily throughout the period to 2012 and beyond with increased production from the WCSB, followed by the arrival of Mackenzie Delta gas in 2010.

North American Dry Gas Productive Capacity (billion cubic feet per day)

Figure 1



Updated April-1 2005 40916-39

Table 1

North American Dry Gas Productive Capacity Outlook

(billion cubic feet per day)

	<u>2000</u>	<u>2001</u>	<u>2004</u>	<u>2005</u>	<u>2007</u>	<u>2010</u>	<u>2012</u>
US Lower 48	53.9	54.9	52.7	52.5	52.4	51.8	50.4
Alaska	1.4	1.4	1.3	1.3	1.1	1.0	0.9
Total US Wet	55.3	56.3	54.0	53.8	53.5	52.8	51.3
Total US Dry	52.7	53.7	51.5	51.3	51.1	50.4	48.9
Canadian Dry	16.1	17.1	17.0	17.2	17.6	18.8	19.5
Total North America	68.8	70.8	68.5	68.5	68.7	69.2	68.4

Source: Cambridge Energy Research Associates.

Updated April 2005.

LOWER-48 WET GAS OUTLOOK

From 2004 through 2009, US lower-48 wet gas capacity is expected to be on an undulating plateau at around 52.5 Bcf per day. The addition of 0.8 Bcf per day from the eastern Gulf of Mexico in 2008 will slow the decline in the Gulf of Mexico for one year, interrupting the gradual decline in the US Lower 48. Increased capacity in the Rockies, the Gulf Coast, the Mid-Continent, and the eastern United States offsets an increase in the capacity decline in the Gulf of Mexico and moderate declines in the Permian and San Juan Basins and the West Coast (see Figures 2 and 3 and Table 2). After 2010 the decline spreads to all regions except the Rockies, increasing the overall decline, interrupted only by gas from Alaska, which is assumed to achieve first flow in 2015.

WHAT COULD THROW THIS PROJECTION OF CAPACITY OFF?

The projection of US lower-48 capacity could be thrown off by two key events:

First, if the high rate of unconventional gas drilling actually adds more capacity than expected, not only in the current hot spots but also in evolving ones, and if the expected declines in the Permian and San Juan Basins actually become smaller or reverse, US lower-48 capacity could increase slightly, by 0.5 to 1.0 Bcf per day, in the next few years. This trend has been occurring in some areas and has been noticed in successive updates that show smaller decline rates than originally expected. One obstacle to the growth of unconventional





Figure 3 Regional Wet Gas Capacity—US Lower 48

gas much beyond our outlook is that the rig count is nearing full utilization. In some cases increases in rigs have required operators to make deals with drilling companies involving the construction of new rigs. Thus the upside from unconventional gas is limited in magnitude and timing.

On the other hand, a second event that could affect the outlook is the concern of some companies over surging drilling and completion costs and a shortage of experienced drilling crews. Some companies could actually curtail some drilling activity in response to these issues. Earlier this year as many as six active operators had made tentative decisions to possibly curtail some drilling. Should more companies trim drilling in response to higher costs, the smaller capacity growth in some areas could result in a more rapid US lower-48 decline than is currently expected. At this point, companies have the incentive to develop efficiencies to moderate the effect of higher costs.

COMPANY STRATEGY TOWARD GAS

In North America there have been two opposite strategies toward developing gas production. The supermajors have continued to sell off onshore assets to the eager and receptive large independents, opting instead to emphasize the deepwater Gulf of Mexico and the shallow-water shelf, focusing on the deep and ultradeep shelf gas plays. Also supermajors are involved with liquefied natural gas (LNG) development. It is noteworthy that seven of the top ten North American natural gas producers are pursuing LNG development in North America.

	Wet Gas Capacity Outlook by Region (billion cubic feet per day)													
	<u>2000</u>	<u>2002</u>	<u>2003</u>	<u>2004</u>	<u>2005</u>	<u>2006</u>	<u>2007</u>	<u>2008</u>	<u>2009</u>	<u>2010</u>	<u>2012</u>	70		
Gulf of Mexico ¹	14.64	13.57	13.13	12.55	11.80	11.04	10.29	10.05	9.45	8.87	7.85			
Gulf Coast ²	13.68	13.08	13.22	13.20	13.33	13.57	13.80	14.01	14.09	14.11	13.87	100		
Mid-Continent ³	7.96	7.84	7.80	7.85	7.91	8.01	8.12	8.14	8.20	8.26	8.20			
Rocky Mountains ⁴	5.08	6.51	7.04	7.12	7.50	7.90	8.28	8.63	8.94	9.19	9.60	ETOS:		
Permian Basin ⁵	4.92	4.82	4.76	4.62	4.55	4.49	4.42	4.33	4.24	4.15	3.91	5		
San Juan Basin ⁶	4.21	4.06	4.05	4.05	4.02	3.96	3.90	3.84	3.78	3.72	3.60			
Eastern US ⁷	2.35	2.41	2.39	2.45	2.57	2.69	2.79	2.85	2.85	2.86	2.71	Þ		
West Coast	1.04	1.00	0.93	0.87	0.84	0.81	0.78	0.75	0.72	0.69	0.62			
Total Lower 48	53.88	53.29	53.32	52.71	52.52	52.47	52.38	52.60	52.27	51.85	50.36			
Alaska	1.40	1.40	1.35	1.30	1.25	1.20	1.15	1.10	1.05	1.00	0.90			
Total US	55.28	54.69	54.67	54.01	53.77	53.67	53.53	53.70	53.32	52.85	51.26	Part		

Table 2

Source: Cambridge Energy Research Associates.

1 Includes Louisiana, Texas, and Alabama.

2 Includes Texas RRC 1-6, Louisiana, Mississippi, Alabama onshore, and Florida.

3 Includes Texas RRC 7B, 9, 10; Kansas, Oklahoma, and Arkansas.

4 Includes Montana, North Dakota, North Colorado, Utah, Wyoming, and Arizona.

5 Includes southeast New Mexico, and Texas RRC 7C, 8, 8A.

6 Includes northwest New Mexico, and Southern Colorado.

7 Includes Michigan, Kentucky, Ohio, New York, Pennsylvania, Virginia, and West Virginia. Updated April 2005.

8³⁵., eje j 79 171

70

C

50

<u>अल्पाई</u>

 \frown

191

7
C	E	R	A	P	R	-	V	A	Т	R	E	Р	0	R	-

Large independents are generally targeting the unconventional gas resource plays in both the United States and Canada, with several of the leading companies pulling back from the Gulf of Mexico. The Rockies and East Texas are the leading areas of interest, with acquisition and strong drilling activity as a dominant strategy (see Table 3). In response to high prices, some companies are increasing previously-established drilling budgets. These companies are also seeking evolving unconventional hot spots, especially possible tight reservoirs and gas shales in older fields in mature areas made economic with the current

Table 3

Recent E&P Trends Affecting the Gas Capacity Forecast

General

- Emphasis continues on long-lived unconventional reserves (coal seam, tight sands/shales with a prolonged low
 decline after steep initial decline) with "gas mining," which involves low-risk, closely spaced wells, highly efficient
 exploitation not economic at previous low prices (i.e., Rockies, East Texas)
- Supermajors shift to LNG and continue to exit from onshore areas
- · Reduced exploration with emphasis on low-risk exploitation of onshore niche areas
- Large independents continue to focus on the purchase of disposed assets of supermajors in core areas or entry into
 desired new areas
- Strong emphasis on the Rockies, with rig activity well above record 2001 levels and companies continuing to pursue entry, but access is still a problem

Gulf of Mexico

- Exploration emphasis is shifting to Atwater Foldbelt area and ultradeepwater Eocene (Lower Tertiary) oil play with much exploration and appraisal drilling
- Continued slowed pace of sanction of deepwater discoveries
- · Shift to deep and ultradeep shelf play by supermajors and others
- No pickup in shallow-shelf drilling to 15,000 feet, likely due to lack of prospects
- · Gas drilling in deep water is centered on short cycle time prospects near existing infrastructure or the eastern Gulf

Source: Cambridge Energy Research Associates.

C E		A	р	R		V	A	Ĩ	£.	R	E	P	0	R	Ĩ
-----	--	---	---	---	--	---	---	---	----	---	---	---	---	---	---

high gas price and advances in fracturing technology. Companies that have not accumulated a large inventory of undrilled locations will have difficulty maintaining current production levels. In the Gulf of Mexico some of these companies are participating in both the deep water and the deep and ultradeep shelf plays.

The dominant large independents are emphasizing the exploitation of undrilled locations in onshore resource plays while placing much less emphasis on exploration.

With so much attention directed to unconventional gas, should gas prices drop because of weather, a slowdown in the economy, and sudden large supplies of LNG, the ensuing downturn in drilling activity would be similar to the 2001 to 2002 period. It is noteworthy that drilling increases are constrained by the lack of available rigs, and thus the drilling upside is relatively insensitive to price while the drilling downside is highly sensitive to price. A sudden drop in wellhead capacity, due to a loss of so many new wells with high initial production levels, would allow the decline rate to take over, restoring higher prices and thus supporting drilling again after a short time.

CONTEXT AND METHODOLOGY

CERA projects US wellhead capacity on a wet gas basis, i.e., with the natural gas liquids (NGLs) left in the gas.

Productive capacity refers to peak annual capacity, excluding the effects of seasonal and temporary factors such as weather (hurricanes in the Gulf and well freeze-offs) and regional price distortions related to local demand and distribution problems. At recent price levels, gas producers have a strong incentive to get their gas to market, and gas production is very close to gas capacity. On average, only about 2 percent of gas capacity is expected to be unavailable at any given point in the future. CERA's projection of gas capacity is based on the analysis of three basic components: existing production, including expected declines; recent significant discoveries; and the expectation of production from future discoveries based on the current and expected pace of exploration, recent discovery experience, and the timing required for development (see Figure 4). CERA's price outlooks further anticipate changes to downstream infrastructure, which may constrain wellhead supply from reaching markets. Inclusion of future discoveries differentiates CERA's outlook from others. This stepped methodology applies mostly to the Gulf of Mexico as most of the onshore resource plays depend not on exploration, but on a continued high level of drilling and on company decisions to increase drilling to exploit large, undeveloped resource plays.

The near-term outlook is activity-based by basin combined with recent production levels, modified by current and expected near-term activity levels, superimposed on CERA's expectation of reserves and discovery potential, the elimination of the effect of temporary disruptions, and the pace and timing of development of expected new production. Some industry analysts, when examining scattered individual quarterly company production comparisons with previous quarters, have called for declines of up to 5 percent over the same time periods, disregarding the reason for the decline. CERA believes that company production comparisons provide a misleading picture of the future.





Figure 4

UNCONVENTIONAL GAS: THE OFFSET TO THE INCREASED DECLINE IN THE GULF OF MEXICO

Unconventional gas plays, consisting of tight sands, shale gas, and CSM, are on track to constitute close to half of the total gas production in the United States by 2012. Most of the areas with strong drilling activity (at least 75 percent of the current gas-related rig count) are targeting unconventional gas, the only component of North American supply that is currently increasing (see Figure 5). The 5.1 Bcf per day increase from 2004 to 2012 in unconventional gas is sufficient to offset the 4.4 Bcf per day decline in the Louisiana/Texas portion of the Gulf of Mexico to 2012.

US lower-48 unconventional gas capacity began to increase rapidly in the 1990s from 5.9 Bcf per day in 1990 to 16.0 Bcf per day in 2000, led primarily by CSM in the San Juan and Powder River Basins and tight sands in all areas, spurred by Section 29 tax incentives. Since 2000, unconventional gas capacity has increased from 16.0 Bcf per day to 19.4 Bcf per day in 2004, with further growth to 24.6 Bcf per day projected for 2012 (see Table 4).



Figure 5 Rapid Growth of US Lower-48 Unconventional Gas



Table 4

US Unconventional Gas Outlook

(billion cubic feet per day)

Total	5.89	12.81	16.04	17.89	19.43	21.56	23.80	24.56
Shale gas	0.41	0.88	1.40	1.42	1.60	1.90	2.20	2.40
Tight sands	4.93	9.19	10.68	11.89	13.00	14.50	16.20	16.60
Coal seam methane	0.55	2.74	3.96	4.58	4.83	5.16	5.40	5.56
	<u>1990</u>	<u>1996</u>	<u>2000</u>	<u>2002</u>	<u>2004</u>	<u>2007</u>	<u>2010</u>	<u>2012</u>

Source: Cambridge Energy Research Associates. Updated April 2005.

C I	9 8 احق معل	R	A	5.ª	R	CARDING	V	A	ţ	E	R		P.	0	R	and a local
-----	--------------------------	---	---	-----	---	---------	---	---	---	---	---	--	----	---	---	-------------

Unconventional gas plays tend to provide huge, steady, long-lived reserves at low risk and at predictable production rates; however, they can be difficult to produce because of low reservoir permeability and the production mechanism. Unconventional reserves are not generally the result of recent discoveries but include existing reservoirs, reservoirs that were too high cost to exploit not so many years ago but have become economic thanks to the recent higher prices combined with new technology. The gas reservoirs in many cases are sorption reservoirs where gas is stored by absorption to the matrix, and the gas must be released by a drop in pressure before it can be produced. These wells are generally low capacity producers and have a characteristic production profile of a rapid initial decline within the first year followed by low decline rates, resulting in long-lived reserves. This profile compares with the typical conventional gas reservoir profile of a higher initial production followed by a rapid decline to reservoir depletion. The reservoirs have homogeneous characteristics and are typically continuous with large areal extent and in most cases are controlled by geologic trap-forming factors as in conventional oil and gas fields.

The initial development of unconventional gas reserves was inspired by US tax credits in the 1980s. Unconventional resources in the United States are almost twice the size of conventional reserves (185 trillion cubic feet [Tcf]) and are estimated by the US Geologic Survey at 350 Tcf, of which 185 Tcf occurs in tight sands. The Rockies are thought to contain 169 Tcf of unconventional gas. There are numerous estimates of reserves and potential resources of unconventional gas, and many of these estimates include some likely uneconomic resources. It is important to note that the volume of unconventional reserves and resources exceeds known proved reserves and conventional undiscovered potential. However, evolving areas of probable unconventional gas are becoming apparent as companies seek new areas and overlooked possible reservoirs in existing fields with the potential to produce gas (see Figure 6).

Much of the increase in expectations from unconventional gas has resulted from finding the optimum well spacing to recover more reserves utilizing updated multistage fracturing technology. Thus the Jonah field in Wyoming has had well spacing (the density of wells) reduced from the original 160 acres to 20 acres, and an application to go to 10 acres is awaiting approval. The reduction in spacing, increasing the density of development wells, and development of the entire package of sands in this field have increased gas recovery from 20–40 percent to 60–75 percent of the reserve. Likewise the Pinedale field, also in Wyoming, has had spacing reduced to 20 acres, resulting in numerous undrilled locations.

The projected strong growth in unconventional gas depends upon continued innovations in drilling, reduced well spacing, completion technology, overall land access, and removal of delays in well permitting.

It is important to look at the production response to increases in gas-related rig activity. From 1999 to 2001 an increase of almost 700 rigs, with unconventional gas the predominant target, resulted in an increase in capacity of only 1.7 Bcf per day due largely to three plays: Bossier Sand, Powder River CSM and the deepwater Gulf of Mexico. Note that





Figure 6 Evolving US Lower-48 Unconventional Gas Hot Spots

the deepwater play was a long lead time play with several large fields coming onstream in this time frame and thus was not a response to heightened drilling activity. Thus only the Rockies registered a significant supply response to activity in the 1999–2001 drilling boom. For the most recent increase in rig activity of 585 rigs between 2002 and mid-2005, which resulted in a new record for US gas-related drilling, productive capacity in 2005 is projected to have declined by 0.9 Bcf per day from the 2002 low point in drilling, largely due to the decline in the Gulf of Mexico (-1.8 Bcf per day), partially offset by a 1.0 Bcf per day increase in the Rockies. The increase in the Rockies and the surprising smaller-than-expected declines in other areas can be attributed solely to increased unconventional gas. It is evident that even very strong drilling activity targeting unconventional gas does not add large new production because of the small individual well production capabilities. Many wells are required to add small amounts of production.

Even though it is evident that more drilling is required to increase production, the lack of rig and personnel availability will moderate any increase in drilling.

UNCONVENTIONAL GAS—TIGHT GAS SANDS

By far the largest component of unconventional gas is tight gas sands, which reached 13 Bcf per day in 2004 and is projeccted to reach 16.6 Bcf per day in 2012. Tight gas sands are fine-grained sandstone formations with permeability of less than 0.1 millidarcy. They are characteristically of widespread areal extent, usually overpressured, and occur in most geologic basins. The most active areas of tight sand development are East Texas and north Louisiana, the San Juan Basin, the Green River Basin in Wyoming, the Permian Basin of West Texas, and southeast New Mexico (see Figure 7). In the late 1990s, South Texas produced the most tight gas (2.4 Bcf per day), followed by the Rockies (1.6 Bcf per day), East Texas (1.5 Bcf per day), the San Juan Basin (1.4 Bcf per day), and the Permian Basin (0.9 Bcf per day).

Also many of the largest US gas fields involve tight sands. These include the Carthage field in East Texas, the Jonah and Pinedale fields in the Green River Basin in Wyoming, the Wattenburg field in Colorado, and much of the San Juan Basin.

Important tight gas sand reservoir targets include the Cotton Valley/Bossier sand play in East Texas, the overpressured Lance sand in the Greater Green River and Wind River Basins in Wyoming, the Williams Fork/Mesaverde in the Rulison and Mamm Creek fields in the Piceance Basin in Colorado, the Vicksburg and the Lobo (Wilcox) field in South Texas, the Mesaverde and Dakota formations in the San Juan Basin, and the Canyon sand in the Ozona field in the Permian Basin.



UNCONVENTIONAL GAS—SHALE GAS

Shale gas is an increasing component of unconventional gas, occurring principally in the rapidly expanding Barnett play in north central Texas, the Devonian Antrim play in Michigan, the Devonian shales of the Appalachian Basin, and smaller pockets in the Illinois and San Juan Basins. The Newark East field near Fort Worth, Texas, producing from the Barnett, is among the largest US gas fields.

Gas shales are self-sourcing, organic-rich reservoirs requiring careful stimulation to produce. Completions utilize a new "light sand frac," horizontal drilling, and the refracturing of previously completed wells. The presence of natural fracturing in the reservoir is a major aid to economic production.

The Barnett is a relatively deep, overpressured, Mississippian age, fractured shale. Drilling and completion technology have evolved with more horizontal wells using water and foam with a small amount of propant. The play is expanding rapidly to the south into Tarrant and Johnson counties and west of Fort Worth, with production expected to continue to increase from 0.40 Bcf per day in 2000 to 0.86 Bcf per day in 2004 and 1.15 Bcf per day in 2012.

The Lewis Shale in the San Juan Basin, while not treated as standalone target, is a secondary target with the surrounding conventional reservoirs and is a solid contributor to production.

Production from the Antrim shale in Michigan has slowly declined from 0.5 Bcf per day around 2000 to 0.4 Bcf per day in 2004, even though some drilling is still occurring. Production from the Devonian shale in the Appalachian Basin will continue to increase slowly from 0.3 Bcf per day in 2000.

Future shale gas production is projected to grow from 1.6 Bcf per day in 2004 to 2.4 Bcf per day in 2012, led by the existing plays but also by other evolving gas shales that have not yet been fully assessed. Some examples include the Fayetteville shale in Arkansas and a Barnett shale lookalike in the Palo Duro Basin in the Texas Panhandle.

UNCONVENTIONAL GAS—COAL SEAM METHANE

A major segment of unconventional gas is CSM, which became the major component in the San Juan Basin in the early 1990s (see Table 5). CSM is also the dominant gas in the Black Warrior Basin of Alabama. The most recent new significant area of CSM production is the Powder River Basin of Wyoming, whose development is currently stalled by environmental concerns and slow drilling permit approvals.*

US CSM capacity increased from 3.96 Bcf per day in 2000 to 4.83 Bcf per day in 2004 and is projected to reach 5.56 Bcf per day in 2012. The growth in CSM is moderated by a gradual decline in the San Juan Basin. Most of the US CSM is produced in the Rockies; however, capacity in this area will remain at around 4.0 Bcf per day to 2012 as slowly declining San

^{*}Coal seam methane is also known as coalbed methane.

North American Coal Seam Methane Capacity													
nited States Recoverable Average Well Capacity Outlook (Bcf per day)													
<u>Gas (Tcf)</u>	Depth (feet)	2000	2002	2003	2004	2005	2006	<u>2007</u>	2010	<u>2012</u>			
10.0	2,600	2.70	2.52	2.51	2.52	2.48	2.43	2.39	2.20	2.05			
20.0	700–1,500	0.41	0.91	0.95	0.91	0.95	1.00	1.10	1.30	1.40			
10.0	650–6,000				0.01	0.01	0.02	0.03	0.05	0.07			
3.5	1,500	0.10	0.20	0.25	0.27	0.30	0.32	0.34	0.35	0.32			
5.5	3,500	0.20	0.28	0.28	0.25	0.23	0.22	0.21	0.19	0.17			
1.2	1,800	0.27	0.32	0.33	0.33	0.33	0.32	0.31	0.28	0.26			
		0.04	0.07	0.11	0.15	0.19	0.22	0.25	0.30	0.32			
2.0		0.10	0.12	0.18	0.22	0.26	0.30	0.35	0.50	0.70			
		0.14	0.16	0.17	0.17	0.18	0.18	0.18	0.18	0.17			
		3.96	4.58	4.78	4.83	4.93	5.01	5.16	5.35	5.46			
								—	0.05	0.10			
		3.96	4.58	4.78	4.83	4.93	5.01	5.16	5.40	5.56			
50.0	2,000-3,000			0.04	0.09	0.20	0.35	0.50	1.10	1.25			
otal North America 3.96 4.58 4.82 4.92 5.13 5.36 5.66 6.50 6.81													
	Recoverable <u>Gas (Tcf)</u> 10.0 20.0 10.0 3.5 5.5 1.2 2.0 50.0	Recoverable Average Well Gas (Tcf) Depth (feet) 10.0 2,600 20.0 700–1,500 10.0 650–6,000 3.5 1,500 5.5 3,500 1.2 1,800 2.0 50.0 2,000–3,000	Recoverable Average Well 2000 Gas (Tcf) Depth (feet) 2000 10.0 2,600 2.70 20.0 700–1,500 0.41 10.0 650–6,000 3.5 1,500 0.10 5.5 3,500 0.20 1.2 1,800 0.27 0.04 0.04 0.11 2.0 0.10 5.5 3,500 0.27 0.04 0.20 0.10 0.14 3.96 3.96 3.96 50.0 2,000–3,000 3.96 3.96	Recoverable Average Well 2000 2002 10.0 2,600 2.70 2.52 20.0 7001,500 0.41 0.91 10.0 650-6,000 3.5 1,500 0.10 0.20 5.5 3,500 0.20 0.28 1.2 1,800 0.27 0.32 2.0 0.10 0.12 5.5 3,500 0.20 0.28 1.2 1,800 0.27 0.32 0.04 0.07 2.0 0.10 0.12 2.0 0.10 0.12 0.14 0.16 3.96 4.58 3.96 4.58 50.0 2,000-3,000 3.96 4.58 50.0 2,000-3,000 3.96 4.58	Recoverable Gas (Tcf)Average Well Depth (feet) 2000 2002 2003 10.02,6002.702.522.5120.0700-1,5000.410.910.9510.0650-6,0003.51,5000.100.200.255.53,5000.200.280.281.21,8000.270.320.330.040.070.110.120.182.00.100.120.180.140.160.173.964.584.780.043.964.584.82	Recoverable Gas (Tcf)Average Well Depth (feet) 2000 $2002200320032004200410.02,6002.702.522.512.5220.0700-1,5000.410.910.950.9110.0650-6,0000.013.51,5000.100.200.250.275.53,5000.200.280.280.251.21,8000.270.320.330.332.00.100.120.180.220.140.160.170.170.172.00.100.120.180.220.140.160.170.175.02,000-3,0000.040.093.964.584.824.92$	Recoverable Gas (Icf)Average WellCapacity Outlook (Bc 2,600 0.0 $2,600$ 2.002 2003 2004 2005 10.0 $2,600$ 2.70 2.52 2.51 2.52 2.48 20.0 $700-1,500$ 0.41 0.91 0.95 0.91 0.95 10.0 $650-6,000$ $$ $$ 0.01 0.01 3.5 $1,500$ 0.10 0.20 0.25 0.27 0.30 5.5 $3,500$ 0.20 0.28 0.28 0.25 0.23 1.2 $1,800$ 0.27 0.32 0.33 0.33 0.33 2.0 $$ 0.10 0.12 0.18 0.22 0.26 0.14 0.16 0.17 0.17 0.18 2.0 $$ 0.10 0.12 0.18 0.22 0.26 0.14 0.16 0.17 0.17 0.18 3.96 4.58 4.78 4.83 4.93 $$ $$ $$ 0.04 0.09 0.20 3.96 4.58 4.78 4.83 4.93 50.0 $2,000-3,000$ $$ $$ 0.04 0.09 0.20 3.96 4.58 4.82 4.92 5.13	Recoverable Gas (Tcf)Average WellCapacity Outlook (Bcf per day) $Gas (Tcf)$ Depth (feet) 2000 2002 2003 2004 2005 2006 10.0 $2,600$ 2.70 2.52 2.51 2.52 2.48 2.43 20.0 $700-1,500$ 0.41 0.91 0.95 0.91 0.95 1.00 10.0 $650-6,000$ $$ $$ -0.01 0.01 0.02 3.5 $1,500$ 0.10 0.20 0.25 0.27 0.30 0.32 5.5 $3,500$ 0.20 0.28 0.28 0.25 0.23 0.22 1.2 $1,800$ 0.27 0.32 0.33 0.33 0.33 0.32 2.0 $$ 0.10 0.12 0.18 0.22 0.26 0.30 0.14 0.16 0.17 0.17 0.18 0.18 $$ <t< td=""><td>Recoverable Gas (Icf)Average Well2000 2000$2002$ 2003$2004$ 2004$2005$ 2005$2006$ 2007$2007$ 200710.02,6002.702.522.512.522.482.432.3920.0700-1,5000.410.910.950.910.951.001.1010.0650-6,0000.010.010.020.033.51,5000.100.200.250.270.300.320.345.53,5000.200.280.280.250.230.220.211.21,8000.270.320.330.330.330.320.310.040.070.110.150.190.220.252.00.100.120.180.220.260.300.350.140.160.170.170.180.180.18</br></td></t<> <td>Recoverable Gas (Icf)Average Well$2000$$2002$$2003$$2004$$2005$$2006$$2007$$2010$10.02,6002.702.522.512.522.482.432.392.2020.0700-1,5000.410.910.950.910.951.001.101.3010.0650-6,0000.010.010.020.030.053.51,5000.100.200.250.270.300.320.340.355.53,5000.200.280.280.250.230.220.210.191.21,8000.270.320.330.330.330.320.310.282.00.100.120.180.220.260.300.350.502.00.100.120.180.220.260.300.350.502.00.100.120.180.220.260.300.350.500.140.160.170.170.180.180.180.183.964.584.784.834.935.015.165.350.050.040.090.200.350.501.1050.02,000-3,0000.040.090.200.350.501.103.964.58</td>	Recoverable Gas (Icf)Average Well 2000 2000 2002 	Recoverable Gas (Icf)Average Well 2000 2002 2003 2004 2005 2006 2007 2010 10.02,6002.702.522.512.522.482.432.392.2020.0700-1,5000.410.910.950.910.951.001.101.3010.0650-6,0000.010.010.020.030.053.51,5000.100.200.250.270.300.320.340.355.53,5000.200.280.280.250.230.220.210.191.21,8000.270.320.330.330.330.320.310.282.00.100.120.180.220.260.300.350.502.00.100.120.180.220.260.300.350.502.00.100.120.180.220.260.300.350.500.140.160.170.170.180.180.180.183.964.584.784.834.935.015.165.350.050.040.090.200.350.501.1050.02,000-3,0000.040.090.200.350.501.103.964.58			

Table 5

***Includes Appalachian, Forest City, Illinois, Piceance Basins, and Southeast Kansas. Updated April 2005.

16

٢ 1755

370

and a

"Q

X-ADVECT

77

D

277

70

5.4.3 TŪ \circ

Sec.

mang

Juan and Uinta Basin production is offset by expected increases in the Powder River, the Green River, and the Raton Basins. The decline in the San Juan Basin has been reduced by well reentry, new technology, and infill drilling resulting from well downspacing. The Anadarko and Arkoma Basins of Oklahoma are also the scene of current active CSM development.

The ultimate potential of the Powder River Basin remains a wild card in the CSM outlook. Increasing production from the Big George coal seam is critical in offsetting production from the already-declining Wyodak coals. Recent well performance data on Big George wells show low decline rates, which will be favorable to the outlook if sustained. However, Big George coal wells produce more water per thousand cubic feet (Mcf) than other Powder River CSM wells, potentially intensifying environmental concerns over water disposal.

It is expected that CSM will continue to be developed from many of the shallow coal fields in the Appalachians, the Forest City and Illinois Basins, southeast Kansas, and the low-ranked coals along the Gulf Coast. In addition, considerable CSM potential exists in Alaska. However, required changes in shallow-gas leasing procedures have currently stalled activity.

The production profile for CSM wells is similar to that for other unconventional gases except for the timing of initial peak output, which depends on the length of the dewatering process prior to peak production. Some CSM plays, such as the Horseshoe Canyon coals in Alberta, are dry, allowing gas production to ramp up quickly.

CSM in Canada has begun to ramp up quite rapidly from initial production of 0.04 Bcf per day in 2003 to a projected 0.20 Bcf per day in 2005, reaching 1.25 Bcf per day in 2012. The early ramp-up is occurring from the dry Horseshoe Canyon coals in south central Alberta. Production from the deeper, wet Manville coals is expected to provide much of the growth toward 2010.*

THE ROLE OF THE GULF OF MEXICO: THE MAJOR DETERMINANT OF FUTURE US GAS CAPACITY

Capacity in the Texas/Louisiana portion of the Gulf of Mexico is projected to decline by 4.4 Bcf per day, from 11.9 Bcf per day in 2004 to 7.5 Bcf per day in 2012, reflecting the peaking of the deep water in 2005 and subsequent decline (with the exception of 2008) through 2012 and beyond (see Figure 8). This is an increase in the expected decline in the Gulf of Mexico by 0.8–1.0 Bcf per day toward the end of the decade compared with our previous outlook.

The Gulf of Mexico continues to experience increasing aggregate decline rates due to the projected early peaking of the deep water, with the exception of a single year, 2008, in response to the development of a series of eight nonassociated gas discoveries in the eastern Gulf (see Figure 9). The deepwater portion of the Gulf has turned disappointing because

^{*}See the CERA Private Report The Giant Awakens-Coalbed Methane in Canada.



of the recent lack of nonassociated gas discoveries, slow company sanctioning of field development, the probable lack of associated gas in the recent spate of large ultradeepwater Eocene oil discoveries stretching from Texas to south of the Mississippi Delta, more rapid declines experienced in some fields, poor discovery appraisal results, and the removal of some earlier discoveries from expected development until at least after 2010 for economic reasons.* Recent major company mergers and acquisitions tend also to delay development decisions.

In the deep water (Flexure Trend), most of the larger recent nonassociated gas discoveries are in production: the Canyon Express, Nakika, and the Nansen/Boomvang/ Falcon group of fields. In the next few years the last of the larger associated gas fields will come into production: Thunderhorse, Mad Dog, Holstein, Atlantis, K2, and K2 North.

Capacity in the deepwater Gulf of Mexico is projected to peak at 5.31 Bcf per day in 2004–05 before declining to 4.06 Bcf per day, with the exception of 2008, when the Independence platform, receiving gas from eight nearby discoveries, is expected to ramp up to 0.85 Bcf per day, interrupting the decline in deepwater production for one year (see Figure 10).

*See the CERA Private Report Worldwide Liquids Capacity Outlook to 2010-Tight Supply or Excess of Riches?

.

Figure 9 US Gulf of Mexico Flexure Trend and Associated Plays



Capacity on the shallow-water shelf is expected to continue its decline through 2012 and beyond. Since the 2001 peak year of drilling activity with 158 active rigs, activity on the shelf has been lower by 30 to 70 rigs, thus significantly reducing the availability of new production to offset declining existing production. High decline rates in the small discoveries will perpetuate the decline on the shelf.

	C	44.4	R	A	P	R		V	A		E	R	E	Р	0	R	Т
--	---	------	---	---	---	---	--	---	---	--	---	---	---	---	---	---	---

Fi	aı	ı٣	e	1	0
	g		5	I.	v

Components of Louisiana and Texas Flexure Trend Gas Productive Capacity

(water depth >600 feet)



Source: Cambridge Energy Research Associates. *Includes Eastern Gulf. Updated April 2005 40916-1

A secondary play on the shelf—the deep shelf gas play—is rapidly moving forward. Although the play has long been known to exist, advances in the interpretation of 3D seismic followed by several large discoveries have reignited interest in the deep structures characteristic of this play. After having abandoned this area to pursue deepwater prospects, supermajors, other large companies, and large independents have returned to the shelf and amassed large acreage positions in recent lease sales and in joint ventures with existing smaller company leaseholders.

The deep gas play, below 15,000 feet, has captured much of the interest on the shallowwater shelf, with numerous wells being drilled and drilling companies upgrading and building new rigs to meet the demand for this deep drilling. Companies are pursuing targets of 30 to 200 Bcf compared with conventional shelf targets above 15,000 feet of less than 10 Bcf. The shift to the deep play is a major reason for the slow pickup in shallow-water jackup activity, owing to the limited availability of rigs capable of deep drilling much below 15,000 feet. Deep drilling is characterized by high temperatures and pressures, longer driling time, and high drilling costs.

.

CERA PRIVATE REP	ORT
------------------	-----

Recent deep shelf discoveries include the JB Mountain and Mound Point fields, the Tarantula subsalt field, and the Deep Tern and Hurricane Upthrown discoveries. Many of the wells on the deep shelf produce between 10 and 30 million cubic feet (MMcf) per day. The 20 productive wells drilled in 2004 could lead to production of 0.3 to 0.4 Bcf per day within a year.

Production from the deep shelf has increased only slowly from 0.8 Bcf per day in 1995 to 1.1 Bcf per day in the late 1990s and early 2000s and to 1.45 Bcf per day in 2002. Due to the rapid decline in existing wells, growth in production is likely to remain labored, at least through 2010, and will do little to slow the decline on the shelf.

A new target on the shelf is the ultradeep shelf play below 25,000 feet for lower Miocene to Cretaceous reservoirs and large structures with reserves possibly exceeding 1–5 Tcf. Several unsuccessful wells have been drilled, but the primary result so far has been successful penetration below 25,000 feet. Currently the Blackbeard West wildcat is targeting a large structure at 32,000 feet. Several other wells are drilling toward 25,000 feet. Any significant production from the ultradeep shelf is not likely until after 2010, as production technology will have to be perfected.

REGIONAL HIGHLIGHTS

A discussion of the important plays in the individual region follows.

Gulf Coast

Stretching from South Texas across Louisiana, this area has significant potential for discovery. Deep (below 12,000 feet), high-capacity, high-cost wells targeting the Wilcox, Yegua, and Frio formations are common, especially in South Texas and south Louisiana. Capacity is projected to increase by 0.67 Bcf per day, from 13.20 Bcf per day in 2004 to 14.11 Bcf per day in 2010, before declining to 13.87 Bcf per day in 2012 (see Table 6). The main activity involves reworking wells in existing fields and drilling to deeper zones in existing fields.

The Gulf Coast region is also the location of two important tight gas plays: the Cotton Valley/Bossier Sand in Texas RRC-5 and 6 in East Texas and the Cotton Valley in north Louisiana. Production in these areas is projected to increase by 0.5 Bcf per day, from 4.37 Bcf per day in 2004 to 4.87 Bcf per day in 2012. Production in South Texas RRC-4 and south Louisiana is currently declining in response to the slow recovery of drilling from the 2002 lows. These areas are considered deep, expensive wildcat plays, and companies have redirected their emphasis to low-risk, shallower, unconventional gas drilling. However, recently drilling activity in these areas has increased closer to the 2001 peak levels. CERA's outlook reflects this recent increase in activity, which could lead to a moderate increase in capacity from these highly prospective areas from 5.32 Bcf per day in 2004 to 5.70 Bcf per day in 2012.

							Table	6									
					W	et Gas Ca	pacity Oı	ıtlook by	Region								pm
							(Bcf per o	lay)									23
Gulf of Mexico	<u>1995</u> 16.62	<u>1997</u> 15.76	<u>1999</u> 15.01	<u>2000</u> 14.64	<u>2001</u> 14.69	<u>2002</u> 13.57	<u>2003</u> 13.13	<u>2004</u> 12.55	<u>2005</u> 11.80	<u>2006</u> 11.04	<u>2007</u> 10.29	<u>2008</u> 10.05	<u>2009</u> 9.45	<u>2010</u> 8.87	<u>2011</u> 8.18	<u>2012</u> 7.85	A
Alabama Offshore	1.00	1.14	1.04	0.94	0.89	0.84	0.79	0.70	0.64	0.59	0.54	0.50	0.45	0.42	0.38	0.35	
Louisiana Offshore	11.92	11.33	11.43	11.26	11.33	10.48	9.75	9.29	8.81	8.30	7.80	7.80	7.40	7.00	6.40	5.90	36357
Texas Offshore	3.70	3.29	2.54	2.44	2.47	2.25	2.59	2.56	2.35	2.15	1.95	1.75	1.60	1.45	1.40	1.60	
Subtotal TX/LA Offshore	15.62	14.62	13.97	13.70	13.80	12.73	12.34	11.85	11.16	10.45	9.75	9.55	9.00	8.45	7.80	7.50	æ
Gulf Coast	13.25	14.01	13.34	13.68	13.76	13.08	13.22	13.20	13.33	13.57	13.80	14.01	14.09	14.11	14.04	13.87	untis
Texas District 1	0.25	0.32	0.31	0.28	0.28	0.25	0.26	0.28	0.27	0.27	0.27	0.28	0.28	0.28	0.27	0.26	
Texas District 2	0.63	0.70	0.84	0.87	0.85	0.76	0.75	0.76	0.75	0.74	0.73	0.71	0.70	0.69	0.68	0.67	~
Texas District 3	2.30	2.50	2.21	2.35	1.98	1.63	1.60	1.62	1.64	1.66	1.68	1.70	1.70	1.65	1.60	1.55	_
Texas District 4	3.32	3.60	3.30	3.46	3.66	3.74	3.63	3.32	3.25	3.30	3.40	3.50	3.50	3.50	3.45	3.35	فتنك
Texas District 5	0.48	0.58	0.60	0.75	0.92	1.02	1.27	1.42	1.48	1.55	1.58	1.60	1.62	1.64	1.65	1.65	mej
Texas District 6	1.62	1.51	1.53	1.58	1.62	1.55	1.58	1.69	1.75	1.80	1.83	1.85	1.87	1.89	1.90	1.90	
Louisiana Onshore-South	2.82	2.70	2.61	2.56	2.56	2.26	2.12	2.00	2.02	2.05	2.10	2.15	2.20	2.25	2.30	2.35	popos
Louisiana Onshore-North	1.01	1.25	1.13	1.10	1.12	1.10	1.18	1.26	1.30	1.32	1.33	1.34	1.35	1.35	1.35	1.32	
Mississippi	0.29	0.32	0.30	0.24	0.30	0.31	0.37	0.40	0.43	0.45	0.46	0.47	0.48	0.49	0.50	0.50	
Alabama Onshore	0.51	0.51	0.49	0.47	0.45	0.45	0.45	0.44	0.43	0.42	0.41	0.40	0.38	0.36	0.34	0.32	æ
Florida	0.02	0.02	0.02	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01	0.01	0.01	0.00	0.00	
Nid Continent	0.91	0 12	7 02	7 06	7 02	7 9/	7 90	7 85	7 01	Q A1	8 1 2	<u>8</u> 1 <i>1</i>	8 20	8 76	Q 75	8 20	PT71
Toyas District 7P	9.01	9.13	0.16	0.16	0.16	0.16	0.15	0.14	0.15	0.01	0.12	0.14	0.20	0.20	0.23	0.20	nça
Texas District 0	0.20	0.19	0.10	0.10	0.10	0.10	0.15	0.14	0.15	0.10	0.17	1.02	1.02	1 12	1.17	0.20	
Toyas District 10	U.29 1 70	0.50	U.ST 1 00	0.40 1 A0	0.JT 1.0/	0.07	0.00	0.00	V.00 0.00	0.73 1 01	0.70 1 A2	1.05	1.00	1.15	1.14	1.15	0
IEXAS DISUICE TO	1.20 2.10	1.20 1.95	1.09	1.00	1.04	0.90 1 75	0.94	0.97 1 10	U.99 1 AC	1.01	1.05	1.00	1.07	1.09	0.75	0.70	
Nd115d5 Okłaboma	Z.19 5 21	1.00 5.00	1.5Z	1.44	1.32	1.20	1.10	1.10	1.00	1.00	0.99	0.90 A AF	U.OJ 1 17	0.00	U.70 450	0.70	R
UKIdHUHId	5.51 0.54	5.00 0.50	4.37	4.41	4.43 0.46	4.33 0.45	4.3U 0.46	4.3U 0.49	4.33	4.4U 0.51	4.4) 0.50	4.40 0.50	4.4/ 0.5/	4.47 0.55	4.JU 0.55	4.JU 0.55	
AIKalisas	0.54	0.59	0.4/	0.47	0.40	0.45	0.40	0.46	0.49	0.51	0.52	0.23	0.54	دد.0	0.55	0.55	-

Wet Gas Capacity Outlook by Region (continued)	(TT)												
(Bcf per day)													
<u>1995 1997 1999 2000 2001 2002 2003 2004 2005 2006 2007 2008 2009 2010 2011 2012</u>	Þ												
Rocky Mountains 3.98 4.06 4.63 5.08 6.02 6.51 7.04 7.12 7.50 7.90 8.28 8.63 8.94 9.19 9.41 9.60													
Arizona 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.0													
Northern Colorado (2) 0.80 0.82 0.87 0.94 1.11 1.37 1.54 1.61 1.67 1.75 1.82 1.87 1.90 1.90 1.90 1.90	raçoy												
Montana 0.16 0.17 0.17 0.19 0.22 0.24 0.24 0.26 0.28 0.30 0.32 0.34 0.35 0.35 0.33 0.31	•2												
North Dakota 0.14 0.15 0.15 0.15 0.16 0.16 0.15 0.15 0.16 0.17 0.18 0.18 0.18 0.18 0.17 0.16	33												
Utah 0.77 0.77 0.73 0.74 0.78 0.75 0.73 0.74 0.78 0.82 0.85 0.88 0.90 0.90 0.90 0.87													
Wyoming 2.10 2.14 2.70 3.05 3.74 3.98 4.37 4.35 4.60 4.85 5.10 5.35 5.60 5.85 6.10 6.35	Januar												
Permian 5.21 5.02 4.69 4.92 4.97 4.82 4.76 4.62 4.55 4.49 4.42 4.33 4.24 4.15 4.03 3.91	for an												
Southeast New Mexico 1.51 1.43 1.32 1.53 1.65 1.61 1.57 1.57 1.59 1.59 1.57 1.53 1.49 1.45 1.40 1.35													
Texas District 7C 1.05 1.08 0.92 0.91 0.92 0.91 0.95 0.86 0.82 0.78 0.76 0.74 0.72 0.70 0.68 0.66	1												
Texas District 8 2.07 1.93 1.86 1.85 1.75 1.66 1.58 1.51 1.44 1.40 1.35 1.30 1.25 1.20 1.15 1.10	end												
Texas District 8A 0.58 0.59 0.63 0.65 0.64 0.66 0.68 0.70 0.72 0.74 0.76 0.78 0.80 0.80 0.80	•												
San Juan 3.77 3.90 4.18 4.21 4.11 4.06 4.05 4.05 4.02 3.96 3.90 3.84 3.78 3.72 3.66 3.60	777												
Southern Colorado 0.73 0.97 1.11 1.11 1.15 1.23 1.27 1.26 1.24 1.21 1.18 1.15 1.12 1.09 1.06 1.03													
Northwest New Mexico 3.04 2.93 3.07 3.10 2.96 2.83 2.78 2.79 2.78 2.75 2.72 2.69 2.66 2.63 2.60 2.57	IJ												
Fastern IIS 2 51 2 33 2 34 2 35 2 37 2 41 2 39 2 45 2 57 2 69 2 79 2 85 2 85 2 86 2 79 2 71													
Michigan 0.84 0.80 0.70 0.67 0.64 0.60 0.56 0.54 0.51 0.49 0.47 0.45 0.43 0.41 0.39 0.37	73												
Kentucky 0.77 0.73 0.71 0.77 0.71 0.77 0.74 0.75 0.77 0.30 0.37 0.47 0.45 0.41 0.57 0.57	5												
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-												
New York 0.06 0.05 0.05 0.05 0.08 0.10 0.11 0.15 0.20 0.20 0.20 0.25 0.25 0.24 0.24 0.25 0.25	0												
$\begin{array}{cccccccccccccccccccccccccccccccccccc$													
$Virginia \qquad 0.52 0.23 0.37 0.40 0.42 0.42 0.47 0.43 0.47 0.49 0.50 0.50 0.50 0.50 0.47 0.4$	Т												
West Virginia 0.54 0.50 0.50 0.57 0.52 0.51 0.52 0.52 0.55 0.25 0.25 0.25 0.25 0.25													
$\begin{array}{cccccccccccccccccccccccccccccccccccc$													

																	179
							Table	6									20
					Wet Gas	Capacity	Outlook (Bcf per	by Regio _{day)}	on (contir	ued)							2
	<u>1995</u>	<u>1997</u>	<u>1999</u>	<u>2000</u>	<u>2001</u>	<u>2002</u>	<u>2003</u>	<u>2004</u>	<u>2005</u>	<u>2006</u>	<u>2007</u>	<u>2008</u>	<u>2009</u>	<u>2010</u>	<u>2011</u>	<u>2012</u>	שי
West Coast California Oregon	0.84 0.83 0.01	0.84 0.83 0.01	1.07 1.06 0.01	1.04 1.03 0.01	1.04 1.03 0.01	1.00 0.99 0.01	0.93 0.92 0.01	0.87 0.86 0.01	0.84 0.83 0.01	0.81 0.80 0.01	0.78 0.77 0.01	0.75 0.74 0.01	0.72 0.71 0.01	0.69 0.68 0.01	0.65 0.65 0.00	0.62 0.62 0.00	8,67 201
Total Lower 48	55.99	55.05	53.18	53.88	54.88	53.29	53.32	52.71	52.52	52.47	52.38	52.60	52.27	51.85	51.01	50.36	A
Alaska North	1.40	1.40	1.40	1.40	1.40	1.40	1.35	1.30	1.25	1.20	1.15	1.10	1.05	1.00	0.95	0.90	
South	1.40	1.40	1.40	1.40	1.40	1.40	1.35	1.30	1.25	1.20	1.15	1.10	1.05	1.00	0.95	0.90	1775
Total US	57.39	56.45	54.58	55.28	56.28	54.69	54.67	54.01	53.77	53.67	53.53	53.70	53.32	52.85	51.96	51.26	R
Total US, Dry (3)	54.75	53.85	52.07	52.74	53.69	52.17	52.20	51.53	51.30	51.20	51.07	51.23	50.87	50.42	50.42	50.42	777
Source: Cambridge Energy Rese 1 Includes Illinois, Indiana, Mary 2 Includes New Mexico's portior	arch Associates. /land, Missouri, I 1 of the Raton Ba	Nebraska, Nev asin.	ada, South Dal	kota, and Tenn	essee.												-9
3 Dry/wet factor 95.4 percent. Updated April 2005.																	OR

w

Rockies

The Rockies (excluding the San Juan Basin) are expected to provide the greatest offset to the ongoing significant production decline in the Gulf of Mexico. Capacity is projected to increase 2.48 Bcf per day, from 7.12 Bcf per day in 2004 to 9.60 Bcf per day in 2012, representing nearly 35 percent growth from this important region. Wyoming will show the largest growth, adding 2.0 Bcf per day, followed by northern Colorado, adding 0.29 Bcf per day. Record drilling activity in the Rockies supports the projected growth in capacity. More than three quarters of the resource base in the Rockies is unconventional gas. Numerous plays occur in Wyoming, including the tight sands in the Jonah and Pinedale fields and CSM production in the Powder River and other basins in the southern portion of the state.

Production in both the Jonah and Pinedale fields in Wyoming is projected to reach 1.0 Bcf per day within the next five years. Further downspacing in the Jonah field has been held up pending approval of the infill program. CSM production in the Powder River Basin actually declined in 2004 due to poorer-than-expected performance in the shallower Wyodak coals and the slow well permitting process. Eventual development of the deeper Big George coals should reverse this decline and add to production. Record production in the Madden field in the Wind River Basin largely offset these declines in Wyoming in 2004. Developments in the Washakie Basin are also contributing to the growth in Wyoming.

Northern Colorado has three important plays. Production from tight sands in the Piceance Basin should reach 1.0 Bcf per day by 2012, the Wattenburg field in the Denver Julesburg Basin should at least maintain current capacity levels, and CSM in the Raton Basin is expected to increase. Production in Utah is projected to increase slightly in response to infill drilling programs in the Greater Natural Buttes field and to associated gas from the recent significant Covenant oil discovery on the Central Utah Overthrust Belt.

Permian Basin

Capacity in the Permian Basin is projected to decline slowly by 0.71 Bcf per day, from 4.62 Bcf per day in 2004 to 3.91 Bcf per day in 2012. The continuous decline since 2000 has been mitigated by the Montoya-Devonian horizontal play, the Morrow play in southeast New Mexico, tight sand drilling in the Canyon Sand play in the Ozona field area, and other pockets of infill and tight sand activity.

San Juan Basin

Capacity in the San Juan Basin (including southwest Colorado) is projected to slowly decline by 0.45 Bcf per day, from 4.05 Bcf per day in 2004 to 3.60 Bcf per day in 2012. Production in the San Juan Basin is 60 percent CSM. The downsizing of well spacing in the Fruitland coals has reduced the previously projected decline." Conventional" gas production is largely unconventional tight sand gas, and even though numerous undrilled locations exist due to downspacing, this production has remained in a narrow band for the past five years.

Mid-Continent

The Mid-Continent area includes three important areas of increasing capacity: the Barnett shale gas play in north central Texas, the Granite Wash play in the Texas Panhandle, and the Anadarko/Arkoma Basins in Oklahoma. Capacity in the Mid-Continent region is projected to increase slightly from 7.85 Bcf per day in 2004 to 8.26 Bcf per day in 2010 before declining slightly to 8.20 Bcf per day in 2012.

The core Barnett shale area is in RRC-9 and is projected to continue to increase from 0.86 Bcf per day in 2004 to 1.15 Bcf per day in 2012. The Barnett play is expanding into RRC-5 and RRC-7B to the south and west of Fort Worth, Texas, with companies actively acquiring acreage and drilling horizontal wells.

Capacity in Oklahoma is projected to increase slightly from 4.30 Bcf per day in 2005 to 4.5 Bcf per day in 2012. This increase, reflecting the recent high rate of drilling, contrasts with the decline evident since the late 1990s and the reversal of CERA's previous outlooks for a continuing decline. In the Anadarko Basin production from the shallow portion (less than 15,000 feet) is expected to continue to decline; however, deep production (at greater than 15,000 feet) and CSM from the Arkoma Basin are expected to increase.

Capacity in Arkansas is projected to increase moderately assuming the Fayetteville gas shale play materializes as expected. Capacity in the Texas Panhandle (RRC-10) has reversed a recent decline and is projected to increase slowly from 0.97 Bcf per day in 2004 to 1.10 Bcf per day in 2012 from the tight Granite Wash reservoir in the Buffalo Wallow field. Rig count in this area has soared to 63 rigs from 30 rigs in late 2003.

Capacity in Kansas is expected to continue to decline from 1.1 Bcf per day in 2004 to 0.7 Bcf per day in 2012 as the Hugoton field continues to decline.

Eastern United States

Capacity in the eastern United States (largely Michigan, West Virginia, and Pennsylvania) is projected to show slight growth, from 2.45 Bcf per day in 2004 to 2.86 Bcf per day in 2010, with a slight decline to 2.71 Bcf in 2012. New York, with the deep Trenton–Black River play, and West Virginia lead the growth, offset somewhat by the declining Antrim shale production in Michigan.

GAS-RELATED DRILLING

Since the recent low point of 591 rigs reached in April 2002, activity recovered rapidly into mid-2003 before embarking on a slower increase to record levels above the 2001 high of 1,068 rigs in the fall of 2004. After hovering at 1,090 rigs until early March 2005, rig activity has jumped to 1,175 rigs in mid-April 2005, with large increases in Wyoming, South Louisiana, East Texas, and New Mexico (see Figure 11).

1.200 1,150 1,100 1.050 1,000 950 900 850 800 Gas 750 Rig 700 650 Count 600 550 500 450 400 350 300 250 0 Jan lar lar Jan .lan .lar Jar .lan Jan Jan 1996 1998 2002 2004 1995 1997 1999 2000 2001 2003 2005 2006 Source: Baker Hughes Inc. Note: A secondary peak of 1,068 rigs was attained in July 2001. Updated May 2005 40916-2

Figure 11 US Gas-related Drilling

Continuing the lackluster recovery from the 2002 low pace of rig activity, activity in the Gulf of Mexico is hovering at around 90 rigs, 68 rigs below the 2001 peak of 158 rigs and only a few rigs above the 2002 low point. Most of the drop in activity has occurred in the shallow-water shelf, thus contributing to its rapid decline in production. Also between 2001 and October 2004, 47 jackups have left the Gulf.

There is a close coincidence between the leading areas of rig activity and areas with strong projected growth in supply. All of these areas involve unconventional gas, mostly tight sands and some shales (see Table 7). Rig activity in these areas is close to the 2002–05 peaks and in most cases is substantially above the record highs of 2001.

The current rig situation in many areas onshore involves long delays in obtaining a rig as most drilling companies are close to full rig utilization. Compounding the rig availability situation is the lack of manpower for rig crews and for service companies that are falling behind in meeting the soaring demand for well fracing. Evidence of the tight market are the deals between producers and drilling companies to utilize newly built rigs as they become available. Some drilling companies are now building new rigs, and this trend should accelerate as the duration of the drilling boom becomes more obvious. (The shallow CSM wells in the Powder River Basin are drilled with truck-mounted rigs, which are excluded from the active rig count.)

CERA PRIVATE REPO) R T
-------------------	-------

Table 7	
---------	--

Active Rig Count—Leading Areas of Current Activity (includes oil and gas wells)

	1997–98	1999	2001	200	2–05	Latest Count
	Weekly High	Weekly Low	High	Low	High	(April 15, 2005)
Rockies			_ _			<u>, , , , , , , , , , , , , , , , , , , </u>
Colorado	19	7	41	22	73	71
Montana	16	4	13	5	25	26
Wyoming	49	18	66	29	87	76
Total	84	29	120	56	185	173
Texas South and East	396	171	572	251	594	594
RRC 5 (Bossier)	31	9	73	28	73	70
RRC 6 (E.Texas)	45	15	58	21	92	92
RRC 9 (Barnett)	10	4	38	25	49	35
RRC 10 (N. Texas)	31	8	27	6	63	61
North Louisiana		—	_	13	49	45
Oklahoma	109	46	160	68	178	153
Source: Baker Hughes Inc. Updated April 2005.						
Memo:						
Gulf of Mexico	141	84	158	86	125	91

Offshore, the movement of jackups out of the region has tightened the market as deep shelf drilling has become the dominant target. Deep and ultradeepwater rigs are mostly contracted with day rates exceeding \$270,000. At this time there are only 14 rigs worldwide, consisting of drillships and floaters capable of drilling in 10,000 feet of water, and only an additional 22 rigs that can drill in 8,000 feet of water. Two new 7,500-foot water depth semisubmersible rigs are being delivered this spring. Most of the other current newbuilds are shallow-water jackups capable of drilling deep shelf wells. There are also plans to return several departed deepwater rigs to the Gulf of Mexico.

	C	E	R	A	P	R		V	Å	T	Ε	R	E	P	0	i s	ł
--	---	---	---	---	---	---	--	---	---	---	---	---	---	---	---	-----	---

In the deep and ultradeep water, the demand for rigs to continue exploration, appraisal, development, and the test of the large number of leases expiring in the next three years will begin to exceed the supply. This could be a factor in the timing to develop some of the recent ultradeepwater oil discoveries. Numerous large Lower Tertiary oil discoveries have been made recently, inspiring active exploration and appraisal activity along the trend. Since these wells are drilled below 27,000 feet, they require around six months to drill, thus severely limiting the number of wells that can be drilled per year per rig.

ACCESSIBILITY

Increased access to areas with a large potential resource base is critical to the ability to continue to moderate the decline in capacity. US conventional resources offshore that are now off limits are on both coasts and in most of the eastern Gulf of Mexico. These areas have been subject to congressionally imposed drilling moratoria and leasing prohibitions since the early 1980s and currently in effect until 2012—thus preventing industry from exploiting resources that total as much as 79 Tcf (see Figure 12). Even though economic quantities of gas are known to exist in each area, none of this gas is included in the outlook to 2012 owing to CERA's expectation that the intractability of opening these lands pushes their potential production beyond 2012.



	C	E	R	А	<u>G</u>	R	al can	V	A	T		R	E	P	0	R	T
--	---	---	---	---	----------	---	--------	---	---	---	--	---	---	---	---	---	---

Draft legislation circulating in the US Congress would expand state control over offshore energy development and eliminate the blanket moratoria. Virginia has passed legislation, recently rejected by the governor, seeking to end the federal moratorium off its coast and allow seismic activity by working with the state congressional delegation. States would claim up to half of the revenues depending on their distance from the coast. However, the Carolinas, Florida, and California continue to oppose efforts to reinstate drilling.

This outlook assumes that drilling restrictions onshore, mostly in the Rockies, involving an additional potential resource of 69 Tcf, are not materially eased in the near future. An additional 56 Tcf of reserves are available for exploitation; however, these resources are subject to restrictions and delays. Most of this gas is unconventional and would provide an important extension to the tight sand and CSM plays currently under development. In addition, the Bureau of Land Management well permitting process to drill wells on federal lands has been subject to considerable delays, thus delaying new production. An ongoing effort to streamline procedures has been moving forward. However, without radical revision to the National Environmental Policy Act, requirements for environmental impact statements and delays in oil and gas developments will continue to be an aspect of working in the Rockies. In addition, most of the unconventional gas production requires hydraulic facing, and this can be at odds with the Safe Drinking Water Act. Growing western populations and ongoing conflicts between resource- and recreation-based economies also contribute to the difficulties in developing Rockies resources. Currently the focal points in land access issues in the Rockies are the Jack Morrow subbasin in the Green River Basin and the Roan Plateau in the Piceance Basin. In addition, a number of areas in Montana, Utah, and New Mexico also have prospective lands that are off limits to some degree. This is an important factor since it is the Rockies that are being counted on to play the dominant role in offsetting the decline in the Gulf of Mexico.

CANADIAN DRY GAS OUTLOOK: SLOWLY INCREASING GROWTH ACCELERATES BY 2010

Canadian dry gas capacity is projected to increase slowly from 17.0 Bcf per day in 2004 to 18.0 Bcf per day in 2009 before accelerating to 19.5 Bcf per day in 2012 with production from the Mackenzie Delta (see Figure 13). The peak of 17.1 Bcf per day in 2001 was coincident with the attainment of the 0.5 Bcf per day peak from the Sable Island project in the eastern Canadian offshore and peak production from the Ladyfern field in northeast British Columbia. The decline to the 16.6 Bcf per day low point in 2003 was largely the result of the rapid decline in the Ladyfern field from 0.7 Bcf per day to less than 0.2 Bcf per day by 2003.

Production from the WCSB is projected to increase 1.3 Bcf per day, from 16.6 Bcf per day in 2004 to 17.9 Bcf per day in 2012. Most of the increase is related to unconventional gas—tight sands and carbonates, and CSM—plus consecutive years of record gas-related drilling, which is likely to continue (although the drilling season came to an abrupt early end by early March 2005 due to an early warmup).

Offsetting the rapid decline in the shallow gas in eastern Alberta and southeastern Saskatchewan is a series of plays in the deeper portions of the WCSB (see Figure 14). Among these plays are the Greater Sierra Jean Marie reef play in northeastern British Columbia consisting of tight carbonates and requiring underbalanced horizontal drilling,



Figure 13 Canadian Dry Gas Productive Capacity

the Cutbank Ridge play, the prolific Monkman Deep Paleozoic play, Saddle Hills Devonian, "blowdowns" in economically depleted oil fields accounting for over 0.2 Bcf per day of production through 2006, and the Pembina Nisku area revival. The Greater Sierra and Cutbank Ridge plays are tight gas unconventional resource plays located in the Deep Basin and are undergoing active development. Shale gas plays in Canada have not yet been exploited, although several shales have considerable potential, including the Nordegg, the Doig Phosphate, the upper Doig, and the Duvernay. In addition, the recent significant discovery, Tay River, in the foothills area could establish a deep productive trend.

A large contributor to the growth of the WCSB is CSM in southern Alberta. After initial production in 2003, Canadian CSM is projected to increase from an average annual production rate of 0.09 Bcf per day in 2004 to 0.5 Bcf per day in 2007 and 1.25 Bcf per day in 2012, assuming the commercial exploitation of the Mannville coals. Initial production has been largely from the dry, low-cost, shallow Horseshoe Canyon/Belly River coals requiring little time to ramp up to average well production of 0.1 to 0.15 MMcf per day, utilizing the existing infrastructure. With time the CSM play will spread to the deeper, water-wet Mannville coals, which, upon dewatering, will be the major source of CSM. Drilling in the CSM play is expected to almost triple in 2005, with 3,000 wells compared with 1,250 wells in 2004. British Columbia also has attractive CSM potential, although environmental concerns have stalled leasing in the coal-rich Elk Valley in the south near the US border.



Figure 14 Major North American Gas Exploration and Development Hot Spots



It has been estimated that unconventional gas production in the WCSB is currently around 1.5 Bcf per day. With the emphasis on tight reservoirs in the Deep Basin and the anticipated growth of CSM, unconventional gas in Canada could be close to 4 Bcf per day by 2012.

EASTERN CANADIAN OFFSHORE

The eastern Canadian offshore Sable Island project commenced production in 2000, increasing to a 0.5 Bcf per day peak in 2001 from three initial fields (Thebaud, Venture, and North Triumph) (see Figure 15). However, production from these fields has been declining



Figure 15 Nova Scotia Offshore Significant Discovery Areas

at unexpectedly high rates, and several bouts of reserve writedowns have occurred. Even with the development of the nearby Alma and South Venture fields, each adding 0.13 and 0.15 Bcf per day in late 2003 and late 2004, respectively, production has struggled to remain at around 0.4 to 0.45 Bcf per day. The last field to be developed, Glenelg, has been deemed uneconomic. In mid-2006 new compression will be added to boost output from around 0.4 Bcf per day to 0.45 Bcf per day. The deep Panuke discovery, containing some sulfur, which will require high-cost offshore processing, is not as yet sanctioned for development. As more reserves are sought to render Panuke economic, the discovery is assumed in this outlook to be developed by 2009, increasing production in the Sable area to 0.5 Bcf per day in 2010.

CERA PRIVATE REPO	R T
-------------------	-----

In the area around the Sable fields numerous single well discoveries were made in the 1980s. These discoveries could eventually be connected to slow the decline in the Sable complex. The Chebucto and Citnalta discoveries are the largest.

Since 2000, exploration drilling in the Sable area has not been successful, with the exception of the Annapolis G-24 Wildcat, drilled in 2002 in 5,500 feet of water. Companies are allowing leases to expire, with only a few appraisal wells expected to be drilled in 2005.

Initial associated gas production from the Hibernia area, offshore Newfoundland, is projected to commence by around 2015 from the Whiterose field. CERA expects that this gas will not be brought to market by pipeline; rather, it may be transported as compressed natural gas. The other producing oil fields, Hibernia and Terra Nova, contain significant reserves of associated gas.

To the north on the Labrador shelf, several large discoveries were made in the late 1970s and early 1980s. These discoveries are Bjarni, Gudrid, and Snorri, with combined resources of 5 Tcf. The threat of iceberg scouring of the shallow shelf bottom presents a development concern.

MACKENZIE DELTA/BEAUFORT SEA

In the Mackenzie Delta in northwest Canada, 6 Tcf of proved gas reserves, discovered in the 1970s, is assumed to be available in three main fields: Taglu, Parsons Lake, and Niglintgak. In 2001 the Tuk M-18 discovery was made, testing 30 MMcf per day with reserves of 0.5 Tcf. Since 1999, much of the area, including the offshore Beaufort Sea, has been leased with significant work commitments requiring numerous wildcats to be drilled in the next few years. This past winter several wildcats were drilled. This gas awaits a pipeline to access the market.

The companies that constitute the Mackenzie Delta Producers Group have submitted to the Canadian National Energy Board a proposal for regulatory review of the 760-mile Mackenzie Valley pipeline extending to northern Alberta. Recently, the negotiations have not been going smoothly, putting pressure on the schedule for completion of the pipeline but assuming all the approvals are reached by late 2006, a decision will then be made whether to proceed, and pipeline construction could begin in 2007 with completion by early 2010. Initial production is projected for late 2010, ramping up to 1.2 Bcf per day in 2011. The recent quickening pace of the Alaska gas pipeline project could pressure the Mackenzie pipeline, potentially resulting in delays.*

^{*}See the CERA Alert Mackenzie Valley Pipeline Delayed: Will History Repeat Itself?

BRITISH COLUMBIA OFFSHORE

An attempt to open the British Columbia offshore to exploration in the Queen Charlotte Basin is under way. Although the BC government is ready to open the area, the federal government is delaying a decision while unanswered questions are addressed. The controversy involves a 33-year old Canadian government freeze on the offshore. It is possible the fate of the moratorium could be determined in 2006; if it is lifted, the establishment of a regulatory regime and local agreements could allow seismic by 2010 and drilling by 2012.

CONCLUSION

North American gas capacity from 2004 to 2012 will remain on an undulating plateau, with a steeper decline evident past 2010. The 2.6 Bcf per day decline in the United States from 2004 to 2012 is mostly offset by a 2.5 Bcf per day increase in Canada. Generally flat North American capacity contrasts with a growing appetite for gas consumption, primarily in the power sector, which translates into increasing LNG import requirements.

DOE Meeting on BALANCING NATURAL GAS SUPPLY AND DEMAND

Natural Gas Supply Overview

December 19-20, 2005

Natural Gas Supply Overview Overview of Presentation

BACKGROUND

- NPC North American Supply Outlook
- Supply Update Actual vs NPC

KEY ISSUES

- Gulf of Mexico
- Western Canadian Sedimentary Basin
- Non-Conventional Gas Basins
- Arctic Gas
- LNG Imports
- Access Considerations
- Technology Considerations

RECOMMENDATIONS & PROGRESS

Key NPC Study Supply Recommendations

Natural Gas Supply Overview North American Supply Outlook – NPC 2003 Study (Reactive Path)



- In the 1990's, natural gas production increased steadily to meet growing demand.
- After 2000, natural gas supplies became "tight" and have continued to "tighten".
- · Looking forward, traditional sources of gas supply are expected to remain essentially constant.

3

• LNG and Arctic natural gas will be essential for meeting future growth of demand.

Natural Gas Supply Overview Supply Update – Actual vs NPC Projections (Reactive Path)



- Conventional gas production had been declining, partially offset by non-conventional gas development.
- Production declines are most noticeable in offshore Gulf of Mexico (GOM) (Shelf and Deepwater) and Western Canada.
- Non-conventional gas production is higher due to increased drilling and new/expanded tight gas and gas shale plays.

4

Natural gas drilling is at record levels; change in type of reserves added (higher R/P reserves) has limited production response.

Natural Gas Supply Overview Gulf of Mexico Basin (Shelf/Deepwater)



- GOM shelf gas production has declined by nearly 2.6 Bcf/day, since 2001 (excluding effect of hurricanes); opportunities continue to become smaller; hurricanes reduced gas production by an additional 1.7 Bcf/day in 2005.
- Deepwater projects are delayed and less gas prone than expected.
- Sustaining GOM production will be challenging given recent disappointing exploration results.

Natural Gas Supply Overview Western Canada Sedimentary Basin (WCSB)



- WCSB gas production has fallen below NPC expectations, despite record gas well drilling.
- Well productivity continues its long-term decline, dominated by shallow in-fill drilling.
- Non-conventional gas supplies are less developed than in the US; industry is beginning to develop CBM, consistent with NPC expectations.

Natural Gas Supply Overview Non-Conventional Basins





U.S.	Natural	Gas	Well	Drilling
------	---------	-----	------	----------

	Total Gas Wells	Actual Unconventional Gas Wells	NPC Unconventional Gas Wells*				
	(Wells/Yr)	(Wells/Yr)	(Wells/Yr)				
2001	22,100	12,700	10,600				
2002	16,200	11,900	9,900				
2003	19,700	13,400	9,400				
2004	22,700	14,100	9,200				

*Includes "low perm conventional"

- Non-conventional gas is being intensely developed with activity levels above NPC • expectations.
- Resource base for certain gas shale and tight gas basins may be higher than in ٠ NPC assessments.
- CBM development is in-line with NPC expectations; permitting constraints are ٠ limiting pace of drilling.

Natural Gas Supply Overview Arctic Gas





Mackenzie Gas Project

- NPC Assumptions - 2009 start-up at 1 Bcf/d; expansion to 1.5 Bcf/d in 2015
- Outlook - Project scope consistent; start-up timing likely delayed by 2 years

Alaska Gas Pipeline

- NPC Assumptions - 2013 start-up at 2.5 Bcf/d; full volume of 4 Bcf/d in 2014
- Outlook - Project scope consistent; start-up timing likely 1-2 years later
Natural Gas Supply Overview LNG Imports



*Source: EIA AEO 2005.

**Includes LNG plants at Altamira and Baja in Mexico.

- Recent LNG terminal expansions and developments are in line with NPC expectations.
- Considerable differences exist on the longer-term outlook for LNG imports.

Natural Gas Supply Overview Access Considerations



- Studies are underway to update access restrictions and resources impacted.
- Pace of permitting has slowed development in the Rockies, particularly for Powder River CBM.

Technology Progress "Levers" (% Annual Improvement)

	NPC Study*	AEO 2006
Drilling Efficiency	1.81%	0.89%
EUR/Well (Technology Effect)		
New Field Discoveries (Onshore)	0.87%	0%
Unconventional Gas Wells	0.87%	0%/0.25%**
Operating Efficiency	1.00%	0.52%

*Average parameters.

**0% for mature gas plays; 0.25% for immature gas plays.

- For two decades, progress in natural gas E&P technology countered the effects of resource maturity and depletion.
- For the past several years, the pace of technology progress in natural gas E&P technology appears to have slowed.
- The decline in technology progress is reflected in reductions in the "technology levers" used in recent EIA gas supply models, compared to those used by the NPC Study.

11

Increase Supply Diversity

- Increase Access and Reduce Permitting Impediments to Development of Lower-48
 Natural Gas Resources
 - + Administration efforts to expedite lease sales and permitting (NGOs in opposition)
 - Implementation in state/field offices limited by lack of resources
 - Lack of progress on access to OCS

• Enact Enabling Legislation for an Alaska Gas Pipeline

- + Alaska Natural Gas Pipeline Act enacted (October, 2004)
- + State of Alaska negotiations with ANS producers are well advanced

Process LNG Project Permit Applications within 1 Year

- + FERC demonstrating progress toward permitting efficiency
- + Center for LNG (CLNG) has provided LNG education and advocacy
- + EPAct gives FERC primary authority for LNG terminals
- Evaluate the appropriateness of funding levels for natural supply R&D
 - + EPAct authorizes R&D program for ultra-deep and unconventional gas resources, plus marginal wells and methane hydrates.

Policy recommendations in NPC Study remain sound, important and timely. Should more robust recommendations be pursued?

EXTENDED RANGE FORECAST OF ATLANTIC SEASONAL HURRICANE ACTIVITY, INDIVIDUAL MONTHLY ACTIVITY AND U.S. LANDFALL STRIKE PROBABILITY FOR 2006

We foresee an active Atlantic basin tropical cyclone season in 2006; however, we have reduced our projection for 2006 hurricane activity from our earlier forecasts. Landfall probabilities for the 2006 hurricane season are projected to be above their long-period averages for the East Coast and near their long-period averages for the Gulf Coast.

(as of 3 August 2006)

By Philip J. Klotzbach¹ and William M. Gray²

with special assistance from William Thorson³

This forecast as well as past forecasts and verifications are available via the World Wide Web at <u>http://hurricane.atmos.colostate.edu/Forecasts</u>

Emily Wilmsen, Colorado State University Media Representative, (970-491-6432) is available to answer various questions about this forecast

Department of Atmospheric Science Colorado State University Fort Collins, CO 80523 Email: <u>amie@atmos.colostate.edu</u>

¹ Research Associate

² Professor Emeritus of Atmospheric Science

³ Research Associate

Forecast Parameter and 1950-2000 Climatology (in parentheses)	Issue Date 6 Dec 2005	Issue Date 4 April 2006	Issue Date 31 May 2006	Observed Activity Through July 2006	Forecast Activity After 1 August	Total Seasonal Forecast
Named Storms (NS) (9.6)	17	17	17	2	13	15
Named Storm Days (NSD) (49.1)	85	85	85	5.5	69.5	75
Hurricanes (H) (5.9)	9	9	9	0	7	7
Hurricane Days (HD) (24.5)	45	45	45	0	35	35
Intense Hurricanes (IH) (2.3)	5	5	5	0	3	3
Intense Hurricane Days (IHD) (5.0)	13	13	13	0	8	8
Net Tropical Cyclone Activity (NTC) (100%)	195	195	195	6	134	140

ATLANTIC BASIN SEASONAL HURRICANE FORECAST FOR 2006

POST 1-AUGUST PROBABILITIES FOR AT LEAST ONE MAJOR (CATEGORY 3-4-5) HURRICANE LANDFALL ON EACH OF THE FOLLOWING COASTAL AREAS:

- 1) Entire U.S. coastline 73% (average for last century is 52%)
- U.S. East Coast Including Peninsula Florida 64% (average for last century is 31%)
- 3) Gulf Coast from the Florida Panhandle westward to Brownsville 26% (average for last century is 30%)
- 4) Above-average major hurricane landfall risk in the Caribbean

Notice of Author Changes

By William Gray

The order of the authorship of these forecasts has been reversed from Gray and Klotzbach to Klotzbach and Gray. After 22 years (since 1984) of making these forecasts, it is appropriate that I step back and have Phil Klotzbach assume the primary responsibility for our project's seasonal, monthly and landfall probability forecasts. Phil has been a member of my research project for the last five years and has been second author on these forecasts for the last four years. I have greatly profited and enjoyed our close personal and working relationships.

Phil is now devoting more time to the improvement of these forecasts than I am. I am now giving more of my efforts to the global warming issue and in synthesizing my projects' many years of hurricane and typhoon studies.

Phil Klotzbach is an outstanding young scientist with a superb academic record. I have been amazed at how far he has come in his knowledge of hurricane prediction since joining my project five years ago. I foresee an outstanding future for him in the hurricane field. I expect he will make many new forecast innovations and skill improvements in the coming years. I plan to continue to be closely involved in the issuing of these forecasts for the next few years.

ABSTRACT

Information obtained through July 2006 indicates that the 2006 Atlantic hurricane season will be more active than the average 1950-2000 season; however, we have reduced our prediction from our earlier forecasts. We estimate that 2006 will have about 7 hurricanes (average is 5.9), 15 named storms (average is 9.6), 75 named storm days (average is 49.1), 35 hurricane days (average is 24.5), 3 intense (Category 3-4-5) hurricanes (average is 2.3) and 8 intense hurricane days (average is 5.0). The probability of U.S. major hurricane landfall is estimated to be about 40 percent above the long-period average. Landfall probabilities are based upon our expectation for another active season as well as analysis of our new steering current predictors for the East Coast and Gulf Coast of the United States.

We expect Atlantic basin Net Tropical Cyclone (NTC) activity in 2006 to be about 140 percent of the long-term average. This early August forecast is based on a newly devised extended range statistical forecast procedure which utilizes 57 years of past global reanalysis data. Analog predictors are also utilized. This 3 August forecast reduces our forecast from our early December 2005, early April 2006 and late May 2006 predictions due to small changes in June-July atmospheric and oceanic fields that indicate conditions are less favorable for tropical cyclone development in the tropical Atlantic. These changes include above-average tropical Atlantic sea level pressure, above-average tropical Atlantic trade wind strength and a decreasing trend in tropical Atlantic sea surface temperatures. Sea surface temperatures have also risen slightly in the eastern equatorial Pacific. We expect an active hurricane season for the Atlantic basin, but we do not foresee nearly as active a season as was experienced in 2004 and 2005. Seasonal updates of our 2006 forecast will be issued on 1 September and 3 October. A seasonal summary and forecast verification will be issued in late November.

<u>Acknowledgment</u>

We are grateful to the National Science Foundation (NSF) and Lexington Insurance Company (a member of the American International Group (AIG)) for providing partial support for the research necessary to make these forecasts. We also thank the GeoGraphics Laboratory at Bridgewater State College (MA) for their assistance in developing the Landfalling Hurricane Probability Webpage (available online at <u>http://www.e-transit.org/hurricane</u>).

The second author gratefully acknowledges valuable input to his CSU research project over many years by former graduate students and now colleagues Chris Landsea, John Knaff and Eric Blake. We also thank Professors Paul Mielke and Ken Berry of Colorado State University for much statistical analysis and advice over many years.

DEFINITIONS

Atlantic Basin - The area including the entire North Atlantic Ocean, the Caribbean Sea, and the Gulf of Mexico.

 $\underline{EI Niño} - (EN) A 12-18$ month period during which anomalously warm sea surface temperatures occur in the eastern half of the equatorial Pacific. Moderate or strong El Niño events occur irregularly, about once every 3-7 years on average.

<u>Hurricane</u> – (H) A tropical cyclone with sustained low-level winds of 74 miles per hour (33 ms⁻¹ or 64 knots) or greater.

<u>Hurricane Day</u> – (HD) A measure of hurricane activity, one unit of which occurs as four 6-hour periods during which a tropical cyclone is observed or estimated to have hurricane intensity winds.

<u>Hurricane Destruction Potential</u> – (HDP) A measure of a hurricane's potential for wind and storm surge destruction defined as the sum of the square of a hurricane's maximum wind speed (in 10^4 knots²) for each 6-hour period of its existence.

Intense Hurricane - (IH) A hurricane which reaches sustained low-level winds of at least 111 mph (96 knots or 50 ms⁻¹) at some point in its lifetime. This constitutes a category 3 or higher on the Saffir/Simpson scale (also termed a "major" hurricane).

Intense Hurricane Day – (IHD) Four 6-hour periods during which a hurricane has an intensity of Saffir/Simpson category 3 or higher.

Named Storm – (NS) A hurricane or a tropical storm.

<u>Named Storm Day</u> – (NSD) As in HD but for four 6-hour periods during which a tropical cyclone is observed (or is estimated) to have attained tropical storm intensity winds.

<u>NTC</u> – <u>Net Tropical Cyclone Activity</u> – Average seasonal percentage mean of NS, NSD, H, HD, IH, IHD. Gives overall indication of Atlantic basin seasonal hurricane activity.

ONR - Previous year October-November SLPA of subtropical Ridge in eastern Atlantic between 20-30°W.

<u>QBO</u> – Quasi-<u>B</u>iennial <u>O</u>scillation – A stratospheric (16 to 35 km altitude) oscillation of equatorial east-west winds which vary with a period of about 26 to 30 months or roughly 2 years; typically blowing for 12-16 months from the east, then reversing and blowing 12-16 months from the west, then back to easterly again.

<u>Saffir/Simpson (S-S) Category</u> – A measurement scale ranging from 1 to 5 of hurricane wind and ocean surge intensity. One is a weak hurricane; whereas, five is the most intense hurricane.

<u>SLPA</u> – <u>Sea Level Pressure Anomaly</u> – The deviation of sea level pressure from observed long-term average conditions.

 $\underline{SOI} - \underline{S}$ outhern \underline{O} scillation \underline{I} ndex – A normalized measure of the surface pressure difference between Tahiti and Darwin.

 $\underline{SST(s)} - \underline{Sea} \underline{Surface} \underline{T}emperature(s)$

 $\underline{SSTA(s)} - \underline{Sea} \underline{Surface} \underline{T}emperature(s) \underline{A}nomalies$

<u>Tropical Cyclone</u> – (TC) A large-scale circular flow occurring within the tropics and subtropics which has its strongest winds at low levels; including hurricanes, tropical storms and other weaker rotating vortices.

<u>Tropical Storm</u> – (TS) A tropical cyclone with maximum sustained winds between 39 (18 ms⁻¹ or 34 knots) and 73 (32 ms⁻¹ or 63 knots) miles per hour.

 $\underline{ZWA} - \underline{Z}$ onal \underline{W} ind \underline{A} normaly - A measure of the upper level (~200 mb) west to east wind strength. Positive anomaly values mean winds are stronger from the west or weaker from the east than normal.

1 knot = 1.15 miles per hour = 0.515 meters per second

1 Introduction

This is the 23rd year in which the CSU Tropical Meteorology Project has made forecasts of the upcoming season's Atlantic basin hurricane activity. Our forecast team has shown that a sizable portion of the year-to-year variability of Atlantic tropical cyclone (TC) activity can be hindcast with skill exceeding climatology. These forecasts are based on a statistical methodology derived from 57 years of past data and a separate study of analog years which have similar precursor circulation features to the current season. Qualitative adjustments are added to accommodate additional processes which may not be explicitly represented by our statistical analyses. These evolving forecast techniques are based on a variety of climate-related global and regional predictors previously shown to be associated with the forthcoming seasonal Atlantic basin tropical cyclone activity and landfall probability. We believe that seasonal forecasts must be based on methods that show significant hindcast skill in application to long periods of prior data. It is only through hindcast skill that one can demonstrate that seasonal forecast skill is possible. This is a valid methodology provided that the atmosphere continues to behave in the future as it has in the past.

A variety of atmosphere-ocean conditions interact with each other to cause yearto-year and month-to-month hurricane variability. The interactive physical linkages between these many physical parameters and hurricane variability are complicated and cannot be well elucidated to the satisfaction of the typical forecaster making short range (1-5 days) predictions where changes in the momentum fields are the crucial factors. Seasonal and monthly forecasts, unfortunately, must deal with the much more complicated interaction of the energy-moisture fields with the momentum fields.

We find that there is a rather high (50-60 percent) degree of year-to-year hurricane forecast potential if one combines 4-5 semi-independent atmospheric-oceanic parameters together. The best predictors (out of a group of 4-5) do not necessarily have the best individual correlations with hurricane activity. The best forecast parameters are those that explain the portion of the variance of seasonal hurricane activity that is not associated with the other variables. It is possible for an important hurricane forecast parameter to show little direct relationship to a predictand by itself but to have an important influence when included with a set of 4-5 other predictors.

In a five-predictor empirical forecast model, the contribution of each predictor to the net forecast skill can only be determined by the separate elimination of each parameter from the full five predictor model while noting the hindcast skill degradation. When taken from the full set of predictors, one parameter may degrade the forecast skill by 25-30 percent, while another degrades the forecast skill by only 10-15 percent. An individual parameter that, through elimination from the forecast, degrades a forecast by as much as 25-30 percent may, in fact, by itself, show much less direct correlation with the predictand. A direct correlation of a forecast parameter may not be the best measure of the importance of this predictor to the skill of a 4-5 parameter forecast model. This is the nature of the seasonal or climate forecast problem where one is dealing with a very complicated atmospheric-oceanic system that is highly non-linear. There is a maze of changing physical linkages between the many variables. It is impossible to understand how all these processes interact with each other. It follows that any seasonal or climate

forecast scheme showing significant hindcast skill must be empirically derived. No one can completely understand the full complexity of the atmosphere-ocean system or develop a reliable scheme for forecasting the myriad non-linear interactions in the full-ocean atmosphere system.

2 Newly-Developed 1 August Forecast Scheme

We have recently developed a new 1 August statistical seasonal forecast scheme for prediction of Net Tropical Cyclone (NTC) activity. This scheme was developed on NOAA/NCEP reanalysis data from 1949-1989. It was then tested on independent data from 1990-2005 to insure that the forecast shows similar skill in this later forecast period. As a rule, predictors were only added to the scheme if they explained an additional three percent of the variance of NTC in both the dependent period (1949-1989) and the independent period (1990-2005)

The pool of four predictors for this new extended range forecast is given and defined in Table 1. The location of each of these new predictors is shown in Fig. 1. Strong statistical relationships can be extracted via combinations of these predictive parameters (which are available by the end of July), and quite skillful Atlantic basin forecasts of NTC activity for the season can be made if the atmosphere and ocean continue to behave in the future as they have during the hindcast period of 1949-2005. Sixty percent of the variance in NTC is explained over the 1949-2005 period, and on independent data (1900-1948), using the same equations and predictors, 49 percent of the variance is explained. This is comparable to what would be expected with independent data as a jackknife regression technique on the 1949-2005 period indicated 52 percent of the variance could be explained. This gives us increased confidence that the new statistical scheme should be of considerable value in the future.

Our statistical forecast for the other predictors (i.e., named storms, hurricanes) is then adjusted by the predicted statistical value of NTC. For example, if a typical season has 10 named storms and the predicted NTC value is 120%, the predicted number of named storms for the season would be 12 (10 * 120%).

7

August Seasonal Forecast Predictors



Figure 1: Location of predictors for the 1 August forecast for the 2006 hurricane season.

Table 1: Listing of 1 August 2006 predictors for this year's hurricane activity. A plus (+) means that positive values of the parameter indicate increased hurricane activity this year, and a minus (-) means that positive values of the parameter indicate decreased hurricane activity this year. The combination of these four predictors calls for about an average hurricane season.

Predictor	Values for 2006 Forecast
1) June-July SST (20-40°N, 15-35°W) (+)	+0.8 SD
2) June-July SLP (10-25°N, 10-60°W) (-)	+0.3 SD
3) June-July NINO3 Index (5°S-5°N, 90-150°W) (-)	+0.3 SD
4) Pre-1 August Named Storm Days – South of 23.5°N, East of 75°W	0

Table 2 shows our statistical forecast for the 2006 hurricane season and the comparison of this forecast with climatology (average season between 1950-2000). Our statistical forecast is calling for about average activity this year, which adds additional support for the reduction of our forecast from our previous early-season predictions.

Table 2: 1 August statistical forecast for 2006.

	Statistical Forecast
Predictands and Climatology	Numbers
Named Storms (NS) – 9.6	10.0
Named Storm Days (NSD) - 49.1	51.1
Hurricanes (H) – 5.9	6.1
Hurricane Days (HD) – 24.5	25.5
Intense Hurricanes (IH) – 2.3	2.4
Intense Hurricane Days (IHD) – 5.0	5.2
Net Tropical Cyclone Activity (NTC) – 100	104

2.1 Physical Associations among Predictors Listed in Table 1

Brief descriptions of our 1 August predictors follow:

Predictor 1. June-July SST in the Northeastern Subtropical Atlantic (+)

(20°-40°N, 15-35°W)

Warm sea surface temperatures in this area in June-July correlate very strongly with anomalously warm sea surface temperatures in the tropical Atlantic throughout the upcoming hurricane season. Anomalously warm sea surface temperatures are important for development and intensification of tropical cyclones by infusing more latent heat into the system (Goldenberg and Shapiro 1998). In addition, associated with anomalously warm June-July SSTS are weaker trade winds. Weaker trade winds cause less evaporation and upwelling of the sea surface which therefore feeds back into keeping the tropical Atlantic warm. In addition, weaker trade winds imply that there is less vertical wind shear across the tropical Atlantic. Weak wind shear is favorable for tropical cyclone development and intensification (Gray 1968, Gray 1984a, Goldenberg and Shapiro 1996, Knaff et al. 2004). Lastly, there is a strong positive correlation (~0.5) between anomalously warm June-July SSTs in the subtropical northeastern Atlantic and low sea level pressures in the tropical Atlantic and Caribbean during the heart of the hurricane season. Low sea level pressures imply decreased subsidence and enhanced mid-level moisture. Both of these conditions are favorable for tropical cyclogenesis and intensification (Knaff 1997).

Predictor 2. June-July SLP in the Tropical Atlantic (-)

(10-25°N, 10-60°W)

Low sea level pressure in the tropical Atlantic in June-July implies that early summer conditions in the tropical Atlantic are favorable for an active tropical cyclone season with increased vertical motion, decreased stability and enhanced mid-level moisture. There is a strong auto-correlation (r > 0.5) between June-July sea level pressure anomalies and August-October sea level pressure anomalies in the tropical Atlantic. Low sea level pressure in the tropical Atlantic also correlates quite strongly (r > 0.5) with reduced trade winds (weaker easterlies) and anomalously easterly upper-level winds (weaker westerlies). The combination of these two features implies weaker vertical wind shear and therefore more favorable conditions for tropical cyclone development in the Atlantic (Gray 1968, Gray 1984a, Goldenberg and Shapiro 1996).

Predictor 3. June-July Nino3 Index (-)

(5°S-5°N, 90-150°W)

Cool sea surface temperatures in the Nino3 region during June-July imply that a La Niña event is currently present. In general, positive or negative anomalies in the Nino3 region during the early summer persist throughout the remainder of the summer and fall. El Niño conditions shift the center of the Walker Circulation eastward which causes increased convection over the central and eastern tropical Pacific. This increased convection in the central and eastern Pacific manifests itself in anomalous upper-level westerlies across the Caribbean and tropical Atlantic, thereby increasing vertical wind shear and reducing Atlantic basin hurricane activity. The relationship between ENSO and Atlantic hurricane activity has been well-documented in the literature (e.g., Gray 1984a, Goldenberg and Shapiro 1996, Elsner 2003, Bell and Chelliah 2006).

Predictor 4. Named Storm Days South of 23.5°N, East of 75°W (+)

Most years do not have named storm formations in June and July in the tropical Atlantic; however, if deep tropical formations do occur, it indicates that a very active hurricane season is likely. For example, the six years with the most named storm days in the deep tropics in June and July (since 1949) are 1966, 1969, 1995, 1996, 1998 and 2005. All six of these seasons were very active. When storms form in the deep tropics in the early part of the hurricane season, it indicates that conditions are already very favorable for TC development. In general, the start of the hurricane season is restricted by thermodynamics (warm SSTs, unstable lapse rates), and therefore deep tropical activity early in the hurricane season implies that the thermodynamics are already quite favorable for TC development. Also, this predictor's correlation with seasonal NTC is 0.53 over the 1949-2005 period, and when tested on independent data (1900-1948), the correlation actually improves to 0.63, which gives us increased confidence in its use as a seasonal predictor.

2.2 Hindcast Skill

Table 3 shows the degree of hindcast variance (r^2) explained by our new 1 June forecast scheme based on our 41-year developmental dataset (1949-1989), our skill on the independent dataset (1990-2005), and our skill over the entire dataset (1949-2005).

Note that the scheme generally shows improved skill in the independent dataset, which lends increased confidence in its use.

Table 3: Variance (r^2) explained for our new 1 August forecast scheme for NTC in the developmental dataset (1949-1989), in the independent dataset (1990-2005), and over the entire dataset (1949-2005).

Variable	Variance (r ²) Explained	Variance (r ²) Explained	Variance (r ²) Explained
	Developmental Dataset	Independent Dataset	Entire Dataset
	(1949-1989)	(1990-2005)	(1949-2005)
NTC	0.52	0.76	0.60

3 Predictions of Individual Monthly Atlantic TC Activity for August, September and October

A new aspect of our climate research is the development of TC activity predictions for individual months. There are often monthly periods within active and inactive Atlantic basin hurricane seasons which do not conform to the overall season. For example, 1961 was an active hurricane season (NTC of 222), but there was no TC activity during August; 1995 had 19 named storms, but only one named storm developed during a 30-day period during the peak of the hurricane season between 29 August and 27 September. By contrast, the inactive season of 1941 had only six named storms (average 9.3), but four of them developed during September. During the inactive 1968 hurricane season, three of the eight named storms formed in June (June average is 0.5).

We have conducted new research to see how well various sub-season or individual monthly trends of TC activity can be forecast. This effort has recently been documented in papers by Blake and Gray (2004) for August and Klotzbach and Gray (2003) for September. These reports show that it is possible to develop skillful prediction schemes for August-only and September-only Atlantic basin tropical cyclone activity. We have also developed a separate October forecast scheme. On average, August, September, and October have about 26%, 48%, and 17% or 91% of the Atlantic basin's NTC activity. Initial August-only forecasts have now been made by Blake for the last six years (2000-2005), and the verification of these forecasts looks promising. The verification of the September-only and October-only forecasts also appears to show skill.

3.1 Independent August-Only Statistical Forecast

Figure 2 and Table 4 list the predictors used in the August-only hindcast (Blake and Gray 2004) for each of the seven different forecast parameters. The table also shows hindcast skill for the 51-year period 1950-2000, as well as the independent jackknife hindcast skill over this period. Table 5 gives the predictor values for August 2006. Table 6 gives our independent statistical prediction for August 2006. These predictors indicate well above-average activity for August 2006. The most skillful August predictors, in

general, call for a very active month, so we are calling for considerable activity during the month.



Predictor Map

Figure 2: Global map showing locations of August-only TC predictors. Table 4 provides a listing and description of these predictors. The numbers in the boxes are keyed to the descriptions given in Table 4. The numbers in parentheses beneath each box indicate how many individual parameters (NS, NSD, etc.) are obtained from each predictor.

Table 4: Listing of predictors chosen for each forecast parameter and the total hindcast variance explained by these predictors for the August-only forecast. The name and atmospheric parameter utilized in each predictor is given below – where the number is keyed to Fig. 2.

	Number			Estimated
Forecast	of	Predictors Chosen	Variability Explained by	Independent Forecast
Parameter	Predictors	From Table	Hindcast (r ²) (1949-1999)	Skill (Jackknife)
NS	5	3, 6, 7, 9, 11	0.55	0.41
NSD	5	1, 2, 3, 8, 10	0.71	0.61
Η	4	1, 2, 8, 10	0.57	0.47
HD	5	3, 4, 8, 9, 10	0.69	0.59
IH	5	1, 3, 5, 8, 12	0.68	0.59
IHD	5	1, 4, 5, 6, 9	0.78	0.72
NTC	5	1, 2, 8, 10, 12	0.74	0.66

Table 5: August 2006 predictors. The sign of the predictor associated with increased tropical cyclone activity is in parentheses.

	2006	
	Observed	Effect on 2006 Hurricane
Predictors	Values	Season
Galapagos July 200 mb V (-)	-1.0 SD	Enhance
Bering Sea July SLP (-)	-1.0 SD	Enhance
Atlantic Ocean July SLP (-)	+0.8 SD	Suppress
SE Pacific July 200 mb U (-)	+0.2 SD	Suppress
S. Indian Ocean July 500 mb Geo Ht. (-)	+1.0 SD	Suppress
Coral Sea July 200 mb U (+)	+0.7 SD	Enhance
Galapagos July 200 MB U (-)	-0.5 SD	Enhance
North Greenland June 200 MB U (+)	+0.5 SD	Enhance
Northwest Pacific June SLP (+)	+1.0 SD	Enhance
S. Atlantic Ocean April SLP (-)	-0.6 SD	Enhance
Scandinavia February SLP (-)	+0.4 SD	Suppress
SW USA January SLP (-)	-1.0 SD	Enhance

Table 6: Independent August-only prediction of 2006 hurricane activity based on Blake and Gray (2004). August climatology is shown in parentheses.

	Statistical	
Parameter	Model	Qualitative Adjustment
NS	3.3 (2.8)	4
NSD	21.1 (11.8)	22
Н	2.9 (1.6)	3
HD	8.1 (5.7)	11
IH	0.7 (0.6)	1
IHD	2.0 (1.2)	3
NTC	53.6 (26.1)	50

3.2 Independent September-Only Statistical Forecast

Figure 3 and Table 7 list the predictors used in the September-only hindcast (Klotzbach and Gray 2003) for each of the seven different forecast parameters. The table also shows hindcast skill for the 51-year period 1950-2000, as well as the independent jackknife hindcast skill over this period. Table 8 gives the predictor values for September 2006. Table 9 gives our independent statistical prediction for September 2006. Predictor values for September 2006 are mixed, so our final forecast is calling for slightly above-average activity for the month.



Figure 3: Predictors selected for the end of July forecast of September tropical cyclone activity. The numbers in each area are keyed to the description given in Table 7.

Table 7: Listing of predictors chosen for each forecast parameter and the total hindcast variance explained by these predictors for the September-only forecast. The name and atmospheric parameter utilized in each predictor is given below – where the number is keyed to Fig. 3.

	Number			Estimated
Forecast	of	Predictors Chosen	Variability Explained by	Independent Forecast
Parameter	Predictors	From Table	Hindcast (r ²) (1950-2000)	Skill (Jackknife)
NS	3	2, 3, 5	0.29	0.19
NSD	5	2, 3, 4, 5, 8	0.54	0.44
Η	3	2, 3, 8	0.38	0.28
HD	5	2, 3, 4, 5, 8	0.60	0.51
IH	5	1, 2, 3, 5, 9	0.63	0.53
IHD	4	3, 4, 5, 9	0.63	0.54
NTC	5	2, 3, 4, 5, 9	0.75	0.68

Table 8: September 2006 predictors. The sign of the predictor associated with increased tropical cyclone activity is in parentheses.

	2006	
	Observed	Effect on 2006 Hurricane
Predictors	Values	Season
South Atlantic April 1000 mb U (-)	-0.6 SD	Enhance
Northeast Asia July 200 mb Geo Ht. (+)	+1.3 SD	Enhance
Tropical Atlantic July 1000 MB U (+)	-0.8 SD	Suppress
West Africa February 1000 mb U (-)	+0.8 SD	Suppress
Northeast Siberia April 200 mb U (-)	-1.2 SD	Enhance
Central Africa May 200 mb V (+)	-1.4 SD	Suppress
West Pacific Jan-Feb 200 mb U (-)	+0.4 SD	Suppress

Table 9: Independent September-only prediction of 2006 hurricane activity based on Klotzbach and Gray (2003). September climatology is shown in parentheses.

Parameter	Statistical Model	Qualitative Adjustment
NS	4.1 (3.4)	5
NSD	20.8 (21.7)	25
Н	2.2 (2.4)	3
HD	7.2 (12.3)	12
IH	1.8 (1.3)	2
IHD	1.5 (3.0)	4
NTC	48.3 (48.0)	60

3.3 Independent October-Only Statistical Forecast

Through examination of the NCEP/NCAR reanalysis, we have discovered four predictors that in combination explain about 50 percent of the October cross-validated variance in Net Tropical Cyclone activity for the hindcast period of 1950-2001. We are currently unable to find combinations of predictors that explain large amounts of variance for the individual tropical cyclone parameters (i.e., named storms, hurricane days, etc.). Therefore, our October forecast consists of predicting NTC and consequently increasing or decreasing October's values for the other parameters accordingly. For example, if October NTC was 150 percent of normal and a typical October had two named storms, we would forecast three named storms for October. The predictors utilized in our initial October prediction are displayed graphically in Figure 4. Table 10 gives the predictor values for October 2006. Table 11 gives our independent statistical prediction for October 2006. In general, predictors for October 2006 indicate slightly below-average activity for the month, and our final forecast for October 2006 is in line with our statistical prediction.

OCTOBER PREDICTORS



Figure 4: Location of 1 August predictors for October tropical cyclone activity.

Table 10: October 2006 predictors. The sign of the predictor associated with increased tropical cyclone activity is in parentheses.

	2006	
	Observed	Effect on 2006 Hurricane
Predictors	Values	Season
Tropical Atlantic June-July SLP (-)	+0.5 SD	Suppress
Sub-Tropical Atlantic July 200 MB Ht. (+)	+1.1 SD	Enhance
South Pacific July 200 MB U (+)	-0.9 SD	Suppress
NW North America Previous Nov SLP (-)	0.0 SD	Neutral

Table 11: Independent October-only prediction of 2006 hurricane activity. October climatology is shown in parentheses.

Parameter	Statistical Model	Qualitative Adjustment
NS	1.4 (1.7)	2
NSD	7.3 (9.0)	11
Н	0.9 (1.1)	1
HD	3.6 (4.4)	4
IH	0.2 (0.3)	0
IHD	0.6 (0.8)	0
NTC	14.6 (18.0)	15

3.4 Monthly Prediction Summary

Table 12 summarizes our individual monthly predictions and our monthly adjustments to these predictions. Based on jackknifed hindcast data from 1950-2000, the sum of the August, September, and October forecasts explains 79% of the variance in seasonal TC activity.

Table 12: August, September and October 2006 individual statistical model predictions and qualitative adjustments. The monthly climatology is given in parentheses.

	August		September		October		3- Month	
Forecast	Model	August	Model	September	Model	October	Model	3 – Month
Parameter	Prediction	Adjustment	Prediction	Adjustment	Prediction	Adjustment	Sum	Adjusted Sum
NS	3.3 (2.8)	4	4.1 (3,4)	5	1.4 (1.7)	2	8.8	11
NSD	21.1 (11.8)	22	20.8 (21.7)	25	7.3 (9.0)	11	49.2	58
Н	2.9 (1.6)	3	2.2 (2.4)	3	0.9 (1.1)	1	6.0	7
HD	8.1 (5.7)	11	7.2 (12.3)	12	3.6 (4.4)	4	18.9	27
IH	0.7 (0.6)	1	1.8 (1.3)	2	0.2 (0.3)	0	2.7	3
IHD	2.0 (1.2)	3	1.5 (3.0)	5	0.6 (0.8)	0	4.1	8
NTC	53.6 (26.1)	50	48.3 (48.0)	60	14.6 (18.0)	15	116.5	125

4 Analog-Based Predictors for 2006 Hurricane Activity

Certain years in the historical record have global oceanic and atmospheric trends which are substantially similar to 2006. These years also provide useful clues as to trends in activity that the upcoming 2006 hurricane season may bring. For this early August forecast, we project atmospheric and oceanic conditions for August through October 2006 and determine which of the prior years in our database have distinct trends in key environmental conditions which are similar to current June-July 2006 conditions. Table 13 lists our analog selections.

We select prior hurricane seasons since 1949 which have similar atmosphericoceanic conditions to those currently being experienced. Analog years for 2006 were selected primarily on how similar they are to conditions that are currently observed such as slightly above-average tropical and North Atlantic sea surface temperatures and neutral to slightly warm ENSO conditions.

There were five hurricane seasons since 1949 with characteristics similar to what we observed in June-July and what we project for August-October. The best analog years that we could find for the 2006 hurricane season are 1953, 1958, 1980, 2001 and 2003. We anticipate that 2006 will have comparable seasonal hurricane activity to what was experienced in the average of these five years. We believe that the 2006 Atlantic basin hurricane season will be somewhat above average.

Year	NS	NSD	Η	HD	IH	IHD	NTC
1953	14	64.50	6	18.00	4	6.75	127
1958	10	55.50	7	30.25	5	9.50	144
1980	11	60.00	9	38.25	2	7.25	130
2001	15	64.25	9	25.50	4	4.25	134
2003	16	79.25	7	32.75	3	16.75	174
Mean	13.2	64.7	7.6	29.0	3.6	8.9	141
2006 Forecast	15	75	7	35	3	8	140

Table 13: Best analog years for 2006 with the associated hurricane activity listed for each year.

5 Comparison of Forecast Techniques

Table 14 provides a comparison of our statistical and analog forecast techniques along with the final adjusted forecast and climatology. Column 1 gives activity prior to 1 August. Column 2 gives the 3-month sum of our monthly forecasts. Column 3 is our adjusted final after 1 August forecast, Column 4 is our analog scheme, column 5 is the total season adjusted forecast and column 6 is the 1950-2000 climatology.

Table 14: Comparison of our post-1 August 2006 statistical and analog forecast techniques along with our final adjusted forecast and the 1950-2000 climatology.

	(1)	(2)	(3)	(4)	(5)	(6)
	Pre-1	Sum of 3	After-1 Aug	Total Season	Total Season	
Forecast	Aug	Individual Adjusted	Adjusted	Analog	Adjusted	1950-2000
Parameter	Activity	Monthly Forecasts	Final Forecast	Forecast	Forecast	Climatology
NS	2	11	13	13.2	15	9.6
NSD	5.5	58	69.5	64.7	75	49.1
Н	0	7	7	7.6	7	5.9
HD	0	27	35	29.0	35	24.5
IH	0	3	3	3.6	3	2.3
IHD	0	8	8	8.9	8	5.0
NTC	6	125	134	141	140	100

6 Discussion

6.1 Reasons for Reduction of the 2006 Hurricane Seasonal Forecast

We have reduced our forecast from our earlier predictions issued in early December, early April and late May. There have been no large changes in any particular atmospheric and oceanic predictor that have caused us to do this. There has, however, been a combination of changes in the ocean/atmosphere system that indicate to us that this season is no longer likely to be as active as our earlier predictions indicated. Physical features which have become less favorable for an active hurricane season are as follows:

1) An increase in sea level pressure values in the tropical Atlantic. Higher sea level pressure values indicate increased stability in the tropical Atlantic which inhibits tropical cyclogenesis.

2) An increase in strength of the trade winds in the tropical Atlantic. Stronger trade winds drive increased evaporation and upwelling which cools Atlantic sea surface temperatures. In addition, stronger trades usually indicate increased vertical wind shear in the tropical Atlantic.

3) A decrease in tropical Atlantic sea surface temperatures. Cooler Atlantic SSTAs (sea surface temperature anomalies) provide less latent heat (i.e., less fuel) for developing tropical cyclones.

4) An increase in Pacific eastern equatorial SSTAs. Sea surface temperatures have still not reached El Niño levels; however, increased warming implies a shift in tropical convection towards the dateline. This eastward-shifted convection often increases vertical wind shear over the tropical Atlantic.

The fact that we have had only two tropical storms during June-July does not necessarily impact our forecast for the upcoming season. There have been many active hurricane seasons (e.g., 1950, 2004, etc.) that had no activity in June and July. Last year (2005) was an unusually active early season with seven named storms and two major hurricanes before August 1. Last year broke most existing single season hurricane records.

6.2 Brief Comments on the 2005 Hurricane Season

The year of 2005 was a very unusual year, not only for Atlantic hurricanes but also for other global circulation features. We consider 2005 to be within the range of natural variation. For example, 1933 had 21 named storms and would likely have had 4-5 more storms if satellite data had been available. Also, the tremendous economic damage from last year's storms would have been only about 30-50% as much if the levees in New Orleans had not been breached.

7 Post 1-August Landfall Probabilities for 2006

7.1 Introduction

A significant focus of our recent research involves efforts to develop forecasts of the probability of hurricane landfall along the U.S. coastline. Whereas individual hurricane landfall events cannot be accurately forecast months in advance, the total seasonal probability of landfall can be forecast with statistical skill. With the observation that, statistically, landfall is a function of varying climate conditions, a probability specification has been developed through statistical analyses of all U.S. hurricane and named storm landfall events during the 20th century (1900-1999). Specific landfall probabilities can be given for all tropical cyclone intensity classes for a set of distinct U.S. coastal regions.

As shown in Table 15, NTC is a combined measure of the year-to-year mean of six indices of hurricane activity, each expressed as a percentage difference from the long-term average. Long-term statistics show that, on average, the more active the overall Atlantic basin hurricane season is, the greater the probability of U.S. hurricane landfall. For example, landfall observations during the 20th century show that a greater number of intense hurricanes strike the United States coastline in years of above-average NTC.

Table 15: NTC activity in any year consists of the seasonal total of the following six parameters expressed in terms of their long-term averages. A season with 10 NS, 50 NSD, 6 H, 25 HD, 3 IH, and 5 IHD would then be the sum of the following ratios: 10/9.6 = 104, 50/49.1 = 102, 6/5.9 = 102, 25/24.5 = 102, 3/2.3 = 130, 5/5.0 = 100, divided by six, yielding an NTC of 107.

	1950-2000 Average	
1)	Named Storms (NS)	9.6
2)	Named Storm Days (NSD)	49.1
3)	Hurricanes (H)	5.9
4)	Hurricane Days (HD)	24.5
5)	Intense Hurricanes (IH)	2.3
6)	Intense Hurricane Days (IHD)	5.0

7.2 Steering Current Prediction

We have considerably improved the statistical skill of our landfall probability forecasts through the inclusion of three April-May predictors of mid-latitude steering flow for the Florida Peninsula and the East Coast and two predictors of mid-latitude steering flow for the Gulf Coast. Based on data from the NCEP/NCAR reanalysis, using a combination of our NTC forecast and the predictors listed in Tables 16 and 17 and displayed in Figures 5 and 6, we are able to hindcast approximately 30 percent of the variance in hurricane landfall for the Gulf Coast and approximately 50 percent of the variance in hurricane landfall for the Florida Peninsula and the East Coast over the period 1950-2004. As evidenced by hurricane landfall activity in 2004 and 2005 compared with the earlier period of 1995-2003, the strength of midlatitude westerly winds related to the position of the Bermuda High, is vitally important in determining how likely storms are to make landfall along either the East Coast or the Gulf Coast. The predictors listed in Tables 16 and 17 give us some degree of skill in predicting the mid-level steering flow during the hurricane season, and therefore add skill to our landfall probabilities beyond that specified by the combination of NTC and SSTA*. New research is finding that SSTA* does not add much additional skill beyond NTC and the steering current predictors, and therefore we have now discontinued the inclusion of SSTA* in our

landfall probability calculations. We are currently unable to find any June-July mid-level steering flow indicators that improve upon our 1 June hindcast skill for landfall probability.

Table 16: Listing of steering current predictors for the East Coast and Florida Peninsula. The sign of the predictor associated with increased landfall is in parentheses.

Predictor	Values for 2006 Forecast
1) April-May 500 MB Ht. (35-50°N, 60-80°W) (+)	+0.1 SD
2) April-May SLP (20-40°S, 70-110°W) (+)	+0.2 SD
3) April-May 500 MB Ht. (70-85°N, 20°W-100°E) (+)	+1.9 SD

Table 17: Listing of steering current predictors for the Gulf Coast. The sign of the predictor associated with increased landfall is in parentheses.

Predictor	Values for 2006 Forecast
1) May 500 MB Ht. (10-25°S, 20-60°W) (+)	-0.8 SD
2) April-May 500 MB Ht. (40-55°S, 120°E-170°W) (+)	+0.1 SD

East Coast Landfall Predictors



Figure 5: Listing of steering current predictors for the East Coast and Florida Peninsula.

Gulf Coast Landfall Predictors



Figure 6: Listing of steering current predictors for the Gulf Coast.

7.3 Steering Current Predictor Physical Relationships

Brief descriptions of how we believe our steering current predictors relate to the steering currents likely to be present during the hurricane season are as follows:

East Coast Predictors:

Predictor 1. April-May 500 MB Geopotential Height in the Northeast United States and Canada (+)

(35-50°N, 60-80°W)

Anomalously high heights in the northeast United States during April-May, i.e., anomalous mid-level ridging, are associated with increased likelihood of hurricane landfalls along the East Coast and Florida Peninsula during the upcoming hurricane season. High heights in April-May tend to persist through August-October with an autocorrelation between the two periods of approximately 0.40. Easterly mid-level zonal wind anomalies associated with this anomalous ridging tend to drive tropical cyclones further west across the East Coast of the United States and inhibit early recurvature into the westerlies.

Predictor 2. April-May 500 MB Geopotential Height off the West Coast of South America (+)

(20-40°S, 70-110°W)

Anomalous ridging off the west coast of South America during April-May is commonly associated with strong equatorial east winds over the eastern Pacific and cold water upwelling. Such cold water upwelling is associated with a positive Southern Oscillation Index (SOI) and hence a La Niña event. La Niña events tend to persist from late May through the summer/fall period. In general, United States hurricane landfalls, especially in the Southeast, are more likely when the SOI is positive (Elsner 2003).

Predictor 3. April-May 500 MB Geopotential Height in the Arctic (+)

(70-85°N, 20°W-100°E)

Anomalously high heights in the Arctic are associated with a negative phase of the Arctic Oscillation (AO) (Thompson and Wallace 2000). A negative phase of the AO is associated with weaker westerlies across the North Atlantic. Stronger westerlies tend to steer storms away from the United States; whereas, weaker westerlies favor landfall along the East Coast of the United States (Xie et al. 2005).

Gulf Coast Predictors:

Predictor 1. May 500 MB Geopotential Height off the East Coast of South America (+)

(10-25°S, 20-60°W)

Anomalously high heights off the east coast of South America in May are strongly correlated with anomalous mid- and upper-level warming throughout the entire tropics during the spring. By the summer/fall period, a strong ridge develops over the southeastern United States with positive values of this predictor in May. This ridging tends to advect storms further west into the Gulf of Mexico and prevent early recurvature into the North Atlantic.

Predictor 2. April-May 500 MB Geopotential Height off the South Coast of New Zealand (+)

(40-55°S, 120°E-170°W)

Anomalous ridging off the south coast of New Zealand is associated with a positive value of the Antarctic Oscillation (AAO) (Thompson and Wallace 2000) and a stronger midlatitude zonal circulation in the Southern Hemisphere. By August-October, when April-May values of this predictor are positive, La Niña conditions tend to be seen in the tropical Pacific, and there is also anomalous ridging seen over the southeastern United States. Anomalous ridging over the southeastern United States tends to enhance United States Gulf Coast landfall.

7.4 2006 Landfall Probabilities

Landfall probabilities for the 2006 season are calculated based upon values of the steering current predictors listed in the previous section and NTC. Landfall probabilities for the East Coast are quite high this year, due to a combination of predicted above-average NTC values and favorable steering currents for East Coast landfall. In general, a negative North Atlantic Oscillation (NAO) and Arctic Oscillation (AO) increases the likelihood of East Coast landfall, and both of these indices have been predominately negative so far this spring (Xie et al. 2005). Two of the three predictors utilized in our East Coast steering current model relate to the NAO and AO, especially Predictor 3, which as can be seen in Table 16, has very high values this year. The odds of a major hurricane making landfall along the East Coast are more than twice the climatological average value this year.

For the Gulf Coast, landfall probabilities are slightly below the climatological average. Steering current parameters for the Gulf Coast are mixed, with one of the predictors being slightly positive and the other predictor being moderately negative. However, it is to be noted that Gulf Coast landfall probabilities are still near their climatological averages (based on predicted high values of NTC), and therefore, coastal residents should prepare for a 26% probability of a landfalling major hurricane along the Gulf Coast.

Table 18 displays the landfall probabilities for the 2006 season.

Please visit our website at <u>http://www.e-transit.org/hurricane</u> for landfall probabilities for 11 U.S. coastal regions, 55 subregions and 205 coastal and near-coastal counties from Brownsville, Texas to Eastport, Maine.

Table 18: Estimated probability (expressed in percent) of one or more U.S. landfalling tropical storms (TS), category 1-2 hurricanes (HUR), category 3-4-5 hurricanes, and total hurricanes and named storms along the entire U.S. coastline, along the Gulf Coast (Regions 1-4), and along the Florida Peninsula and the East Coast (Regions 5-11) for 2006. The long-term mean annual probability of one or more landfalling systems during the 20th century is given in parentheses.

Coastal Region	TS	Category 1-2	Category 3-4-5	All HUR	Named Storms
Entire U.S. (Regions 1-11)	85% (80%)	67% (68%)	73% (52%)	91% (84%)	99% (97%)
Gulf Coast (Regions 1-4)	57% (59%)	33% (42%)	26% (30%)	51% (61%)	79% (83%)
Florida plus East Coast	64% (51%)	47% (45%)	64% (31%)	81% (62%)	93% (81%)
(Regions 5-11)					

8 Is Global Warming Responsible for the Large Upswing in 2004-2005 US Hurricane Landfalls?

8.1 Background

The U.S. landfall of major hurricanes Dennis, Katrina, Rita and Wilma in 2005 and the four Florida landfalling hurricanes of 2004 (Charley, Frances, Ivan and Jeanne) has raised questions about the possible role that global warming may be playing in these last two unusually destructive seasons.

The global warming arguments have been given much attention by many media references to recent papers claiming to show such a linkage. Despite the global warming of the sea surface of about 0.4° C that has taken place over the last two decades, global numbers of hurricanes and their intensity have not shown increases over the past twenty years (Klotzbach 2006). In addition, we have no valid physical theory as to why small changes of global average sea surface temperature (SST) should bring about increases in Atlantic basin hurricane activity. In the past century, Atlantic basin hurricane activity has been above-average both when global SST has been increasing (from the middle 1920s through the middle 1940s) and when global SST has been decreasing (from the middle 1940s).

The Atlantic has seen a very large increase in major hurricanes during the last 11year period of 1995-2005 (average 4.0 per year) in comparison to the prior 25-year period of 1970-1994 (average 1.5 per year). This large increase in Atlantic major hurricanes is primarily a result of a multi-decadal increase in strength in the Atlantic Ocean thermohaline circulation (THC) which is not directly related to global temperature increase. Changes in ocean salinity are believed to be the driving mechanism. These multi-decadal changes have also been termed the Atlantic Multi-Decadal Oscillation (AMO). It should also be noted that during this same time period, activity in the Northeast Pacific basin has decreased considerably. When activity in these two basins (the North Atlantic and the Northeast Pacific) is summed together, there has been virtually no trend in major hurricanes.

There have been similar past periods (1940s-1950s) when the Atlantic was just as active as in recent years. For instance, when we compare Atlantic basin hurricane numbers of the last 15 years with an earlier 15-year period (1950-1964), we see little difference in hurricane frequency or intensity even though global surface temperatures were cooler and there was a general global cooling during 1950-1964 as compared with global warming during 1990-2004.

8.2 Discussion

There is no physical basis for assuming that global hurricane intensity or frequency is necessarily related to global mean surface temperature changes of less than $\pm 0.5^{\circ}$ C. As the ocean surface warms, global upper air temperatures warm as well to maintain conditionally unstable lapse-rates and global rainfall rates at their climatological values. Seasonal and monthly variations of sea surface temperature (SST) within individual storm basins show only very low correlations with monthly, seasonal, and yearly variations of hurricane activity (Shapiro and Goldenberg 1998, Klotzbach 2006). Other factors such as tropospheric vertical wind shear, surface pressure, low level

vorticity, mid-level moisture, etc. play more dominant roles in explaining hurricane variability than do surface temperatures. Although there has been a general global warming over the last 30 years and particularly over the last 10 years, the SST increases in the individual tropical cyclone basins have been smaller than the overall global warming (about half) and, according to the observations, have not brought about any significant increases in global major tropical cyclones except for the Atlantic which, as has been discussed, has multi-decadal oscillations driven primarily by changes in Atlantic salinity. No credible observational evidence is currently available that directly associates global surface temperature change with changes in global hurricane frequency and intensity.

Most Southeast coastal residents probably do not know how fortunate they had been in the prior 38-year period (1966-2003) leading up to 2004-2005 when there were only 17 major hurricanes (0.45/year) that crossed the U.S. coastline. In the prior 40-year period of 1926-1965, there were 36 major hurricanes (0.90/year or twice as many) that made U.S. landfall. It is understandable that coastal residents were not prepared for the great upsurge in landfalling major hurricanes in 2004-2005. For many years, we had been warning that the southeastern United States should expect great increases in hurricane-spawned destruction in future years.

We should interpret the last two years of unusually large numbers of U.S. landfalling hurricanes as natural but very low probability years. During 1966-2003, U.S. hurricane landfall numbers were substantially below the long-term average. In the last two seasons, they have been much above the long-term average. Although the 2004 and 2005 hurricane seasons have had an unusually high number of major landfall events, the overall Atlantic basin hurricane activity has not been much more active than five of the recent hurricane seasons since 1995 (e.g., 1995-1996, 1998-1999, 2003). What has made the 2004-2005 seasons so unusually destructive is the higher percentage of major hurricanes that moved over the U.S. coastline. These landfall events were not primarily a function of the overall Atlantic basin net major hurricane numbers, but rather of the favorable broad-scale Atlantic upper-air steering currents which were present the last two seasons. It was these favorable Atlantic steering currents which caused so many of the major hurricanes which formed to come ashore.

It is rare to have two consecutive years with such a strong simultaneous combination of high amounts of major hurricane activity together with especially favorable steering flow currents. The historical records and the laws of statistics indicate that the probability of seeing another two consecutive hurricane seasons like 2004-2005 is very low. Even though we expect to see the current active period of Atlantic major hurricane activity continue for another 15-20 years, it is statistically unlikely that the coming 2006 and 2007 hurricane seasons, or the seasons which follow, will have the number of U.S. landfalling major hurricane events that we have seen in 2004-2005.

9 Forecast Theory and Cautionary Note

Our forecasts are based on the premise that those global oceanic and atmospheric conditions which preceded comparatively active or inactive hurricane seasons in the past provide meaningful information about similar trends in future seasons. It is important that the reader appreciate that these seasonal forecasts are based on statistical schemes which, owing to their intrinsically probabilistic nature, will fail in some years. Moreover, these forecasts do not specifically predict where within the Atlantic basin these storms will strike. The probability of landfall for any one location along the coast is very low and reflects the fact that, in any one season, most U.S. coastal areas will not feel the effects of a hurricane no matter how active the individual season is. However, it must also be emphasized that a low landfall probability does not insure that hurricanes will not come ashore. Regardless of how active the 2006 hurricane season is, a finite probability always exists that one or more hurricanes may strike along the U.S. coastline or in the Caribbean and do much damage.

10 Forthcoming Updated Forecasts of 2006 Hurricane Activity

We will be issuing seasonal updates of our 2006 Atlantic basin hurricane forecasts on **Friday 1 September** and **Tuesday 3 October 2006**. The 1 September and 3 October forecasts will include separate forecasts and updates of September-only and October-only Atlantic basin tropical cyclone activity. A verification and discussion of all 2006 forecasts will be issued in late November 2006. Table 19 displays our forecast schedule for the remainder of the 2006 hurricane season. Our first seasonal hurricane forecast for the 2007 hurricane season will be issued in early December 2006. All of these forecasts will be made available on the web at: <u>http://hurricane.atmos.colostate.edu/Forecasts</u>.

	Based on Data						
Forecast Date	Through	Upcoming Forecasts and Updates					
1 September 2006	August 2006	August	Updated	Updated	Updated		
	-	Verification	September	October	Seasonal		
			Forecast	Forecast	Forecast		
3 October 2006	September		September	Updated	Updated		
	2006		Verification	October	Seasonal		
				Forecast	Forecast		
Late November 2006		Verification of all Forecasts					

Table 19: Timetable of upcoming forecasts and updates for the 2006 hurricane season.

11 Acknowledgments

Besides the individuals named on page 4, there have been a number of other meteorologists that have furnished us with data and given valuable assessments of the current state of global atmospheric and oceanic conditions. These include Brian McNoldy, Arthur Douglas, Richard Larsen, Todd Kimberlain, Ray Zehr, and Mark DeMaria. In addition, Barbara Brumit and Amie Hedstrom have provided excellent manuscript, graphical and data analysis and assistance over a number of years. We have profited over the years from many in-depth discussions with most of the current and past NHC hurricane forecasters. The second author would further like to acknowledge the encouragement he has received for this type of forecasting research application from Neil Frank, Robert Sheets, Robert Burpee, Jerry Jarrell, former directors of the National Hurricane Center (NHC), and from the current director, Max Mayfield and their forecast staffs. Uma Shama and Larry Harman of Bridgewater State College, MA have provided assistance and technical support in the development of the Landfalling Hurricane Probability Webpage. We also thank Bill Bailey of the Insurance Information Institute for his sage advice and encouragement.

The financial backing for the issuing and verification of these forecasts has in part been supported by the National Science Foundation and by the Research Foundation of Lexington Insurance Company (a member of the American International Group). We also thank the GeoGraphics Laboratory at Bridgewater State College for their assistance in developing the Landfalling Hurricane Probability Webpage.

12 Citations and Additional Reading

- Blake, E. S., 2002: Prediction of August Atlantic basin hurricane activity. Dept. of Atmos. Sci. Paper No. 719, Colo. State Univ., Ft. Collins, CO, 80 pp.
- Blake, E. S. and W. M. Gray, 2004: Prediction of August Atlantic basin hurricane activity. *Wea. Forecasting*, 19, 1044-1060.
- DeMaria, M., J. A. Knaff and B. H. Connell, 2001: A tropical cyclone genesis parameter for the tropical Atlantic. *Wea. Forecasting*, 16, 219-233.
- Elsner, J. B., 2003: Tracking hurricanes. Bull. Amer. Meteor. Soc., 84, 353-356.
- Elsner, J. B., G. S. Lehmiller, and T. B. Kimberlain, 1996: Objective classification of Atlantic hurricanes. J. Climate, 9, 2880-2889.
- Goldenberg, S. B., C. W. Landsea, A. M. Mestas-Nunez, and W. M. Gray, 2001: The recent increase in Atlantic hurricane activity: Causes and Implications. *Science*, 293, 474-479.
- Goldenberg, S. B. and L. J. Shapiro, 1996: Physical mechanisms for the association of El Niño and West African rainfall with Atlantic major hurricane activity. *J. Climate*, 9, 1169-1187.
- Gray, W. M., 1984a: Atlantic seasonal hurricane frequency: Part I: El Niño and 30 mb quasi-biennial oscillation influences. *Mon. Wea. Rev.*, 112, 1649-1668.
- Gray, W. M., 1984b: Atlantic seasonal hurricane frequency: Part II: Forecasting its variability. Mon. Wea. Rev., 112, 1669-1683.
- Gray, W. M., 1990: Strong association between West African rainfall and US landfall of intense hurricanes. *Science*, 249, 1251-1256.
- Gray, W. M., and P. J. Klotzbach, 2003, 2004 and 2005: Forecasts of Atlantic seasonal and monthly hurricane activity and US landfall strike probability. Available online at http://hurricane.atmos.colostate.edu

- Gray, W. M., C. W. Landsea, P. W. Mielke, Jr., and K. J. Berry, 1992: Predicting Atlantic seasonal hurricane activity 6-11 months in advance. *Wea. Forecasting*, 7, 440-455.
- Gray, W. M., C. W. Landsea, P. W. Mielke, Jr., and K. J. Berry, 1993: Predicting Atlantic basin seasonal tropical cyclone activity by 1 August. *Wea. Forecasting*, 8, 73-86.
- Gray, W. M., C. W. Landsea, P. W. Mielke, Jr., and K. J. Berry, 1994a: Predicting Atlantic basin seasonal tropical cyclone activity by 1 June. *Wea. Forecasting*, 9, 103-115.
- Gray, W. M., J. D. Sheaffer and C. W. Landsea, 1996: Climate trends associated with multi-decadal variability of intense Atlantic hurricane activity. Chapter 2 in "Hurricanes, Climatic Change and Socioeconomic Impacts: A Current Perspective", H. F. Diaz and R. S. Pulwarty, Eds., Westview Press, 49 pp.
- Gray, W. M., 1998: Atlantic Ocean influences on multi-decadal variations in El Niño frequency and intensity. Ninth Conference on Interaction of the Sea and Atmosphere, 78th AMS Annual Meeting, 11-16 January, Phoenix, AZ, 5 pp.
- Henderson-Sellers, A., H. Zhang, G. Berz, K. Emanuel, W. Gray, C. Landsea, G. Holland, J. Lighthill, S-L. Shieh, P. Webster, and K. McGuffie, 1998: Tropical cyclones and global climate change: A post-IPCC assessment. *Bull. Amer. Meteor. Soc.*, 79, 19-38.
- Klotzbach, P. J., 2002: Forecasting September Atlantic basin tropical cyclone activity at zero and onemonth lead times. Dept. of Atmos. Sci. Paper No. 723, Colo. State Univ., Ft. Collins, CO, 91 pp.
- Klotzbach, P. J., 2006: Trends in global tropical cyclone activity over the last twenty years (1986-2005). *Geophys. Res. Lett.*, 33, L10805, doi:10.1029/2006GL025881.
- Klotzbach, P. J. and W. M. Gray, 2003: Forecasting September Atlantic basin tropical cyclone activity. *Wea. and Forecasting*, 18, 1109-1128.
- Klotzbach, P. J. and W. M. Gray, 2004: Updated 6-11 month prediction of Atlantic basin seasonal hurricane activity. *Wea. and Forecasting*, 19, 917-934.

Knaff, J. A., 1997: Implications of summertime sea level pressure anomalies. J. Climate, 10, 789-804.

- Knaff, J. A., 1998: Predicting summertime Caribbean sea level pressure. *Wea. and Forecasting*, 13, 740-752.
- Landsea, C. W., 1991: West African monsoonal rainfall and intense hurricane associations. Dept. of Atmos. Sci. Paper, Colo. State Univ., Ft. Collins, CO, 272 pp.
- Landsea, C. W., 1993: A climatology of intense (or major) Atlantic hurricanes. Mon. Wea. Rev., 121, 1703-1713.
- Landsea, C. W. and W. M. Gray, 1992: The strong association between Western Sahel monsoon rainfall and intense Atlantic hurricanes. J. Climate, 5, 435-453.
- Landsea, C. W., W. M. Gray, P. W. Mielke, Jr., and K. J. Berry, 1992: Long-term variations of Western Sahelian monsoon rainfall and intense U.S. landfalling hurricanes. J. Climate, 5, 1528-1534.
- Landsea, C. W., W. M. Gray, K. J. Berry and P. W. Mielke, Jr., 1996: June to September rainfall in the African Sahel: A seasonal forecast for 1996. 4 pp.
- Landsea, C. W., N. Nicholls, W.M. Gray, and L.A. Avila, 1996: Downward trends in the frequency of intense Atlantic hurricanes during the past five decades. *Geo. Res. Letters*, 23, 1697-1700.

- Landsea, C. W., R. A. Pielke, Jr., A. M. Mestas-Nunez, and J. A. Knaff, 1999: Atlantic basin hurricanes: Indices of climatic changes. *Climatic Changes*, 42, 89-129.
- Landsea, C.W. et al., 2005: Atlantic hurricane database re-analysis project. Available online at <u>http://www.aoml.noaa.gov/hrd/data_sub/re_anal.html</u>
- Mielke, P. W., K. J. Berry, C. W. Landsea and W. M. Gray, 1996: Artificial skill and validation in meteorological forecasting. *Wea. Forecasting*, 11, 153-169.
- Mielke, P. W., K. J. Berry, C. W. Landsea and W. M. Gray, 1997: A single sample estimate of shrinkage in meteorological forecasting. *Wea. Forecasting*, 12, 847-858.
- Pielke, Jr. R. A., and C. W. Landsea, 1998: Normalized Atlantic hurricane damage, 1925-1995. Wea. Forecasting, 13, 621-631.
- Rasmusson, E. M. and T. H. Carpenter, 1982: Variations in tropical sea-surface temperature and surface wind fields associated with the Southern Oscillation/El Niño. *Mon. Wea. Rev.*, 110, 354-384.
- Seseske, S. A., 2004: Forecasting summer/fall El Niño-Southern Oscillation events at 6-11 month lead times. Dept. of Atmos. Sci. Paper No. 749, Colo. State Univ., Ft. Collins, CO, 104 pp.
- Shapiro, L. J. and S. B. Goldenberg, 1998: Atlantic sea surface temperatures and tropical cyclone formation. J. Climate, 11, 578-590.
- Xie, L., L. J. Pietrafesa, J. M. Morrison, and T. Karl, 2005: Climatological and interannual variability of North Atlantic hurricane tracks. J. Climate, 18, 5370-5381.

13 Verification of Previous Forecasts

•

Table 20: Summary verification of the authors' six previous years of seasonal forecasts for Atlantic TC activity between 2000-2005.

2020		U	pdate	Update	Update		
2000	8 Dec.	1999 7	April	7 June	4 August	Obs	
No. of Hurricanes	7	7		8	7	8	
No. of Named Storms	11	1	1	12	11	14	
No. of Hurricane Days	25	2:	5	35	30	32	
No. of Named Storm Days	55	5:	\$ 	65	55	66	
Hurr. Destruction Potential	85	8:	5	100	90	85	
Intense Hurricanes	3	3		4	3	3	_
Intense Hurricane Days	6	6		8	6	5.25	5
Net Tropical Cyclone Activity	125	1:	25	160	130	134	
				** • .	.		
2001	7 Dec	2000 6	pdate Ameil	Update 7 June	∪pdate 7 August	01-	
2001	7 Dec.	2000 6	April	/ June	/ August	Obs	i.
No. of Humicanes	3	0		10	10	9	
No. of Named Storms	9	10	-	12	12	15	
No. of Named Stame Days	20	2.		30	30	27	
No. of Named Storm Days	45	51) -	50	60	63	
Intense Humisenes	65	0.	b	/5	/5	/1	
Intense Hurricanes	2	2		5	3	4	
Net Transcal Custome Astistic	4	4	0	5	5	3	
Net Hopical Cyclone Activity	90	10	10	120	120	142	
		I indate	7 Tau d	oto TL	ndata	T In data	1
2002	7 Dec. 2001	5 April	0pu 31 λ	ate ∪] ∕[av 7	August	Opdate 2 Sent	Obe
No. of Hurricanes	8	7		14y /	August	2 3001.	005.
No. of Named Storms	0	12	11	4		3	4
No. of Hurricone Dave	25	12	25	9	, ,	°	12
No. of Named Storm Days	70	50	25	12	-	10	54
Hurr Destruction Potential	00	85		33	,	25	34
Intense Hurricones	30	2	2	33	>	25	31
Intense Hurricone Davs	7	5	2	2		2	25
Net Tropical Cyclone Activity	140	125	100	4)	45	2.5
The Proproat Cyclone Fleat Add		120	100		, 		00
		Update	Update	Update	Undate	Undate	1
2003	6 Dec. 2002	4 April	30 May	6 August	3 Sept.	2 Oct.	Obs.
No. of Hurricanes	8	8	8	8	7	8	7
No. of Named Storms	12	12	14	14	14	14	14
No. of Hurricane Days	35	35	35	25	25	35	32
No. of Named Storm Days	65	65	70	60	55	70	71
Hurr, Destruction Potential	100	100	100	80	80	125	129
Intense Hurricanes	3	3	3	3	3	2	3
Intense Hurricane Davs	8	8	8	5	9	15	17
Net Tropical Cyclone Activity	140	140	145	120	130	155	173
				·			
		Update	Update	Update	Update	Update	1
2004	5 Dec. 2003	2 April	28 May	6 August	3 Sept.	1 Oct.	Obs.
No. of Hurricanes	7	8	8	7	8	9	9
No. of Named Storms	13	14	14	13	16	15	14
No. of Hurricane Days	30	35	35	30	40	52	46
No. of Named Storm Days	55	60	60	55	70	96	90
Intense Hurricanes	3	3	3	3	5	6	6
Intense Hurricane Days	6	8	8	6	15	23	22
Net Tropical Cyclone Activity	125	145	145	125	185	240	229
2005	2 Dag 2004	Update	Update	Update	Update 2. Sant	Update	0
No. of Humissner	5 Dec. 2004	1 April 7	o i iviay	J August	2 Sept.	3 Uct.	UDS.
No. of Nomed Storms	0	/ 12	5 16	10	10	11	15
No. of Hurricone David	11	15	15	20	∠U 45	20	2/ 51
No. of Named Storm Dava			4.7	17	40	40	∎ 31
INU. ULINAIIICU SIUTIII D'AYS	25	55	75	05	05	100	125
Intense Hurricones	25 55 3	65 3	75	95	95	100	125
Intense Hurricanes Intense Hurricane Days	25 55 3	65 3 7	75 4 11	95 6 18	95 6 15	100 6 13	125 7 16 75
Intense Hurricanes Intense Hurricane Days Net Tropical Cyclone Activity	25 55 3 6 115	65 3 7 135	75 4 11 170	95 6 18 235	95 6 15 220	100 6 13 215	125 7 16.75 275